Some Inequalities Concerning Power Series and Their Interaction With Univalent Function Theory

Alawiah Ibrahim

B.Sc (Hons), M.Sc (Mathematics)
College of Engineering and Science, Victoria University, Melbourne, Australia.

Submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

Abstract

Power series are fundamental in the study of Geometric Function Theory. In fact, they constitute a major part in Complex Analysis. The purpose of this dissertation is to employ analytic functions defined by power series in two different directions. The first of these discussed in Part I, mainly contributes to Analytic Inequalities in Real and Complex Analysis, while the second direction discussed in Part II, deals with Analytic and Univalent Function Theory.

Part I of this dissertation is devoted to some inequalities concerning power series with real or nonnegative coefficients and convergent on an open disk. The main purpose of this part is to derive new inequalities for functions defined by convergent power series, which are related to the celebrated classical inequalities in Real and Complex Analysis such as the Cauchy-Bunyakovsky-Schwarz, Young's, Hölder's and Jensen type inequalities. In particular, we obtain new and better inequalities for the power series functions, which provide generalizations, refinements and improvements on the earlier results published by Cerone and Dragomir [89], Dragomir and Ionescu [138], Dragomir and Sándor [136] and others. Further, particular inequalities are obtained by applying the results for the fundamental functions such as exponential, logarithm, trigonometric and hyperbolic functions. Some inequalities involving special functions such as polylogarithm, hypergeometric, Bessel and modified Bessel functions of the first kind are established as well.

Part II is mainly concerned with analytic functions in a unit disk and normalized by the conditions f(0) = f'(0) - 1 = 0. The functions in the class S, which are analytic, univalent and normalized in a unit disk and have power series representations, are the center of the study of Univalent Function Theory. The properties of functions in the class S and its subclasses such as the starlike, spirallike, convex, close-to-convex, etc., have been widely investigated by various researchers in the past three decades. In the other direction, new subclasses of functions, which are defined involving operators, were introduced. The Sălăgean operator [383] is one of the famous differential operators, which was used by

0. Abstract

vii

Opoola [343] to introduce new subclasses of S. To provide some new properties for certain subclasses of functions introduced by Opoola is the main objective throughout the second part of this dissertation. In this study, we introduce the special subclass of Opoola's functions and then investigate some properties and coefficients inequalities of functions in this class. Finally, we conclude Part II by providing the Fekete-Szegő theorem concerning subclasses of analytic and univalent functions.

Declaration

"I, Alawiah Ibrahim, declare that the PhD thesis entitled "Some Inequalities Concerning Power Series and Their Interaction with Univalent Function Theory" is no more than 100,000 words in length including quotes and exclusive of tables, figures, appendices, bibliography, references and footnotes. This thesis contains no material that has been submitted previously, in whole or in part, for the award of any other academic degree or diploma. Except where otherwise indicated, this thesis is my own work".

Signature Date

Acknowledgements

All praises belong to Almighty Allah (SWT), who gave me the strength, patience and ability to come up with this thesis for the completion of my PhD studies. Peace be upon to our Holy Prophet Muhammad (SAW) and his honorable family. This thesis also would not have been possible without the valid support and cooperation of a number of people closed to me, who believed in me and my undertakings. Thus, I would like to express my appreciation to them.

First and foremost, I would like to express my sincere appreciation to my principal supervisor Professor Sever S. Dragomir for his untiring supervision, wonderful guidance, support and advice throughout the whole period of my candidature. I have no doubt that this thesis would not been produced without his ideas, assistance and abundant time. I am deeply thankful for his patience, understanding and encouraging, and his personal advice have provided a strong motivation for me to work continuously and complete this thesis successfully within the time. His breadth of experience and knowledge, enthusiasm, dedication and hard work have much impressed on me. Indeed, I am very proud of having the opportunity of working with him.

My warmest gratitude also goes to my associated supervisor Professor Maslina Darus for her constant support and encouraging advice. I also thank her for giving a wonderful idea and suggestion, when I was at the stage of finding the place and field to pursue my study. I also wish to extend my thanks to all my colleagues at School of Mathematical Sciences, UKM, who always pray for me to be successful in these studies.

This study would not be possible without the financial help from the Ministry of Higher Education of Malaysia (MOHE) and my employer, National University of Malaysia (UKM). I gratefully acknowledge to these institutions for providing the funding in terms of the scholarships, and hence giving me the opportunity to pursue my study.

I also would like to acknowledge the College of Science and Engineering, Victoria University for providing me all necessary facilities and excellent research environment that allowed me to complete this thesis successfully. For all the academic and support staff itself, especially Kylie Boggetto, Lyn Allis and Anne Young and the Postgraduate Research Coordinator Dr. Gitesh Raikundalia, thank you very much for your generous support, kindness and cooperation during my time as a research student. Thanks also to the Postgraduate Research Centre for organizing high-quality workshops and research training that contributed to my personal and professional development.

Most of all, I reserve the greatest gratitude for my beloved Husband Azman Mohd Zain, who had sacrificed his time and career accompanying me in pursuing this 'endless' journey over these many years. 'Abang' thank you very much for your love, patience, support and understanding. Thanks also for always being there for me and for always believing in me. My dearest children Tasneem Batrisyia and Adam Nawfal, they are a source of my inspiration, strength and joy, although they are too young to understand what this task is all about. My greatest thanks also to my parents, parents-in-law, brothers and sisters, whose unconditional love, support and prayer are invaluable to me. Thank you for your patience with me being far from you for many years.

Last but not least, my copious thanks to all my friends who have contributed directly or indirectly to my study and completion of this dissertation. To all Malaysians, thanks a lot for sharing some great moments in Footscray, Melbourne. May Allah bless all of you.

List of Publications

This dissertation contains the results from a number of the author's research papers that have been published in refereed publications or have been submitted for publications:

- (1) Chapter 3 of this dissertation contains some of the results from the following research papers:
 - "Power series inequalities via Buzano's result and applications" with S. S. Dragomir, *Integral Transforms and Special Functions*, Vol. **22**, Issue 12, 2011, pp. 867-878.
 - "Power series inequalities via a refinement of the Schwarz inequality" with S. S. Dragomir, *Integral Transforms and Special Functions*, Vol. **23**, Issue 10, 2012, pp. 769-781.
 - "Some inequalities for power series with applications" with S. S. Dragomir and M. Darus, *Integral Transforms and Special Functions*, Vol. **24**, No. 5, 2013, pp. 364-376.
- (2) Chapter 4 of this dissertation contains some of the results from the following research papers:
 - "Power series inequalities via Young's inequality with applications" with S. S. Dragomir and M. Darus, Journal of Inequalities and Application **2013**, 2013, pp. 314 326.
 - "Power series inequalities related to Young's inequality and applications" with S. S. Dragomir and M. Darus, *Integral Transforms and Special Functions*, Vol. **24**, No. 9, 2013, pp. 700-714.
 - "Inequalities for power series with positive coefficients" with S. S. Dragomir, P. Cerone and M. Darus, *Journal of Inequalities and Special Functions*, Vol. **3**, Issue 2, 2012, pp. 1-15.

- (3) Chapter 5 of this dissertation contains some of the results from the author's research papers, which are mentioned in (1) and (2) above.
- (4) All the results that contain in Chapter 3 and some of the results from Chapter 5, also have been incorporated in the following survey paper:
 - "A survey on Cauchy-Bunyakovsky-Schwarz inequality for power series" with S. S. Dragomir. This paper has been accepted as a chapter in the book of H. M. Srivastava Honorary Volume by Springer.
- (5) Chapter 7 of this dissertation contains the results from the following author's research papers:
 - "Coefficient inequalities for a certain subclass of analytic functions" with M. Darus, S. S. Dragomir and O. A. F-. Joseph, *Journal of Inequalities and Special Functions* (accepted). Preprint available at: *RGMIA Research Report Collection*, Vol. **14**, 2011, Article no. 89, 9 pp, http://www.rgmia.org/issues.php.
 - -"Fekete-Szegő problem for concave univalent functions defined by Sălăgean operator" with M. Darus and S. S. Dragomir (submitted). Preprint available at: *RGMIA Research Report Collection*, Vol. **14**, 2011, Article no. 88, 11 pp, http://www.rgmia.org/issues.php.
 - "Some properties for a certain subclass of univalent functions" with M. Darus, S. S. Dragomir and O. A. F-. Joseph, to appear.

List of Symbols

The following lists are standard symbols, notations and abbreviations that are used in this dissertation.

$\mathbb{R}, \ \mathbb{R}^+, \ \mathbb{R}^-$	The set of real, positive real and negative real numbers
\mathbb{C}	The complex numbers
$\mathbb{Z},~\mathbb{Z}^+,~\mathbb{Z}^-$	The set of integers, positive integers and negative integers
$\mathbb{Z}_0^+,\mathbb{Z}_0^-$	The set of nonnegative integers and nonpositive integers
\mathbb{N}	The set of natural numbers
\mathbb{N}_0	The set of zero and natural numbers
\mathbb{K}	The field of real or complex number
\mathbf{U}	The convex set
\mathbf{X}	The linear space
D	The open unit disk
$D\left(0,R\right)$	The open disk
$D\left(z_{0},R\right)$	The open disk with center at z_0 and radius R
$\partial D,\ \overline{D}$	The boundary of D , the closure of D
$\Gamma\left(s\right)$	The gamma function
$\Psi\left(s ight)$	The digamma (psi) function
$\Psi_{n}\left(s ight)$	The polygamma function
$\gamma\left(lpha,z ight)$	The incomplete gamma function
$\Gamma\left(\alpha,z\right)$	The complementary incomplete gamma function
$B\left(x,y\right)$	The incomplete beta function
$\zeta\left(s\right)$	The Riemann zeta function
$\beta\left(s ight)$	The beta function
$\eta\left(s ight)$	The eta function
$\zeta\left(s,q\right)$	The Hurwitz zeta function
$Li_{n}\left(s\right)$	The polylogarithm function
$_2F_1$	The Gauss hypergeometric function
$_pF_q$	The generalized hypergeometric function
$J_{\alpha}\left(z\right)$	The Bessel function of the first kind

$I_{lpha}\left(z ight)$	The modified Bessel function of the first kind
$(a)_n$	The Pochhammer symbol
B_n	The Bernoulli numbers
$E_{n,k}$	The Eulerian numbers
$H_{n,r}$	The generalized harmonic number
$Z^{'}$	The Apèry's constant
G	The Catalan's constant
n!, n!!	The factorial function, double factorial function
$\binom{a}{n}$	The binomial coefficient
sgn(x)	The signum (sign) function
$\mathcal{C}\left[a,b\right]$	The function space
$\langle \cdot, \cdot \rangle$	The inner product
$(V,\langle\cdot,\cdot\rangle)$	The inner product spac
\mathbf{X}	The linear space
$\operatorname{Re}\left(z\right)$	The real part of z
$\mathrm{Im}\left(z ight)$	The imaginary part of z
${\cal P}$	The class of functions with a positive real part
$\mathcal{H}\left(D\right)$	The class of analytic functions on the unit disk D
A	The class of analytic and normalized functions on a unit disk
S	The class of analytic, univalent and normalized on a unit disk
S^*	The class of starlike functions
C	The class of convex functions
$\mathcal K$	The class of close-to-convex functions
k(z)	The Koebe function
m(z)	The Möbius transformation
γ	The Jordan curve
$S^{*}\left(lpha ight)$	The class of starlike functions of order α
$C\left(lpha ight)$	The class of convex functions of order α
C_{α}	The class of α -convex functions
$\mathcal{B}\left(\alpha\right)$	The class of Bazilevč functions of order α
$\mathcal{B}\left(\alpha,\beta\right)$	The class of Bazilevč functions or order α and type β
C_{lpha}	The class of α -convex functions
$S^*\left(\alpha\right)$	The class of starlike functions of order α
$C(\alpha)$	The class of convex functions of order α
$D^{\hat{n}}f^{'}$	The Sălăgean differential operator
$D_{\lambda}^{n}f$	The Al-Oboudi differential operator
	The end of proof

Contents

Abstract			vi	
D	Declaration v Acknowledgements			
\mathbf{A}				
Li	st of	Publications	xi	
Li	st of	Symbols	xiii	
1	Intr	roduction	1	
	1.1	General Background	1	
		1.1.1 Power Series	1	
		1.1.2 Inequalities Theory	7	
		1.1.3 Univalent Function Theory	10	
	1.2	Basic Properties of Power Series	13	
	1.3	Power Series Representations	17	
	1.4	Motivation	20	
	1.5	Thesis Outline	23	
I P		NALYTIC INEQUALITIES IN REAL AND COM- K ANALYSIS	26	
2	Ele	mentary and Some Classical Inequalities	27	
	2.1	Symbols and Notations	28	
	2.2	Elementary Inequalities in Analysis	30	
		2.2.1 Basic Inequalities	31	

CONTENTS

		2.2.2	Triangle Inequality	33
		2.2.3	Mean Inequalities	34
		2.2.4	Convexity Inequalities	37
		2.2.5	Normed, Linear and Inner Product Spaces	41
	2.3	The (CBS)-Type Inequalities	44
		2.3.1	(CBS)-Inequalities for Real and Complex Numbers $\ \ .$	44
		2.3.2	(CBS)-Inequalities in Inner Product Spaces	46
		2.3.3	(CBS)-Inequalities for Power Series	47
		2.3.4	Some Results Related to (CBS)-Type	49
	2.4	Young	g's Inequality and Its Variants	51
	2.5	Hölde	r's Inequality	53
	2.6	Jenser	Types Inequalities and Their Reverses	56
	2.7	Other	Inequalities	59
3	Pov	ver Ser	ries and the Cauchy-Bunyakovsky-Schwarz Inequality	61
	3.1	Introd	luction and Preliminary Results	62
	3.2	Power	Series Inequalities Via Buzano's Results	65
	3.3	Power	Series Inequalities Via a Refinement of the Schwarz Inequality	76
	3.4	Other	Refinements of the (CBS)-type	87
4	Mo	re Inec	qualities on Power Series with Real Coefficients	98
	4.1	Some	Results Related to Young's Inequality	99
		4.1.1	Introduction and Preliminary Results	99
		4.1.2	Power Series Inequalities Via Young's Inequality	100
		4.1.3	Further Improvements of Hölder's Inequality for Power Series	112
	4.2		Results Via Convexity and Jensen's Type Inequalities	
		4.2.1	Introduction and Preliminary Results	126
		4.2.2	Power Series Inequalities Via Convexity	128
		4.2.3	Power Series Inequalities Via Jensen Type	135
5	App	olicatio	ons to Special Functions	142
5.1 Definitions and Basic Con		Defini	tions and Basic Concepts	142
		5.1.1	Gamma, Zeta and Related Functions	
		5.1.2	Polylogarithm Functions	

	•••
CONTENTS	111
CONTENTS	111

		5.1.3 Hypergeometric Functions	151	
		5.1.4 Bessel and Modified Bessel Functions	155	
	5.2	Inequalities for Polylogarithm Functions	158	
	5.3	•		
	5.4	Inequalities for Bessel Functions		
II A		OME INEQUALITIES INVOLVING ANALYTIC UNIVALENT FUNCTIONS	: 1 73	
6	Eler	mentary Theory of Univalent Functions	174	
	6.1	Basic Concepts	175	
	6.2	Functions With a Positive Real Part	178	
	6.3	Analytic and Univalent Functions	180	
	6.4	Subclasses of Analytic and Univalent Functions	181	
	6.5	Some Classical Results	186	
	6.6	Sălăgean Differential Operator and Related Subclasses	189	
7	Pro	operties for Certain Subclasses of Analytic Functions	192	
	7.1	Introduction	193	
	7.2	The Properties	195	
	7.3	Coefficient Bounds	205	
	7.4	Fekete-Szegő Inequalities	210	
		7.4.1 Introduction and Preliminary results	210	
		7.4.2 Main Results and Their Proofs	212	
8	Sun	nmary and Conclusion	224	
	8.1	Summary	224	
	8.2	Main Achievements	226	

Chapter 1

Introduction

Power series is a special type of series of a function, that is of fundamental importance in Complex Analysis. The fact that an analytic function in an open disk can be represented by convergent power series, makes it an important element in the study of Geometric Function Theory. This chapter begins with a concise introduction to the power series. Then, it follows by an introduction to Inequalities Theory in real and complex analysis and Univalent Function Theory, which are the two main parts of this dissertation. Basic properties of the power series and their representations for some fundamental functions are given in the following sections. Finally, this chapter presents the main motivation for writing this dissertation and provides an overview of the outline and content of this dissertation.

1.1 General Background

1.1.1 Power Series

An *infinite series* is, informally speaking, the sum of the terms of an infinite sequence. One of the most important infinite series, which appears in almost all areas of pure and applied mathematics, is called *power series*.

A power series is a series, where a 'variable' is involved in the terms of the series. Roughly speaking, a power series is an infinite series that can be written

in a systematic pattern of change in the powers of variable, and the terms of the series are often produced according to the certain rules. For example,

(i)
$$1 + x + x^2 + x^3 + x^4 + \dots$$

(ii) $x - \frac{x^2}{2} + \frac{x^3}{3} - \frac{x^4}{4} + \dots$
(iii) $(x - 1) - \frac{1}{2}(x - 1)^2 + \frac{1}{3}(x - 1)^3 - \frac{1}{4}(x - 1)^4 + \dots$

are the power series, which are specifically known as the *geometric series*, *Maclau-rin's series* and *Taylor's series*, respectively.

The terminology of the infinite series was introduced in the last third of the 17th century. It probably appeared in print for the first time in 1668, in the second of James Gregory's manuscript entitled "Exercitationes Geometricae", published in London (cf. Gregory [172, p. 10-12]). However, the first mention of the notion of the infinite series dates back to antiquity. Historically, Greek mathematician Archimedes (c. 287 BC - c. 212 BC) produced the first summation of the infinite series with a method that is still used in the area of calculus today. He used the method of exhaustion and Egyption fraction to develop the infinite series by calculating the area under the arc of a parabola. Hence, he obtained a remarkable approximation value of π (223/71 $< \pi < 22/7$), see ([31, p. 233-252], [252, p. 35-37], [340]). The development of the infinite series techniques then commenced with the computing of some more accurate values of π by many mathematicians until the 17th century. Chinese mathematicians, Liu Hui (around 265 AD) and Zu Chongzhi (around 480 AD) created the same method based on the polygon iterative algorithm, but they considered the different number of polygon sides in the circle, to obtain the approximation values of π (see [32, p. 177-178], [73, p. 202-203], [329, p. 66, 100-101]). The infinite series were also exploited for π , most notably by European mathematicians such

¹The method of exhaustion is a method of finding area of shape by inscribing inside it a sequence of polygons whose areas converge to the area containing shape. If the sequence is correctly constructed, the difference in area between the nth polygon and the containing shape will become arbitrary small as n become large^a. ^aAdopted from http://en.wikipedia.org/Method of exhaustion, on 15 May 2013.

²Egyption fraction is the sum of distinct unit fractions, where each fraction has a numerator equal to 1 and a denominator is a positive integer^b. ^bAdopted from http://en.wikipedia.org/Egyption_ fraction, on 15 May 2013.

as François Viète (see [32, p. 187], [428, p. 397-398]), John Wallis [429, p. 1-52, 753-754], James Gregory (see [171, p. 34-39], [172, p. 10-12]), Gottfried Wilhelm Leibniz [32, p. 188-189] and Nicholas Mercator [307]. They found the approximation values of π by developing the infinite series to solve problems under the particular curves. Indeed, they showed that the infinite series techniques were the useful tool in solving the geometrical problems until the 17th century, that is before the rise of calculus. However, in those times, to understand that a sum with infinite number of summands could have a finite result was a great philosophical challenge.

Obtaining the sum of a particular series of numbers is of relatively little interest in comparison with the expanding of a function into a series, whose terms depend on a variable. In modern terminology, this is called the *power series expansion*. The idea of the infinite series expansion of functions was first conceived in India, in the 14th century by Madhava of Sangamagrama (c. 1350 - c. 1425). He laid down the precursors of the modern conception of the power series and discovered a number of the Taylor series expansion of functions, such as sine, cosine, tangent and arctangent (see [43], [106], [168]). Alongside his discovering, Madhava also developed the convergence criteria of the infinite series; his students and followers at the Kerala School of Astronomy and Mathematics, India, further expanded his works and obtained various type of series expansions until the 16th century (see [399], [409, p. 173], [420]).

Madhava's formula of arctangent was rediscovered, in a different way, by the Scottish mathematician James Gregory in 1671, a few years after he published the formal terminology of the infinite series. This discovery is attributed to him, though he also formulated the expansion of power series for the trigonometric functions such as sine, cosine, arcsine and arcosine, and published several Maclaurin's series as well [73, p. 439-445]. After Issac Newton developed the formal methods of modern calculus in 1660s, then Leibniz was able to find the series expansions of sine, arcsine and explored many more infinite series including the Taylor series (see [73, p. 446-447], [278], [279]). However, a general method for constructing the Taylor series expansion for all functions was formally provided by Brook Taylor in 1715, after whom the series are now named (cf. Taylor [416, p. 21-23]). A Maclaurin series was named after Scottish mathematician Colin Maclaurin, who made an extensive use of the special case of the Taylor series in

the 18th century, when the series is expanded around zero. A remarkable contribution on the development of the Series Theory also goes to Swiss Leonhard Euler, who made some important discoveries in this field. He directly proved the power series expansions for the exponential and inverse tangent functions [73, p. 496-497], and provided the ways to express various logarithmic functions by using the power series method. Notably, he introduced the theory of hypergeometric series and q-series.

The discovery of the mathematical quantities such as the exponential, logarithm, trigonometric functions of sine, cosine, tangent, etc. that could be written as the power series expansions, was one of the great mathematical achievements in the mid 17th century. Together with the 'invention' of calculus by English scientist Newton and German mathematician Leibniz in the 1660s, then it led to the development of many important power series in analysis. Later, the theory of infinite series was thoroughly developed and used to work out many significant problems that had eluded solutions with any other approach in mathematics and various fields.

In this section, we state the formal definition of the power series in one complex variable. The definition is adopted from the standard texts of Ahlfors [6, p. 38], Agarwal [5, p. 151], Ponnusamy and Silverman [359, p. 153] and other references which are cited therein. We start with the definition of series as follows.

Definition 1 Let $(a_0, a_1, a_2, ...)$ be an infinite sequence. Then, the sum

$$\sum_{k=0}^{\infty} a_k = a_0 + a_1 + a_2 + \cdots, \qquad (1.2)$$

is called an infinite series or simply a series.

Definition 2 A power series is an infinite series of the form

$$\sum_{k=0}^{\infty} a_k (z - z_0)^k = a_0 + a_1 (z - z_0) + a_2 (z - z_0)^2 + \cdots,$$
 (1.3)

where $z, z_0 \in \mathbb{C}$, $a_k, k \in \{0, 1, 2, ...\}$ represents the coefficient of the k-th term of the series, z_0 is an any fixed point in \mathbb{C} and z varies around z_0 .

If z_0 is equal to zero (i.e., $z_0 = 0$), then the power series (1.3) takes the simpler form as follows:

$$\sum_{k=0}^{\infty} a_k z^k = a_0 + a_1 z + a_2 z^2 + \cdots {1.4}$$

 $\mathbf{5}$

The series (1.3) is often called the power series about z_0 or the power series with respect to the center at z_0 . It usually arises as the *Taylor series* of some known functions [4, p. 65-89]. In other words, many fundamental functions in a real or complex variable can be represented as the Taylor series expansion (as we shall show later in Section 1.3), whereas, the series in (1.4), which involves the simple powers of z, is known as *Maclaurin's series*. It is clearly seen that Maclaurin's series (1.4) is the Taylor series (1.3) about the origin. The simplest form of the power series is called the *geometric series*, namely

$$\sum_{k=0}^{\infty} z^k = 1 + z + z^2 + \cdots$$
 (1.5)

It is the special case of Maclaurin's series (1.4), where the coefficients $a_k = 1$ for all $k \in \{0, 1, 2, ...\}$.

The power series such as the geometric, Maclaurin and Taylor series, are frequently used in mathematical analysis. They provide powerful ways of representing some of the most important functions in a wide range of mathematics, physics, engineering, computer science and other fields. The crucial properties such as the algebraic operations (including the differentiation and integration) that can be done much more readily on the power series form, thus make such series particularly easy to study. In analysis, the solution of linear and nonlinear of ordinary or partial differential equations are often raised in the form of power series (see for instance [2], [33], [100], [113], [163], [200], [276]). Power series also arise under the name of generating function in the study of Combinatorics Theory (see [92], [97, Chapt. 2], [178]), while, the concept of p-adic number, which is closed related to the power series, occurs in the field of Number Theory (see [28], [199], [413]).

Much physical modelling requires mathematical approximations and often involves the series expansions. This method plays an important role in many

diverse areas of applied mathematics including physics and engineering. Particularly, the Taylor series expansions are widely used for approximating functions with a large or a small parameter. Hence, it always can be found in the study of the wave equations (see [258], [290]), the dynamical systems (see [38], [94], [177]), the kinetics equations (see [242], [315], [325], [326]), the quantum mechanics (see [230], [247], [424, p. 96-104]) and the neural networks (see [231], [417]). In addition, the power series method is simpler and it is able to reduce the complicated linear or nonlinear equations in order to find their exact solutions.

The applications of the power series also can be found in the field of computer sciences as well. The excellent way of representing some complicated functions, and also be able to serve the quick approximation and evaluation of functions at the given points, makes the power series a powerful tool in many computer algorithms developments. This means that, the power series expansions serve as a workhorse in many computer algorithms, in which the program runs faster or maybe even is possible. In fact, the numerical algorithm developed by using the power series expansion is much simpler, easier and more efficient with less error determination [205]. In other words, the approximation technique using the power series allows the computer algorithm to approximate functions with more speed, and in a short computational time with a high degree of accuracy (see [251], [256], [257], [275], [323]).

Power series are also useful in chemistry as well as in mathematics, physics, engineering and computer science for a number of reasons. First, the power series in truncated polynomial form provide an excellent tool for fitting experimental data, when there is no model formula available (see [95, Chap. 2], [96, Chap. 1], [258], [290]). Second, the great needs of the power series is for geometric optimization of physical systems: for instance, the quasi-newton method makes use of a two variables Taylor series to approximate the equilibrium geometry of a cluster of atoms [76]. In the field of economics and finance, the geometric series are widely used to represent the present and future value of an annuity³, and to estimate the present value of expected stock dividends or the terminal value of security (see [29], [197], [226, p. 175], [444]). For other applications of the power series in the Theory of Financial, see ([156], [184, p. 416-423], [367, p. 186-191], [425]).

³ is a series of payments made at fixed intervals of time.

In general, the importance of the power series representation is because of its extensive and very effective applications in various fields of pure and applied mathematics. Their effectiveness in error determination, function optimization and definite integral resolution are the evidence of the power series being an enormous tool in physical sciences and computational science, as well as an effective way of representing complicated functions in mathematical analysis.

1.1.2 Inequalities Theory

In [414], Tanner pointed out that "Mathematics begins with inequality". He said in a common language, the basic ideas of "more" and "less" are mathematical ideas of unequal and they are more primitive than the ideas of equal and numerals. However, these basic ideas were not symbolized until the mathematical sign of equality "=" appeared in the 16th century, and the sign of inequalities "<" and ">" in the 17th century. Tanner [414] also believes that to find an equality in a real practice is much more difficult than inequality. His argument seems to be relevant to the statement from the pioneers in Inequalities Theory, Beckenbach and Bellman [55, p. 3], who say in their book "An Introduction to Inequalities" that the fundamental results of mathematics are often inequalities rather than equalities.

The Theory of Inequalities plays an important role in mathematical analysis for finding approximations of numbers, functions, integrals, etc. Around the beginning of the 20th century, numerous inequalities were derived and applied in various branches of mathematics as well as in science and engineering. The pioneers in this field were credited to Hardy, Littlewood and Pólya [185], who transformed this field of inequalities from a collection of isolated formulas into a systematic discipline through the book "Inequalities", which was originally published in 1934. This book is the first devoted solely to the subject of inequalities, which presents a lot of fundamental ideas, problems, results, methods of proving and applications for a large variety of classical and new inequalities. Hence, it has had much influence on research in various branches of mathematical analysis. Some other notable books in this area are "An Introduction to Inequalities" by Beckenbach and Bellman [55], which was published in 1961 and the early one of Mitrinovič [316], "Analytic Inequalities" in 1970. These major books have made

considerable contributions to the field of Inequalities Theory and serve as important references for many mathematicians and to those who use analysis seriously. Since them, an enormous amount of effort has been devoted to the discovery of new types of inequalities and their applications in many part of mathematical analysis, see for instance the books of Mitrinovič, Pečarić and Fink [318], Kapur [237], Kazarinoff [239], Garling [161] and recently, Bullen [79], Cerone and Dragomir [91].

Nowadays, a large number of classical and new inequalities can be found in the literature. The classical inequalities are those associated with the names Gauss, Cauchy, Schwarz, Bunyakovsky⁴, Young, Hölder, Hilbert, Hardy, Littlewood, Pólya, Minkowski, Jensen, Čebyšev⁵, Bessel, Ostrowski, Grüss, Hadamard and others. These type of inequalities have been explored by various researchers, hence new inequalities are created in which to provide their generalizations, refinements, improvements, etc., see for instance ([9], [19], [128], [157], [158], [160], [192], [271], [299], [302], [310], [317], [445], [449]) and the references which are cited therein. The studies related to these classical inequalities remain the active fields and have grown into substantial areas of research with many important applications in various fields of Modern Mathematics.

One of the most important inequalities in analysis, that has attracted the great attention of a large number of researchers in the last few decades, is the famous Cauchy-Bunyakovsky-Schwarz inequality. This inequality is also known in the literature as Cauchy's, Schwarz's or Cauchy-Schwarz's inequality, and it has a long history connected with these three names of the famous mathematicians. Starting with the French mathematician A. L. Cauchy [88], who established the elementary form of the Cauchy-Bunyakovsky-Schwarz inequality for real numbers. His inequality, later called Cauchy's inequality, is contained in his book "Cour d'Analyse Algébrique", published in Paris in 1821. Then, in 1859, the Cauchy's student, V. Y. Bunyakovsky [81] derived a corresponding inequality for classical integrals. After around three decades later, in 1888, the German mathematician H. A. Schwarz [388] independently rediscovered and obtained the same result of Cauchy's inequality for Lebesque integrals, without any reference

⁴There are several different spellings of Bunyakovsky's name can be found in the literature, for example Bunyakovskii, Buniakowski, Bounjakowsky, etc., see MatSciNet.

 $^{^5{\}rm The}$ other spellings for Čebyšev's name are Chebyshev, Čebišev, Tchebycheff, etc., see MatSciNet.

to Bunyakovsky's work. Meanwhile, in the year 1885, Schwarz [388] had already obtained the generalization of Cauchy's inequality in inner product spaces, which is popularly known in the literature as Schwarz's or Cauchy-Schwarz's inequality. The classical Cauchy inequality and its generalizations for integral and in inner product spaces, therefore shows up under the *Cauchy-Bunyakovsky-Schwarz inequality*. For a short history of these type inequalities, see ([387, p. 64-70], [407, p. 10-12]).

The generalization of the classical Cauchy-Bunyakovsky-Schwarz inequality for sums was established by O. L. Hölder [201] in 1889, and it is called *Hölder's inequality*. Although this inequality was first derived in 1888 by L. J. Rogers [378], then it was proved in another way a year later by Hölder through his monograph "Über einen Mittelwertsatz". Hölder's inequality, which is sometimes referred as Rogers's inequality, was built around the two positive real numbers p and q with the conditions that p > 1 and 1/p + 1/q = 1. The Cauchy-Bunyakovsky-Schwarz inequality is the special case of Hölder's inequality when p = q = 2. There are several proofs of Hölder's inequality can be found in the literature, where one of them is the proof via a standard Young's inequality for the product of two real numbers.

Jensen's inequality is another important result in Real and Complex Analysis. It was developed by Danish mathematician J. L. W. V. Jensen [223] in 1906 based on the concept of convexity of functions (see also [222]). Due to the fact that Jensen's inequality provides a source of deriving for many classical results in analysis such as the triangle inequality, arithmetic mean-geometric mean inequality, Hölder's inequality, Minkowski's inequality, etc., through judicious choice of convex or concave function, this type of inequality thus becomes one of the landmarks in the study of Inequalities Theory, and in Convex Analysis as a whole.

The classical inequalities of the Cauchy-Bunyakovsky-Schwarz, Jensen, Young and Hölder inequalities play a crucial role in the Theory of Inequalities. They have attracted the attention of a large number of researchers, and stimulated new research directions and influenced various aspects of mathematical analysis and their applications.

1.1.3 Univalent Function Theory

In Complex Analysis, a geometric function is a function whose range describes certain geometries. A Geometric Function Theory is one of the branches of Complex Analysis, which deals and studies the geometric properties of complex analytic functions. The Theory of Univalent Function is one of the most important subjects in the Geometric Function Theory, where the primary interest in this field is the functions that are analytic⁶ and univalent in a certain complex domain. This theory was founded around the turn of the 20th century, when the first important paper appeared in this area by Koebe [253] in 1907. It followed by Alexander ([15], [16]) and Bieberbach [67] in 1915 and 1916, respectively. Koebe [253] introduced the notion of univalent mapping through his monograph 'Über die uniformisierung beliebiger analytischer kurven', which gives a great contribution to the origin of the Univalent Function Theory.

The concept of univalent mapping has been explored by Alexander [16] in his PhD dissertation entitled 'Functions Which Map the Interior of the Unit Circle Upon Single Region' at Princeton University, USA in 1915. Since then, the study of Univalent Function Theory has attracted the great attention and efforts from many mathematicians: for instance, Lindelöf [287], Plemelj [356], Gronwall [174], Löwner ([293], [294]), Privalov [362], Nevanlinna [332], Landau [274] and Littlewood [288] are among others, who continue the study in this area around those decades (see also [8], [60], [145], [150], [181], [190], [264], [309], [343], [384], [385], [397], [398]) and ([39], [188], [207], [229], [243], [344], [348], [403], [426]) for some further studies and recent results in this field.

Many expositions of a multitude of interesting results in the Theory of Univalent Functions have been published by Montel [324] in his monograph entitled "Leçons sur les Fonctions Univalentes ou Multivalentes", which was published in Paris in 1933. He noted in [324] that the origin of the Theory of Univalent Function comes from the general problem of conformal mapping, which maps the simply connected domain to another simply connected domain in a complex plane \mathbb{C} . which is a powerful impetus to the "Verzerrungssats" of Koebe's and Bieberbach's work. The volumes of material in the filed of Univalent Function

 $^{^6{}m The}$ terms holomorphic and regular are also frequently used for analytic complex-valued functions.

Theory increased rapidly around the middle of the 20th century. According to the two bibliography manuscripts, which were prepared by Bernardi [61] in 1966 and 1975 respectively, there have been 570 authors extensively working in the field of Univalent and Multivalent Functions Theory since 1907. Hence, they were able to contribute 1694 research articles and monographs, with around 1563 of the research papers published within the following ten years. These two volumes of bibliography have provided valuable references to the many researchers on this subject. The "Les Functions Multivalents" is another remarkable survey monograph concerned with univalent and multivalent functions was produced in Paris, in 1983 by Biernacki [68].

Along with these, there are also many texts contributing to the field of Univalent Functions Theory and its related topics. For instance, the books of Goluzin [175], Nehari [330], Jenkins [221], Hayman [189] and Ahlfors [7], which cover this field in great detail by providing comprehensive surveys and some open problems. Other recent books that cover this subject in depth are the three different books entitled "Univalent Functions", and produced individually by Pommerenke [357] in 1975, Goodman [169] and Duren [144] in 1983. Other useful monographs that contribute to this area are (see for instance [183], [308], [320], [406]).

One of the most classical and remarkable results in Geometric Functions Theory is *Riemann Mapping Theorem*. Earlier than Koebe, in 1851, Riemann [374] has provided this important result in Geometric Function Theory by proving that there always exists a unique analytic function, which maps a simply connected domain onto another simply connected domain in a complex plane. Koebe [253] initiated the study of univalent functions in 1907, then in view of the Riemann Mapping Theorem, the study of the properties of analytic and univalent functions began.

The class A of all functions that are analytic in a simply connected domain in a complex plane and satisfy certain conditions of normalization, was introduced. Further, the the class of functions $S \subset A$ consisting of all analytic, univalent and normalized functions in the domain of definition, becomes the center of the study of Univalent Function Theory. This class of functions has attracted great attention from various researchers around the world. In order to fulfill the requirement of the Riemann Mapping Theorem, the domain of definition is

generally selected as the open unit disk $D = \{z \in \mathbb{C} : |z| < 1\}$, which is one of the simplest examples of simply connected domain in a complex plane. All of the analytic functions are conformal, in the sense that their angles are preserved under the mapping. Thus, we simply understand that all the analytic functions in the class S, map the unit disk D conformally and one-to-one onto another simply connected domain in a complex plane. The most interesting about this function is that, it can be represented by a power series expansion with real or complex coefficients and convergent on the unit disk D.

The crucial property of such functions is that, the image domain f(D) will describe various nice geometries, such as starlike (or star-shaped), spirallike, close-to-star, convex, close-to-convex, etc. The functions map the unit disk Donto those geometrical domains with some interesting properties, which can be characterized in several ways, such as by analytical and geometrical conditions. The subclasses of S; such as the class of starlike, convex, close-to-convex functions, etc., were defined and some properties of functions in this subclasses exhaustively studied by several authors in the second half of the 20th century. In [144, p. 40-44], Duren provided an analytical description of functions in the class of starlike and convex (see also Pommerenke [357, p. 42-44]). Kaplan [235] introduced the class of close-to-convex functions and studied their properties of functions in this class. Some more general subclasses of S, such as the class of λ —convex, the class of close-to-convex of order α and type β , the class of Bazilević functions, etc., have been studied by numerous authors (see for instance [54], [108], [110], [164], [182], [219], [206], [309], [319], [328], [336], [376], [400], [418]). Libera [283] introduced an integral operator and investigated some properties of starlike functions under this operator. Meanwhile, Sălăgean [383] studied the class of analytic functions defined by differential operator. These works opened new ways to study the operators in Geometric Function Theory. Hence, afterwards, many studies have been conducted by many authors, which attempt to generalize and define various subclasses of analytic functions involving the integral and differential operators (see for instance [62], [337], [343], [344], [382], [383]).

Generally, an analytic and univalent function, which is defined in a unit disk of a complex domain, lie in a complex plane by various properties and conditions. Therefore, a general problem in the study of Geometric Function Theory is to

investigate the properties of such functions in various subclasses of S. As a part of the study of geometric functions, the necessary and sufficient conditions for the existing functions to be univalent in the unit disk, is one of the fundamental studies in the Theory of Univalent Function.

1.2 Basic Properties of Power Series

One of the most important applications of the power series is that, it provides a useful way of representing some fundamental functions in real and complex analysis. Therefore, the convergent property of the power series plays an important role to guarantee that it converges back to the given function in its domain. Generally, the power series (1.4) may be converge or diverge in a certain domain in a complex plane.

Theorem 3 ([5, p. 38]) Suppose that the power series is given by (1.4). Then, one of the following happens:

- (a) The series (1.4) converges only when z = 0.
- (b) The series (1.4) converges absolutely for all $z \in \mathbb{C}$.
- (c) There is a number R, R > 0 such that the series (1.4) converges absolutely for all $z \in \mathbb{C}$ with |z| < R, and it diverges for all |z| > R.

The number R described in Theorem 3 is known as the radius of convergence of the power series (1.4). Obviously, by allowing R=0 in the case (i) and $R=\infty$ in the case (ii) as above, we can consider that every power series has a positive radius of convergence. The set of all $z \in \mathbb{C}$ such that |z| < R is called the domain or disk of convergence of the series (1.4). The set of all $z \in \mathbb{C}$ with |z| = R is called the circle of convergence of the series (1.4), and nothing is claimed about the convergence on this circle, only the continuity of the series along the line segment from the origin to the point z.

Power series are classified not only whether they converge or diverge, but also by the properties of the terms a_k (i.e., absolute or conditional convergence) and by the types of convergence of the series (i.e., pointwise or uniform). The following results describe various properties of the power series.

Theorem 4 The power series (1.4) is said to be convergent absolutely if the series of absolute values, i.e., $\sum_{k=0}^{\infty} |a_k z^k|$ converges. It is said to be conditionally convergent (or semi convergent) if the series (1.4) is convergent but not absolutely convergent.

Theorem 5 ([359, p. 174]) If the power series (1.4) converges when $z = z_1$ $(z_1 \neq 0)$, then it is absolutely convergent at each point z in the open disk $|z| < |z_1|$.

Theorem 6 ([5, p. 151]) If z_1 is a point inside the circle of convergence |z| = R of the power series (1.4), then the series converges absolutely and uniformly in the close disk $|z| < |z_1|$.

Theorem 7 ([166, p. 28]) Let (1.4) be a power series with radius of convergence R, R > 0. If $0 < \rho < R$, then the power series (1.4) converges normally, absolutely and uniformly for $|z| \le \rho$. The power series (1.4) diverges for $|z| > \rho$.

Theorem 8 ([166, p. 28]) Let (1.4) be a power series and convergent on |z| < R, R > 0. Then the function f(z) which is represented by the power series (1.4) continuous all $z \in \mathbb{C}$ with |z| < R.

Several methods to determine the convergent of the infinite series are discussed in the literature, for instance, the comparison test, the integral test, the ratio test, the roots test, the Cauchy-Hadamard formula, etc., (see [5, p. 152], [70, p. 451-461], [98, p. 125-134], [166, p. 31], [358, p. 331-336]). The *Ratio Test* is commonly used, fairly simple and applicable in the case of the power series with real or nonnegative coefficients. The following result provides a rule for the convergence test of the power series by using this test. The power series given by (1.4) converges absolutely if

$$\lim_{k \to \infty} \left| \frac{a_{k+1} z^{k+1}}{a_k z^k} \right| < 1, \tag{1.6}$$

provided that the limit exists. This property is equivalent to

$$|z| \le R = \lim_{k \to \infty} \left| \frac{a_k}{a_{k+1}} \right|,\tag{1.7}$$

where R is the radius of convergence of the power series (1.4). Thus, we can determine the value R of radius of convergence as well through this result (1.7).

Remark 9 Throughout this study, we mostly considered the simplest and the most widely used form of the power series (1.4), which is convergent in the open disk |z| < R, R > 0.

The algebraic operations (such as addition, multiplication, division), differentiation and integration, can be applied for the power series as well, where the resulting new power series are convergent inside of their common domain of convergence. Let, the two power series f(z) be of the form (1.4) and $g(z) = \sum_{k=0}^{\infty} b_k z^k$, which are convergent for all $|z| < R_f$ and $|z| < R_g$, respectively. Then, the following rules of algebra apply (see [4, p. 15], [5, p. 155], [26, p. 14], [93], [276], [359, p. 186-189]):

- (i) The two convergent power series are equal if their corresponding terms are identical. That is, if f(z) = g(z), wherever the two series converge, then $a_k = b_k$ for each $k \in \{0, 1, 2, ...\}$.
- (ii) The two or more power series can be combined through termwise by addition or subtraction, that is,

$$f(z) \pm g(z) = \sum_{k=0}^{\infty} (a_k \pm b_k) z^k,$$
 (1.8)

for all $z \in \mathbb{C}$ such that $|z| < \min\{R_f, R_g\}$. For a real number λ , the multiple of a series converges with the same radius of convergence and

$$\lambda f(z) = \sum_{k=0}^{\infty} \lambda a_k z^k. \tag{1.9}$$

(iii) If $g(z) \neq 0$ and $b_0 \neq 0$ in the disk $|z| < R_g$, then the quotient

$$\frac{f(z)}{g(z)} = \sum_{k=0}^{\infty} c_k z^k,\tag{1.10}$$

where the coefficients satisfy the equation $a_k = \sum_{j=0}^{\infty} b_{k-j} c_j$, and the radius of convergence of the series (1.10) is $R = \min\{R_f, R_q\}$.

(iv) The Cauchy products of f(z) and g(z) is defined by

$$f(z)g(z) = \sum_{k=0}^{\infty} \left[\sum_{j=0}^{k} a_j b_{k-j} \right] z^k = \sum_{k=0}^{\infty} \left[\sum_{j=0}^{k} a_{k-j} b_k \right] z^k, \tag{1.11}$$

which converges with the radius of convergence $R = \min\{R_f, R_g\}$. The double sum of the infinite series of the coefficients,

$$\sum_{k=0}^{\infty} \sum_{j=0}^{k} a_j b_{k-j} = \sum_{k=0}^{\infty} \sum_{j=0}^{k} a_{k-j} b_k,$$
(1.12)

is known as the *convolution* of the sequences $\{a_k\}$ and $\{b_k\}$, $k \in \{0, 1, 2, ...\}$. Some other basic rules of the multiplication of the series of coefficients are

(a)
$$c \sum_{k=0}^{\infty} a_k = \sum_{k=0}^{\infty} (ca_k), \quad c \in \mathbb{R},$$

(b) $\left(\sum_{j=0}^{\infty} a_j\right) \left(\sum_{k=0}^{\infty} b_k\right) = \sum_{j=0}^{\infty} a_j \left(\sum_{k=0}^{\infty} b_k\right) = \sum_{j=0}^{\infty} \sum_{k=0}^{\infty} a_j b_k,$
(c) $\left(\sum_{j=0}^{\infty} a_j\right)^2 = \sum_{j=0}^{\infty} \sum_{k=0}^{\infty} a_j a_k.$ (1.13)

(v) For a convergent power series, it always converges to a continuous function. Thus, the power series can always be differentiated and integrated termwise within in its domain of convergence, that is [5, p. 153],

$$\frac{df(z)}{dz} = \sum_{k=1}^{\infty} k a_k z^{k-1} = \sum_{k=0}^{\infty} (k+1) a_{k+1} z^k, \quad |z| < R_f$$
 (1.14)

and

$$\int_{0}^{z} f(t)dt = \sum_{k=0}^{\infty} \frac{a_k}{k+1} z^{k+1} = \sum_{k=1}^{\infty} \frac{a_{k-1}}{k} z^k, \quad |z| < R_f, \tag{1.15}$$

by assuming that the summation \sum , differentiation $\frac{d}{dz}$ and integration \int are interchanged. Clearly, both of the newly derived series in (1.14) and (1.15) have the same radius of convergence as the original series (1.4).

(vi) Let f(z) and g(z) be the two power series, which are both convergent in the unit disk |z| < 1. Then, the *convolution* or *Hadamard product* of f(z) and g(z) is denoted by f * g, and it is defined by the power series

$$(f * g)(z) = \sum_{k=1}^{\infty} a_k b_k z^k, \quad |z| < 1,$$
 (1.16)

where, the geometric series

$$h(z) = \sum_{k=1}^{\infty} z^k = \frac{z}{1-z}, \ |z| < 1$$
 (1.17)

is the identity function under this convolution (1.16), i.e., (f * h)(z) = f(z). The *integral convolution*, which is denoted by $f \circledast g$, is defined by

$$(f \circledast g)(z) = \sum_{k=1}^{\infty} \frac{a_k b_k}{k} z^k = \int_0^z \frac{h(t)}{t} dt,$$
 (1.18)

see ([144, p. 246-247], [357, p. 49]).

1.3 Power Series Representations

There are many fundamental real or complex functions that can be represented by the power series expansions with real or complex coefficients. In this section, we give some examples of the functions and their corresponding power series expansions in complex variable.

(1) Geometric series and their variants:

$$\frac{1}{1-z} = \sum_{k=0}^{\infty} z^k, (1.19)$$

$$\frac{1}{1+z} = \sum_{k=0}^{\infty} (-1)^k z^k, \tag{1.20}$$

$$\frac{z}{1-z} = \sum_{k=1}^{\infty} z^k, \tag{1.21}$$

$$\frac{1}{(1-z)^2} = \sum_{k=0}^{\infty} kz^{k-1}, \tag{1.22}$$

$$\frac{z}{(1-z)^2} = \sum_{k=0}^{\infty} kz^k, (1.23)$$

$$\sqrt{1+z} = \sum_{k=0}^{\infty} \frac{(-1)^k (2k)!}{2^{2k} (1-2k) (k!)^2} z^k,$$
 (1.24)

for all $z \in \mathbb{C}$ such that |z| < 1, where k! denotes the factorial function.

(2) Exponential functions and their variants:

$$\exp(z) = \sum_{k=0}^{\infty} \frac{1}{k!} z^k$$
, for any $z \in \mathbb{C}$, (1.25)

$$\exp(-z) = \sum_{k=0}^{\infty} \frac{(-1)^k}{k!} z^k, \text{ for any } z \in \mathbb{C},$$
 (1.26)

$$\exp(\sqrt{z}) = \sum_{k=0}^{\infty} \frac{1}{(2k)!} z^k, \text{ for any } z \in \mathbb{C},$$
 (1.27)

$$z\exp(z) = \sum_{k=0}^{\infty} \frac{1}{(k-1)!} z^k, \text{ for any } z \in \mathbb{C},$$
 (1.28)

$$(z+z^2)\exp(z) = \sum_{k=0}^{\infty} \frac{k}{(k-1)!} z^k, \text{ for any } z \in \mathbb{C}.$$
 (1.29)

(3) Natural logarithmic functions:

$$\ln\left(\frac{1}{1+z}\right) = \sum_{k=1}^{\infty} \frac{(-1)^{k+1}}{k} z^k, \text{ for } |z| < 1, \tag{1.30}$$

$$\ln\left(\frac{1}{1-z}\right) = \sum_{k=1}^{\infty} \frac{1}{k} z^k, \text{ for } |z| < 1.$$
 (1.31)

(4) Trigonometric and the inverse of trigonometric functions:

$$\sin(z) = \sum_{k=0}^{\infty} \frac{(-1)^k}{(2k+1)!} z^{2k+1}, \text{ for all } z \in \mathbb{C},$$
 (1.32)

$$\cos(z) = \sum_{k=0}^{\infty} \frac{(-1)^k}{(2k)!} z^{2k}, \quad \text{for all } z \in \mathbb{C}, \tag{1.33}$$

$$\tan(z) = \sum_{k=0}^{\infty} \frac{(4)^k (4^k - 1) |B_{2k}|}{(2k)!} z^{2k-1}, \text{ for } |z| < \frac{\pi}{2}, \quad (1.34)$$

$$\arcsin(z) = \sum_{k=0}^{\infty} \frac{(2k)!}{2^{2k} (k!)^2 (2k+1)} z^{2k+1}, \text{ for } |z| \le 1, \qquad (1.35)$$

$$\arccos(z) = \frac{\pi}{2} - \sum_{k=0}^{\infty} \frac{(2k)!}{2^{2k} (k!)^2 (2k+1)} z^{2n+1}, \text{ for } |z| < 1, (1.36)$$

$$\arctan(z) = \sum_{k=0}^{\infty} \frac{(-1)^k}{2k+1} z^{2k+1}, \text{ for } |z| < 1,$$
 (1.37)

where B_{2k} denotes the *Bernoulli numbers*⁷ (see [25, p. 12], [102, p. 201], [305, p. 50-53]).

(5) Hyperbolic and the inverse of hyperbolic functions:

$$\sinh(z) = \sum_{k=0}^{\infty} \frac{1}{(2k+1)!} z^{2k+1}, \text{ for all } z \in \mathbb{C},$$
 (1.38)

$$\cosh(z) = \sum_{k=0}^{\infty} \frac{1}{(2k)!} z^{2k}, \text{ for all } z \in \mathbb{C},$$
(1.39)

$$\tanh(z) = \sum_{k=1}^{\infty} \frac{4^k (4^k - 1) B_{2k}}{(2k)!} z^{2k-1}, \quad \text{for } |z| < \frac{\pi}{2}, \quad (1.40)$$

$$\coth(z) = \sum_{k=1}^{\infty} \frac{4^k B_{2k}}{(2k)!} z^{2k-1}, \quad \text{for } |z| < \pi, \tag{1.41}$$

$$\operatorname{arsinh}(z) = \sum_{k=0}^{\infty} \frac{(-1)^k (2k)!}{4^k (k!)^2 (2k+1)} z^{2k+1}, \quad \text{for } |z| < 1, \quad (1.42)$$

$$\operatorname{artanh}(z) = \sum_{k=0}^{\infty} \frac{1}{2k+1} z^{2k+1}, \quad \text{for } |z| < 1.$$
 (1.43)

Instead of defining the trigonometric and hyperbolic functions of sine, cosine, sinh and cosh by their power series expansions, it is possible and might be useful to define them directly in terms of the exponential functions as follows:

$$\sin(z) = \frac{1}{2i} (e^{iz} - e^{-iz}), \qquad \cos(z) = \frac{1}{2} (e^{iz} + e^{-iz}),$$

$$\sinh(z) = \frac{1}{2} (e^{z} - e^{-z}), \qquad \cosh(z) = \frac{1}{2} (e^{z} + e^{-z}),$$
(1.44)

⁷The first few Bernoulli numbers are $B_0 = 1$, $B_1 = \pm 1/2$, $B_2 = 1/6$, $B_3 = 0$, $B_4 = -1/30$, $B_5 = 0$, $B_6 = 1/42...$, which are derived from the recursion formula: $B_{2k} = \frac{1}{2} - \frac{1}{2k+1} \sum_{m=0}^{k-1} \binom{2k+1}{2m} B_{2m}$ for $k \ge 1$ and $B_{2k+1} = 0$, $k \ge 1$..

for all $z \in \mathbb{C}$.

Some of the above series can be found in the Handbooks of Abramowitz and Stegun [4, p. 68-69, 81-85] and Bronshtein et al. [77, p. 698-699]. The concept of the power series is also widely used in the Theory of Special Functions, in which, are defined the special functions of Polylogarithm, hypergeometric, Bessel and modified Bessel functions, etc. that we will discuss later in Chapter 5. Nowadays, there are numerous aids of modern computer software, such as MAPLE, MATHEMATICA, MATLAB, etc. that can be easily used in finding the series expansion of a given function.

Some numerical infinite series, which are given in the following might be useful for the subsequence chapters:

(a)
$$\sum_{k=0}^{\infty} \frac{1}{2^k} = 2$$
, (b) $\sum_{k=0}^{\infty} \frac{(-1)^k}{2^k} = \frac{2}{3}$,
(c) $\sum_{k=1}^{\infty} \frac{(-1)^{k-1}}{k} = \ln 2$, (d) $\sum_{k=0}^{\infty} \frac{1}{(k+1)^2} = \sum_{k=1}^{\infty} \frac{1}{k^2} = \frac{\pi^2}{6}$,
(e) $\sum_{k=1}^{\infty} \frac{1}{k^4} = \frac{\pi^4}{90}$, (f) $\sum_{k=1}^{\infty} \frac{(-1)^{k-1}}{(2k-1)} = \frac{\pi}{4}$
(g) $\sum_{k=1}^{\infty} \frac{1}{k^6} = \frac{\pi^6}{945}$, (h) $\sum_{k=1}^{\infty} \frac{1}{(nk)^2} = \frac{\pi^2}{6n^2}$, $n > 0$.

The following section discusses the main motivation for writing this dissertation.

1.4 Motivation

In the early 1820s, Cauchy [88] published his famous inequality involving two sequences of real numbers, which is popularly known in the literature as Cauchy's inequality. Long after Cauchy, in 1885, Schwarz [389] provided the generalization of Cauchy's inequality in an inner product space, which is later known as Schwarz's or Cauchy-Schwarz's inequality. In 1888, Bunyakovsky interfered and proved the corresponding version of Cauchy's inequality for integrals. Since

them, the inequalities, which are now referred to the Cauchy-Bunyakovsky-Schwarz type inequalities, have attracted the attention of a large number of researchers, and hence, have stimulated new research directions and influenced various aspects of mathematical analysis and applications.

As noted in the celebrated monograph "Analytic Inequalities" written by Mitrinović [316] in 1970, various remarkable results related to the celebrated Cauchy-Bunyakovsky-Schwarz inequality were established. Also, a large number of research papers have been published since 1970 providing various improvements, refinements and generalizations of this type of the famous inequality; see for instance [19], [20], [78], [116], [128], [138], [147], [269], [302], [408], [445] and the references cited therein. All those results that have been established in the literature, are mainly concerned with the finite sequences and finite sums. Consequently, much effort has been made by several authors to extend the inequalities to infinite sequences and sums, functions and integrals, in order to provide more general results that are useful in various fields of Modern Mathematics and its applications (see [22], [49], [52], [173]).

A revival of the interest in this area occurred in recent years with several results in analytic inequalities involving functions defined by the convergent power series. The approach of the power series functions in the Inequalities Theory has been explored by several researchers, see particularly [89], [122], [123], [173]. This research is mainly motivated by Cerone and Dragomir's work in [89] in 2007, who established some inequalities concerning functions defined by the convergent power series with real or nonnegative coefficients by utilising a refinement of the Cauchy-Bunyakovsky-Schwarz inequality, which is known in the literature as the de Bruijn inequality. They also obtained some particular inequalities by applying the results for fundamental functions of interest such as the exponential, logarithm, trigonometric and hyperbolic functions. Applications for special function such as polylogarithm functions are provided as well in [89].

In this study, we employ the techniques developed in [89], to derive new and better inequalities for functions defined by the power series with real or nonnegative coefficients, and convergent on an open disk. Utilising the classical results that have been available in the literature such as Buzano's inequality [82] and Schwarz's result in inner product spaces due to Dragomir [118], we provided

some refinements and improvements of some known results, which are related to the celebrated Cauchy-Bunyakovsky-Schwarz inequality for functions defined by the power series. Besides Buzano's and Schwarz's results, the other crucial tools that have been used for our investigations are Young's, Hölder's and Jensen type inequalities, as well as their refinements, reverses and counterparts. Particular examples that are related to some fundamental real or complex functions such as exponential, logarithm, trigonometric and hyperbolic functions are presented. Some applications for special functions such as polylogarithm, hypergeometric, Bessel and modified Bessel functions of the first kind are presented as well in the first part of this dissertation.

Geometric Function Theory is the branch of Complex Analysis, which deals with the geometric properties of analytic functions, founded around the turn of the 20th century. The cornerstone of this theory is the Theory of Univalent Functions, which is mainly concerned with the analytic and univalent functions f(z) in a unit disk, and normalized by the conditions f(0) = 0 and f'(0) = 1. It is a fact that all functions in the class S, which are analytic and univalent function in a unit disk can be represented by convergent power series with real or complex coefficients.

Conjecture Bieberbach [67] is one of the major problems in the field of Geometric Function Theory, which asserts the modulus of the k-th coefficient in the power series expansion of functions in the class S is less than or equal to k for all $k \geq 2$. In spite of this famous coefficient estimate that was completely solved by de Brange [75] in 1984, it suggests various approaches and directions for the study of Univalent Function Theory, and Geometric Function Theory as a whole. Hence, various subclasses of analytic and univalent functions such as starlike, spirallike, convex, close-to-convex, etc., have been introduced by numerous researchers in the last few decades and developed many interesting properties of functions in these classes.

In the other direction, the new subclasses of analytic functions, which are defined involving operators were introduced. The Sălăgean differential operator D^n , $n \in \mathbb{N}_0$, is one of the famous operators, which was established by Sălăgean in [383]. In [343], Opoola introduced and studied the new class $T_n^{\alpha}(\beta)$, $\alpha > 0$, $0 \le \beta < 1$, $n \in \mathbb{N} \cup \{0\}$ of analytic functions, which satisfies certain conditions

involving the Sălăgean differential operator. Regarding this class, we investigate in second part of this dissertation, some properties of functions in this class. In order to stimulate further investigation in this field and the related topics, we introduce the class $\widetilde{T}_n^{\alpha}(\beta)$ and $T_{n,\lambda}^{\alpha}(\beta)$, $\lambda \geq 1$, and provide new properties and coefficient inequalities of functions in these classes. The properties of the generalized Sălăgean operators $D^n f^{\alpha}$ and $D_{\lambda}^n f^{\alpha}$ are also established. We conclude the second part of this dissertation with some results on the Fekete-Szegö theorem concerning analytic and univalent functions.

Functions in the class of analytic and univalent functions S and its subclasses, which map the unit disk or half-plane conformally onto another simply connected domain in a complex plane, have found many applications in engineering, physics, electronics, medicines and other branches of applied mathematics (see [153], [238]).

1.5 Thesis Outline

This dissertation is partitioned into two main parts, where each part is further divided into several chapters that have a number of sections and subsections. Overall, this dissertation contains eight chapters including Chapter 1 as above.

In the first chapter, we provide a brief introduction to the power series, Inequalities Theory and Univalent Function Theory. The power series that are the most important elements throughout this dissertation, are also discussed in this chapter. Some examples of analytic functions with their corresponding power series expansions are given as well for useful references in the subsequent chapters.

Part I consists of Chapter 2 - Chapter 5, and mainly deals with Analytic Inequalities in Real and Complex Analysis. The main purpose of this part is to establish some inequalities concerning the power series with real or nonnegative coefficients. Chapter 2 provides some basic concepts and introductory materials in Inequalities Theory including the triangle, means and convexity inequalities. Some of the most important and classical inequalities such as the Cauchy-Bunyakovsky-Schwarz, Young's, Hölder's and Jensen type inequalities

1. Introduction 24

are emphasized in order to provide an essential background for the subsequent chapters.

In chapter 3, we derive new and better inequalities for functions defined by the power series, which are related to the celebrated Cauchy-Bunyakovsky-Schwarz inequality. In particular, we provide some refinements and improvements of the Cauchy-Bunyakovsky-Schwarz inequality by utilising Buzano's and Schwarz's results in inner product spaces, and the technique based on the continuity properties of modulus. Particular inequalities that are related to some fundamental functions of interest such as the exponential, logarithm, trigonometric and hyperbolic functions are obtained.

More inequalities for functions defined by the power series are established in Chapter 4. To develop some inequalities for the power series by utilising the standard of Young's inequality, the refinement and the reverse of Young's inequality, the Jensen type inequality and one of its reverses due to Dragomir and Ionescu [139], is the main purpose of this chapter. Applications for some fundamental functions of interest are also presented.

Chapter 5 is devoted to some inequalities for special functions. In this chapter, we employ some results established in Chapter 3 and Chapter 4 to obtain particular inequalities involving special functions such as polylogarithm, hypergeometric, Bessel and modified Bessel functions of the first kinds.

Part II consists of Chapter 6 - Chapter 7, devotes to the study of Univalent Function Theory. In this part, we mainly concern with the class of functions, which are analytic and univalent in a unit disk, and normalized by a standard conditions. To investigate some properties of certain subclasses of analytic and univalent functions is the main objective throughout this part.

Chapter 6 is written as a reference point for the subsequent chapter. Some fundamental facts and basic concepts regarding analytic and univalent functions are provided. We emphasize the class A of analytic functions and the class S of analytic, univalent and normalized functions in the unit disk D. Some basic properties and coefficient inequalities of functions in the class S and its subclasses are presented. We also present the well-known Sălăgean differential operators and their related subclasses at the end of this chapter.

1. Introduction 25

Chapter 7 presents some new properties and coefficients inequalities for functions in certain subclasses S. The properties of the generalized Sălăgean differential operator [383] are also given. Some results of the Fekete-Szegö functional concerning analytic and univalent functions are included at the end of this chapter.

Finally, we summarize the work of this dissertation in Chapter 8, which contains the summary and main achievements of the work.

The various theorems, corollaries, lemmas, propositions, remarks and examples are numbered in order, consecutively throughout the thesis, whereas, the equations are numbered consecutively within each chapter. The end of the proof of a theorem, corollary or lemma is indicated by a solid square ' \blacksquare '.

Part I ANALYTIC INEQUALITIES IN REAL AND COMPLEX ANALYSIS

Chapter 2

Elementary and Some Classical Inequalities

Chapter 2 provides some essential background for the subsequent chapters of the first part of this dissertation. This includes the introductory materials and basic inequalities in Real and Complex Analysis. The elementary inequalities such as triangle, means and convexity inequalities are discussed in Section 2.2. The basic properties of normed, linear and inner product spaces are also addressed.

We emphasize in Section 2.3 the basic inequalities of the classical Cauchy-Bunyakovsky-Schwarz type in real and complex numbers, and in inner product spaces. The corresponding version of the Cauchy-Bunyakovsky-Schwarz inequality for functions defined by the power series is also given. Some results related to these types of inequalities such as the de Bruijn inequality, the Buzano's and Schwarz's results are mentioned as a foundation for the next chapters. Young's inequality and its variants, and Hölder's inequality and its generalizations to the power series functions are briefly discussed in Section 2.4 and Section 2.5, respectively. Inequalities of the Jensen type and their reverses are given in Section 2.6. Some other well-known inequalities in analysis are provided at the end of this chapter.

All the results are given without proof. Some of their proof can be found in the literature, see for instance, the books of Beckenbach and Bellman [55], Bullen [79], Hardy, Littlewood and Pólya [185], Mitrinović [316], Mitrinović, Pečarić and

Fink [318] and the numerous textbooks devoted to the Inequalities Theory and their related topics.

2.1 Symbols and Notations

We start this chapter by introducing some of the symbols and notations that we shall use in this dissertation. First, let z, z_0 be two points in a complex plane \mathbb{C} and R is a positive real number. Then, the set of all points that satisfies the inequality $|z - z_0| < R$ is called the *open disk* (or simply the *disk*) with center z_0 and radius R [44, p. 13]. It is denoted by $D(z_0, R)$, i.e.,

$$D(z_0, R) = \{ z \in \mathbb{C} : |z - z_0| < R \}.$$
(2.1)

The boundary is denoted by $\partial D(z_0, R)$, where $|z - z_0| = R$, while, the closure $\overline{D}(z_0, R)$ contains the set of all points given by (2.1), including its boundary points. If the center of the open disk is zero (i.e., $z_0 = 0$), then from (2.1) we have

$$D(0,R) = \{ z \in \mathbb{C} : |z| < R \}. \tag{2.2}$$

The open unit disk (or simply unit disk) is the disk centered at the origin and the radius is 1 (see [357, p. 10]), i.e.,

$$D = D(0,1) = \{ z \in \mathbb{C} : |z| < 1 \}. \tag{2.3}$$

Let n be a nonnegative integer and $a \in \mathbb{C}$ such that $a \neq \{..., -2, -1\}$. Then, the *Pochhammer symbol* $(a)_n$ is defined by (see [4, p. 256], [404])

$$(a)_n = \begin{cases} 1, & \text{for } n = 0; \\ a(a+1)\cdots(a+n-1), & \text{for } n \ge 1. \end{cases}$$
 (2.4)

This symbol, which is also known as the *rising factorial* or the *shifted factorial* function, can also be expressed in the following form (see [26, p. 86]):

$$(a)_n = \prod_{j=1}^n (a+j-1) = \frac{\Gamma(a+n)}{\Gamma(a)}, \quad n = 1, 2, \dots,$$
 (2.5)

for $a, a + n \in \mathbb{C} \setminus \{\dots, -2, -1\}$, where $\Gamma(z)$ is the gamma function defined by $\Gamma(z) = \int_0^\infty t^{z-1} e^t dt$, for any $z \in \mathbb{C}$ such that $\operatorname{Re}(z) > 0$. Other notations such as $a^{\overline{n}}$, $a^{(n)}$ and (a, n) may be found in the literature, which represent the Pochhammer symbol as well.

The $(a)_n$ has simple values at the arguments 0 and 1, i.e.,

$$(0)_0 = 1$$
, and $(1)_n = n!$ (2.6)

Some other useful formulas associated with the Pochhammer symbol are given in the following:

$$(a)_{-n} = \frac{(-1)^n}{(1-a)_n}, \qquad (a)_n = \frac{n!}{(a-n)!} = \frac{(a+n-1)!}{(a-1)!},$$

$$(a+1)_n = \frac{(a+n)!}{n!}, \qquad (a)_{n+k} = (a+k)_k (a)_n,$$

$$(2.7)$$

where n and k are nonnegative integers. The Pochhammer symbol is widely used in the Theory of Special Functions, particularly to represent the coefficients of hypergeometric functions.

A binomial series is an infinite series defined by

$$(1+z)^{\alpha} = \sum_{k=0}^{\infty} {\alpha \choose k} z^k, \qquad (2.8)$$

where α is an arbitrary complex number. The series in the right hand-side of (2.8) converges absolutely for all $z \in D$, with the generalized binomial coefficients defined by

$$\binom{\alpha}{k} = \frac{\alpha!}{(\alpha - k)!k!},\tag{2.9}$$

for any $\alpha \in \mathbb{C}$ and $k \in \mathbb{N}_0$. Some well-known formulas of binomial coefficients are given as follows:

$$\begin{pmatrix} \alpha \\ 0 \end{pmatrix} = \begin{pmatrix} \alpha \\ \alpha \end{pmatrix} = 1, \qquad \qquad \begin{pmatrix} \alpha \\ 1 \end{pmatrix} = \begin{pmatrix} \alpha \\ \alpha - 1 \end{pmatrix} = \alpha,
\begin{pmatrix} \alpha \\ k \end{pmatrix} = \begin{pmatrix} \alpha \\ \alpha - k \end{pmatrix}, \qquad \begin{pmatrix} -\alpha \\ k \end{pmatrix} = (-1)^k \begin{pmatrix} \alpha + k - 1 \\ k \end{pmatrix},$$
(2.10)

which holding for all $\alpha \in \mathbb{C}$ and $k \in \mathbb{N}_0$.

A real signum or real sign function is a function denoted by y = sgn(x) (read 'y is equal to sign of x'), and it is defined by the equations

$$sgn(x) = \begin{cases} -1, & \text{for } x < 0; \\ 0, & \text{for } x = 0; \\ 1, & \text{for } x > 0. \end{cases}$$
 (2.11)

In the other interactive way, the function y = sgn(x) is related to the function y = |x|, where its moment's reflection shows that for each real number a,

$$|a| = a \operatorname{sgn}(a)$$
 or equivalent to $a = |a| \operatorname{sgn}(a)$. (2.12)

On the other hand, the equation in (2.12) provides another characterization of the absolute value of a real number a. In the complex plane, a *complex sign* function is defined by

$$sgn(z) = \begin{cases} \frac{z}{|z|}, & \text{for } z \in \mathbb{C} \setminus \{0\}; \\ 0, & \text{for } z = 0. \end{cases}$$
 (2.13)

The two positive real numbers p, q such that

$$\frac{1}{p} + \frac{1}{q} = 1, (2.14)$$

are called the *conjugate exponent*. For p = 1, the conjugate exponents of p is $q = \infty$. If p and q are integers, then the only pair of the conjugate exponents is 2, 2.

2.2 Elementary Inequalities in Analysis

We shall begin this section with some basic inequalities in Real and Complex Analysis. These include the triangle inequality, mean inequalities and the inequalities involving convex functions.

2.2.1 Basic Inequalities

There are many basic facts, which are fundamental in the Theory of Inequalities. First, we state the two simple propositions called *axioms* (see [55, p. 7], [239, p. 2]):

Axiom 10 A given real number x satisfies precisely one of the following: x = 0, x > 0 or x < 0.

Axiom 11 If x and y are positive numbers, then the sum x + y and the product xy are positive numbers.

Consequently, there are obvious variations and extensions of these axioms, for example,

Axiom 12 If $x, y \in \mathbb{R}$, then one and only one of the following relationships holds:

$$x = y, \qquad x > y \qquad or \qquad x < y, \tag{2.15}$$

where the symbols ">" and "<" represent the strict inequalities.

The following fundamental rules of algebra also hold for all the real numbers $x, y, z \in \mathbb{R}$ (see [55, p. 7-21]):

- (a) If $x \ge y$ and $y \ge z$, then $x \ge z$.
- (b) If $x \leq y$, then $x + a \leq y + a$ for any $a \in \mathbb{R}$.
- (c) If $x \ge y$ and a > 0, then $ax \ge ay$ and $\frac{x}{a} \ge \frac{y}{a}$. If a < 0, then $ax \le ay$ and $\frac{x}{a} \le \frac{y}{a}$.
- (d) If x > y and a > b, then x + a > y + b for any $a, b \in \mathbb{R}$.
- (e) If x > y > 0 and a, b are positive integers, then $x^{a/b} > y^{a/b}$.

The next result is of fundamental importance and frequently used in dealing with inequalities.

Theorem 13 ([55, p. 10]) The product xy of a positive number x and negative number y is a negative number, while, the product of two negative numbers x and y is a positive number.

We observe that the square of any real number can never be negative. Thus, we obtain one of the simplest and most useful results in the Theory of Inequalities as follows.

Theorem 14 ([55, p. 11]) Any real number x satisfies the inequality $x^2 \ge 0$, with the equality holds if and only if x = 0.

Accordingly, for any two real numbers a and b, we have the quantity that $(a-b)^2 \ge 0$, which implies

$$\frac{a^2 + b^2}{2} \ge ab, (2.16)$$

with equality occurring in (2.16) if and only if a = b. If a > 0 and b = 1 in (2.16), then we have

$$a + \frac{1}{a} \ge 2. \tag{2.17}$$

The inequality (2.16) can be written in the form [55, p. 48]

$$\frac{x+y}{2} \ge \sqrt{xy},\tag{2.18}$$

by replacing a^2 by x and b^2 by y, with x and y being positive real numbers. Clearly, the equality in (2.18) occurs if and only if x = y. This inequality is often called the *arithmetic mean-geometric mean inequality* for the two positive real numbers (we shall discuss further this inequality in Section 2.2.3).

Analogously, for complex numbers, $|z - w|^2 \ge 0$, we have

$$\frac{|z|^2 + |w|^2}{2} \ge \operatorname{Re}(z\overline{w}), \qquad (2.19)$$

for any $z, w \in \mathbb{C}$. The equality in (2.19) is achieved if and only if z = w.

The inequalities given by (2.16) and (2.19) are of fundamental importance in the study of Analytic Inequalities in Real and Complex Analysis. They have a much richer interpretation in the Theory of Inequalities as well. In fact, these bounds are the sources and the methods of proof of many important classical inequalities in analysis.

2.2.2 Triangle Inequality

If z and w are the two points in the complex plane, then we can associate the vectors from the origin to the each point. Now, we consider the triangle with the vertices are the origin, z and w, with the lengths of its sides are |z|, |w| and |z-w| (or |w-z|). Thus, using the geometrical fact that the sum of the lengths of two sides of a triangle is greater than or equal to the third side, implies the inequality

$$|z| + |w| \ge |z - w|, \tag{2.20}$$

with the sign of equality holding in (2.20) if and only if $\operatorname{Re}(z\overline{w}) = |zw|$. Since w represents an arbitrary complex number, then the inequality (2.20) still holds if w is replaced by -w, thus we obtain

$$|z| + |w| > |z + w|, (2.21)$$

for all $z, w \in \mathbb{C}$, in which, the equality occurs again in (2.21) if and only if $\operatorname{Re}(z\overline{w}) = |zw|$. The inequality (2.21) is popularly known as the *triangle inequality* for two complex numbers. On the same basis, we also have the inequality

$$|z - w| \ge ||z| - |w||, \tag{2.22}$$

which is concerned with the length of the difference of the two complex numbers. We note that, if w is replaced by -w in (2.22), it then yields

$$|z + w| \ge ||z| - |w||. \tag{2.23}$$

The inequality (2.23) is called the reverse triangle inequality, which provides the lower bounds instead of the upper bounds of |z+w|. Thus, for any $z, w \in \mathbb{C}$ we have

$$||z| - |w|| \le |z \pm w| \le |z| + |w|, \tag{2.24}$$

which obviously follows from the inequalities (2.20) - (2.23).

The triangle inequality in (2.21) can be generalized for n complex numbers $z_1, z_2, \ldots, z_n \in \mathbb{C}$ as follows:

$$\left| \sum_{j=1}^{n} z_j \right| \le \sum_{j=1}^{n} |z_j|, \tag{2.25}$$

with the equality occurring in (2.25) if and only if the ratio of any nonzero terms is positive [4, p. 11]. In terms of their weighted version, the following inequality holds:

$$\left| \sum_{j=1}^{n} \alpha_j z_j \right| \le \sum_{j=1}^{n} \alpha_j \left| z_j \right|, \tag{2.26}$$

where $\alpha_j > 0$ for $j \in \{1, 2, ..., n\}$ such that $\sum_{j=1}^n \alpha_j = 1$. The last two types of the triangle inequalities are the most important and frequently used in mathematical analysis.

2.2.3 Mean Inequalities

A mean is a notion of average for collection of numbers, data, etc. For instance, the arithmetic mean is the simplest type of means that is commonly used in Probability and Statistics Theory to measure the central tendency of numbers or data. Besides this mean, there are several types of means occurring in various contexts of measuring the central tendency of numbers, such as geometric mean and harmonic mean. The values of these means have an important relationship, that we shall emphasize in this section. Before that, we shall recall the basic definitions of the arithmetic, geometric and harmonic means for n positive real numbers.

Let $a = (a_1, a_2, ..., a_n)$ be a sequence of positive real numbers. Then, we denote the arithmetic mean, geometric mean and harmonic mean for n numbers by AM, GM and HM respectively, which are defined by the following equations (see [77, p. 19-20], [239, p. 16]):

$$AM(a) = \frac{1}{n} \sum_{j=1}^{n} a_j, \quad GM(a) = \left(\prod_{j=1}^{n} a_j\right)^{1/n}, \quad HM(a) = n \left(\sum_{j=1}^{n} \frac{1}{a_j}\right)^{-1}.$$
(2.27)

The value of these means always lies between the lowest and the highest of its numbers. In the same manner, we can generalize these means into their weighted versions. Let us fix a set of positive real numbers $w_j > 0$ for $j \in \{1, 2, ..., n\}$. Then, the weighted version of the AM, GM and HM are defined by the following expressions:

$$AM(w,a) = \sum_{j=1}^{n} w_j a_j \left(\sum_{j=1}^{n} w_j\right)^{-1}, \quad GM(w,a) = \left(\prod_{j=1}^{n} a_j^{w_j}\right)^{1/\sum_{j=1}^{n} w_j}$$
(2.28)

and

$$HM(w,a) = \sum_{j=1}^{n} w_j \left(\sum_{j=1}^{n} \frac{w_j}{a_j} \right)^{-1},$$
 (2.29)

where the w_j 's, $j \in \{1, 2, ..., n\}$ are called the weights. The ordinary means mentioned in (2.27) are the special case of these weighted means, when the weights w_j are equal for all $j \in \{1, 2, ..., n\}$.

The most important in the Theory of Inequalities is the relations between these means. As mentioned in the book of Kapur [237, p. 1] (see also [55, p. 54]), the relation $HM \leq AM \leq GM$ holds for n positive real numbers. More precisely, we have the following result:

Theorem 15 ([55, p. 54]) For any n nonnegative real numbers $a_1, a_n, ..., a_n$,

$$\frac{1}{n} \sum_{j=1}^{n} a_j \ge \left(\prod_{j=1}^{n} a_j \right)^{\frac{1}{n}}.$$
 (2.30)

The sign of equality holds in (2.30) if and only if all the numbers a_1, a_2, \ldots, a_n are equal.

Since there is the obvious relation between the arithmetic mean and the harmonic mean, that is,

$$\operatorname{HM}(a) = \left[\operatorname{AM}\left(\frac{1}{a}\right)\right]^{-1},\tag{2.31}$$

where $1/a = (1/a_1, 1/a_2, ..., 1/a_n)$ and $a_j, j \in \{1, 2, ..., n\}$ are positive real numbers, we can derive a similar inequality between the harmonic mean and geometric mean as follows:

$$\left(\prod_{j=1}^{n} a_j\right)^{1/n} \ge n \left(\sum_{j=1}^{n} \frac{1}{a_j}\right)^{-1},\tag{2.32}$$

for $a_j > 0$, $j \in \{1, 2, ..., n\}$. Again, the case of equality occurs in (2.32) if and only if $a_1 = a_2 = \cdots = a_n$. The inequalities given by (2.30) and (2.32) are well-known in the literature as the arithmetic mean-geometric mean inequality and the geometric mean-harmonic mean inequality for n numbers respectively, or they are simply denoted by (AM-GM)-inequality and (GM-HM)-inequality. These inequalities are widely used in Optimization Theory, for instance, in solving of the optimization problems for two or more dimensional spaces, where the maximum or minimum achieved at the point of the equality of a given inequality occurs (see [55, Chapt. 5], [237, Chapt. 2]).

It can be clearly seen from (2.32) that the harmonic mean is bounded by the geometric mean, which in turn is bounded by the arithmetic mean (2.30). Hence, this string of the fundamental inequalities can be expressed as follows (see [4, p. 10], [237, p. 1]):

$$\frac{1}{n} \sum_{j=1}^{n} a_j \ge \left(\prod_{j=1}^{n} a_j \right)^{1/n} \ge n \left(\sum_{j=1}^{n} \frac{1}{a_j} \right)^{-1}, \tag{2.33}$$

for any positive real numbers a_1, a_2, \ldots, a_n . The equalities occur in (2.33) if and only if $a_1 = a_2 = \cdots = a_n$. The double inequality in (2.33) is often called the arithmetic mean-geometric mean-harmonic mean inequality, or simply denoted as the (AM-GM-HM)-inequality. The weighted means of the (AM-GM-HM)-inequality also holds [237, p. 225], namely

$$\sum_{j=1}^{n} w_j a_j \ge \prod_{j=1}^{n} a_j^{w_j} \ge \left(\sum_{j=1}^{n} \frac{w_j}{a_j}\right)^{-1}, \tag{2.34}$$

for $w_j > 0$, $j \in \{1, 2, ..., n\}$ such that $\sum_{j=1}^n w_j > 0$. The simplest case of (2.33) for two positive real numbers is that

$$\frac{x+y}{2} \ge \sqrt{xy} \ge \frac{2}{\frac{1}{x} + \frac{1}{y}},\tag{2.35}$$

for any x, y > 0, where the first inequality in (2.35) has been mentioned in the previous section as (2.18). Hence, we also have from (2.34) that [91, p. 4]

$$\frac{\alpha a + \beta b}{\alpha + \beta} \le a^{\frac{\alpha}{\alpha + \beta}} b^{\frac{\beta}{\alpha + \beta}} \le \frac{\alpha + \beta}{\frac{\alpha}{\alpha} + \frac{\beta}{b}},\tag{2.36}$$

for the weighted version of the (AM-GM-HM)-inequality, with a, b > 0 and α , $\beta \geq 0$ such that $\alpha + \beta > 0$. The equality holds in (2.36) if and only if a = b. We note that by letting $p = (\alpha + \beta)/\alpha$, $q = (\alpha + \beta)/\beta$, $x = a^{\alpha/(\alpha+\beta)}$ and $b = y^{\beta/(\alpha+\beta)}$ in the first inequality of (2.36), it reduces to

$$xy \le \frac{x^p}{p} + \frac{y^q}{q},\tag{2.37}$$

for any $x, y \ge 0$, with p, q > 1 such that 1/p + 1/q = 1. This inequality (2.37) is one the most important results in Analytic Inequalities, which provides a principal tool for deriving some classical inequalities in analysis; such as the Cauchy-Bunyakovsky-Schwarz's, Hölder, Minkowski inequalities, etc. It also happens to be a special case of the classical Young's inequality for the product of two real numbers. In the other sense, the result in (2.37) becomes one of the most important tools of our investigations in Chapter 4.

2.2.4 Convexity Inequalities

Convexity is a simple and natural notion of functions, which can be traced back to Archimedes (c. 250 B.C), in connection with his famous estimate of the value of π . It has a central role in the study of Convex Function Theory. Convex functions, which are extensively treated in various textbooks on calculus (see for instance [333], [352, Chapt. 2], [360], [375], [377]), play an extremely important role in many branches of mathematics as well as their applications in various fields.

The subject of convex functions was studied by Danish mathematician J. Jensen ([222], [223]) in the 1900s, although this class of functions was already treated by Ch. Hermite [195], O. Hölder [201], J. Hadamard [176] and O. Stolz [410]. In the third chapter of Hardy, Littlewood and Pólya's famous work [185] and the sixth chapter of Steela's work [407], they revealed that the convexity and its generalizations, play a crucial rule in the Theory of Inequalities. Many classical inequalities such as the triangle, Cauchy-Bunyakovsky-Schwarz, Hölder, Minkowski inequalities, etc., are closely related to the *convexity* of certain underlying functions.

Roughly speaking, there are several basic properties of convex functions that made them so widely used in theoretical and applied mathematics. As noted in [334, p. 11], the basic ideas of convex functions begins in the context of real-valued functions of a real variable. So, we have the following property of convex functions defined on an interval (see also [318, p. 1], [375, p. 2]).

Definition 16 Let I be an interval in \mathbb{R} . Then, the real-valued function $f: I \to \mathbb{R}$ is said to be convex if for all $x, y \in I$ and $\lambda \in [0, 1]$,

$$f(\lambda x + (1 - \lambda)y) \le \lambda f(x) + (1 - \lambda)f(y). \tag{2.38}$$

It is called strictly convex provided that the inequality in (2.38) is strict for $x \neq y$ and $\lambda \in (0,1)$. If the sign of inequality in (2.38) is reversed, then the function f is concave (strictly concave). In other words, we say f is concave (strictly concave) if and only if -f is convex (strictly convex). If f is both convex and concave, then f is said to be an affine function.

Geometrically, the property in (2.38) on the other hand, requires that the entire secant line joining the points (x, f(x)) and (y, f(y)) on the graph of a convex function f lie above (or on) the graph of function between x_1 and x_2 , if λ assumes all real values in [0, 1].

Remark 17 The definition of convexity of functions (Definition 16) has a natural generalization to the real-valued functions defined on arbitrary linear space \mathbf{X} , in which, it requires that the domain of f(x) be a convex and open set \mathbf{U} . In other words, we may define the function f(x) to be convex on $\mathbf{U} \subseteq \mathbf{X}$, then the inequality (2.38) holds true for all $x, y \in \mathbf{U}$ and $\lambda \in [0, 1]$.

Many convex functions are well-known in analysis. For instance, $f(x) = x^2$, $f(x) = \exp(x)$ and f(x) = |x|, are convex functions on any open interval $I \subset \mathbb{R}$. In elementary calculus, the following simple criteria are commonly used to verify the convexity of a given function (see [104, p. 70], [318, p. 2], [375, p. 10-11]).

Theorem 18 ([375, p. 10-11]) (i) Suppose f(x) is differentiable on the interval I. Then, the function f(x) is convex (strictly convex) on I if and only if f'(x) is increasing (strictly increasing). (ii) Suppose the second derivative f''(x) exists on I. Then, f(x) is convex on I if and only if $f''(x) \ge 0$ for each $x \in I$, and if f''(x) > 0 on I, then f(x) is strictly convex on the interval.

These criteria give sufficient conditions that functions be convex on an interval I, however such functions need to be continuous everywhere on this interval. Checking that a given function is convex or not, is not an easy task, but there are several useful criteria of convexity properties available in the literature. The simplest one of the basic properties of convex functions in one dimensional case is due to Jensen [223], see also ([79, p. 48], [318, p.3], [334, p. 14]).

Theorem 19 ([223]) Let $f: I \to \mathbb{R}$ be a continuous real-valued function. Then, f is convex if and only if f is midpoint convex (or simply midconvex), that is, for all points $x, y \in I$, the inequality

$$f\left(\frac{x+y}{2}\right) \le \frac{f(x) + f(y)}{2} \tag{2.39}$$

is valid.

The function f is said to be strictly midconvex, if for all pairs of points (x, y), $x \neq y$, then the strict inequality holds in (2.39). If the sign of inequality in (2.39) is reversed, then we shall say the function f is concave. The functions, which fulfil the *Jensen's functional inequality* (2.39), are also known as the *Jensen-convex* or simply *J-convex* functions. The inequality in (2.39) is obviously the special case of (2.38) for $\lambda = 1/2$. In addition, Hardy, Littlewood and Pölya [185, p. 96] showed that every convex function is a Jensen-convex.

One of the most important aspects in the Theory of Inequalities is that, the convexity inequalities have been generalized, extended and refined in several ways. For instance, the following refinement of the inequality (2.39) was proved by Hadamard in [176].

Theorem 20 ([176]) Let $f: I \to \mathbb{R}$ be a convex function, where I = [a, b]. Then, the inequalities

$$f\left(\frac{a+b}{2}\right) \le \frac{1}{b-a} \int_a^b f(x)dx \le \frac{f(a)+f(b)}{2} \tag{2.40}$$

are valid.

The inequality (2.40) is called the *Hadamard's inequality*. Besides the midconvex, the other forms of convexity of functions are log-convex and quasiconvex. We shall say that a positive function $f: I \to \mathbb{R}^+$ is logarithmic convex, or for short, log-convex, if log f is a convex function, or equivalently, if $x, y \in I$ and $\lambda \in [0, 1]$, one has the inequality (see [79, p. 48], [318, p. 3], [352, p. 7])

$$f(\lambda x + (1 - \lambda)y) \le f^{\lambda}(x)f^{1-\lambda}(y). \tag{2.41}$$

It is said to be a positive log-concave if the sign of inequality in (2.41) is reversed. Note that, if f and g are convex functions and g is increasing, then the convolution of f and g, i.e., $f \circ g$ is convex. Moreover, since $f = \exp(\log f)$, it follows that a log-convex function is convex, but the converse may not necessarily be true [352, p. 7].

The function $f: I \to \mathbb{R}$ is said to be *quasi-convex* if for all $x, y \in I$ and all $\lambda \in [0, 1]$, then (see [318, p. 3])

$$f(\lambda x + (1 - \lambda)y) \le \max[f(x), f(y)]. \tag{2.42}$$

Further, the concept of *m*-convexity was introduced by Toader [421] to provide a generalization of the convexity of functions (2.38). It states that, a function $f:[0,b] \to \mathbb{R}$ is said to be *m*-convex if it satisfies the following condition:

$$f(\lambda x + m(1 - \lambda)y) \le \lambda f(x) + m(1 - \lambda)f(y), \tag{2.43}$$

where $m \in [0, 1], x, y \in [0, b]$ and $\lambda \in [0, 1]$.

Another extension of the property in (2.39) was discovered by Jensen in his work ([222], [223]), which is known in the literature as *Jensen's inequality*. We shall discuss this inequality in detail in Section 2.6).

2.2.5 Normed, Linear and Inner Product Spaces

Suppose that **X** is a *linear space* over the real or complex number field \mathbb{K} . The inner product on **X** is a real-valued function $\langle \cdot, \cdot \rangle^1 : \mathbf{X} \times \mathbf{X} \to \mathbb{K}$, which satisfies the following conditions (see [161, p. 14], [439, p. 3]). For all $x, y, z \in \mathbf{X}$ and $\alpha, \beta \in \mathbb{K}$,

(i) Positive definiteness:
$$\langle x, x \rangle \ge 0$$
 and $\langle x, x \rangle = 0$ if and only if $x = 0$. (2.44)

(ii) Hermitian condition: $\langle x, y \rangle = \overline{\langle y, x \rangle}$.

(iii) Linearity:
$$\langle \alpha x + \beta y, z \rangle = \alpha \langle x, z \rangle + \beta \langle y, z \rangle$$
 and $\langle x, \alpha y + \beta z \rangle = \overline{\alpha} \langle x, y \rangle + \overline{\beta} \langle x, z \rangle$.

The notation $\langle x, y \rangle$ is called the *inner product* of x and y. The linear space \mathbf{X} equipped with the inner product $\langle \cdot, \cdot \rangle$ defined on \mathbf{X} , is called an *inner product* space or pre-Hilbert space. The $(\mathbf{X}, \langle \cdot, \cdot \rangle)$ is the standard notation for the inner product spaces \mathbf{X} .

One example of the inner product spaces is $(\mathbb{C}^n, \langle \cdot, \cdot \rangle)$, with its usual inner product defined by

$$\langle z, w \rangle = \sum_{k=1}^{n} z_k \overline{w}_k,$$
 (2.45)

for any $z, w \in \mathbb{C}^n$ and \overline{w}_k , $k \in \{1, 2, ..., n\}$ are the complex conjugate of w_k . For the real inner product space $(\mathbb{R}^n, \langle \cdot, \cdot \rangle)$, the inner product corresponds to the standard *dot product* of the two vectors $x, y \in \mathbb{R}^n$, namely

$$\langle x, y \rangle = \sum_{k=1}^{n} x_k y_k. \tag{2.46}$$

Consider the complex-valued functions $f, g : [a, b] \to \mathcal{C} \subset \mathbb{C}$, which are continuous on the bounded interval [a, b]. Then, the expression

$$\langle f, g \rangle = \int_{a}^{b} f(x) \overline{g(x)} dx,$$
 (2.47)

¹In other disciplines, the notation $\langle \cdot | \cdot \rangle$ may be used for an inner product.

defines the inner product on the complex function space C[a, b].

There are many examples of useful inner products that occur in a great variety of mathematical contexts. For instance, if we fix a set of positive real numbers $q_k > 0$, $k \in \{1, 2, ..., n\}$, then we can easily define an inner product on \mathbb{C}^n with the weighted sums, i.e.,

$$\langle z, w \rangle = \sum_{k=1}^{n} q_k z_k \overline{w}_k. \tag{2.48}$$

Also, if $w : [a, b] \to \mathbb{R}$ is a real-valued continuous function such that w(x) > 0 for all $x \in [a, b]$, then we might define the weighted inner product on C[a, b] as well by setting

$$\langle f, g \rangle = \int_{a}^{b} f(x) \overline{g(x)} w(x) dx.$$
 (2.49)

The definitions of the weighted inner products given by (2.48) and (2.49), satisfy all of the properties (i) - (iii) as mentioned in (2.44) that one requires for the inner product.

Another family of the linear spaces, which is of great important in mathematical analysis, is the family of $normed\ space$. Every inner product induces a norm on X by the following identity:

$$||z|| = \langle z, z \rangle^{1/2},$$
 (2.50)

for all $z \in \mathbf{X}$. We denote the above norm by $\|\cdot\|$. In general, a norm on a linear space \mathbf{X} is a function $\|\cdot\| : \mathbf{X} \to \mathbb{K}$ satisfying the following properties:

(i) ||v|| > 0 for all nonzero $v \in \mathbf{X}$.

(ii)
$$||cv|| = |c| ||v||$$
 for all $c \in \mathbb{C}$ and all $v \in \mathbf{X}$. (2.51)

(iii) $||v + w|| \le ||v|| + ||w||$ for all $v, w \in \mathbf{X}$ (the triangle inequality).

Consequently, from the above examples (2.45) - (2.49), thus we have

$$||x||^2 = \sum_{k=1}^n |x_k|^2$$
, or $||x||^2 = \sum_{k=1}^n q_k |x_k|^2$, $q_k > 0$ (2.52)

on the complex inner product space, and

$$||f||^2 = \int_a^b |f(x)|^2 dx$$
, or $||f||^2 = \int_a^b |f(x)|^2 w(x) dx$ (2.53)

on the function space C[a, b].

It is called a *normed linear space* $(\mathbf{X}, \|\cdot\|)$ for a linear space \mathbf{X} equipped with the norm $\|\cdot\|$ and a *normed space* $(V, \|\cdot\|)$ is a vector space with a norm, where V is a real or complex vector space and $\|\cdot\|$ is a norm defined on V.

Let $a = (a_1, a_2, \dots, a_n)$ be an *n*-tuple of real numbers and $1 \le p < \infty$. Then, we can write

$$||a||_p = \left(\sum_{k=1}^n |a_k|^p\right)^{1/p},$$
 (2.54)

where for $p = \infty$ we simply set $||a||_{\infty} = \max_{1 \le k \le n} |a_k|$. The quantity $||a||_p$ is called the *p*-norm or the l^p -norm of the *n*-tuple of real numbers.

The function $a \mapsto \|a\|_p$, where $\|a\|_p$ is defined by (2.54) indeed satisfies the properties required by the definition of the norm. Specifically, the function $\|a\|_p$, $p \ge 1$ needs to satisfy the following properties:

(i)
$$||a||_p = 0$$
 if and only if $a = 0$,

(ii)
$$\|\alpha a\|_p = |\alpha| \|a\|_p$$
 for all $\alpha \in \mathbb{R}$, (2.55)

(iii)
$$||a+b||_p \le ||a||_p + ||b||_p$$
 for all real *n*-tuple a and b ,

where the third property in (2.55) appears as Minkowski's inequality.

Remark 21 The function (2.54) reduces to simple norm (2.52) for the case p = 2.

The following section discusses the classical Cauchy-Bunyakovsky-Schwarz inequality that is one of the famous and important inequalities in analysis.

2.3 The (CBS)-Type Inequalities

The (CBS)-inequality is a short-term of the Cauchy-Bunyakovsky-Schwarz inequality, is also known in the literature as Cauchy's, Schwarz's or Cauchy-Schwarz's inequality. For simplicity, we shall refer to it throughout this dissertation as the (CBS)-inequality. This inequality plays an important role in different branches of Modern Mathematics including Hilbert Spaces Theory, Probability and Statistics Theory, Classical Real and Complex Analysis, Numerical Analysis, Qualitative Theory of Differential Equations and their applications (see [63], [66], [125], [135], [246], [270], [300], [302], [355], [380]).

Several versions of the classical (CBS)-inequality have been found in the literature, starting from Cauchy [88], who published the (CBS)-inequality in elementary form. Then, Bunyakovsky [81] derived the corresponding inequality for integrals, and later, it was rediscovered by Schwarz [388]. Schwarz [389] also discovered the corresponding version of the (CBS)-inequality in inner product spaces. For more information about the name and history of these type of inequalities, see ([387], [407]). In any case, the discrete version of the (CBS)-inequality is stated in the following section.

2.3.1 (CBS)-Inequalities for Real and Complex Numbers

In the following, we state the elementary inequality of the (CBS)-inequality for sequences of real numbers, which it is also known in the literature as *Cauchy's inequality* [318, p. 83] (see also [55, p. 66], [91, p. 13], [117, p. 1]).

Theorem 22 ([318, p. 83]) If $a = (a_1, a_2, ..., a_n)$ and $b = (b_1, b_2, ..., b_n)$ are sequences of real numbers, then

$$\left(\sum_{k=1}^{n} a_k b_k\right)^2 \le \left(\sum_{k=1}^{n} a_k^2\right) \left(\sum_{k=1}^{n} b_k^2\right), \tag{2.56}$$

with equality holding in (2.56) if and only if the sequences a and b are proportional, i.e., there is a real number $r \in \mathbb{R}$ such that $a_k = rb_k$ for each $k \in \{1, 2, ..., n\}$.

The following version of the (CBS)-inequality for complex numbers also holds (see [117, p. 2], [318, p. 84]).

Theorem 23 If $\mathbf{a} = (a_1, a_2, \dots, a_n)$ and $\mathbf{b} = (b_1, b_2, \dots, b_n)$ are sequences of complex numbers, then

$$\left| \sum_{k=1}^{n} a_k b_k \right|^2 \le \left(\sum_{k=1}^{n} |a_k|^2 \right) \left(\sum_{k=1}^{n} |b_k|^2 \right), \tag{2.57}$$

with equality holding in (2.57) if and only if the sequences a and \overline{b} are proportional, i.e., there is a complex number $c \in \mathbb{C}$ such that $a_k = c\overline{b}_k$ for any $k \in \{1, 2, ..., n\}$.

These types of the (CBS)-inequality have been proved in several ways, see for instance the works of Mitrinovic et al. [318, p. 83], Dragomir [117, p. 1-2], Cerone and Dragomir [91, p. 13-14], Bullen [79, p. 183] and Steele [407, p. 4-6].

An analogous statement of the (CBS)-inequality is the Cauchy-Schwarz inequality for convergent infinite series and integrals [77, p. 31], i.e.,

$$\left(\sum_{k=1}^{\infty} a_k b_k\right)^2 \le \left(\sum_{k=1}^{\infty} a_k^2\right) \left(\sum_{k=1}^{\infty} b_k^2\right) \tag{2.58}$$

and

$$\left[\int_{a}^{b} f(x) g(x) dx \right]^{2} \le \int_{a}^{b} \left[f(x) \right]^{2} dx \int_{a}^{b} \left[g(x) \right]^{2} dx, \tag{2.59}$$

for f and g are square integrable functions on (a, b).

Let $(p_1, p_2, ..., p_n)$ be a sequence of nonnegative real numbers. Then, we can state the weighted version of the (CBS)—inequality as follows:

$$\left| \sum_{k=1}^{n} p_k a_k b_k \right|^2 \le \sum_{k=1}^{n} p_k |a_k|^2 \sum_{k=1}^{n} p_k |b_k|^2, \tag{2.60}$$

for any $a_k, b_k \in \mathbb{C}$, $p_k \geq 0$, $k \in \{1, 2, ..., n\}$. Hence, if in (2.60) we choose $p_k = 1$ for all $k \in \{1, 2, ..., n\}$, then we recapture the inequality (2.57).

Remark 24 By the (CBS)-inequality for real numbers (2.56) and the generalized triangle inequality for complex numbers (2.25), we also have

$$\left| \sum_{k=1}^{n} a_k b_k \right|^2 \le \left(\sum_{k=1}^{n} |a_k b_k| \right)^2 \le \sum_{k=1}^{n} |a_k|^2 \sum_{k=1}^{n} |b_k|^2, \tag{2.61}$$

for $a_k, b_k \in \mathbb{C}, k \in \{1, 2, ..., n\}$.

2.3.2 (CBS)-Inequalities in Inner Product Spaces

In 1885, Schwarz [389] gave another proof of the (CBS)-inequality for sums by extending it to an inner product space. The result states that if x and y are vectors in an inner product space $(H, \langle \cdot, \cdot \rangle)$ over a real or complex number \mathbb{K} , then

$$\left|\left\langle x,y\right\rangle \right|^{2}\leq\left\langle x,x\right\rangle \left\langle y,y\right\rangle ,$$
 (2.62)

for any $x, y \in H$, with the equality holding in (2.62) if and only if the vectors x and y are linearly dependent, i.e., there exists a nonzero constant $\lambda \in \mathbb{K}$ such that $x = \lambda y$. The inequality (2.62) is mainly known in the literature as Schwarz's or Cauchy-Schwarz's inequality (see [161, p. 15]).

We note that, if we consider the norm of $x, y \in H$ as defined by (2.50), then the inequality (2.62) can be written in the form

$$|\langle x, y \rangle|^2 \le ||x||^2 ||y||^2$$
 (2.63)

or, equivalently,

$$|\langle x, y \rangle| \le ||x|| \, ||y|| \,, \tag{2.64}$$

with the case of equality holding in (2.63) or (2.64) if and only if there exists a scalar $\lambda \in \mathbb{C}$ such that $y = \lambda x$. Also, we note that the inequality (2.62) reduces to the elementary (CBS)-inequality (2.56) by considering the real inner product space $(\mathbb{R}^n, \langle \cdot, \cdot \rangle)$, with its usual inner product as defined by (2.46).

2.3.3 (CBS)-Inequalities for Power Series

We now generalize the situation of the celebrated (CBS)-inequality for real or complex numbers and sequences into the following case. Let f(z) be an analytic function defined by the power series

$$f(z) = \sum_{n=0}^{\infty} a_n z^n, \qquad (2.65)$$

which is convergent on the open disk $D(0,R) \subset \mathbb{C}$, R > 0. If the coefficients $a_n, n \in \{1, 2, ...\}$ in (2.65) are complex numbers and applying the well-known (CBS)-inequality for complex numbers (2.57), then we can deduce that

$$|f(z)|^2 = \left| \sum_{n=0}^{\infty} a_n z^n \right|^2 \le \sum_{n=0}^{\infty} |a_n|^2 \sum_{n=0}^{\infty} |z|^{2n} = \frac{1}{1 - |z|^2} \sum_{n=0}^{\infty} |a_n|^2, \quad (2.66)$$

for any $z \in D(0, R) \cap D(0, 1)$, where R is the radius of convergence of f(z). The above inequality (2.66) gives some information about the magnitude of the function f(z) provided the numerical series $\sum_{n=0}^{\infty} |a_n|^2$ is convergent and z is not too close to the boundary of the open disk D(0, 1).

In practice, many usual fundamental complex functions can be represented by the power series with real or nonnegative coefficients (see Section 1.3). Thus, if we assume the coefficients in the power series (2.65) are nonnegative and utilizing the weighted version of the (CBS)-inequality for complex numbers (2.60), then we can state the following inequality:

$$|f(zw)|^2 = \left| \sum_{n=0}^{\infty} a_n z^n w^n \right|^2 \le \sum_{n=0}^{\infty} a_n |z|^{2n} \sum_{n=0}^{\infty} a_n |w|^{2n} = f(|z|^2) f(|w|^2), \quad (2.67)$$

for any $z, w \in \mathbb{C}$ with $zw, |z|^2, |w|^2 \in D(0, R)$. In particular, if $w = a \in \mathbb{R}$, then we get from (2.67)

$$|f(az)|^2 \le f(a^2) f(|z|^2), \qquad (2.68)$$

for any $z \in \mathbb{C}$ with az, a^2 , $|z|^2 \in D(0, R)$.

Specifically, for functions, which are defined by the power series (2.65) with real coefficients a_n , $n \in \{0, 1, 2, ...\}$, we can naturally construct another power series, where the coefficients of the new series are given by the absolute values of the coefficients of the original series, namely

$$f_A(z) = \sum_{n=0}^{\infty} |a_n| z^n,$$
 (2.69)

for all $z \in D(0, R)$, where $a_n = |a_n| sgn(a_n)$, $n \in \{0, 1, 2, ...\}$. The sgn(x) is the real signum function defined by (2.11). It is obvious that the new power series $f_A(z)$ given by (2.69) has the same radius of convergence as the original power series f(z) given by (2.65).

As some natural examples of functions and their transformations that may be useful for later applications, we can point out that if the functions g(z), h(z), k(z) and l(z) are respectively given by (1.20), (1.30), (1.32) and (1.33), then the corresponding functions constructed by the use of the absolute values of the coefficients (2.69) are obviously

$$g_A(z) = \frac{1}{1-z}, \qquad h_A(z) = \ln\left(\frac{1}{1-z}\right),$$

$$k_A(z) = \sinh(z), \qquad l_A(z) = \cosh(z).$$
(2.70)

These functions are defined on the same domain as their generating functions respectively.

In a similar way, the following inequality holds for any $z, w \in \mathbb{C}$ such that zw, $|z|^2$, $|w|^2 \in D(0, R)$, by utilising the weighted version of the (CBS)-inequality for complex numbers (2.60):

$$|f(zw)|^{2} = \left| \sum_{n=0}^{\infty} |a_{n}| \operatorname{sgn}(a_{n}) z^{n} w^{n} \right|^{2} \leq \sum_{n=0}^{\infty} |a_{n}| |z|^{2n} \sum_{n=0}^{\infty} |a_{n}| |w|^{2n}$$

$$= f_{A}(|z|^{2}) f_{A}(|w|^{2}). \tag{2.71}$$

In particular, if $w = a \in \mathbb{R}$, then we get from (2.71)

$$|f(az)|^2 \le f_A(a^2) f_A(|z|^2)$$
 (2.72)

for all $z \in \mathbb{C}$ with $az, a^2, |z|^2 \in D(0, R)$.

2.3.4 Some Results Related to (CBS)-Type

It is well-known that the classical (CBS)-type inequalities have been improved, generalized, refined and applied by a remarkable large number of researchers in the last few decades. For the details, see particularly the survey paper of Dragomir [127], the relevant chapters in the books of Dragomir ([117], [120, Chapt. 1], [121, Chapt. 2]) and numerous references, which are cited therein.

In this section, we provide some of the results related to the (CBS)-type inequalities as important tools for our investigation in the next chapters. First, in 1960, N. G. de Bruijn [78] established the following refinement of the classical (CBS)-inequality for a sequence of real numbers and the second sequence of complex numbers (see also [117, p. 48], [121, p. 43], [318, p. 89]).

Theorem 25 (de Bruijn [78]) Let $(a_1, a_2, ..., a_n)$ be n-tuple of real numbers and $(z_1, z_2, ..., z_n)$ an n-tuple of complex numbers. Then,

$$\left| \sum_{k=1}^{n} a_k z_k \right|^2 \leq \frac{1}{2} \sum_{k=1}^{n} a_k^2 \left(\sum_{k=1}^{n} |z_k|^2 + \left| \sum_{k=1}^{n} z_k^2 \right| \right)$$

$$\left(\leq \sum_{k=1}^{n} a_k^2 \cdot \sum_{k=1}^{n} |z_k|^2 \right).$$
(2.73)

Equality holds in (2.73) if and only if, $a_k = \text{Re}(\lambda z_k)$ for $k \in \{1, 2, ..., n\}$, where λ is a complex number such that the quantity $\lambda^2 \sum_{k=1}^n z_k^2$ is a nonnegative real number.

This inequality (2.73) is popularly known as the *de Bruijn inequality*. The proof of Theorem 25 can also be found in the work of Mitrinović, Pečarić and Fink [318, p. 89] and Dragomir [117, p. 48]. The weighted version of the de Bruijn inequality also holds, namely

$$\left| \sum_{k=1}^{n} p_k a_k z_k \right|^2 \le \frac{1}{2} \sum_{k=1}^{n} p_k a_k^2 \left(\sum_{k=1}^{n} p_k |z_k|^2 + \left| \sum_{k=1}^{n} p_k z_k^2 \right| \right), \tag{2.74}$$

where $p_k \ge 0$, $a_k \in \mathbb{R}$, $z_k \in \mathbb{C}$, $k \in \{1, 2, \dots, n\}$.

In [82], M. L. Buzano obtained the extension of the celebrated (CBS)-inequality (2.63) for a real or complex inner product space $(H, \langle \cdot, \cdot \rangle)$. The result is stated as follows:

Theorem 26 (Buzano [82]) Let H be a complete inner product space and let x, a, b be elements of H with $x \neq 0$. Then,

$$|\langle x, a \rangle \langle x, b \rangle| \le \frac{1}{2} (\|a\| \|b\| + |\langle a, b \rangle|) \|x\|^2,$$
 (2.75)

with equality holds in (2.75) if and only if there exists a scalar $\lambda \in \mathbb{K}$ such that $x = \lambda a$.

It is clear that for a = b, the above inequality (2.75) becomes the (CBS)-inequality (2.62) in inner product spaces. A similar inequality to (2.75) has been independently obtained by Richard [373] for H is a real inner product space (see also [361]), namely

$$\frac{1}{2} \|a\|^2 (|\langle a, b \rangle| - \|a\| \|b\|) \le \langle x, a \rangle \langle x, b \rangle \le \frac{1}{2} \|x\|^2 (|\langle a, b \rangle| - \|a\| \|b\|). \tag{2.76}$$

Further, in 1985, Dragomir [118] proved the following refinement of the (CBS)-inequality (2.63) in inner product space $(H; \langle \cdot, \cdot \rangle)$ over the real and complex number field \mathbb{K} .

Theorem 27 (Dragomir [118]) For any $x, y \in H$ and $e \in H$ with ||e|| = 1, the following refinement of the (CBS)-inequality holds:

$$||x|| ||y|| \ge |\langle x, y \rangle - \langle x, e \rangle \langle e, y \rangle| + |\langle x, e \rangle \langle e, y \rangle| \ge |\langle x, y \rangle|. \tag{2.77}$$

Remark 28 If in the first inequality of (2.77), we choose e = z/||z||, $z \in H \setminus \{0\}$, then we get

$$||x|| ||y|| ||z||^2 - |\langle x, z \rangle \langle z, y \rangle| \ge |\langle x, y \rangle ||z||^2 - \langle x, z \rangle \langle z, y \rangle|,$$
 (2.78)

for any $x, y, z \in H$.

2.4 Young's Inequality and Its Variants

In 1912, W. H. Young [440] presented the following integral inequality:

Theorem 29 ([440]) Let $f : [0, c] \to R$ be a real-valued, continuous and strictly increasing function on [0, c] with c > 0. If f(0) = 0, $x \in [0, c]$ and $y \in [0, f(c)]$, then

$$xy \le \int_0^x f(t) dt + \int_0^y f^{-1}(t) dt,$$
 (2.79)

where $f^{-1}(t)$ is the inverse function of f(t). Equality occurs in (2.79) if and if y = f(x) or equivalently, $x = f^{-1}(y)$.

The above inequality (2.79) has an easy geometric interpretation shown in [422], so that some monographs simply refer to it and omit the proof. The above result also can be found in the work of Hardy, Littlewood and Pólya [185, p. 111-113], but there was no analytic proof until Diaz and Metcalf [114] supplied it in 1970. The complete proof is also given in [316, p. 48-50], [432] and for more information on Young's inequality, we may refer to [318, p. 379-389] and the references which are cited therein.

The inequality (2.79) provides an important tool in deriving other classical inequalities. For instance, we shall obtain the (AM-GM)-inequality given in (2.18) by choosing the identity function f(x) = x in (2.79). The most useful consequence of the Theorem 29 is the inequality

$$xy \le \frac{x^q}{q} + \frac{y^p}{p},\tag{2.80}$$

for x, y are nonnegative numbers and p, q > 1 with 1/p + 1/q = 1, where it can be easily derived from the inequality (2.79) by choosing the exponential function $f(t) = t^{p-1}$ and q = p/(p-1). The equality holds in (2.80) if and only if $x^q = y^p$. This classical inequality (2.80) is called *Young's inequality* for the scalar products of real numbers. It appears as Theorem 61 in the remarkable work of Hardy, Littlewood and Pòlya [185, p. 61], but it is not credited to any individual.

This form of Young's inequality (2.80) can be used to establish some important inequalities such as the (CBS), Hölder and Minkowski inequalities [422]. The inequality (2.80) is itself, a special case of the (AM-GM)-inequality (2.18) when p = q = 2.

Now, let $a = x^q$, $b = y^p$, v = 1/q and 1 - v = 1/p. Then, the inequality (2.80) can be written in the form

$$a^{v}b^{1-v} \le va + (1-v)b,$$
 (2.81)

where $a, b \ge 0$ and $v \in [0, 1]$, with the equality holds in (2.81) if and only if a = b. This inequality gives us another useful form of the classical Young's inequality for two scalar products. We note that, if v = 1/2, then we obtain from (2.81) the basic result of the arithmetic-geometric mean inequality (2.18). This shows that, the inequality (2.81) is the generalizations of the weighted (AM-GM)-inequality (2.36), as it is mentioned earlier in Section 2.2.3.

In [250], Kittaneh and Manasrah defined the Heinz means as

$$H_v(a,b) = \frac{a^v b^{1-v} + a^{1-v} b^v}{2},$$
 (2.82)

for $a, b \ge 0, v \in [0, 1]$ and showed from the inequalities (2.80), (2.81) and (2.18) that the Heinz means interpolates between the arithmetic mean and geometric mean as follows:

$$\sqrt{ab} \le H_v(a,b) \le \frac{a+b}{2},\tag{2.83}$$

for $a, b \ge 0$ and $v \in [0, 1]$. Again, the equality holds in (2.81) if and only if a = b.

The famous Young's inequality (2.81) has been studied by several authors, see [10], [11], [157], [158], [198], [250] and the references cited therein. For instance, Hirzallah and Kittaneh [198] obtained the refinement of the inequality (2.81) as follows:

$$[va + (1 - v)b]^{2} - (a^{v}b^{1-v})^{2} \ge r^{2}(a - b)^{2},$$
(2.84)

for $a, b \ge 0$, $v \in [0, 1]$ and $r = \min\{v, 1 - v\}$. Kittaneh and Manasrah [250] provided the refinement of Young's inequality (2.81) in the following form:

$$va + (1 - v)b - a^{v}b^{1-v} \ge r\left(\sqrt{a} - \sqrt{b}\right)^{2},$$
 (2.85)

for all $a, b \ge 0$, $v \in [0, 1]$ and $r = \min\{v, 1 - v\}$. The following inequality has been proved by Aldaz [10],

$$2v\left(\frac{a+b}{2} - \sqrt{ab}\right) \leq va + (1-v)b - \left(a^vb^{1-v}\right)^2$$

$$\leq 2(1-v)\left(\frac{a+b}{2} - \sqrt{ab}\right), \tag{2.86}$$

for any $a, b \ge 0$ and $v \in [0, 1/2]$, which provided a refinement and a reverse of Young's inequality as well. This inequality (2.86) has been further studied by the same author in ([11], [12]) and Furuichi in [158].

A generalization of Young's inequality (2.81) was given by Furuichi in [159], namely

$$\sum_{j=1}^{n} p_j a_j - \prod_{j=1}^{n} a_j^{p_j} \ge n p_{\min} \left(\frac{1}{n} \sum_{j=1}^{n} a_j - \prod_{j=1}^{n} a_j^{1/n} \right), \tag{2.87}$$

for $a_j, p_j \ge 0, j \in \{1, 2, ..., n\}$ with $\sum_{j=1}^n p_j = 1$ and $p_{\min} = \min \{p_1, p_2, ..., p_n\}$. This inequality (2.87) becomes an equality if and only if $a_1 = a_2 = \cdots = a_n$. Note that, for n = 2, the inequality (2.87) reduces to Kittaneh and Manasrah's result (2.85).

Other generalizations of Young's inequality can be found in [13] and [14]. See also [21], [72, p. 205], [119], [160], [204], [259], [310], [335], [369], [423], [441] and the references cited therein, for some improvements of Young's inequality and their recent advances.

2.5 Hölder's Inequality

The classical Hölder's inequality for sequences of real numbers was first derived in 1888 by L. J. Rogers [378]. Then a year later, it was rediscovered in another way by O. L. Hölder [201] and named after him.

Hölder's inequality, which is also known is the literature as Rogers's inequality, was built around the two real numbers p and q that satisfy the conditions p > 1 and 1/p + 1/q = 1. This inequality asserts that for all nonnegative numbers a_k , $b_k \in \mathbb{R}$, $k \in \{1, 2, ..., n\}$ and p > 1 with 1/p + 1/q = 1, one has the bound [55, p. 68]

$$\sum_{k=1}^{n} a_k b_k \le \left(\sum_{k=1}^{n} a_k^p\right)^{1/p} \left(\sum_{k=1}^{n} b_k^q\right)^{1/q}, \tag{2.88}$$

with the equality holding in (2.88) if and only if the sequences (a_k^p) and (b_k^q) for $k \in \{1, 2, ..., n\}$ are proportional, i.e., there exists a real number $r \in \mathbb{R}$ such that $b_k^{1/p} = ra_k^{1/q}$ for all $k \in \{1, 2, ..., n\}$. The inequality (2.88) is reversed if p < 1 $(p \neq 0)$.

Hölder's inequality holds for complex numbers as well [318, p. 105], namely

$$\left| \sum_{k=1}^{n} a_k b_k \right| \le \left(\sum_{k=1}^{n} |a_k|^p \right)^{1/p} \left(\sum_{k=1}^{n} |b_k|^q \right)^{1/q}, \tag{2.89}$$

for all $a_k, b_k \in \mathbb{C}$, $k \in \{1, 2, ..., n\}$, p > 1 and q satisfying 1/p + 1/q = 1. The equality occurs in (2.89) if and only if the sequences $\{|a_k|^p\}$ and $\{|b_k|^q\}$, $k \in \{1, 2, ..., n\}$ are proportional and the $\arg(a_k b_k)$ is independent of k.

The following inequality also holds, which is called the weighted version of Hölder's inequality,

$$\left| \sum_{k=1}^{n} m_k a_k b_k \right| \le \left(\sum_{k=1}^{n} m_k |a_k|^p \right)^{1/p} \left(\sum_{k=1}^{n} m_k |b_k|^q \right)^{1/q}, \tag{2.90}$$

where $m_k \geq 0$, $a_k, b_k \in \mathbb{C}$, $k \in \{1, 2, ..., n\}$ and p > 1 such that 1/p + 1/q = 1. Again, the equality occurs in (2.90) if and only if the sequences $\{|a_k|^p\}$ and $\{|b_k|^q\}$ for $k \in \{1, 2, ..., n\}$ are proportional the $\arg(a_k b_k)$ is independent of k. We note that, if all the weights m_k for $k \in \{1, 2, ..., n\}$ are equal, then (2.90) reduces to the inequality (2.89).

Several proofs of Hölder's inequality (2.88) can be found in the literature. For instance, Tolsted in [422] (see also [301, p. 457], [381, p. 63-64]) showed that

Hölder's inequality (2.88) can be easily proved by using the standard Young's inequality (2.80).

The above inequality (2.88) can be written in the simpler form as follows:

$$\left| \sum_{k=1}^{n} a_k b_k \right| \le \|a\|_p \|b\|_q, \tag{2.91}$$

where for 1 , the pair <math>(p,q) are the usual conjugate exponents and the $||x||_p$ is the *p*-norm of *n*-tuple of real numbers x as defined by (2.54).

The inequality (2.88) also appears in ([4, p. 11], [55, p. 19-21], [185, p. 24, 26], [239, p. 67, 71], [318, p. 99]. One of the standard proofs of Hölder's inequality is based on Young's inequality (2.80) for the scalar products (see [422]). Alternative proof is by utilising the Jensen type inequality for the convex function $f(x) = x^p$, p > 1. We note that, the (CBS)-inequality (2.56), which is established in the previous section, is the special case of Hölder's inequality (2.88) with p = q = 2. In other words, we have to remark that Hölder's inequality is one of the remarkable extensions of the (CBS)-inequality.

The inequality (2.89) is also valid for countable infinite pairs of numbers, where from the convergence of the series on the right-hand side $\sum_{k=1}^{\infty} |a_k b_k|$, the convergent of the left-hand side follows.

Remark 30 Hölder's inequality (2.89) is also valid for countable infinite pairs of numbers, i.e.,

$$\left| \sum_{k=1}^{\infty} a_k b_k \right| \le \left(\sum_{k=1}^{\infty} |a_k|^p \right)^{1/p} \left(\sum_{k=1}^{\infty} |b_k|^q \right)^{1/q}, \tag{2.92}$$

where from the convergence of the series on the right-hand side, the convergence of the left-hand side follows [77, p. 32].

If now, we consider an analytic function f(z) defined by the real power series (2.65) with real or nonnegative coefficients and convergent on the interval (-R, R), R > 0, and applying the weighted version of the Hölder inequality for real numbers, then we can state that

$$f(xy) = \sum_{n=0}^{\infty} a_n x^n y^n \le \left(\sum_{n=0}^{\infty} a_n x^{pn}\right)^{1/p} \left(\sum_{n=0}^{\infty} a_n y^{qn}\right)^{1/q}$$
$$= f^{1/p}(x^p) f^{1/q}(y^q)$$
(2.93)

for any $x, y \in \mathbb{R}$ such as $xy, x^p, y^q < R$. A similar result can be obtained by applying the complex power series (2.65) with real coefficients for the weighted version of the Hölder inequality (2.90), namely

$$|f(xy)| = \left| \sum_{n=0}^{\infty} a_n x^n y^n \right| \le \left(\sum_{n=0}^{\infty} |a_n| |x|^{pn} \right)^{1/p} \left(\sum_{n=0}^{\infty} |a_n| |y|^{qn} \right)^{1/q}$$

$$= f_A^{1/p} (|x|^p) f_A^{1/q} (|y|^q), \qquad (2.94)$$

for any $x, y \in \mathbb{C}$ with $xy, |x|^p, |y|^q \in D(0, R)$ and $f_A(z)$ is the power series defined by (2.69).

Various extensions, generalizations, refinements, etc. of the Hölder inequality have been obtained by many authors, see for instance [3], [87], [112], [192], [248], [271], [317], [365], [436], [437], [438] and the numerous references which are cited therein.

2.6 Jensen Types Inequalities and Their Reverses

Many important classical inequalities depend upon convexity. One of them is Jensen's inequality. Jensen's inequality, which was proved by Jensen ([222], [223]), concerned with the bounding of convex functions of sums or integrals (see also [79, p. 31], [91, p. 43], [317, p.], [334, p. 12]). It is one of the most important inequalities and has had a tremendous impact on many different fields such as Probability and Statistics Theory, Information Theory, Control and Systems Theory, etc. Since its discovery in 1906 [223], Jensen's inequality has been proven to be one of the most useful inequalities in mathematical analysis, because

it implies many of other classical results in Inequalities Theory. The triangle, (AM-GM), (CBS), Young, Hölder, Minkowski, Schnnon, Ky Fan inequalities, etc., can be obtained as particular cases of Jensen's inequality. For classical and contemporary developments related to Jensen's inequality, we may refer to the works of ([56], [101], [119], [140], [318, Chapt. I]) and the references, which are cited therein.

There are several types of Jensen's inequality that appear in the literature. The classical form of Jensen's inequality involving real numbers and weights is stated as follows (see [79, p. 31], [317, p. 6]). Let f be a convex function on an interval $I \subset \mathbb{R}$ and p_j , $j \in \{1, 2, ..., n\}$ be nonnegative scalars with $P_n := \sum_{j=1}^n p_j > 0$, $n \geq 2$. Then,

$$f\left(\frac{1}{P_n}\sum_{j=1}^n p_j x_j\right) \le \frac{1}{P_n}\sum_{j=1}^n p_j f\left(x_j\right),\tag{2.95}$$

for $x_j \in I$ such that $x_j \geq 0$ for each $j \in \{1, 2, ..., n\}$, with equality holds in (2.95) if and only if $x_j = x_k$ for all $j, k \in \{1, 2, ..., n\}$. If f(x) is strictly convex for all the distinct points x_j and all weights p_j , $j \in \{1, 2, ..., n\}$ are positive, then the strict inequality holds in (2.95). The above inequality (2.95) is known as the weighted version of Jensen inequality with p_j , $j \in \{1, 2, ..., n\}$ are called the weights.

The reverse of the Jensen inequality holds, namely (see [79, p. 43], [318, p. 6], [427])

$$f\left(\frac{1}{P_n}\sum_{j=1}^n p_j x_j\right) \ge \frac{1}{P_n}\sum_{j=1}^n p_j f(x_j),$$
 (2.96)

for a convex function f on I, $x_j \in I$, $j \in \{1, 2, ..., n\}$, $p_1 > 0$, $p_j < 0$ for $2 \le j \le n$ and $P = \frac{1}{P_n} \sum_{j=1}^n p_j x_j \in I$. On the other hand, the inequality in (2.96), which provides the lower bound instead of the upper bound, particularly holds for concave functions f(x) on I.

We note that, if n=2, then we have from (2.95) the inequality

$$f\left(\frac{px+qy}{p+q}\right) \le \frac{pf(x)+qf(y)}{p+q},\tag{2.97}$$

for any $x, y \in I$, where $p, q \ge 0$ such that p + q > 0. Thus, the inequality in (2.39) as shown in the previous section, is the special case of (2.95) when p = q = 1. Another remarkable particular case is obtained when $p_j = 1/n$ for all $j \in \{1, 2, ..., n\}$. That is, the inequality (2.95) becomes [79, p. 32]

$$f\left(\frac{1}{n}\sum_{j=1}^{n}x_{j}\right) \leq \frac{1}{n}\sum_{j=1}^{n}f\left(x_{j}\right),\tag{2.98}$$

holding for a convex function f on the interval I and $x_j \in I$, $j \in \{1, 2, ..., n\}$. Again, the sign of inequality reverses in (2.98) for concave functions. Obviously, the inequality (2.98) holds for every Jensen-convex function f on I. For an analytic proof of the Jensen type inequalities, we may refer to the works of ([185, p. 71], [352, p. 53]).

Further, we have the following inequality

$$f\left(\sum_{j=1}^{n} q_j x_j\right) \le \sum_{j=1}^{n} q_j f\left(x_j\right), \tag{2.99}$$

which holds for a convex function $f: I \to \mathbb{R}, x_j \in I, j \in \{1, 2, ..., n\}$ and q_j are probabilities, namely $q_j \geq 0$ for $j \in \{1, 2, ..., n\}$ such that $\sum_{j=1}^n q_j = 1$. The equality holds in (2.99) if and only if all the $x_j, j \in \{1, 2, ..., n\}$ are equal, or if f(x) reduces to a linear function. In a real sense, the Jensen type inequality given by (2.99) is the extended property of the inequality in (2.38), in which it defines a convex function for the case n = 2.

The forms of Jensen's discrete inequality given by (2.98), (2.95) and (2.99) as well as their reverses (2.96) allow us to derive some other important classical results in Inequalities Theory. Through judicious choice of convex (or concave) functions f, we can derive, for instance, the generalized triangle inequality (2.25) for real numbers, by choosing in (2.99) the convex function $f(x) = |x|, x \in \mathbb{R}$. Jensen's inequality (2.96) reduces to the celebrated (CBS)-inequality (2.56) for the convex function $f(x) = x^2$, and noting that $x_j = b_j/a_j$ and $p_j = a_j^2$ for $j \in \{1, 2, ..., n\}$. Taking the function $f(x) = x^p$, $q_j = b_j^q$ and $x_j = \left(a_j^p/b_j^q\right)^{1/p}$ with $a_j, b_j > 0$ for $j \in \{1, 2, ..., n\}$ and p, q > 0 such that 1/p + 1/q = 1, we can prove the famous Hölder's inequality.

Some studies related to Jensen's inequality concerning generalizations, extensions, refinements, reverses, counterparts, etc., can be found in (see for example [27], [56], [57], [80, p. 139-142], [101], [126], [129], [130], [139], [273], [318, p. 1-20]). Also, see [286], [322], [351] for further references.

2.7 Other Inequalities

In this section, we state some other important classical inequalities in analysis, which are Minkowski's and Čebyšev's inequalities. *Minkowski's inequality* was established in 1896 by Hermann Minkowski [313] in his book "Geometrie der Zahlen" (Geometry of Numbers). It states that for p > 1, one has the bound

$$\left(\sum_{k=t}^{n} |a_k + b_k|^p\right)^{1/p} \le \left(\sum_{k=1}^{n} |a_k|^p\right)^{1/p} + \left(\sum_{k=1}^{n} |b_k|^p\right)^{1/p}, \tag{2.100}$$

for any $a_k, b_k \in \mathbb{C}$, $k \in \{1, 2, ..., n\}$. Equality holds in (2.100) if and only if there exist a constant $\lambda \in \mathbb{R}$ such that $|b_k| = \lambda |a_k|$ for all $k \in \{1, 2, ..., n\}$. If p < 1 $(p \neq 0)$, then the inequality sign in (2.100) is reversed. This inequality arises as Theorem 25 in the work of Hardy, Littlewood and Pòlya [185, p. 31], see also [4, p. 11], [55, p. 69], [185, p. 31], [239, p. 69-70] and the references cited therein. We note that the special case of (2.100) with p = 2 is the triangle inequality for arbitrary complex numbers.

The inequality (2.100) can be generalized for their weighted version as follows,

$$\left(\sum_{k=1}^{n} m_k |a_k + b_k|\right)^{1/p} \le \left(\sum_{k=1}^{n} m_k |a_k|^p\right)^{1/p} + \left(\sum_{k=1}^{n} m_k |b_k|^p\right)^{1/p}, \quad (2.101)$$

for any $a_k, b_k \in \mathbb{C}$, $m_k \geq 0$ for $k \in \{1, 2, ..., n\}$, p > 1 and again the equality holds in (2.101) if and only if the sequences $\{a_k\}$ and $\{b_k\}$ are proportional.

Čebyšev's inequality provides the lower bound of $\sum_{k=1}^{n} a_k b_k$, which says that if $a_1 \leq a_2 \leq \cdots \leq a_n$ and $b_1 \leq b_2 \leq \cdots \leq b_n$ be real numbers, then

$$\left(\sum_{k=1}^{n} a_k\right) \left(\sum_{k=1}^{n} b_k\right) \le n \sum_{k=1}^{n} a_k b_k, \tag{2.102}$$

with equality occurring if and only if, either all the a_k , $k \in \{1, 2, ..., n\}$ are equal or all the b_k are equal. It is important to order, either the $a_k s$ and $b_k s$ in both descending and ascending order in the inequality (2.102).

Remark 31 The inequality (2.102) is also true in the case when $a_1 \ge a_2 \ge \cdots \ge a_n$ and $b_1 \ge b_2 \ge \cdots \ge b_n$, but if $a_1 \le a_2 \le \cdots \le a_n$ and $b_1 \ge b_2 \ge \cdots \ge b_n$ (or the reverse), then the sign in inequality (2.102) is reversed.

In terms of its weighted version, Čebyšev's inequality can be written as follow:

$$\left(\sum_{k=1}^{n} a_k m_k\right) \left(\sum_{k=1}^{n} b_k m_k\right) \le \sum_{k=1}^{n} a_k b_k m_k, \tag{2.103}$$

for $a_1 \leq a_2 \leq \cdots \leq a_n$, $b_1 \leq b_2 \leq \cdots \leq b_n$ be real numbers and m_1, m_2, \ldots, m_n be nonnegative real numbers such that $m_1 + m_2 + \ldots + m_n = 1$. The equality occurs if and only if $a_1 = a_2 = \ldots = a_n$ or $b_1 = b_2 = \ldots = b_n$. Note that, if in (2.103) we choose $m_1 = m_2 = \ldots = m_n = 1/n$, then we get the inequality (2.102).

Chapter 3

Power Series and the Cauchy-Bunyakovsky-Schwarz Inequality

Chapter 3 is devoted to some inequalities concerning the power series that are related to the (CBS)-type inequalities. It is well-known that the classical (CBS)-type inequalities have been generalized, extended, refined and applied by a remarkable large number of researchers for different and various motivations (see for instance [117], [120, Chapt. 1], [121, Chapt. 2], [127]). In this chapter, we provide some generalizations, improvements and refinements of the (CBS)-inequality for functions defined by the power series with real or nonnegative coefficients and convergent on an open disk.

We start this chapter with some known results on the power series inequalities proved by Dragomir [122], Cerone and Dragomir [89]. In Section 3.2, we derive new and better inequalities for functions defined by the power series with real coefficients, by utilizing Buzano's result in inner product spaces. Particular inequalities are obtained by applying the results for some fundamental functions of interest, such as the exponential, logarithm, trigonometric and hyperbolic functions.

In Section 3.3, we utilize a refinement of the celebrated (CBS)-inequality in inner product spaces established by Dragomir in [118], to develop some other

inequalities for functions defined by the power series with real and nonnegative coefficients. Some applications for elementary complex functions of interest are also presented.

We end this chapter with some more inequalities involving the power series functions that have been developed by utilising a different technique based on the continuity properties of modulus. Those results have natural applications for fundamental functions as well, which is a subsequent aim of this section.

All the results contained in this chapter, are mainly taken from the author's research papers in collaboration with Dragomir and Darus (see [208], [209], [214]). The same results can also be found in the survey research paper published by the author and Dragomir (see [217]).

3.1 Introduction and Preliminary Results

The (CBS)-inequality is one of the most important inequalities in analysis. Nowadays, a large number of results concerning new proofs, noteworthy extensions, generalizations, refinements, etc., of the classical (CBS)-inequality have been published in the literature. Most of the results are discrete and involve finite sums. A few attempts to extend those results to the infinite series have been considered by several authors (see [49], [51], [89], [122], [123], [173]). For instance, in [122], Dragomir provided the generalization of the (CBS)-inequality for functions defined by the real power series (2.65) with nonnegative coefficients and convergent on the interval (-R, R), R > 0. The result is stated as follows (see also [117, p. 19]):

Theorem 32 (Dragomir [122]) Let $f: (-R,R) \to \mathbb{R}$, $f(x) = \sum_{k=0}^{\infty} \alpha_k x^k$ with $\alpha_k \geq 0$, $k \in \mathbb{N}$. If $a = (a_1, a_2, \dots, a_n)$ and $b = (b_1, b_2, \dots, b_n)$ are sequences of real numbers such that $a_k b_k$, a_k^2 , $b_k^2 \in (-R,R)$ for any $j \in \{1, 2, \dots, n\}$, then one has the inequality

$$\left(\sum_{k=1}^{n} f(a_k b_k)\right)^2 \le \sum_{k=1}^{n} f(a_k^2) \sum_{k=1}^{n} f(b_k^2).$$
 (3.1)

We note that for f(x) = x, $x \in \mathbb{R}$ in (3.1), we simply recapture the classical (CBS)-inequality for real numbers (2.56).

Particular inequalities of (3.1) for some fundamental functions are also given in [122]. For instance, if a and b are sequences of real numbers, then one has the inequality

$$\left(\sum_{j=1}^{n} \exp\left(a_{j}b_{j}\right)\right)^{2} \leq \sum_{j=1}^{n} \exp\left(a_{j}^{2}\right) \sum_{j=1}^{n} \exp\left(b_{j}^{2}\right),$$

$$\left(\sum_{j=1}^{n} \sinh\left(a_{j}b_{j}\right)\right)^{2} \leq \sum_{j=1}^{n} \sinh\left(a_{j}^{2}\right) \sum_{j=1}^{n} \sinh\left(b_{j}^{2}\right),$$

$$\left(\sum_{j=1}^{n} \cosh\left(a_{j}b_{j}\right)\right)^{2} \leq \sum_{j=1}^{n} \cosh\left(a_{j}^{2}\right) \sum_{j=1}^{n} \cosh\left(b_{j}^{2}\right).$$

$$(3.2)$$

Other inequalities concerning the power series (2.65) with real or nonnegative coefficients, have been established by Cerone and Dragomir in [89]. On utilising the known result that has been available in the literature, which is called the de Bruijn inequality (2.73), they obtained some refinements of the (CBS)-inequality for functions defined by the power series (2.68) and (2.72). The results are summarized as follows:

Theorem 33 (Cerone and Dragomir [89]) Let $f(z) = \sum_{k=0}^{\infty} a_k z^k$ be an analytic function defined by a power series with nonnegative coefficients a_k , $k \geq 0$ and convergent on the open disk $D(0,R) \subset \mathbb{C}$, R > 0. If a is a real number and z a complex number such that az, a^2 , z^2 , $|z|^2 \in D(0,R)$, then

$$|f(az)|^2 \le \frac{1}{2}f(a^2)[f(|z|^2) + |f(z^2)|].$$
 (3.3)

Cerone and Dragomir [89] have also proved an analogous inequality of (3.3) for functions defined by the complex power series with real coefficients.

Theorem 34 (Cerone and Dragomir [89]) Let $f(z) = \sum_{k=0}^{\infty} a_k z^k$ be a function defined by a power series with real coefficients and convergent on the open

disk $D(0,R) \subset \mathbb{C}$, R > 0. If $a \in \mathbb{R}$ and $z \in \mathbb{C}$ are such that az, a^2 , $|z|^2 \in D(0,R)$, then

$$|f(az)|^2 \le \frac{1}{2} f_A(a^2) \left[f_A(|z|^2) + |f_A(z^2)| \right].$$
 (3.4)

Particularly, the results given by (3.3) and (3.4) provide the improvements of the de Bruijn inequality (2.73) as well as the refinements of the inequalities (2.68) and (2.72) of the (CBS)-type for functions defined by the complex power series with real or nonnegative coefficients and convergent on the open disk.

Particular inequalities for various examples of fundamental functions such as the exponential, logarithm, trigonometric and hyperbolic functions are also pointed out in [89]. For instance,

$$\left[1 - |z|^{2} + |1 - z|\right] |1 - az|^{2} \ge 2 \left(1 - a^{2}\right) \left(1 - |z|^{2}\right) |1 - z^{2}|,$$

$$\left|\ln\left(1 - az\right)\right|^{2} \le \frac{1}{2} \ln\left(\frac{1}{1 - a^{2}}\right) \left[\ln\left(\frac{1}{1 - |z|^{2}}\right) + \left|\ln\left(1 - z^{2}\right)\right|\right],$$

$$\left|\sin^{-1}\left(az\right)\right|^{2} \le \frac{1}{2} \sin^{-1}\left(a^{2}\right) \left[\sin^{-1}\left(|z|^{2}\right) + \left|\sin^{-1}\left(z^{2}\right)\right|\right],$$

$$\left|\tanh^{-1}\left(az\right)\right|^{2} \le \frac{1}{2} \tanh^{-1}\left(a^{2}\right) \left[\tanh^{-1}\left(|z|^{2}\right) + \left|\tanh^{-1}\left(z^{2}\right)\right|\right],$$
(3.5)

for any $a \in (-1,1)$ and $z \in D(0,1)$. Applications for special function such as polylogarithm, are given as well.

Motivated by the results given by (3.3), (3.4) and some other results established in [89], we derive new and better inequalities for functions defined by the power series with real or nonnegative coefficients, by utilising Buzano's and Schwarz's result in complex inner product spaces, and a different technique based on the continuity properties of modulus. Natural applications for fundamental functions of interest such as the exponential, logarithm, trigonometric and hyperbolic are also provided.

3.2 Power Series Inequalities Via Buzano's Results

In [124], Dragomir has observed that from [78], on utilizing Buzano's inequality (2.75) in the complex inner product space $(H; \langle \cdot, \cdot \rangle)$, where the inner product is defined by (2.48), one can obtain the following discrete inequality:

$$\left| \sum_{j=1}^{n} p_{j} c_{j} \overline{x_{j}} \sum_{j=1}^{n} p_{j} x_{j} \overline{b_{j}} \right|$$

$$\leq \frac{1}{2} \left[\left(\sum_{j=1}^{n} p_{j} |c_{j}|^{2} \sum_{j=1}^{n} p_{j} |b_{j}|^{2} \right)^{1/2} + \left| \sum_{j=1}^{n} p_{j} c_{j} \overline{b_{j}} \right| \right] \sum_{j=1}^{n} p_{j} |x_{j}|^{2},$$

$$(3.6)$$

where $p_j \geq 0$, $x_j, b_j, c_j \in \mathbb{C}$, $j \in \{1, ..., n\}$. If we take in (3.6) $b_j = \overline{c_j}$ for $j \in \{1, 2, ..., n\}$, then we get

$$\left| \sum_{j=1}^{n} p_{j} c_{j} \overline{x_{j}} \sum_{j=1}^{n} p_{j} c_{j} x_{j} \right| \leq \frac{1}{2} \left[\sum_{j=1}^{n} p_{j} \left| c_{j} \right|^{2} + \left| \sum_{j=1}^{n} p_{j} c_{j}^{2} \right| \right] \sum_{j=1}^{n} p_{j} \left| x_{j} \right|^{2}, \tag{3.7}$$

for any $p_i \ge 0$, x_i , $c_i \in \mathbb{C}$, $j \in \{1, 2, ..., n\}$.

As pointed out in [124] and also in [120, p. 49], if x_j , $j \in \{1, 2, ..., n\}$, are real numbers, then (3.6) generates the de Bruijn refinement of the celebrated weighted (CBS)-inequality, namely

$$\left| \sum_{j=1}^{n} p_j x_j z_j \right|^2 \le \frac{1}{2} \sum_{j=1}^{n} p_j x_j^2 \left[\sum_{j=1}^{n} p_j |z_j|^2 + \left| \sum_{j=1}^{n} p_j z_j^2 \right| \right], \tag{3.8}$$

where $p_j \geq 0$, $x_j \in \mathbb{R}$, $z_j \in \mathbb{C}$, $j \in \{1, 2, ..., n\}$. Therefore, the Buzano's result (3.6) may be regarded as a generalization of the de Bruijn inequality (2.73) as well.

First, we prove the following result [208] that has been obtained on utilising Buzano's result (3.6), see also [217].

Theorem 35 (Ibrahim and Dragomir [208]) Let $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be a power series with nonnegative coefficients a_n and convergent on the open disk D(0,R). If $x, \alpha, \beta \in \mathbb{C}$ so that $\alpha \overline{x}, \overline{\beta} x, |\alpha|^2, \beta^2, a \overline{\beta}, |x|^2 \in D(0,R)$, then

$$\left| f\left(\alpha \overline{x}\right) f\left(\overline{\beta} x\right) \right| \leq \frac{1}{2} \left[\left[f\left(|\alpha|^2\right) f\left(|\beta|^2\right) \right]^{1/2} + \left| f\left(\alpha \overline{\beta}\right) \right| \right] f\left(|x|^2\right). \tag{3.9}$$

Proof. On utilizing the inequality (3.6) for the choices $p_n = a_n$, $c_n = \alpha^n$, $x_n = x^n$ and $b_n = \beta^n$, $n \ge 0$, we have

$$\left| \sum_{n=0}^{m} a_n \alpha^n (\overline{x})^n \sum_{n=0}^{m} a_n (\overline{\beta})^n x^n \right|$$

$$\leq \frac{1}{2} \left[\left(\sum_{n=0}^{m} a_n |\alpha|^{2n} \sum_{n=0}^{m} a_n |\beta|^{2n} \right)^{1/2} + \left| \sum_{n=0}^{m} a_n \alpha^n (\overline{\beta})^n \right| \right] \sum_{n=0}^{m} a_n |x|^{2n}, \quad (3.10)$$

for any $m \geq 0$.

Since $\alpha \overline{x}$, $\overline{\beta}x$, $|\alpha|^2$, $|\beta|^2$, $\alpha \overline{\beta}$, $|x|^2$ belong to the convergence disk D(0,R), hence the series in (3.10) are convergent and letting $m \to \infty$, we deduce the desired inequality (3.9).

A particular case of interest is as follows:

Corollary 36 Let f(z) be as in Theorem 35 and $z, x \in \mathbb{C}$ with $z\overline{x}, zx, |z|^2, |z|^2 \in D(0, R)$. Then,

$$|f(z\overline{x}) f(zx)| \le \frac{1}{2} \left[f(|z|^2) + |f(z^2)| \right] f(|x|^2). \tag{3.11}$$

Proof. This follows from Theorem 35 by choosing $\alpha = z$ and $\beta = \overline{z}$.

Remark 37 In particular, if $x = a \in \mathbb{R}$, then from (3.11) we deduce the inequality (3.3) [89].

The above result (3.9) has some natural applications for particular complex functions of interest as follows:

(1) If we apply the inequality (3.9) for the function f(z) given by (1.19), then we get

$$\left| \frac{1}{1 - \alpha \overline{x}} \cdot \frac{1}{1 - \overline{\beta} x} \right|$$

$$\leq \frac{1}{2} \left[\left(\frac{1}{1 - |\alpha|^2} \cdot \frac{1}{1 - |\beta|^2} \right)^{1/2} + \left| \frac{1}{1 - \alpha \overline{\beta}} \right| \right] \frac{1}{1 - |x|^2},$$
(3.12)

for any x, α , $\beta \in D(0,1)$. This is equivalent with

$$2\left(1-|x|^{2}\right)\left|1-\alpha\overline{\beta}\right|\sqrt{\left(1-|\alpha|^{2}\right)\left(1-|\beta|\right)^{2}}$$

$$\leq\left|1-\alpha\overline{x}\right|\left|1-\overline{\beta}x\right|\left[\left|1-\alpha\overline{\beta}\right|+\sqrt{\left(1-|\alpha|^{2}\right)\left(1-|\beta|\right)^{2}}\right],\tag{3.13}$$

for $x, \alpha, \beta \in D(0,1)$. In particular, if $\beta = \overline{\alpha}$, then we get from (3.13) that

$$2(1 - |x|) (1 - |\alpha|^2) |1 - \alpha^2|$$

$$\leq |1 - \alpha \overline{x}| |1 - \alpha x| \left[|1 - \alpha^2| + 1 - |\alpha|^2 \right], \tag{3.14}$$

for any $x, \alpha \in D(0,1)$.

(2) If we apply (3.9) for the exponential function f(z) given by (1.25), then we get the inequality

$$\left| \exp\left(\alpha \overline{x} + \overline{\beta}x\right) \right|$$

$$\leq \frac{1}{2} \left[\left(\exp\left(|\alpha|^2 + |\beta|^2 \right) \right)^{1/2} + \left| \exp\left(\alpha \overline{\beta} \right) \right| \right] \exp\left(|x|^2 \right),$$
(3.15)

for any $\alpha, \beta, x \in \mathbb{C}$. In particular, if $\alpha = \overline{\beta}$, then we get from (3.15) that

$$|\exp(2\alpha \operatorname{Re}(x))| \le \frac{1}{2} \left[\left(\exp(2|\alpha|^2) \right)^{1/2} + \left| \exp(\alpha^2) \right| \right] \exp(|x|^2), \quad (3.16)$$

for any $\alpha, x \in \mathbb{C}$.

(3) If we apply (3.9) for the Koebe function f(z) defined by (1.23), then we get

$$\left| \frac{\alpha \overline{\beta} |x|^2}{(1 - \alpha \overline{x})^2 (1 - \overline{\beta} x)^2} \right| \leq \frac{1}{2} \left(\frac{|\alpha \beta|}{(1 - |\alpha|^2) (1 - |\beta|^2)} + \left| \frac{\alpha \overline{\beta}}{(1 - \alpha \overline{\beta})^2} \right| \right) \frac{|x|^2}{(1 - |x|^2)^2}, \tag{3.17}$$

for any x, α , $\beta \in D(0,1)$. If we simplify the above inequality (3.17), then we have

$$\frac{1 - |x|^2}{\left| (1 - \alpha \overline{x}) \left(1 - \overline{\beta} x \right) \right|}$$

$$\leq \left(\frac{1}{2 \left(1 - |\alpha|^2 \right) \left(1 - |\beta|^2 \right)} + \frac{1}{2 \left| 1 - \alpha \overline{\beta} \right|^2} \right)^{1/2}, \tag{3.18}$$

for any α , β , $x \in D(0,1)$. In particular if $\beta = \overline{\alpha}$, then we get from (3.18) that

$$\frac{1 - |x|^2}{|(1 - \alpha \overline{x})(1 - \alpha x)|} \le \left(\frac{1}{2(1 - |\alpha|^2)^2} + \frac{1}{2|1 - \alpha^2|^2}\right)^{1/2},\tag{3.19}$$

for any α , $x \in D(0,1)$.

(4) If we apply the same inequality (3.9) for the hyperbolic function f(z) given by (1.39), then we obtain

$$\left|\cosh\left(\alpha \overline{x} + \overline{\beta}x\right) + \cosh\left(\alpha \overline{x} - \overline{\beta}x\right)\right|$$

$$\leq \left(\left[\frac{1}{2}\left(\cosh\left(|\alpha|^2 + |\beta|^2\right) + \cosh\left(|\alpha|^2 - |\beta|^2\right)\right)\right]^{1/2}$$

$$+ \left|\cosh\left(\alpha \overline{\beta}\right)\right| \cosh\left(|x|^2\right), \tag{3.20}$$

for any $x, \alpha, \beta \in \mathbb{C}$. In particular, for $\beta = \overline{\alpha}$, we get that from (3.20)

$$|\cosh(2\alpha \operatorname{Re}(x)) + \cosh(2i\alpha \operatorname{Im}(x))|$$

$$\leq \left[\cosh(|\alpha|^2) + \left|\cosh(\alpha^2)\right|\right] \cosh(|x|^2), \qquad (3.21)$$

that holds for any α , $x \in \mathbb{C}$.

We also obtain the analogous result to (3.9), which connects the power series f(z) with its transform f_A [208] (see also [217]).

Theorem 38 (Ibrahim and Dragomir [208]) Let $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be a function defined by a power series with real coefficients and convergent on the open disk $D(0,R) \subset \mathbb{C}$, R > 0. If α , β , x are complex numbers such that $\alpha \overline{x}$, $\overline{\beta}x$, $\alpha \overline{\beta}$, $|\alpha|^2$, $|\beta|^2$, $|x^2| \in D(0,R)$, then

$$\left| f\left(\alpha \overline{x}\right) f\left(\overline{\beta} x\right) \right| \leq \frac{1}{2} \left(\left[f_A\left(\left|\alpha\right|^2\right) f_A\left(\left|\beta\right|^2\right) \right]^{1/2} + \left| f_A\left(\alpha \overline{\beta}\right) \right| \right) f_A\left(\left|x\right|^2\right). \tag{3.22}$$

Proof. By choosing $p_n = |a_n|$, $c_n = \alpha^n$, $b_n = \beta^n$ and $x_n = sgn(a_n) x^n$, $n \ge 0$ in (3.6), where sgn(t) is a real sign function defined by (2.11), we have

$$\left| \sum_{n=0}^{m} a_n \left(\alpha \overline{x} \right)^n \sum_{n=0}^{m} a_n \left(\overline{\beta} x \right)^n \right|$$

$$= \left| \sum_{n=0}^{m} |a_n| \operatorname{sgn} \left(a_n \right) \alpha^n \left(\overline{x} \right)^n \sum_{n=0}^{m} |a_n| \operatorname{sgn} \left(a_n \right) x^n \left(\overline{\beta} \right)^n \right|,$$

$$\leq \frac{1}{2} \left(\left[\sum_{n=0}^{m} |a_n| \left(|\alpha|^2 \right)^n \sum_{n=0}^{m} |a_n| \left(|\beta|^2 \right)^n \right]^{1/2} + \left| \sum_{n=0}^{m} |a_n| \left(\alpha \overline{\beta} \right)^n \right| \right)$$

$$\times \sum_{n=0}^{m} |a_n| \left(|x|^2 \right)^n, \tag{3.23}$$

for any $\alpha, \beta, x \in \mathbb{C}$ with $\alpha \overline{x}, \overline{\beta} x, \alpha \overline{\beta}, |\alpha|^2, |\beta|^2, |x|^2 \in D(0, R)$. Now, taking the limit as $m \to \infty$ in (3.23) and noticing that all the involved series in (3.23) are convergent, then we deduce the desired inequality (3.22).

Remark 39 Taking in (3.22) $\alpha = z \in \mathbb{C}$, $\beta = \overline{z}$ and $x = a \in R$ will produce the inequality (3.4) [89].

In what follows, we provide some applications of the inequality (3.22) for particular functions of interest.

(1) If we take the function f(z) given by (1.20), then we have the transform $f_A(z)$ given by (1.19). Applying Theorem 38, we get the following inequality

$$2\left|1 - \alpha\overline{\beta}\right| \left(1 - |x|^2\right) \left[\left(1 - |\alpha|^2\right) \left(1 - |\beta|^2\right)\right]^{1/2}$$

$$\leq |1 + \alpha\overline{x}| \left|1 + \overline{\beta}x\right| \left(\left|1 - \alpha\overline{\beta}\right| + \left[\left(1 - |\alpha|^2\right) \left(1 - |\beta|^2\right)\right]^{1/2}\right), \quad (3.24)$$

for any $\alpha, \beta, x \in D(0,1)$. In particular, if $\alpha = \overline{\beta}$, then from (3.24) we obtain

$$2|1 - \alpha^{2}|(1 - |\alpha|^{2})(1 - |x|^{2})$$

$$\leq |1 + \alpha x||1 + \alpha \overline{x}|(1 - |\alpha|^{2} + |1 - \alpha^{2}|), \qquad (3.25)$$

for any $\alpha, x \in D(0, 1)$.

(2) For the exponential function f(z) given by (1.26), we have the transform

$$f_A(z) = \sum_{n=0}^{\infty} \frac{z^n}{n!} = e^z, \ z \in \mathbb{C}.$$
 (3.26)

Utilising the inequality (3.22) we obtain

$$\frac{1}{\left|\exp\left(\alpha \overline{x} + \overline{\beta}x\right)\right|} \le \frac{1}{2} \left(\exp\left[\frac{1}{2}\left(|\alpha|^2 + |\beta|^2\right) + |x|^2\right] + \exp\left(|x|^2\right) \left|\exp\left(\alpha \overline{\beta}\right)\right|\right), \quad (3.27)$$

for any $\alpha, \beta, x \in \mathbb{C}$. In particular, if $\alpha = \overline{\beta}$ in (3.27), then we get

$$\frac{1}{\left|\exp\left(2\alpha\operatorname{Re}\left(x\right)\right)\right|} \le \frac{1}{2} \left[\exp\left(\left|\alpha\right|^{2} + \left|\beta\right|^{2}\right) + \exp\left(\left|x\right|^{2}\right) \left|\exp\left(\alpha^{2}\right)\right|\right], \quad (3.28)$$

for any $\alpha, x \in \mathbb{C}$.

(3) If in (3.22) we choose the trigonometric function f(z) given by (1.33), then the transform $f_A(z)$ is given by (1.39). Applying the inequality (3.22) will produce the following result:

$$\left|\cos\left(\alpha \overline{x}\right)\cos\left(\overline{\beta}x\right)\right|$$

$$\leq \frac{1}{2} \left(\left[\cosh\left(|\alpha|^2 \right) \cosh\left(|\beta|^2 \right) \right]^{1/2} + \left| \cosh\left(\alpha \overline{\beta} \right) \right| \right) \cosh\left(|x|^2 \right), \qquad (3.29)$$

for any α , β , $x \in \mathbb{C}$. In particular, if we choose $\alpha = \overline{\beta}$ in (3.29), then we obtain the inequality

$$\left|\cos\left(\alpha \overline{x}\right)\cos\left(\alpha x\right)\right| \le \frac{1}{2} \left[\cosh\left(\left|\alpha\right|^{2}\right) + \left|\cosh\left(\alpha^{2}\right)\right|\right] \cosh\left(\left|x\right|^{2}\right), \quad (3.30)$$

for any α , $x \in \mathbb{C}$.

Next, we have proved the following result [208], which connects the two power series, one having positive coefficients (see also [217]).

Theorem 40 (Ibrahim and Dragomir [208]) Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$ and $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be two power series with $g_n \in \mathbb{C}$ and $a_n > 0$ for $n \geq 0$. If f and g are convergent on $D(0, R_1)$ and $D(0, R_2)$ respectively, and the numerical series $\sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n}$ is convergent, then we have the inequality:

$$|g(z)g(\overline{z})| \le \frac{1}{2} \sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n} \left[f(|z|^2) + \left| f(z^2) \right| \right], \tag{3.31}$$

for any $z \in \mathbb{C}$ with $z, z^2, |z|^2 \in D(0, R_1) \cap D(0, R_2)$.

Proof. On utilizing the inequality (3.7) for the choices $p_n = a_n$, $c_n = z^n$, $x_n = g_n/a_n$, $n \ge 0$, we have

$$\left| \sum_{n=0}^{m} g_n z^n \sum_{n=0}^{m} \overline{g_n} z^n \right| \le \frac{1}{2} \left[\sum_{n=0}^{m} a_n \left(|z|^2 \right)^n + \left| \sum_{n=0}^{m} a_n \left(z^2 \right)^n \right| \right] \sum_{n=0}^{m} \frac{|g_n|^2}{a_n}, \quad (3.32)$$

for any $m \geq 0$. We observe that

$$\sum_{n=0}^{m} \overline{g_n} z^n = \overline{\sum_{n=0}^{m} g_n(\overline{z})^n} \quad \text{and then} \quad \left| \sum_{n=0}^{m} \overline{g_n} z^n \right| = \left| \sum_{n=0}^{m} g_n(\overline{z})^n \right|. \tag{3.33}$$

Replacing (3.33) into the inequality (3.32) we get

$$\left| \sum_{n=0}^{m} g_n z^n \sum_{n=0}^{m} g_n \left(\overline{z} \right)^n \right| \le \frac{1}{2} \sum_{n=0}^{m} \frac{|g_n|^2}{a_n} \left[\sum_{n=0}^{m} a_n \left(|z|^2 \right)^n + \left| \sum_{n=0}^{m} a_n \left(z^2 \right)^n \right| \right]. \quad (3.34)$$

Since $z, z^2, |z|^2 \in D(0, R_1) \cap D(0, R_2)$, hence the series in (3.34) are convergent and letting $m \to \infty$, we deduce the desired inequality (3.31).

Remark 41 If the coefficients g_n , $n \ge 0$ are real, then by (3.31) we recapture the result of Cerone and Dragomir in [89], namely

$$|g(z)|^2 \le \frac{1}{2} \sum_{n=0}^{\infty} \frac{g_n^2}{a_n} \left[f(|z|^2) + |f(z^2)| \right],$$
 (3.35)

for any $z, z^2, |z|^2 \in D(0, R_1) \cap D(0, R_2)$.

As natural consequence of Theorem 40, the following corollaries hold for particular choices of functions f(z). First, we have

Corollary 42 Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$ be a power series with complex coefficients and convergent on the open disk D(0,R). If the numerical series $\sum_{n=0}^{\infty} |g_n|^2$ is convergent, then

$$|g(z)g(\overline{z})| \le \frac{1}{2} \sum_{n=0}^{\infty} |g_n|^2 \left[\frac{1 - |z|^2 + |1 - z^2|}{\left(1 - |z|^2\right)|1 - z^2|} \right],$$
 (3.36)

for any $z \in D(0,1) \cap D(0,R)$.

Proof. The proof simply follows from (3.31) for the function f(z) given by (1.19). \blacksquare

If we consider the series expansion

$$\frac{1}{iz}\ln\left(\frac{1}{1-iz}\right) = \sum_{n=0}^{\infty} \frac{i^n}{n+1} z^n; \ z \in D(0,1) \setminus \{0\},$$
 (3.37)

then, on utilising the inequality (3.36) for the choice $g_n = i^n/(n+1)$ and taking into account the identity (d) in (1.45), we can state the following inequality

$$\left| \ln \left(\frac{1}{1 - iz} \right) \ln \left(\frac{1}{1 - i\overline{z}} \right) \right| \le \frac{\pi^2}{12} \left(\frac{|z|^2}{1 - |z|^2} \right) \left(\frac{1 - |z|^2 + |1 - z^2|}{|1 - z^2|} \right), \quad (3.38)$$

for all $z \in D(0,1)$.

Corollary 43 Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$ be a power series with complex coefficients and convergent on the open disk D(0,R). If the numerical series $\sum_{n=0}^{\infty} n! |g_n|^2$ is convergent, then

$$|g(z)g(\overline{z})| \le \frac{1}{2} \sum_{n=0}^{\infty} n! |g_n|^2 \left[\exp\left(|z|^2\right) + \left| \exp\left(z^2\right) \right| \right], \tag{3.39}$$

for any $z \in D(0, R)$.

Proof. The proof follows from Theorem 40 by choosing the exponential function f(z) given by (1.25).

Some applications of the inequality (3.39) are as follows.

(1) If we apply the inequality (3.39) for the function

$$\sin(iz) = \sum_{n=0}^{\infty} \frac{i}{(2n+1)!} z^{2n+1}, \quad z \in \mathbb{C},$$
 (3.40)

then, we obtain the inequality

$$|\sin(iz)\sin(i\overline{z})| \le \frac{1}{2} \sum_{n=0}^{\infty} \frac{n!}{[(2n+1)!]^2} \left[\exp(|z|^2) + \left|\exp(z^2)\right|\right],$$
 (3.41)

for any $z \in \mathbb{C}$.

(2) If we apply the inequality (3.39) for the function

$$\sinh(iz) = \sum_{n=0}^{\infty} \frac{(-1)^n i}{(2n+1)!} z^{2n+1}, \quad z \in \mathbb{C}, \tag{3.42}$$

then, we obtain the inequality

$$\left|\sin(z)\right|^{2} \le \frac{1}{2} \sum_{n=0}^{\infty} \frac{n!}{\left[(2n+1)!\right]^{2}} \left[\exp\left(\left|z\right|^{2}\right) + \left|\exp\left(z^{2}\right)\right|\right],$$
 (3.43)

for any $z \in \mathbb{C}$. Indeed, observing that

$$|\sinh(iz)\sinh(i\overline{z})| = |i\sin(z)\cdot i\sin(\overline{z})| = |\sin(z)\sin(\overline{z})| \quad (3.44)$$
$$= |\sin(z)|^2,$$

and by (3.39) we have

$$|\sinh(iz)\sinh(i\overline{z})| \le \frac{1}{2} \sum_{n=0}^{\infty} \frac{n!}{[(2n+1)!]^2} \left[\exp(|z|^2) + |\exp(z^2)|\right], \quad (3.45)$$

for any $z \in \mathbb{C}$, then we deduce the desired inequality (3.43).

Finally, we obtain the following result [208], which provides the connection between three functions defined by the power series, one having positive coefficients, while the others have complex coefficients (see also [217]).

Theorem 44 (Ibrahim and Dragomir [208]) Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$, $h(z) = \sum_{n=0}^{\infty} h_n z^n$ and $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be three power series with g_n , $h_n \in \mathbb{C}$ and $a_n > 0$ for $n \geq 0$. If f, g and h are convergent on $D(0, R_1)$, $D(0, R_2)$ and $D(0, R_3)$ respectively, and the numerical series $\sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n}$, $\sum_{n=0}^{\infty} \frac{|h_n|^2}{a_n}$ and $\sum_{n=0}^{\infty} \frac{g_n \overline{h_n}}{a_n}$ are convergent, then we have the inequality

$$|g(z)h(z)| \le \frac{1}{2} \left(\left[\sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n} \sum_{n=0}^{\infty} \frac{|h_n|^2}{a_n} \right]^{1/2} + \left| \sum_{n=0}^{\infty} \frac{g_n \overline{h_n}}{a_n} \right| \right) f(|z|^2),$$
 (3.46)

for any $z \in \mathbb{C}$ with $z, |z|^2 \in D(0, R_1) \cap D(0, R_2) \cap D(0, R_3)$.

Proof. Again, on utilising the Buzano inequality (3.6) for the choices $p_n = a_n$, $c_n = g_n/a_n$, $b_n = h_n/a_n$, $x_n = z^n$, $n \ge 0$, we can state that

$$|g(z)h(z)| = \left| \sum_{n=0}^{\infty} a_n \left(\frac{g_n}{a_n} \right) z^n \sum_{n=0}^{\infty} a_n \left(\frac{h_n}{a_n} \right) z^n \right|,$$

$$\leq \frac{1}{2} \left(\left[\sum_{n=0}^{\infty} a_n \left| \frac{g_n}{a_n} \right|^2 \sum_{n=0}^{\infty} a_n \left| \frac{h_n}{a_n} \right|^2 \right]^{1/2} + \left| \sum_{n=0}^{\infty} a_n \left(\frac{g_n}{a_n} \right) \overline{\left(\frac{h_n}{a_n} \right)} \right| \right) \sum_{n=0}^{\infty} a_n \left(|z|^2 \right)^n,$$

$$= \frac{1}{2} \left(\left[\sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n} \sum_{n=0}^{\infty} \frac{|h_n|^2}{a_n} \right]^{1/2} + \left| \sum_{n=0}^{\infty} \frac{g_n \overline{h_n}}{a_n} \right| \right) f\left(|z|^2 \right), \tag{3.47}$$

for any $z \in \mathbb{C}$ with z, $|z|^2 \in D(0, R_1) \cap D(0, R_2) \cap D(0, R_3)$.

Remark 45 In particular, if $g_n = h_n$, then from (3.46) we have

$$|g(z)|^2 \le f(|z|^2) \sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n},$$
 (3.48)

for any z, $|z|^2 \in D(0, R_1) \cap D(0, R_2)$.

Remark 46 Also, if $h_n = \overline{g_n}$, then from (3.46) we get the following inequality

$$|g(z)g(\overline{z})| \le \frac{1}{2} \left(\sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n} + \left| \sum_{n=0}^{\infty} g_n^2 \right| \right) f(|z|^2), \tag{3.49}$$

for any $z, |z|^2 \in D(0, R_1) \cap D(0, R_2)$.

For particular choices of function f(z) in (3.46), the following result holds.

Corollary 47 Let g(z) and h(z) be power series as in Theorem 44. If the numerical series $\sum_{n=0}^{\infty} |g_n|^2$, $\sum_{n=0}^{\infty} |h_n|^2$ and $\sum_{n=0}^{\infty} |g_n\overline{h_n}|$ are convergent, then

$$|g(z)h(z)| \le \frac{1}{2(1-|z|^2)} \left(\left[\sum_{n=0}^{\infty} |g_n|^2 \sum_{n=0}^{\infty} |h_n|^2 \right]^{1/2} + \left| \sum_{n=0}^{\infty} g_n \overline{h_n} \right| \right),$$
 (3.50)

for any $z \in D(0,1) \cap D(0,R_2) \cap D(0,R_3)$.

Proof. This follows from (3.46) for the function f(z) given by (1.19).

If we consider the series expansion given by (3.37) and the series

$$\ln\left(\frac{1}{1+iz}\right) = \sum_{n=1}^{\infty} \frac{(-i)^n}{n} z^n, \ z \in D(0,1),$$
 (3.51)

then on utilizing the inequality (3.50) for the choices $g_0 = h_0 = 0$, $g_n = i^n/(n+1)$, $h_n = (-i)^n/n$, $n \ge 1$, and taking into account that

$$\sum_{n=0}^{\infty} g_n \overline{h_n} = \sum_{n=1}^{\infty} \frac{1}{n(n+1)} = 1,$$
(3.52)

and the identity (d) as mentioned in (1.45), we obtain the following inequality

$$\left| \ln \left(\frac{1}{1 - iz} \right) \ln \left(\frac{1}{1 + iz} \right) \right| \le \frac{\pi^2 + 6}{12} \left(\frac{|z|}{1 - |z|^2} \right), \tag{3.53}$$

for any $z \in D(0,1)$.

In the next section, we establish some other inequalities for functions defined by the power series with nonnegative coefficients and convergent on an open disk. A refinement of the (CBS)-inequality in inner product spaces due to Dragomir [118], is being an important tool for this investigations. Natural applications for some fundamental functions of interest such as the exponential, logarithm, trigonometric and hyperbolic functions are also pointed out.

3.3 Power Series Inequalities Via a Refinement of the Schwarz Inequality

If we write the inequality (2.78) for the particular inner product space $(\mathbb{K}^n; \langle \cdot, \cdot \rangle)$, where the weighted inner product is defined by (2.48), for $x = (x_1, x_2, ..., x_n)$, $y = (y_1, y_2, ..., y_n) \in \mathbb{K}^n$ and $p = (p_1, p_2, ..., p_n)$ with $p_j \geq 0$, $j \in \{1, 2, ..., n\}$, then we get the following discrete inequality

$$\left(\sum_{j=1}^{n} p_{j} |x_{j}|^{2}\right)^{1/2} \left(\sum_{j=1}^{n} p_{j} |y_{j}|^{2}\right)^{1/2} \sum_{j=1}^{n} p_{j} |z_{j}|^{2} - \left|\sum_{j=1}^{n} p_{j} x_{j} \overline{z_{j}} \sum_{j=1}^{n} p_{j} z_{j} \overline{y_{j}}\right| \\
\geq \left|\sum_{j=1}^{n} p_{j} x_{j} \overline{y_{j}} \sum_{j=1}^{n} p_{j} |z_{j}|^{2} - \sum_{j=1}^{n} p_{j} x_{j} \overline{z_{j}} \sum_{j=1}^{n} p_{j} z_{j} \overline{y_{j}}\right|$$
(3.54)

where $p_j \ge 0$, $x_j, y_j, z_j \in \mathbb{K}$, $j \in \{1, 2, ..., n\}$. In particular, if we take in (3.54) $y_j = \overline{x_j}$ for $j \in \{1, 2, ..., n\}$, then we obtain

$$\sum_{j=1}^{n} p_{j} |x_{j}|^{2} \sum_{j=1}^{n} p_{j} |z_{j}|^{2} - \left| \sum_{j=1}^{n} p_{j} x_{j} \overline{z_{j}} \sum_{j=1}^{n} p_{j} x_{j} z_{j} \right| \\
\geq \left| \sum_{j=1}^{n} p_{j} x_{j}^{2} \sum_{j=1}^{n} p_{j} |z_{j}|^{2} - \sum_{j=1}^{n} p_{j} x_{j} \overline{z_{j}} \sum_{j=1}^{n} p_{j} x_{j} z_{j} \right|,$$
(3.55)

for $p_j \ge 0$, $x_j, z_j \in \mathbb{K}$, $j \in \{1, 2, ..., n\}$.

On applying the inequality (3.54) for functions defined by the power series with nonnegative coefficients, we establish the following result [209] (see also [217]).

Theorem 48 (Ibrahim and Dragomir [209]) Let $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be a power series with nonnegative coefficients a_n and convergent on the open disk D(0,R). If $x, y, z \in \mathbb{C}$, so that $|x|^2, |y|^2, |z|^2, x\overline{z}, z\overline{y}, x\overline{y} \in D(0,R)$, then

$$\left[f\left(\left|x\right|^{2}\right)f\left(\left|y\right|^{2}\right)\right]^{1/2}f\left(\left|z\right|^{2}\right) - \left|f\left(x\overline{z}\right)f\left(z\overline{y}\right)\right|$$

$$\geq \left|f\left(x\overline{y}\right)f\left(\left|z\right|^{2}\right) - f\left(x\overline{z}\right)f\left(z\overline{y}\right)\right|. \tag{3.56}$$

Proof. If we choose $p_n=a_n,\ x_n=x^n,\ y_n=y^n$ and $z_n=z^n,\ n\in\{0,1,2,...,m\}$ in (3.54), then we have

$$\left[\sum_{n=0}^{m} a_n (|x|^2)^n\right]^{1/2} \left[\sum_{n=0}^{m} a_n (|y|^2)^n\right]^{1/2} \sum_{n=0}^{m} a_n (|z|^2)^n$$
$$-\left|\sum_{n=0}^{m} a_n (x\overline{z})^n \sum_{n=0}^{m} a_n (z\overline{y})^n\right|$$

$$\geq \left| \sum_{n=0}^{m} a_n (x\overline{y})^n \sum_{n=0}^{m} a_n (|z|^2)^n - \sum_{n=0}^{m} a_n (x\overline{z})^n \sum_{n=0}^{m} a_n (x\overline{y})^n \right|. \tag{3.57}$$

Since $|x|^2$, $|y|^2$, $|z|^2$, $x\overline{z}$, $z\overline{y}$, $x\overline{y}$ belong to the convergence disk D(0,R) and taking the limit as $m \to \infty$ in (3.57), we deduce the desired inequality (3.56).

Some examples for particular functions that are generated by the power series with nonnegative coefficients, are as follows:

(1) If we choose in the above inequality (3.56) for the function f(z) given by (1.19), then we have

$$\frac{\left| (1 - x\overline{z}) (1 - z\overline{y}) \right|}{\left[(1 - |x|^2) (1 - |y|^2) \right]^{1/2} (1 - |z|^2)} - 1$$

$$\geq \left| \frac{(1 - x\overline{z}) (1 - z\overline{y})}{(1 - x\overline{y}) (1 - |z|^2)} - 1 \right|, \tag{3.58}$$

for any $x, y, z \in D(0, 1)$. In particular, for $z = \overline{x}$ in (3.58) we get

$$\frac{\left| (1-x^2) (1-\overline{xy}) \right|}{(1-|x|^2)^{3/2} (1-|y|^2)^{1/2}} - 1 \ge \left| \frac{(1-x^2) (1-\overline{xy})}{(1-x\overline{y}) (1-|x|^2)} - 1 \right|, \tag{3.59}$$

for any $x, y \in D(0, 1)$. Also, if $z = a \in \mathbb{R}$ and $x, y \in \mathbb{C}$, then from (3.58), we obtain the following result:

$$\left| \frac{(1-ax)(1-a\overline{y})}{(1-x\overline{y})(1-a^2)} - 1 \right| \\
\leq \frac{\left| (1-ax)(1-a\overline{y}) \right|}{\left[(1-|x|^2)(1-|y|^2) \right]^{1/2}(1-a^2)} - 1, \tag{3.60}$$

for any $x, y \in D(0, 1)$ and $a \in (-1, 1)$.

(2) If we apply (3.56) for the exponential function f(z) given by (1.25), then we get

$$\exp\left(\frac{|x|^2 + |y|^2}{2} + |z|^2\right) - |\exp\left(x\overline{z} + z\overline{y}\right)|$$

$$\geq |\exp\left(x\overline{y} + |z|^2\right) - \exp\left(x\overline{z} + z\overline{y}\right)|, \tag{3.61}$$

for any $x, y, z \in \mathbb{C}$. In particular, for $z = \overline{x}$ in (3.61), we obtain

$$\left| \exp\left(x\overline{y} + |x|^2\right) - \exp\left(x^2 + \overline{xy}\right) \right|$$

$$\leq \exp\left(\frac{3|x|^2 + |y|^2}{2}\right) - \left| \exp\left(x^2 + \overline{xy}\right) \right|,$$
(3.62)

for any $x, y \in \mathbb{C}$. Also, if $z = a \in \mathbb{R}$ and $x, y \in \mathbb{C}$, then from (3.61) we get

$$\left| \exp\left(x\overline{y} + a^2\right) - \exp\left[a\left(x + \overline{y}\right)\right] \right|$$

$$\leq \exp\left(\frac{\left|x\right|^2 + \left|y\right|^2}{2} + a^2\right) - \left|\exp\left[a\left(x + \overline{y}\right)\right]\right|, \tag{3.63}$$

for any $x, y \in \mathbb{C}$ and $a \in \mathbb{R}$.

(3) For the *Koebe function* defined by (1.23), we get from (3.56) the following inequality:

$$\frac{1}{(1-|x|^2)(1-|y|^2)(1-|z|^2)^2} - \frac{1}{|[(1-x\overline{z})(1-z\overline{y})]^2|}$$

$$\geq \left| \frac{1}{[(1-x\overline{y})(1-|z|^2)]^2} - \frac{1}{[(1-x\overline{z})(1-z\overline{y})]^2} \right|, \tag{3.64}$$

for any $x, y, z \in D(0,1)$. In particular, for $y = \overline{x}$ in (3.64) we get

$$\frac{1}{\left[\left(1-|x|^{2}\right)\left(1-|z|^{2}\right)\right]^{2}} - \frac{1}{\left|\left[\left(1-x\overline{z}\right)\left(1-xz\right)\right]^{2}\right|} \\
\geq \left|\frac{1}{\left[\left(1-x^{2}\right)\left(1-|z|^{2}\right)\right]^{2}} - \frac{1}{\left[\left(1-x\overline{z}\right)\left(1-xz\right)\right]^{2}}\right|, \quad (3.65)$$

for any $x, z \in D(0,1)$. Also, for $z = \overline{x}$, we have from (3.64)

$$\frac{1}{(1-|x|^2)^3 (1-|y|^2)} - \frac{1}{|[(1-x^2)(1-\overline{xy})]^2|}$$

$$\geq \left| \frac{1}{[(1-x\overline{y})(1-|x|^2)]^2} - \frac{1}{[(1-x^2)(1-\overline{xy})]^2} \right|, \quad (3.66)$$

for any $x, y \in D(0, 1)$. If $z = a \in \mathbb{R}$ and $x, y \in \mathbb{C}$, then from (3.64)

$$\frac{1}{(1-|x|^2)(1-|y|^2)(1-a^2)^2} - \frac{1}{|[(1-ax)(1-a\overline{y})]^2|}$$

$$\geq \left| \frac{1}{[(1-x\overline{y})(1-a^2)]^2} - \frac{1}{[(1-ax)(1-a\overline{y})]^2} \right|, \tag{3.67}$$

for any $x, y \in D(0, 1)$ and $a \in (-1, 1)$.

Remark 49 If z = 0, then from (3.56) we obtain

$$[f(|x|^{2}) f(|y|^{2})]^{1/2} - |f(0)| \ge |f(x\overline{y}) - f(0)|, \qquad (3.68)$$

where $f(0) = a_0 > 0$, $|x|^2$, $|y|^2$, $x\overline{y} \in D(0, R)$.

The above result (3.68) also has natural applications for particular functions of interest, which are pointed out as follows:

(1) If we apply the inequality (3.68) for the exponential function f(z) given by (1.25), then we obtain the inequality

$$\exp\left(\frac{|x|^2 + |y|^2}{2}\right) - 1 \ge |\exp(x\overline{y}) - 1|,$$
 (3.69)

for any $x, y \in \mathbb{C}$. Moreover, if $y = \overline{x}$, then from (3.69) we get

$$\exp(|x|^2) - 1 \ge |\exp(x^2) - 1|,$$

for any $x \in \mathbb{C}$.

(2) If we apply the same inequality (3.68) for the trigonometric function f(z) given by (1.33), then we get the following inequality

$$\left[\cos(|x|^2)\cos(|y|^2)\right]^{1/2} - 1 \ge \left|\cos(x\overline{y}) - 1\right|,$$
 (3.70)

for any $x, y \in \mathbb{C}$. Also, if $y = \overline{x}$, then from (3.70) we get

$$\cos(|x|^2) - 1 \ge \left|\cos(x^2) - 1\right|,$$

for any $x \in \mathbb{C}$.

(3) For the function f(x) given by (1.19) and applying the inequality (3.68) we obtain

$$\frac{1}{\left[\left(1-|x|^{2}\right)\left(1-|y|^{2}\right)\right]^{1/2}}-1 \ge \left|\frac{x\overline{y}}{1-x\overline{y}}\right|,\tag{3.71}$$

for any $x, y \in \mathbb{C}$ with $|x|^2, |y|^2, x\overline{y} \in D(0, 1)$.

The following result also holds [209].

Remark 50 If $y = \overline{x}$ in (3.56), then we get

$$f(|x|^2) f(|z|^2) - |f(xz)f(x\overline{z})| \ge |f(x^2) f(|z|^2) - f(xz)f(x\overline{z})|, \qquad (3.72)$$

for $x, z \in \mathbb{C}$ with $|x|^2$, $|z|^2$, $x\overline{z}$, $zx \in D(0,R)$. Moreover, for $z = a \in \mathbb{R}$, from (3.72) we deduce

$$f(|x|^2) f(a^2) - |f(ax)|^2 \ge |f(x^2) f(a^2) - f^2(ax)|,$$
 (3.73)

for any $x \in \mathbb{C}$, $a \in \mathbb{R}$. If we choose in (3.73) a = 1, then we have the inequality

$$f(|x|^2) f(1) - |f(x)|^2 \ge |f(x^2) f(1) - f^2(x)|,$$
 (3.74)

for any $x \in \mathbb{C}$.

Remark 51 If $z = \overline{x}$ in (3.56) then we get

$$\left[f\left(|x|^{2}\right)f\left(|y|^{2}\right)\right]^{1/2}f\left(|x|^{2}\right) - \left|f\left(x^{2}\right)f\left(xy\right)\right|$$

$$\geq \left|f\left(x\overline{y}\right)f\left(|x|^{2}\right) - f\left(x^{2}\right)f\left(\overline{xy}\right)\right|, \tag{3.75}$$

for $x, y \in \mathbb{C}$ with $x^2, xy, |x|^2, |y|^2 \in D(0, R)$.

For some application, we apply the inequality (3.74) for the exponential function f(z) given by (1.25), then we have

$$\exp(|x|^2 + 1) - |\exp(2x)| \ge |\exp(x^2 + 1) - \exp(2x)|,$$
 (3.76)

for any $x \in \mathbb{C}$. Since $|\exp(2x)| \neq 0$, then the inequality (3.76) is equivalent with

$$\frac{\exp(|x|^2 + 1)}{|\exp(2x)|} - 1 \ge |\exp(x - 1)^2 - 1|, \qquad (3.77)$$

for any $x \in \mathbb{C}$.

Also, if we apply the inequality (3.75) for the exponential function f(z) given by (1.25), then we get

$$\exp\left(\frac{3|x|^2 + |y|^2}{2}\right) - \left|\exp\left[x^2 + \overline{xy}\right]\right|$$

$$\geq \left|\exp\left(x\overline{y} + |x|^2\right) - \exp\left(x^2 + \overline{xy}\right)\right|, \tag{3.78}$$

for any $x, y \in \mathbb{C}$. Moreover if $x = a \in \mathbb{R}$, then from (3.78) we obtain

$$\exp\left(\frac{3a^2 + |y|^2}{2}\right) \ge |\exp[a(a+y)]|,$$
 (3.79)

for any $y \in \mathbb{C}$ and $a \in \mathbb{R}$.

The second result on the power series inequalities via a refinement of the (CBS)-inequality in inner product spaces (3.54), is incorporated in the following theorem [209] (see also [217]).

Theorem 52 (Ibrahim and Dragomir [209]) Let $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be a power series with real coefficients a_n and convergent on $D(0,R) \subset \mathbb{C}$, R > 0. If $x, y, z \in \mathbb{C}$, so that $|x|^2$, $|y|^2$, $|z|^2$, $x\overline{z}$, $z\overline{y}$, $x\overline{y} \in D(0,R)$, then

$$\left[f_A\left(|x|^2\right)f_A\left(|y|^2\right)\right]^{1/2}f_A\left(|z|^2\right) - \left|f\left(x\overline{z}\right)f\left(z\overline{y}\right)\right|$$

$$\geq \left|f_A\left(x\overline{y}\right)f_A\left(|z|^2\right) - f\left(x\overline{z}\right)f\left(z\overline{y}\right)\right|. \tag{3.80}$$

Proof. By choosing $p_n = |a_n| \ge 0$, $x_n = x^n$, $y_n = y^n$ and $z_n = sgn(a_n) z^n$, $n \ge 0$ in (3.54), we get

$$\left| \sum_{n=0}^{m} a_{n} (x\overline{z})^{n} \sum_{n=0}^{m} a_{n} (z\overline{y})^{n} \right| \\
= \left| \sum_{n=0}^{m} |a_{n}| sgn (a_{n}) x^{n} (\overline{z})^{n} \sum_{n=0}^{m} |a_{n}| sgn (a_{n}) z^{n} (\overline{y})^{n} \right|, \\
\leq \left(\sum_{n=0}^{m} |a_{n}| |x|^{2n} \right)^{1/2} \left(\sum_{n=0}^{m} |a_{n}| |y|^{2n} \right)^{1/2} \sum_{n=0}^{m} |a_{n}| |sgn (a_{n}) z^{n}|^{2} \\
- \left| \sum_{n=0}^{m} |a_{n}| x^{n} (\overline{y})^{n} \sum_{n=0}^{m} |a_{n}| |sgn (a_{n}) z^{n}|^{2} \\
- \sum_{n=0}^{m} |a_{n}| x^{n} [sgn (a_{n}) (\overline{z})^{n}] \times \sum_{n=0}^{m} |a_{n}| [sgn (a_{n}) z^{n}] (\overline{y})^{n} \right|, \\
= \left(\sum_{n=0}^{m} |a_{n}| (|x|^{2})^{n} \right)^{1/2} \left(\sum_{n=0}^{m} |a_{n}| (|y|^{2})^{n} \right)^{1/2} \sum_{n=0}^{m} |a_{n}| (|z|^{2})^{n} \\
- \left| \sum_{n=0}^{m} |a_{n}| (x\overline{y})^{n} \sum_{n=0}^{m} |a_{n}| (|z|^{2})^{n} - \sum_{n=0}^{m} a_{n} (x\overline{z})^{n} \sum_{n=0}^{m} a_{n} (z\overline{y})^{n} \right|, \quad (3.81)$$

for any $x, y, z \in \mathbb{C}$ with $x\overline{y}, x\overline{z}, z\overline{y}, |x|^2, |y|^2, |z|^2 \in D(0, R)$. Since all the series involved in (3.81) are convergent, by taking the limit as $m \to \infty$ in (3.81), we deduce the desired inequality (3.80).

In what follows, we provide some applications of the inequality (3.80) for particular functions of interest.

(1) If we take the function f(z) given by (1.20), then the transform $f_A(z)$ is given by (1.19). Applying the inequality (3.80), we can state that

$$\left[\frac{1}{1 - |x|^2} \cdot \frac{1}{1 - |y|^2} \right]^{1/2} \left(\frac{1}{1 - |z|^2} \right) - \left| \frac{1}{1 + x\overline{z}} \cdot \frac{1}{1 + z\overline{y}} \right| f_A(z)$$

$$= \frac{1}{\left[(1 - |x|^2) (1 - |y|^2) \right]^{1/2} (1 - |z|^2)} - \frac{1}{\left| (1 + x\overline{z}) (1 + z\overline{y}) \right|}$$

$$= \frac{\left| (1+x\overline{z}) (1+z\overline{y}) \right| - \left[(1-|x|^2) (1-|y|^2) \right]^{1/2} (1-|z|^2)}{\left[(1-|x|^2) (1-|y|^2) \right]^{1/2} (1-|z|^2) |(1+x\overline{z}) (1+z\overline{y})|},
\ge \left| \frac{1}{1-x\overline{y}} \cdot \frac{1}{1-|z|^2} - \frac{1}{1+x\overline{z}} \cdot \frac{1}{1+z\overline{y}} \right|
= \frac{\left| (1+x\overline{z}) (1+z\overline{y}) - (1-x\overline{y}) (1-|z|^2) \right|}{\left| (1-x\overline{y}) (1-|z|^2) \right| |(1+x\overline{z}) (1+z\overline{y})|}.$$
(3.82)

Hence we have

$$\frac{\left| (1+x\overline{z}) (1+z\overline{y}) \right|}{\left[(1-|x|^2) (1-|y|^2) \right]^{1/2} (1-|z|^2)} - 1$$

$$\geq \left| \frac{(1+x\overline{z}) (1+z\overline{y})}{(1-x\overline{y}) (1-|z|^2)} - 1 \right|, \tag{3.83}$$

for any $x, y, z \in D(0,1)$. In particular, if $y = \overline{x}$, $z = a \in \mathbb{R}$, then from (3.83) we get

$$\frac{\left|1+ax\right|^{2}}{\left(1-\left|x\right|^{2}\right)\left(1-a^{2}\right)}-1 \ge \left|\frac{\left(1+ax\right)^{2}}{\left(1-x^{2}\right)\left(1-a^{2}\right)}-1\right|, \tag{3.84}$$

for any $x \in D(0,1)$, $a \in \mathbb{R}$.

(2) For the exponential function f(z) is given by (1.26), we have the transform $f_A(z)$ given by (3.26). Utilising the inequality (3.80), we obtain

$$\exp\left(\frac{|x|^2 + |y|^2}{2} + |z|^2\right) |\exp\left(x\overline{z} + z\overline{y}\right)| - 1$$

$$\geq |\exp\left(x\overline{y} + |z|^2 + x\overline{z} + z\overline{y}\right) - 1|, \qquad (3.85)$$

for any $x, y, z \in \mathbb{C}$. In particular, if $y = \overline{x}$, $z = a \in \mathbb{R}$, then from (3.85) we get

$$\exp(|x|^2 + a^2) |\exp(2ax)| - 1 \ge |\exp(|x|^2 + a + 2ax) - 1|, \qquad (3.86)$$

for any $x \in \mathbb{C}$, $a \in \mathbb{R}$.

Similar result to (3.31) is also obtained that connects the two convergent power series, one having positive coefficients, while the other has complex coefficients [209] (see also [217]).

Theorem 53 (Ibrahim and Dragomir [209]) Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$ and $f(z) = \sum_{n=0}^{\infty} a_n z^n$ be two power series with $g_n \in \mathbb{C}$ and $a_n > 0$, $n \geq 0$. If f and g are convergent on $D(0, R_1)$ and $D(0, R_2)$ respectively, and the numerical series $\sum_{n=0}^{\infty} \frac{|g_n|^2}{a_n}$ is convergent, then we have the inequality

$$\sum_{n=0}^{\infty} \frac{\left|g_n\right|^2}{a_n} f\left(\left|z\right|^2\right) - \left|g(z)g(\overline{z})\right| \ge \left|\sum_{n=0}^{\infty} \frac{\left|g_n\right|^2}{a_n} f\left(z^2\right) - g(z)\overline{g\left(\overline{z}\right)}\right|,\tag{3.87}$$

for any $z \in \mathbb{C}$ with $z, z^2, |z|^2 \in D(0, R_1) \cap D(0, R_2)$.

Proof. On utilizing the inequality (3.55) for the choices $p_n = a_n$, $x_n = z^n$ and $z_n = g_n/a_n$, $n \in \{0, 1, 2, ..., m\}$, we have

$$\sum_{n=0}^{m} a_{n} |z|^{2n} \sum_{n=0}^{m} a_{n} \left| \frac{g_{n}}{a_{n}} \right|^{2} - \left| \sum_{n=0}^{m} a_{n} z^{n} \overline{\left(\frac{g_{n}}{a_{n}}\right)} \sum_{n=0}^{n} a_{n} z^{n} \left(\frac{g_{n}}{a_{n}}\right) \right|$$

$$= \sum_{n=0}^{m} a_{n} (|z|^{2})^{n} \sum_{n=0}^{m} \frac{|g_{n}|^{2}}{a_{n}} - \left| \sum_{n=0}^{m} \overline{g_{n}} z^{n} \sum_{n=0}^{n} g_{n} z^{n} \right|$$

$$\geq \left| \sum_{n=0}^{m} a_{n} z^{2n} \sum_{n=0}^{m} a_{n} \left| \frac{g_{n}}{a_{n}} \right|^{2} - \sum_{n=0}^{m} a_{n} z^{n} \overline{\left(\frac{g_{n}}{a_{n}}\right)} \sum_{n=0}^{n} a_{n} z^{n} \left(\frac{g_{n}}{a_{n}}\right) \right|$$

$$= \left| \sum_{n=0}^{m} a_{n} (z^{2})^{n} \sum_{n=0}^{m} \frac{|g_{n}|^{2}}{a_{n}} - \sum_{n=0}^{m} \overline{g_{n}} z^{n} \sum_{n=0}^{n} g_{n} z^{n} \right|, \qquad (3.88)$$

for any $m \geq 0$. Replacing the equation (3.33) in (3.88), we get

$$\sum_{n=0}^{m} \frac{|g_n|^2}{a_n} \sum_{n=0}^{m} a_n (|z|^2)^n - \left| \sum_{n=0}^{m} g_n (\overline{z})^n \sum_{n=0}^{n} g_n z^n \right|$$

$$\geq \left| \sum_{n=0}^{m} a_n (z^2)^n \sum_{n=0}^{m} \frac{|g_n|^2}{a_n} - \sum_{n=0}^{m} \overline{g_n} z^n \sum_{n=0}^{n} g_n z^n \right|. \tag{3.89}$$

Since $z, z^2, |z|^2 \in D(0, R_1) \cap D(0, R_2)$, the series in (3.89) are convergent, by letting $m \to \infty$, we deduce the desired inequality (3.87).

Remark 54 If the coefficients g_n , $n \ge 0$ are real, then we have the inequality

$$\sum_{n=0}^{\infty} \frac{g_n^2}{a_n} f\left(|z|^2\right) - |g(z)g(\overline{z})| \ge \left|\sum_{n=0}^{\infty} \frac{g_n^2}{a_n} f\left(z^2\right) - g^2(z)\right|,\tag{3.90}$$

for any $z \in \mathbb{C}$ with $z, z^2, |z|^2 \in D(0, R_1) \cap D(0, R_2)$.

The following corollary is the natural consequence of Theorem 53 (see [209], [217]).

Corollary 55 Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$ be a power series with complex coefficients and convergent on the open disk D(0,R). If the numerical series $\sum_{n=0}^{\infty} |g_n|^2$ is convergent, then

$$\left(\frac{1}{1-|z|^2}\right)\sum_{n=0}^{\infty}|g_n|^2-|g(z)g(\overline{z})| \ge \left|\left(\frac{1}{1-z^2}\right)\sum_{n=0}^{\infty}|g_n|^2-g(z)\overline{g(\overline{z})}\right|,\quad (3.91)$$

for any $z \in D(0,1) \cap D(0,R)$.

Proof. It follows from (3.87) for the function f(z) given by (1.19).

If we consider the series expansion given by (3.37), then on utilizing the inequality (3.91) for the choice $g_n = i^n/(n+1)$, and taking into account the equality (d) in (1.45), we can state the following inequality

$$\frac{\pi^2}{6} \left(\frac{|z|^2}{1 - |z|^2} \right) - \left| \ln \left(\frac{1}{1 - iz} \right) \ln \left(\frac{1}{1 - i\overline{z}} \right) \right|
\ge \left| \frac{\pi^2}{6} \left(\frac{z^2}{1 - z^2} \right) + \ln \left(\frac{1}{1 - iz} \right) \ln \left(\frac{1}{1 + iz} \right) \right|,$$
(3.92)

for any $z \in D(0,1)$.

Corollary 56 Let $g(z) = \sum_{n=0}^{\infty} g_n z^n$ be a power series with complex coefficients and convergent on the open disk D(0,R). If the numerical series $\sum_{n=0}^{\infty} n! |g_n|^2$ is convergent, then

$$\sum_{n=0}^{\infty} n! |g_n|^2 \exp\left(|z|^2\right) - |g(z)g(\overline{z})| \ge \left| \sum_{n=0}^{\infty} n! |g_n|^2 \exp\left(z^2\right) - g(z)\overline{g(\overline{z})} \right|, \quad (3.93)$$

for any $z \in D(0, R)$.

Proof. It follows from Theorem 53 by choosing the exponential function f(z) given by (1.25).

If we apply the inequality (3.93) for the trigonometric function f(z) given by (3.40), then we obtain the following inequality

$$\sum_{n=0}^{\infty} \frac{n!}{(2n+1)!} \exp\left(|z|^2\right) - |\sin(iz)\sin(i\overline{z})|$$

$$\geq \left|\sum_{n=0}^{\infty} \frac{n!}{(2n+1)!} \exp\left(z^2\right) - \sin^2(iz)\right|, \tag{3.94}$$

for any $z \in \mathbb{C}$.

More results on the celebrated (CBS)-type inequalities for functions defined by the power series with real coefficients, are discussed in the following section.

3.4 Other Refinements of the (CBS)-type

Utilising a different technique based on the continuity properties of modulus (2.22), in this section, we develop some more inequalities for the power series functions that are related to the (CBS)-type inequality (see [214], [217]). Applications for some fundamental functions are included as well.

We begin with the following result.

Theorem 57 (Ibrahim, Dragomir and Darus [214]) Assume that the power series $f(z) = \sum_{n=0}^{\infty} p_n z^n$ with real coefficients is convergent on the disk D(0,R), R > 0. If $x, z \in \mathbb{C}$ are such that $x, xz, |x||z|^2 \in D(0,R)$, then we have the inequality

$$f_A(|x||z|^2) f_A(|x|) - |f_A(|x|z)|^2$$

$$\geq |f(x)f(x|z|z) - f(xz)f(x|z|)| \geq 0.$$
(3.95)

Proof. For arbitrary complex number z with $z \in D(0, R)$, we have for n, $j \in \mathbb{N}$,

$$|z^{n} - z^{j}|^{2} = |z^{n} - z^{j}| |z^{n} - z^{j}| \ge |z^{n} - z^{j}| ||z|^{n} - |z|^{j}|,$$
 (3.96)

by utilising the continuity properties of modulus (2.22). We also have

$$|z^n - z^j|^2 = |z^n|^2 - 2\operatorname{Re}(z^n\overline{z}^j) + |z^j|^2 = |z|^{2n} - 2\operatorname{Re}(z^n\overline{z}^j) + |z|^{2j},$$
 (3.97)

and

$$|z^{n} - z^{j}| \cdot ||z|^{n} - |z|^{j}| = |z^{n}|z|^{n} + z^{j}|z|^{j} - z^{n}|z|^{j} - z^{j}|z|^{n}|, \qquad (3.98)$$

for any $n, j \in \mathbb{N}$. Utilizing (3.96) we get the inequality

$$|z|^{2n} - 2\operatorname{Re}\left(z^{n}\overline{z}^{j}\right) + |z|^{2j} \ge \left|z^{n}|z|^{n} + z^{j}|z|^{j} - |z|^{j}z^{n} - |z|^{n}z^{j}\right|,$$
 (3.99)

for any $n, j \in \mathbb{N}$. If we multiply the inequality (3.99) by nonnegative quantity, i.e., $|p_n| |x|^n |p_j| |x|^j \ge 0$, where $x \in D(0, R)$ and $n, j \in \mathbb{N}$, then we have

$$|p_{n}| |x|^{n} |z|^{2n} p_{j} |x|^{j} + |p_{n}| |x|^{n} p_{j} |x|^{j} |z|^{2j}$$

$$- 2 \operatorname{Re} \left(|p_{n}| |x|^{n} z^{n} |p_{j}| |x|^{j} \overline{z}^{j} \right)$$

$$\geq \left| p_{n} x^{n} |z|^{n} z^{n} p_{j} x^{j} + p_{n} x^{n} p_{j} x^{j} |z|^{j} z^{j} - p_{n} x^{n} z^{n} p_{j} x^{j} |z|^{j}$$

$$- p_{n} x^{n} |z|^{n} p_{j} x^{j} z^{j} |, \qquad (3.100)$$

for any $n, j \in \mathbb{N}$. Summing over n and j from 0 to k, and utilizing the triangle inequality for the modulus (2.25), we obtain from (3.100) that

$$\sum_{n=0}^{k} |p_{n}| |x|^{n} |z|^{2n} \sum_{j=0}^{k} |p_{j}| |x|^{j} + \sum_{n=0}^{k} |p_{n}| |x|^{n} \sum_{j=0}^{k} |p_{j}| |x|^{j} |z|^{2j}$$

$$- 2 \operatorname{Re} \left(\sum_{n=0}^{k} |p_{n}| |x|^{n} z^{n} \sum_{j=0}^{k} |p_{j}| |x|^{j} (\overline{z})^{j} \right)$$

$$\geq \left| \sum_{n=0}^{k} p_{n} x^{n} |z|^{n} z^{n} \sum_{j=0}^{k} p_{j} x^{j} + \sum_{n=0}^{k} p_{n} x^{n} \sum_{j=0}^{k} p_{j} x^{j} |z|^{j} z^{j}$$

$$- \sum_{n=0}^{k} p_{n} x^{n} z^{n} \sum_{j=0}^{k} p_{j} x^{j} |z|^{j} - \sum_{n=0}^{k} p_{n} x^{n} |z|^{n} \sum_{j=0}^{k} p_{j} x^{j} z^{j} \right|. \tag{3.101}$$

Since

$$\sum_{j=0}^{k} |p_j| |x|^j (\overline{z})^j = \sum_{n=0}^{k} |p_n| |x|^n z^n,$$
 (3.102)

then

$$\operatorname{Re}\left(\sum_{n=0}^{k}|p_{n}||x|^{n}z^{n}\sum_{j=0}^{k}|p_{j}||x|^{j}(\overline{z})^{j}\right) = \left|\sum_{n=0}^{k}|p_{n}||x|^{n}z^{n}\right|^{2}.$$
 (3.103)

Hence, we get the following inequality, by replacing the equality (3.103) into (3.101),

$$\sum_{n=0}^{k} |p_n| |x|^n |z|^{2n} \sum_{n=0}^{k} |p_n| |x|^n - \left| \sum_{n=0}^{k} |p_n| |x|^n z^n \right|^2$$

$$\geq \left| \sum_{n=0}^{k} p_n x^n \sum_{n=0}^{k} p_n x^n |z|^n z^n - \sum_{n=0}^{k} p_n x^n z^n \sum_{n=0}^{k} p_n x^n |z|^n \right|. \tag{3.104}$$

Since all the series whose partial sums are involved in (3.104), are convergent, then by taking the limit as $k \to \infty$ in (3.104), we deduce the desired inequality (3.95).

The particular cases are as follows:

Corollary 58 If $\sum_{n=0}^{\infty} |p_n| < \infty$, i.e., $f_A(1) < \infty$, then for any $\zeta \in \mathbb{C}$ with $|\zeta| = 1$, we have

$$f_A(|z|^2) f_A(1) - |f_A(z)|^2 \ge |f(\zeta)f(\zeta|z|z) - f(\zeta z)f(\zeta|z|)| \ge 0.$$
 (3.105)

In particular, for $\zeta = 1$, we have

$$f_A(|z|^2) f_A(1) - |f_A(z)|^2 \ge |f(1)f(|z|z) - f(z)f(|z|)| \ge 0,$$
 (3.106)

for any $z, |z|^2 \in D(0, R)$.

Some applications of the inequalities (3.95) and (3.106) for particular functions of interest are pointed out as follows:

(1) If we apply the inequality (3.95) for the function f(z) given by (1.19), then we get

$$\frac{|1-z|}{(1-|x|)(1-|x||z|^2)|1-|x|z|^2}$$

$$\geq \left| \frac{1-|z|}{(1-x)(1-xz)(1-x|z|)(1-x|z|z)} \right| (3.107)$$

for any $x, z \in \mathbb{C}$ with $x, |x| |z|^2 \in D(0, 1)$.

(2) If we apply the inequality (3.95) for the function f(z) given by (1.20), then we get the inequality

$$\frac{|1-z|}{(1-|x|)(1-|x||z|^2)|1-|x|z|^2}$$

$$\geq \left| \frac{1-|z|}{(1+x)(1+xz)(1+x|z|)(1+x|z|z)} \right| (3.108)$$

for any $x, z \in \mathbb{C}$ with $x, xz, |x| |z|^2 \in D(0, 1)$.

(3) If we apply the inequality (3.106) for the exponential function f(z) given by (1.25), then we get the inequality

$$\exp(|z|^2 + 1) - |\exp(z)|^2 \ge |\exp(z|z| + 1) - \exp(z + |z|)| \qquad (3.109)$$

for any $z \in \mathbb{C}$.

Remark 59 The inequality (3.95) can be written in the form

$$\det \begin{bmatrix} f_A(|x||z|^2) & f_A(|x|z) \\ f_A(|x|\overline{z}) & f_A(|x|) \end{bmatrix} \ge \det \begin{bmatrix} f(x) & f(xz) \\ f(x|z|) & f(x|z|z) \end{bmatrix}$$
(3.110)

for any $x, z \in \mathbb{C}$ with $x, xz, |x| |z|^2 \in D(0, R)$.

The following result also holds.

Theorem 60 (Ibrahim, Dragomir and Darus [214]) Assume that the power series $f(z) = \sum_{n=0}^{\infty} p_n z^n$ with real coefficients is convergent on the disk D(0, R), R > 0. If $x, z \in \mathbb{C}$ are such that $x, xz, |x||z|^2 \in D(0, R)$, then we have the inequality

$$f_{A}(|x|) f_{A}(|x||z|^{2}) - \operatorname{Re}\left[f_{A}^{2}(|x|z)\right]$$

$$\geq \frac{1}{2} |f(x)f(x|z|z) + f(x)f(x|z|\overline{z}) - f(xz)f(x|z|)$$

$$-f(x|z|) f(x\overline{z})|. \tag{3.111}$$

Proof. If $z \in D(0,R)$, then

$$\begin{aligned}
\left|z^{n} - (\overline{z})^{j}\right|^{2} &= \left|z^{n} - (\overline{z})^{j}\right| \left|z^{n} - (\overline{z})^{j}\right| \geq \left|z^{n} - (\overline{z})^{j}\right| \left||z|^{n} - |z|^{j}\right| \\
&= \left||z|^{n} z^{n} + (\overline{z})^{j} |z|^{j} - |z|^{j} z^{n} - |z|^{n} (\overline{z})^{j}\right|, \tag{3.112}
\end{aligned}$$

for any $n, j \in \mathbb{N}$. We also have

$$|z^{n} - (\overline{z})^{j}|^{2} = |z^{n}|^{2} - 2\operatorname{Re}(z^{n}z^{j}) + |z^{j}|^{2}$$

$$= |z|^{2n} - 2\operatorname{Re}(z^{n}z^{j}) + |z|^{2j}, \qquad (3.113)$$

for any $n, j \in \mathbb{N}$. Utilizing (3.112) we have the inequality

$$|z|^{2n} - 2\operatorname{Re}(z^n z^j) + |z|^{2j} \ge \left| |z|^n z^n + (\overline{z})^j z^n - |z|^j z^n - |z|^n (\overline{z})^j \right|$$
 (3.114)

for any $n, j \in \mathbb{N}$. Now, on utilizing the similar argument to the one in the proof of Theorem 57 above, we deduce the desired result (3.111). The details are omitted. \blacksquare

Corollary 61 If $z = \overline{x}$ in (3.111), then we have

$$f_{A}(|x|) f_{A}(|x|^{3}) - \operatorname{Re}\left[f_{A}^{2}(|x|\overline{x})\right]$$

$$\geq \frac{1}{2} |f(x)f(|x|^{3}) + f(x)f(|x|x^{2}) - f(|x|^{2})f(|x|x)$$

$$-f(|x|x) f(x^{2})| \qquad (3.115)$$

for any $x \in \mathbb{C}$ such that x, |x| x, $|x| x^2 \in D(0, R)$.

In the following, we give some applications of above inequality (3.115) for particular complex functions of interest.

(1) If we take the function f(z) given by (1.20), the we have the transform $f_A(z)$ given by (1.19). Applying the inequality (3.115), we get the following result

$$\frac{1}{(1-|x|)(1-|x|^3)} - \operatorname{Re}\left(\frac{1}{1-|x|}\right)^2$$

$$\geq \frac{1}{2} \left| \frac{2+|x|x^2+|x|^3}{(1+x)(1+|x|^3)(1+|x|x^2)} - \frac{2+x^2+|x|^2}{(1+x^2)(1+|x|x)(1+|x|^2)} \right|, \tag{3.116}$$

for any x, |x| x, $|x| x^2 \in D(0, 1)$.

(2) If we apply the inequality (3.115) for the function f(z) given by (1.25), then we get

$$\exp(|x| + |x|^{3}) - \operatorname{Re}\left[\exp(2|x|\overline{x})\right]$$

$$\geq \frac{1}{2}\left|\exp(x + |x|^{3}) + \exp(x + |x|x^{2}) - \exp(|x|^{2} + |x|x)\right|$$

$$-\exp(|x|x + x^{2})| \qquad (3.117)$$

for any $x \in \mathbb{C}$.

Theorem 62 (Ibrahim, Dragomir and Darus [214]) Assume that the power series $f(z) = \sum_{n=0}^{\infty} p_n z^n$ with real coefficients is convergent on the disk D(0,R), R > 0. If $x, y \in \mathbb{C}$ are such that $|x|^2$, $|y|^2 < R$, then we have the inequality

$$f_{A}(|x|^{2}) f_{A}(|y|^{2}) - |f_{A}(xy)|^{2}$$

$$\geq |f(|x|x)f(|y|\overline{y}) - f(|y|x)f(|x|\overline{y})|. \tag{3.118}$$

Proof. If $x, y \in \mathbb{C}$, then we have

$$\left| x^{n} \left(\overline{y} \right)^{j} - x^{j} \left(\overline{y} \right)^{n} \right|^{2} = \left| x^{n} \left(\overline{y} \right)^{j} - x^{j} \left(\overline{y} \right)^{n} \right| \left| x^{n} \left(\overline{y} \right)^{j} - x^{j} \left(\overline{y} \right)^{n} \right|$$

$$\geq \left| x^{n} \left(\overline{y} \right)^{j} - x^{j} \left(\overline{y} \right)^{n} \right| \left| |x|^{n} |y|^{j} - |x|^{j} |y|^{n} \right|, \qquad (3.119)$$

for any $n, j \in \mathbb{N}$. We have upon simple calculations that

$$|x|^{2n} |y|^{2j} - 2 \operatorname{Re}\left(x^n y^n (\overline{x})^j (\overline{y})^j\right) + |y|^{2n} |x|^{2j}$$

$$\geq \left| |x|^n x^n |y|^j (\overline{y})^j + |y|^n (\overline{y})^n |x|^j x^j - |y|^n x^n |x|^j (\overline{y})^j - |x|^n (\overline{y})^n |y|^j x^j \right|, \tag{3.120}$$

for any $n, j \in \mathbb{N}$.

If we multiply the inequality (3.120) with the positive quantity, i.e., $|p_n| |p_j| \ge 0$, and summing over n and j from 0 to k, then we have

$$\sum_{n=0}^{k} |p_n| |x|^{2n} \sum_{j=0}^{\infty} |p_j| |y|^{2j} + \sum_{n=0}^{k} |p_n| |y|^{2n} \sum_{j=0}^{k} |p_j| |x|^{2j}$$

$$- 2 \operatorname{Re} \left(\sum_{n=0}^{k} |p_n| x^n y^n \sum_{j=0}^{k} |p_j| (\overline{x})^j (\overline{y})^j \right)$$

$$\geq \left| \sum_{n=0}^{k} p_n |x|^n x^n \sum_{j=0}^{k} p_j |y|^j (\overline{y})^j + \sum_{n=0}^{k} p_n |y|^n (\overline{y})^n \sum_{j=0}^{k} p_j |x|^j x^j \right|$$

$$-\sum_{n=0}^{k} p_{n} |y|^{n} x^{n} \sum_{j=0}^{k} p_{j} |x|^{j} (\overline{y})^{j} - \sum_{n=0}^{k} p_{n} |x|^{n} (\overline{y})^{n} \sum_{j=0}^{k} p_{j} |y|^{j} x^{j}$$
(3.121)

Due to the fact that

$$\sum_{n=0}^{k} |p_n| \, x^n y^n \sum_{j=0}^{\infty} |p_j| \, (\overline{x})^j \, (\overline{y})^j = \left| \sum_{n=0}^{k} |p_n| \, x^n y^n \right|^2, \tag{3.122}$$

then the inequality (3.121) is equivalent with

$$\sum_{n=0}^{k} |p_n| |x|^{2n} \sum_{n=0}^{k} |p_n| |y|^{2n} - \left| \sum_{n=0}^{k} |p_n| x^n y^n \right|^2$$

$$\geq \left| \sum_{n=0}^{k} p_n |x|^n x^n \sum_{n=0}^{k} p_n |y|^n (\overline{y})^n - \sum_{n=0}^{k} p_n |y|^n x^n \sum_{n=0}^{k} p_n |x|^n (\overline{y})^n \right|.$$
(3.123)

Since all the series with the partial sums are involved in (3.123), are convergent, then by taking the limit over $k \to \infty$ in (3.123), we deduce the desired result (3.118).

Remark 63 The inequality (3.118) is also equivalent to

$$\det \begin{bmatrix} f_A(|x|^2) & f_A(xy) \\ f_A(\overline{xy}) & f_A(|y|^2) \end{bmatrix} \ge \left| \det \begin{bmatrix} f(|x|x) & f(|y|x) \\ f(|x|\overline{y}) & f(|y|\overline{y}) \end{bmatrix} \right|$$
(3.124)

for any $x, y \in \mathbb{C}$ with $|x|^2$, $|y|^2 < R$.

The inequality (3.118) has some applications for particular complex functions of interest, which will be pointed out as follows.

(1) If we apply the inequality (3.118) for the function f(z) given by (1.19), then we get

$$\frac{1}{(1-|x|^2)(1-|y|^2)} - \frac{1}{|1-xy|^2} \\
\ge \left| \frac{1}{(1-x|x|)(1-|y|\overline{y})} - \frac{1}{(1-|y|x)(1-|x|\overline{y})} \right|$$
(3.125)

for any $x, y \in \mathbb{C}$. In particular, if in (3.125) we choose y = 0, then we obtain the simpler inequality

$$\frac{1}{1 - |x|^2} - 1 \ge \left| \frac{1}{1 - x |x|} - 1 \right| \tag{3.126}$$

for any $x \in \mathbb{C}$.

(2) If we apply the inequality (3.118) for the function f(z) given by (1.25), then we have

$$\exp(|x|^{2} + |y|^{2}) - |\exp(xy)|^{2}$$

$$> |\exp(x|x| + |y|\overline{y}) - \exp(|y|x + |x|\overline{y})|$$
(3.127)

for any $x, y \in \mathbb{C}$. In particular, if in (3.127) we choose y = 0, then we get

$$\exp(|x|^2) - 1 \ge |\exp(x|x|) - 1| \tag{3.128}$$

for any $x \in \mathbb{C}$.

(3) If we take the trigonometric function f(z) given by (1.33) with its transform $f_A(z)$ is given by (1.39), then utilizing the inequality (3.118) for f(z) as above gives

$$\cosh(|x|^{2})\cosh(|y|^{2}) - |\cosh(xy)|^{2}$$

$$\geq |\cos(x|x|)\cos(|y|\overline{y}) - \cos(|y|x)\cos(|x|\overline{y})| \qquad (3.129)$$

for any $x, y \in \mathbb{C}$. In particular, we have, with y = 0 in (3.129),

$$\cosh(|x|^{2}) - 1 \ge |\cos(x|x|) - 1| \tag{3.130}$$

for any $x \in \mathbb{C}$.

We end this section by proving the following results.

Theorem 64 (Ibrahim, Dragomir and Darus [214]) Assume that the power series $f(z) = \sum_{n=0}^{\infty} p_n z^n$ with real coefficients is convergent on the disk

D(0,R), R > 0. If $x, y \in \mathbb{C}$ are such that $|x|^2, |y|^2 < R$, then we have the inequality

$$f_{A}(|x|^{2}) f_{A}(|y|^{2}) - \operatorname{Re}\left[f_{A}^{2}(x\overline{y})\right]$$

$$\geq \frac{1}{2} |f(|x|x)f(|y|\overline{y}) + f(|x|\overline{x})f(|y|y) - f(|x|y)f(|y|\overline{x})$$

$$-f(|y|x)f(|x|\overline{y})|. \tag{3.131}$$

Proof. If $x, y \in D(0, R)$, then we have

$$\left| x^{n} \left(\overline{y} \right)^{j} - \left(\overline{x} \right)^{j} y^{n} \right|^{2} = \left| x^{n} \left(\overline{y} \right)^{j} - \left(\overline{x} \right)^{j} y^{n} \right| \left| x^{n} \left(\overline{y} \right)^{j} - \left(\overline{x} \right)^{j} y^{n} \right|$$

$$\geq \left| x^{n} \left(\overline{y} \right)^{j} - \left(\overline{x} \right)^{j} y^{n} \right| \left| |x|^{n} |y|^{j} - |x|^{j} |y|^{n} \right|$$

$$(3.132)$$

for any $n, j \in \mathbb{N}$. Doing simple calculations we get that

$$|x|^{2n} |y|^{2j} - 2 \operatorname{Re} \left[x^{n} (\overline{y})^{j} x^{j} (\overline{y})^{n} \right] + |x|^{2j} |y|^{2n}$$

$$\geq \left| |x|^{n} x^{n} |y|^{j} (\overline{y})^{j} + |x|^{j} (\overline{x})^{j} |y|^{n} y^{n} - |x|^{n} y^{n} |y|^{j} (\overline{x})^{j}$$

$$- |y|^{n} x^{n} |x|^{j} (\overline{y})^{j} \right|$$
(3.133)

for any $n, j \in \mathbb{N}$. If we multiply the inequality (3.133) with $|p_n| |p_j| \ge 0$, and summing over n and j from 0 to k, then we get

$$\sum_{n=0}^{k} |p_{n}| |x|^{2n} \sum_{j=0}^{k} |p_{j}| |y|^{2j} - 2 \operatorname{Re} \left(\sum_{n=0}^{k} |p_{n}| x^{n} (\overline{y})^{n} \sum_{j=0}^{k} |p_{j}| x^{j} (\overline{y})^{j} \right)
+ \sum_{n=0}^{k} |p_{n}| |y|^{2n} \sum_{j=0}^{k} |p_{j}| |x|^{2j}
\geq \left| \sum_{n=0}^{k} p_{n} |x|^{n} x^{n} \sum_{j=0}^{k} p_{j} |y|^{j} (\overline{y})^{j} + \sum_{n=0}^{k} p_{n} y^{n} |y|^{n} \sum_{j=0}^{k} p_{j} |x|^{j} (\overline{x})^{j}
- \sum_{n=0}^{k} p_{n} |x|^{n} y^{n} \sum_{j=0}^{k} p_{j} |y|^{j} (\overline{x})^{j} - \sum_{n=0}^{k} p_{n} |y|^{n} x^{n} \sum_{j=0}^{k} p_{j} |x|^{j} (\overline{y})^{j} \right|, \quad (3.134)$$

for any $n, j \in \mathbb{N}$.

Since all the series whose partial sums are involved in (3.134), are convergent, then by taking the limit over $k \to \infty$ in (3.134), we deduce the desired result (3.131).

The inequality (3.131) is also a valuable source of particular inequalities for complex functions of interest, that will be outlined in the following.

(1) In (3.131), we take the function f(z) given by (1.25), then we can state that

$$\exp(|x|^{2} + |y|^{2}) - \operatorname{Re}\left[\exp(2x\overline{y})\right]$$

$$\geq \frac{1}{2}\left|\exp(|x|x + |y|\overline{y}) + \exp(|x|\overline{x}) + |y|y\right|$$

$$-\exp(|x|y) + |y|\overline{x}) - \exp(|y|x + |x|\overline{y}), \qquad (3.135)$$

for any $x, y \in \mathbb{C}$. If in (3.135) we choose y = 0, then we obtain the simpler result

$$\exp(|x|^2) - 1 \ge \frac{1}{2} |\exp(|x|x) + \exp(|x|\overline{x}) - 2|,$$
 (3.136)

for any $x \in \mathbb{C}$.

(2) If we apply the inequality (3.131) for the trigonometric function f(z) given by (1.33) with its transform $f_A(z)$ is given by (1.39), then we get

$$\cosh(|x|^{2})\cosh(|y|^{2}) - \operatorname{Re}\left[\cosh^{2}(x\overline{y})\right]$$

$$\geq \frac{1}{2}\left|\cos(|x|x)\cos(|y|\overline{y}) + \cos(|x|\overline{x})\cos(|y|y)\right|$$

$$-\cos(|x|y)\cos(|y|\overline{x}) - \cos(|y|x)\cos(|x|\overline{y}), \qquad (3.137)$$

for any $x, y \in \mathbb{C}$. In particular, if in (3.137) we choose y = 0, then we obtain that

$$\cosh(|x|^2) - 1 \ge \frac{1}{2}|\cos(|x|x) + \cos(|x|\overline{x}) - 2|, \qquad (3.138)$$

for any $x \in \mathbb{C}$.

Chapter 4

More Inequalities on Power Series with Real Coefficients

In the previous chapter, some inequalities concerning the power series have been established by utilizing a different technique based on the continuity properties of modulus, Buzano's inequality and Schwarz's result in inner product spaces. In this chapter, we employ the similar method to develop some more inequalities for functions defined by the power series in a real and complex variable, with the crucial tools for these investigations being Young's, Hölder's and Jensen's type inequalities, as well as their reverses and counterparts.

The main purpose of Chapter 4 is two-fold. The first aim is to derive some inequalities for the power series that are related to the classical Young's inequality. Motivated by the results established by Dragomir and Sàndor [136], we derive new and better inequalities for functions defined by the power series with real coefficients via Young's inequality for the product of two scalars. All the results and their applications for some fundamental functions are presented in Section 4.1.2. More inequalities on the power series functions are obtained by utilising a refinement and a reverse of Young's inequality, see Section 4.1.3. Natural applications for some fundamental functions of interest are also included.

The second purpose of this chapter is discussed in Section 4.2: that is, to develop some inequalities on the power series functions by making use the convexity properties of certain underlying functions. The celebrated Jensen type

inequalities and one of their reverses due to Dragomir and Ionescu [139], are the useful tools that have been used in Section 4.2.3, in order to derive new inequalities for functions defined by the real power series with positive coefficients. Applications for some fundamental functions such as the exponential, logarithm, trigonometric and hyperbolic functions are provided as well.

All the results contained in this chapter are mainly taken from the several research papers published by the author in collaboration with Dragomir, Darus and Cerone (see [213], [215], [216]).

4.1 Some Results Related to Young's Inequality

4.1.1 Introduction and Preliminary Results

It has been shown in the literature that the famous (CBS)-inequality and its generalization, such as Hölder's inequality, can be derived by utilising Young's inequality for the product of two real numbers (see [422], [301, p. 457], [381, p. 63-64]). Consequently, some results concerning the extensions, generalizations, refinements, etc., of the (CBS) or Hölder's inequality, are closely related to Young's inequality as well.

In [122], Dragomir provided the generalization of the (CBS)-inequality by utilising Young's inequality (2.80) as follows (see also [117, p. 11], [136]):

$$\left(\sum_{k=1}^{n} |x_k y_k|\right)^2 \le \frac{1}{p} \sum_{k=1}^{n} |x_k|^p \sum_{k=1}^{n} |y_k|^p + \frac{1}{q} \sum_{k=1}^{n} |x_k|^q \sum_{k=1}^{n} |y_k|^q, \tag{4.1}$$

for $x_k, y_k \in \mathbb{C}$, $k \in \{1, 2, ..., n\}$ and p, q > 1 with 1/p + 1/q = 1. It is clearly seen that this result reduces to the (CBS)-inequality for complex numbers when p = q = 2. Further generalizations of the (CBS)-inequality via Young's inequality (2.80) have been obtained in [136] by Dragomir and Sàndor (see also [117, p. 10]), namely

$$\sum_{k=1}^{n} p_{k} |x_{k}y_{k}| \sum_{k=1}^{n} q_{k} |x_{k}y_{k}|$$

$$\leq \frac{1}{p} \sum_{k=1}^{n} p_{k} |x_{k}|^{p} \sum_{k=1}^{n} q_{k} |y_{k}|^{p} + \frac{1}{q} \sum_{k=1}^{n} q_{k} |x_{k}|^{q} \sum_{k=1}^{n} p_{k} |y_{k}|^{q}, \qquad (4.2)$$

for any $x_k, y_k \in \mathbb{C}$, $p_k, q_k \geq 0$, $k \in \{1, 2, ..., n\}$ and p, q > 1 with 1/p + 1/q = 1. If we assume in (4.2) that $p_k = q_k = 1$ for all $k \in \{1, 2, ..., n\}$, then it reduces to the inequality (4.1).

Motivated by the result (4.2) and other results from [136], we derive some inequalities related to the Hölder's type (2.94) for functions defined by the real and complex power series. The crucial tool that has been used for these investigations is Young's inequality (2.80), as well as a refinement and its reverse. Particular inequalities are also obtained by applying the results for some fundamental real and complex functions such as the exponential, logarithm, trigonometric and hyperbolic functions.

4.1.2 Power Series Inequalities Via Young's Inequality

On utilising Young's inequality (2.80) for the power series, we establish the following result, which improves the Hölder's type (2.94), as well as generalizes the (CBS)-inequality (2.71) for the power series with real coefficients (see [215]).

Theorem 65 (Ibrahim, Dragomir and Darus [215]) Let $f(z) = \sum_{n=0}^{\infty} p_n z^n$ and $g(z) = \sum_{n=0}^{\infty} q_n z^n$ be two power series with real coefficients and convergent on the open disk D(0,R), R > 0. If p > 1 with 1/p + 1/q = 1 and $x, y \in \mathbb{C}$, $x, y \neq 0$ so that $xy, |x|^p, |x|^q, |y|^p, |y|^q \in D(0,R)$, then

$$\frac{1}{p}g_{A}(|x|^{p})f_{A}(|y|^{p}) + \frac{1}{q}f_{A}(|x|^{q})g_{A}(|y|^{q}) \ge |f(xy)g(xy)|$$
(4.3)

and

$$\frac{1}{p}g_{A}(|x|^{p})f_{A}(|y|^{q}) + \frac{1}{q}f_{A}(|x|^{q})g_{A}(|y|^{p}) \ge |f(x|y|^{q-1})g(x|y|^{p-1})|.$$
 (4.4)

Proof. If we choose $x = |x|^j |y|^k$ and $y = |x|^k |y|^j$, $j, k \in \{0, 1, 2, ..., n\}$ in the inequality (2.80), then we have

$$p|x|^{qj}|y|^{qk} + q|x|^{pk}|y|^{pj} \ge pq|xy|^{j}|xy|^{k},$$
 (4.5)

for any $j, k \in \{0, 1, 2, ..., n\}$. Now, if we multiply the above inequality (4.5) with the positive quantities $|p_j| |q_k| > 0$, for $j, k \in \{0, 1, 2, ..., n\}$ and sum over j and k from 0 to n, then we derive

$$p\sum_{j=0}^{n}|p_{j}||x|^{qj}\sum_{k=0}^{n}|q_{k}||y|^{qk}+q\sum_{k=0}^{n}|q_{k}||x|^{pk}\sum_{j=0}^{n}|p_{j}||y|^{pj}$$

$$\geq pq\left|\sum_{j=0}^{n}p_{j}(xy)^{j}\sum_{k=0}^{n}q_{k}(xy)^{k}\right|.$$
(4.6)

Since all the series, whose the partial sums are involved in the inequality (4.6), are convergent on the disk D(0,R) by taking the limit as $n \to \infty$ in (4.6), we deduce the desired result (4.3).

Further, if we choose in (2.80) $x = |x|^j / |y|^j$ and $y = |x|^k / |y|^k$, then we get the inequality

$$p\left(\frac{|x|^{j}}{|y|^{j}}\right)^{q} + q\left(\frac{|x|^{k}}{|y|^{k}}\right)^{p} \ge pq\frac{|x|^{j}}{|y|^{j}}\frac{|x|^{k}}{|y|^{k}},\tag{4.7}$$

for any $|y|^j$, $|y|^k \neq 0$, $j, k \in \{0, 1, 2, ..., n\}$. Simplifying the above inequality (4.7), then we obtain that

$$p|x|^{qj}|y|^{pk} + q|x|^{pk}|y|^{qj} \ge pq|x|^{j}|y|^{(q-1)j}|x|^{k}|y|^{(p-1)k}$$

$$= pq\left|\left(x|y|^{q-1}\right)^{j}\left(x|y|^{p-1}\right)^{k}\right|, \tag{4.8}$$

for any $j, k \in \{0, 1, 2, ..., n\}$. Multiplying the inequality (4.8) by $|p_j| |q_k| \ge 0$, for $j, k \in \{0, 1, 2, ..., n\}$ and summing over j and k from 0 to n, we get

$$p \sum_{j=0}^{n} |p_{j}| |x|^{qj} \sum_{k=0}^{n} |q_{k}| |y|^{pk} + q \sum_{k=0}^{n} |q_{k}| |x|^{pk} \sum_{j=0}^{n} |p_{j}| |y|^{qj}$$

$$\geq pq \left| \sum_{j=0}^{n} p_{j} (x |y|^{q-1})^{j} \sum_{k=0}^{n} q_{k} (x |y|^{p-1})^{k} \right|. \tag{4.9}$$

Since all the series, whose the partial sums are involved in the inequality (4.9), are convergent on the disk D(0,R) and letting $n \to \infty$ in (4.9), we deduce the desired result (4.4).

The following result is the particular case of interest (see [215]).

Corollary 66 If g(z) = f(z) in (4.3) and (4.4), then

$$\frac{1}{p} f_A(|x|^p) f_A(|y|^p) + \frac{1}{q} f_A(|x|^q) f_A(|y|^q) \ge |f(xy)|^2$$
(4.10)

and

$$\frac{1}{p} f_A(|x|^p) f_A(|y|^q) + \frac{1}{q} f_A(|x|^q) f_A(|y|^p) \ge \left| f(x|y|^{q-1}) f(x|y|^{p-1}) \right|$$
(4.11)

respectively, where p > 1, 1/p + 1/q = 1 and $x, y \neq 0$ with xy, $|x|^p$, $|x|^q$, $|y|^p$, $|y|^q \in D(0,R)$. In particular, if y = x in (4.10) and (4.11), then we have

$$\frac{1}{p}f_A^2(|x|^p) + \frac{1}{q}f_A^2(|x|^q) \ge |f(x^2)|^2 \tag{4.12}$$

and

$$f_A(|x|^p) f_A(|x|^q) \ge |f(sgn(x)|x|^q) f(sgn(x)|x|^p)|$$
 (4.13)

respectively, for any $x \in \mathbb{C}$, $x \neq 0$ with x^2 , $|x|^p$, $|x|^q \in D(0,R)$ and sgn(x) is the complex signum function defined by (2.13).

Remark 67 In the particular case p = q = 2 in (4.10), we recapture the (CBS)-type inequality (2.71) for the power series, and we also have from (4.11) that

$$f_A(|x|^2) f_A(|y|^2) \ge |f(x|y|)|^2$$
,

for any $x, y \in \mathbb{C}$ with $x |y|, |x|^2, |y|^2 \in D(0, R)$.

Some applications of the inequalities (4.10) and (4.11) for particular functions of interest are as follows:

(1) If we apply the inequalities (4.10) and (4.11) for the function f(z) given by (1.19), then we get

$$\frac{1}{|1 - xy|^2} \le \frac{1}{p(1 - |x|^p)(1 - |y|^p)} + \frac{1}{q(1 - |x|^q)(1 - |y|^q)}$$
(4.14)

and

$$\frac{1}{\left|1 - x |y|^{q-1} \right| \left|1 - x |y|^{p-1} \right|} \\
\leq \frac{1}{p(1 - |x|^p)(1 - |y|^q)} + \frac{1}{q(1 - |x|^q)(1 - |y|^p)} \tag{4.15}$$

respectively, for any $x, y \in \mathbb{C}$ such that $x, y \neq 0$, xy, $|x|^p$, $|x|^q$, $|y|^p$, $|y|^q \in D(0,1)$ and p > 1 with 1/p + 1/q = 1.

(2) If we apply the inequalities (4.10) and (4.11) for the exponential function f(z) given by (1.25), then we can state that

$$|\exp(xy)|^2 \le \frac{1}{p} \exp(|x|^p + |y|^p) + \frac{1}{q} \exp(|x|^q + |y|^q)$$
 (4.16)

and

$$\left| \exp\left(x \, |y|^{q-1} + x \, |y|^{p-1}\right) \right|$$

$$\leq \frac{1}{p} \exp\left(|x|^p + |y|^q\right) + \frac{1}{q} \exp\left(|x|^q + |y|^p\right)$$
(4.17)

respectively, for any $x, y \in \mathbb{C}$ and p > 1 with 1/p + 1/q = 1.

(3) If we apply the logarithmic function f(z) given by (1.31), then from (4.10) and (4.11) we have the inequalities

$$\left|\ln\left(1 - xy\right)\right|^{2} \le \frac{1}{p}\ln\left(1 - |x|^{p}\right)\ln\left(1 - |y|^{p}\right) + \frac{1}{q}\ln\left(1 - |x|^{q}\right)\ln\left(1 - |y|^{q}\right) \tag{4.18}$$

and

$$\left| \ln \left(1 - x |y|^{q-1} \right) \ln \left(1 - x |y|^{p-1} \right) \right|$$

$$\leq \frac{1}{p} \ln \left(1 - |x|^p \right) \ln \left(1 - |y|^q \right) + \frac{1}{q} \ln \left(1 - |x|^q \right) \ln \left(1 - |y|^p \right) \tag{4.19}$$

respectively, for any $x, y \in \mathbb{C}$ with $x, y \neq 0$, $|x|^p$, $|x|^q$, $|y|^p$, $|y|^q \in D(0,1)$ and p > 1, 1/p + 1/q = 1.

(4) If we consider the trigonometric function f(z) given by (1.32), then we obviously have the transform $f_A(z)$ given by (1.38). Applying the inequalities (4.10) and (4.11) for these functions, we get

$$|\sin(xy)|^2 \le \frac{1}{p}\sinh(|x|^p)\sinh(|y|^p) + \frac{1}{q}\sinh(|x|^q)\sinh(|y|^q)$$
 (4.20)

and

$$\left|\sin\left(x\left|y\right|^{q-1}\right)\sin\left(x\left|y\right|^{p-1}\right)\right|$$

$$\leq \frac{1}{p}\sinh(\left|x\right|^{p})\sinh(\left|y\right|^{q}) + \frac{1}{q}\sinh(\left|x\right|^{q})\sinh(\left|y\right|^{p}) \tag{4.21}$$

respectively, for $x, y \in \mathbb{C}$ and p > 1 with 1/p + 1/q = 1.

(5) Similar results can be obtained for the cosine function f(z) given by (1.33) as well, that is, from (4.10) and (4.11) we get

$$|\cos(xy)|^2 \le \frac{1}{p}\cosh(|x|^p)\cosh(|y|^p) + \frac{1}{q}\cosh(|x|^q)\cosh(|y|^q)$$
 (4.22)

and

$$\left|\cos\left(x\left|y\right|^{q-1}\right)\cos\left(x\left|y\right|^{p-1}\right)\right|$$

$$\leq \frac{1}{p}\cosh(\left|x\right|^{p})\cosh(\left|y\right|^{q}) + \frac{1}{q}\cosh(\left|x\right|^{q})\cosh(\left|y\right|^{p}) \tag{4.23}$$

respectively, for $x, y \in \mathbb{C}$ and p > 1 with 1/p + 1/q = 1.

The second improvement of Hölder's inequality (2.94) for the power series via Young's inequality is incorporated in the following theorem [215].

Theorem 68 (Ibrahim, Dragomir and Darus [215]) Let f(z) and g(z) be as in Theorem 65. Then one has the inequality

$$\frac{1}{p}g_{A}(|x|^{p})f_{A}(|y|^{q}) + \frac{1}{q}f_{A}(|x|^{p})g_{A}(|y|^{q}) \ge |f(|x|^{p-1}|y|^{q-1})g(xy)|$$
(4.24)

and

$$\frac{1}{p} f_A(|x|^p) g_A(|y|^2) + \frac{1}{q} g_A(|x|^2) f_A(|y|^q) \ge \left| f(xy) g(|x|^{2/q} |y|^{2/p}) \right|. \tag{4.25}$$

Proof. If we choose in (2.80) $x = |y|^k / |y|^j$, $y = |x|^k / |x|^j$, $|x|^j$, $|y|^j \neq 0$, j, $k \in \{0, 1, 2, \dots, n\}$, then we have

$$p|y|^{qk}|x|^{pj} + q|x|^{pk}|y|^{qj} \ge pq|x|^{(p-1)j}|y|^{(q-1)j}|xy|^{k}$$

$$= pq\left|\left(|x|^{p-1}|y|^{q-1}\right)^{j}(xy)^{k}\right|$$
(4.26)

for any $j, k \in \{0, 1, 2, \dots, n\}$. Multiplying the inequality (4.26) with $|p_j| |q_k| \ge 0$, and summing over j and k from 0 to n, we obtain that

$$p \sum_{k=0}^{n} |q_{k}| |y|^{qk} \sum_{j=0}^{n} |p_{j}| |x|^{pj} + q \sum_{k=0}^{n} |q_{k}| |x|^{pk} \sum_{j=0}^{n} |p_{j}| |y|^{qj}$$

$$\geq pq \left| \sum_{j=0}^{n} p_{j} (|x|^{p-1} |y|^{q-1})^{j} \sum_{k=0}^{n} q_{k} (xy)^{k} \right|. \tag{4.27}$$

From (2.80), we also have the inequality

$$p \sum_{k=1}^{n} |q_{k}| |x|^{2k} \sum_{j=1}^{n} |p_{j}| |y|^{qj} + q \sum_{j=1}^{n} |p_{j}| |x|^{pj} \sum_{k=1}^{n} |q_{k}| |y|^{2k}$$

$$\geq pq \left| \sum_{j=1}^{n} p_{j} (xy)^{j} \sum_{k=1}^{n} q_{k} \left(|x|^{2/q} |y|^{2/p} \right)^{k} \right|, \tag{4.28}$$

for any $x, y \in \mathbb{C}$, p > 1 with 1/p + 1/q = 1, which was obtained from (2.80) by choosing $x = |x|^{(2/q)k} |y|^j$, $y = |x|^j |y|^{(2/p)k}$, and repeating the same method as above.

Now, since all the series, whose the partial sums are involved in the inequalities (4.27) and (4.28), are convergent on the disk D(0,R) and letting $n \to \infty$ in the both inequalities (4.27) and (4.28), we deduce the desired results (4.24) and (4.25).

The following corollary holds for the particular cases of interest (see [215]).

Corollary 69 If g(z) = f(z) in (4.24) and (4.25), then we get

$$f_A(|x|^p) f_A(|y|^q) \ge |f(xy) f(|x|^{p-1} |y|^{q-1})|$$
 (4.29)

and

$$\frac{1}{p} f_A(|x|^p) f_A(|y|^2) + \frac{1}{q} f_A(|x|^2) f_A(|y|^q) \ge \left| f(xy) f(|x|^{2/q} |y|^{2/p}) \right|$$
(4.30)

respectively, where p > 1, 1/p + 1/q = 1 and $x, y \neq 0$ with xy, $|x|^2$, $|x|^p$, $|x|^{2/q}$, $|y|^2$, $|y|^q$, $|y|^q$, $|y|^{2/p} \in D(0,R)$. In particular, if y = x in (4.29) and (4.30), then we have

$$f_A(|x|^p) f_A(|x|^q) \ge |f(x^2) f(|x|^{pq-2})|$$
 (4.31)

and

$$f_A(|x|^2)\left[\frac{1}{p}f_A(|x|^p) + \frac{1}{q}f_A(|x|^q)\right] \ge |f(x^2)f(|x|^2)|,$$
 (4.32)

for $x \neq 0$ with x^2 , $|x|^2$, $|x|^p$, $|x|^q$, $\in D(0, R)$.

Remark 70 In particular case p = q = 2 in (4.29) or (4.30), we get the inequality

$$f_A(|x|^2) f_A(|y|^2) \ge |f(xy) f(|xy|)|,$$
 (4.33)

for any $x, y \in \mathbb{C}$ with xy, |xy|, $|x|^2$, $|y|^2 \in D(0, R)$.

In what follows, we provide some applications of the inequalities (4.29) and (4.30) for particular functions of interest:

(1) If we apply the inequalities (4.29) and (4.30) for the function f(z) given by (1.19), then we get

$$(1 - |x|^p) (1 - |y|^q) \le |1 - xy| \left| 1 - |x|^{p-1} |y|^{q-1} \right| \tag{4.34}$$

and

$$\frac{1}{|1 - xy| \left| 1 - |x|^{2/q} |y|^{2/p} \right|} \le \frac{1}{p (1 - |x|^p) (1 - |y|^2)} + \frac{1}{q (1 - |x|^2) (1 - |y|^q)}$$
(4.35)

respectively, where p > 1, 1/p + 1/q = 1 and $x, y \neq 0$ with xy, $|x|^2$, $|x|^p$, $|x|^{1/q}$, $|y|^2$, $|y|^q$, $|y|^{1/p} \in D(0,1)$.

(2) If we apply the inequalities (4.29) and (4.30) for the exponential function f(z) given by (1.25), then we can state that

$$\left| \exp\left(xy + |x|^{p-1} |y|^{q-1} \right) \right| \le \exp\left(|x|^p + |y|^q \right)$$
 (4.36)

and

$$\left| \exp\left(xy + |x|^{2/q} |y|^{2/p} \right) \right| \le \frac{1}{p} \exp\left(|x|^p + |y|^2 \right) + \frac{1}{q} \exp\left(|x|^2 + |y|^q \right)$$
 (4.37)

respectively, for any $x, y \in \mathbb{C}$ with $x, y \neq 0$.

(3) If we take the logarithmic function f(z) given by (1.31), then from (4.29) and (4.30) we have

$$\left| \ln (1 - xy) \ln \left(1 - |x|^{p-1} |y|^{q-1} \right) \right| \le \ln (1 - |x|^p) \ln (1 - |y|^q) \tag{4.38}$$

and

$$\left| \ln (1 - xy) \ln \left(1 - |x|^{2/q} |y|^{2/p} \right) \right|$$

$$\leq \frac{1}{p} \ln (1 - |x|^p) \ln \left(1 - |y|^2 \right) + \frac{1}{q} \ln \left(1 - |x|^2 \right) \ln (1 - |y|^q)$$
(4.39)

respectively, where p > 1, 1/p + 1/q = 1 and $x, y \neq 0$ with $xy, |x|^2, |x|^p$, $|x|^{1/q}, |y|^2, |y|^q, |y|^{1/p} \in D(0,1)$.

(4) If we consider the trigonometric function f(z) given by (1.32), then the transform $f_A(z)$ is given by (1.38). Applying the inequalities (4.29) and (4.30) for these functions, we get

$$\left|\sin(xy)\sin(|x|^{p-1}|y|^{q-1})\right| \le \sinh(|x|^p)\sinh(|y|^q)$$
 (4.40)

and

$$\left| \sin(xy) \sin\left(|x|^{2/q} |y|^{2/p}\right) \right|$$

$$\leq \frac{1}{p} \sinh(|x|^p) \sinh(|y|^2) + \frac{1}{q} \sinh(|x|^2) \sinh(|y|^q)$$
(4.41)

respectively, where p > 1, 1/p + 1/q = 1 and $x, y \in \mathbb{C}$ with $x, y \neq 0$.

(5) A similar result can be obtained for the cosine function f(z) given by (1.33) as well, namely

$$\left|\cos(xy)\cos(|x|^{p-1}|y|^{q-1})\right| \le \cosh(|x|^p)\cosh(|y|^q)$$
 (4.42)

and

$$\left|\cos(xy)\cos\left(|x|^{2/q}|y|^{2/p}\right)\right|$$

$$\leq \frac{1}{p}\cosh(|x|^p)\cosh(|y|^2) + \frac{1}{q}\cosh(|x|^2)\cosh(|y|^q) \tag{4.43}$$

respectively, where p > 1, 1/p + 1/q = 1 and $x, y \in \mathbb{C}$ with $x, y \neq 0$.

Further, the following result provides another improvement of Hölder's inequality via Young's inequality [215].

Theorem 71 (Ibrahim, Dragomir and Darus [215]) Let f(z) and g(z) be as in Theorem 65. Then one has the inequality

$$\left| f\left(|x|^{p-1} |y|^{q-1} \right) g\left(|x|^{2/p} |y|^{2/q} \right) \right|
\leq \frac{1}{p} g_A\left(|x|^2 \right) f_A\left(|y|^q \right) + \frac{1}{q} f_A\left(|x|^p \right) g_A\left(|y|^2 \right)$$
(4.44)

and

$$\left| f\left(|x|^{2/q} y \right) g\left(|x|^{2/p} y \right) \right| \le \frac{1}{p} g_A\left(|x|^2 \right) f_A\left(|y|^p \right) + \frac{1}{q} f_A\left(|x|^2 \right) g_A\left(|y|^q \right).$$
 (4.45)

Proof. This follows from the inequality (2.80) by choosing $x = |y|^{(2/q)k} / |y|^j$, $y = |x|^{(2/p)k} / |x|^j$ and $x = |x|^{(2/q)j} |y|^k$, $y = |x|^{(2/p)k} |y|^j$. That is, for any $j, k \in \{0, 1, 2, \ldots n\}$, we have the following inequalities

$$p|x|^{pj}|y|^{2k} + q|x|^{2k}|y|^{qj} \ge pq|x|^{(p-1)j}|y|^{(q-1)j}|x|^{(2/p)k}|y|^{(2/q)k}$$

$$= pq\left|\left(|x|^{(p-1)}|y|^{(q-1)}\right)^{j}\left(|x|^{2/p}|y|^{2/q}\right)^{k}\right| (4.46)$$

and

$$p|x|^{2j}|y|^{qk} + q|x|^{2k}|y|^{pj} \ge pq|x|^{(2/q)j}|y|^{j}|x|^{(2/p)k}|y|^{k}$$

$$= pq\left|\left(|x|^{2/q}y\right)^{j}\left(|x|^{2/p}y\right)^{k}\right|$$
(4.47)

respectively. Repeating the same method as in Theorem 65 for the both inequalities (4.46) and (4.47), we deduce the desired results (4.44) and (4.45).

As a particular case of interest, we can state the following corollary:

Corollary 72 If g(z) = f(z) in (4.44) and (4.45), then

$$\left| f\left(|x|^{p-1} |y|^{q-1} \right) f\left(|x|^{2/p} |y|^{2/q} \right) \right|
\leq \frac{1}{p} f_A\left(|x|^2 \right) f_A\left(|y|^q \right) + \frac{1}{q} f_A\left(|x|^p \right) f_A\left(|y|^2 \right)$$
(4.48)

and

$$\left| f\left(|x|^{2/q} y \right) f\left(|x|^{2/p} y \right) \right| \le f_A \left(|x|^2 \right) \left[\frac{1}{p} f_A \left(|y|^p \right) + \frac{1}{q} f_A \left(|y|^q \right) \right],$$
 (4.49)

where p > 1, 1/p + 1/q = 1 and $x, y \neq 0$ with $|x|^2$, $|x|^p$, $|x|^q$, $|y|^2$, $|y|^p$, $|y|^q \in D(0,R)$. In particular, if y = x in (4.48) and (4.49), then we have

$$|f(|x|^2) f(|x|^{pq-2})| \le f_A(|x|^2) \left[\frac{1}{p} f_A(|x|^q) + \frac{1}{q} f_A(|x|^p) \right]$$
 (4.50)

and

$$\left| f\left(|x|^{2/q} x \right) f\left(|x|^{2/p} x \right) \right| \le f_A\left(|x|^2 \right) \left[\frac{1}{p} f_A\left(|x|^p \right) + \frac{1}{q} f_A\left(|x|^q \right) \right]$$
 (4.51)

for $x \neq 0$, $|x|^2$, $|x|^p$, $|x|^q \in D(0, R)$.

The inequalities (4.48) and (4.49) are also the valuable sources of particular inequalities for some fundamental functions as will be outlined in the following:

(1) If we apply the inequalities (4.48) and (4.49) for the function f(z) given by (1.19), then we get

$$\frac{1}{\left|\left(1-|x|^{p-1}|y|^{q-1}\right)\left(1-|x|^{2/p}|y|^{2/q}\right)\right|} \le \frac{1}{p\left(1-|x|^{2}\right)\left(1-|y|^{q}\right)} + \frac{1}{q\left(1-|x|^{p}\right)\left(1-|y|^{2}\right)}$$
(4.52)

and

$$\frac{1}{\left|\left(1-|x|^{2/q}y\right)\left(1-|x|^{2/p}y\right)\right|} \le \frac{1}{1-|x|^2}\left(\frac{1}{p\left(1-|y|^p\right)} + \frac{1}{q\left(1-|y|^q\right)}\right) \tag{4.53}$$

respectively, for any $x, y \in \mathbb{C}$, $x, y \neq 0$ with $|x|^2$, $|y|^2$, $|x|^p$, $|y|^q$, $|x|^{1/p}$, $|x|^{1/q}$, $|y|^{1/p}$, $|y|^{1/p}$, $|y|^{1/p} \in D(0,1)$.

(2) If we apply the inequalities (4.48) and (4.49) for the exponential function f(z) given by (1.25), then we can state that

$$\left| \exp\left(|x|^{p-1} |y|^{q-1} + |x|^{2/p} |y|^{2/q} \right) \right|$$

$$\leq \frac{1}{p} \exp\left(|x|^2 + |y|^q \right) + \frac{1}{q} \exp\left(|x|^p + |y|^2 \right)$$
(4.54)

and

$$\left| \exp\left[\left(\left| x \right|^{2/q} + \left| x \right|^{2/p} \right) y \right] \right|$$

$$\leq \exp\left(\left| x \right|^2 \right) \left[\frac{1}{p} \exp\left(\left| y \right|^p \right) + \frac{1}{q} \exp\left(\left| y \right|^q \right) \right]$$
(4.55)

respectively, for $x, y \in \mathbb{C}, x, y \neq 0$.

(3) If we apply the logarithmic function f(z) given by (1.31), then from (4.48) and (4.49) we have

$$\left| \ln \left(1 - |x|^{p-1} |y|^{q-1} \right) \ln \left(1 - |x|^{2/p} |y|^{2/q} \right) \right|$$

$$\leq \frac{1}{p} \ln \left(1 - |x|^2 \right) \ln \left(1 - |y|^q \right) + \frac{1}{q} \ln \left(1 - |x|^p \right) \ln \left(1 - |y|^2 \right) \tag{4.56}$$

and

$$\left| \ln \left(1 - |x|^{2/q} y \right) \ln \left(1 - |x|^{2/p} y \right) \right|$$

$$\leq \ln \left(1 - |x|^2 \right) \left[\frac{1}{p} \ln \left(1 - |y|^p \right) + \frac{1}{q} \ln \left(1 - |y|^q \right) \right]$$
(4.57)

respectively, for any $x, y \in \mathbb{C}$, $x, y \neq 0$ with $|x|^2$, $|y|^2$, $|x|^p$, $|y|^q$, $|x|^{1/p}$, $|x|^{1/q}$, $|y|^{1/p}$, $|y|^{1/p}$, $|y|^{1/p}$ $\in D(0,1)$.

(4) If we consider the trigonometric function f(z) given by (1.32), then we have $f_A(z)$ as given by (1.38). Applying the inequalities (4.48) and (4.49) for these functions, we get

$$\left| \sin\left(\left|x\right|^{p-1} \left|y\right|^{q-1}\right) \sin\left(\left|x\right|^{2/p} \left|y\right|^{2/q}\right) \right|$$

$$\leq \frac{1}{p} \sinh\left(\left|x\right|^{2}\right) \sinh\left(\left|y\right|^{q}\right) + \frac{1}{q} \sinh\left(\left|x\right|^{p}\right) \sinh\left(\left|y\right|^{2}\right) \tag{4.58}$$

and

$$\left| \sin\left(|x|^{2/q} y \right) \sin\left(|x|^{2/p} y \right) \right|$$

$$\leq \sinh\left(|x|^2 \right) \left[\frac{1}{p} \sinh\left(|y|^p \right) + \frac{1}{q} \sinh\left(|y|^q \right) \right]$$
(4.59)

respectively, for any $x, y \in \mathbb{C}, x, y \neq 0$.

(5) A similar result can be obtained for the hyperbolic function f(z) given by (1.38) as well, namely

$$\left|\sinh\left(\left|x\right|^{p-1}\left|y\right|^{q-1}\right)\sinh\left(\left|x\right|^{2/p}\left|y\right|^{2/q}\right)\right|$$

$$\leq \frac{1}{p}\sinh\left(\left|x\right|^{2}\right)\sinh\left(\left|y\right|^{q}\right) + \frac{1}{q}\sinh\left(\left|x\right|^{p}\right)\sinh\left(\left|y\right|^{2}\right) \tag{4.60}$$

and

$$\left|\sinh\left(|x|^{2/q}y\right)\sinh\left(|x|^{2/p}y\right)\right|$$

$$\leq \sinh\left(|x|^{2}\right)\left[\frac{1}{p}\sinh\left(|y|^{p}\right) + \frac{1}{q}\sinh\left(|y|^{q}\right)\right]$$
(4.61)

respectively, for any $x, y \in \mathbb{C}, x, y \neq 0$.

More recent studies for different types of Young's inequality can be found in the literature (see for instance [24], [90], [157], [160], [203], [220], [250], [259], [310], [412], [432], [433], [448], [449]).

In the next section, we refine Young's inequality of the form (2.81) and (2.85) of Kittaneh and Manasrah's result [250], then utilising these results, we obtain

further improvements of the Hölder's type inequality (2.94) for function defined by the real power series with real coefficients. In particular, we refine the Hölder's type inequalities from Section 4.1.2. Natural examples for some fundamental real functions are also presented.

4.1.3 Further Improvements of Hölder's Inequality for Power Series

Before we state our results, we first prove the following lemma, which provides the refinement and a reverse of the Young's inequality (2.81) for real numbers as follows (see [216]).

Lemma 73 For any $a, b \ge 0$ and $v \in [0, 1]$, we have

$$2\min\{v, 1 - v\} \left(\frac{a + b}{2} - \sqrt{ab}\right) \le va + (1 - v)b - a^{v}b^{1 - v}$$

$$\le 2\max\{v, 1 - v\} \left(\frac{a + b}{2} - \sqrt{ab}\right). \quad (4.62)$$

Proof. We recall the following result obtained by Dragomir in [119] that provides a refinement and a reverse of the weighted Jensen's discrete inequality (2.95), namely

$$n \min_{j \in \{1,2,\dots,n\}} \{p_j\} \left[\frac{1}{n} \sum_{j=1}^n \Phi(x_j) - \Phi\left(\frac{1}{n} \sum_{j=1}^n x_j\right) \right]$$

$$\leq \frac{1}{P_n} \sum_{j=1}^n p_j \Phi(x_j) - \Phi\left(\frac{1}{P_n} \sum_{j=1}^n p_j x_j\right)$$

$$\leq n \max_{j \in \{1,2,\dots,n\}} \{p_j\} \left[\frac{1}{n} \sum_{j=1}^n \Phi(x_j) - \Phi\left(\frac{1}{n} \sum_{j=1}^n x_j\right) \right], \tag{4.63}$$

where $\Phi: \mathbf{C} \to \mathbb{R}$ is a convex function defined on a convex subset \mathbf{C} of the linear space \mathbf{X} , $\{x_j\}$, $j \in \{1, 2, ..., n\}$ are vectors in \mathbf{C} and $\{p_j\}$, $j \in \{1, 2, ..., n\}$ are nonnegative numbers with $P_n = \sum_{j=1}^n p_j > 0$.

We notice that Furuichi's result (2.87) is the particular case of (4.63), which applied for the convex function $f(t) = \exp(t)$, and denoting the function $\exp(x_j)$ as a_j for $j \in \{1, ..., n\}$. Also, for n = 2, we deduce from (4.63) that

$$2\min\{v, 1 - v\} \left[\frac{\Phi(x) + \Phi(y)}{2} - \Phi\left(\frac{x + y}{2}\right) \right]$$

$$\leq v\Phi(x) + (1 - v)\Phi(y) - \Phi\left[vx + (1 - v)y\right]$$

$$\leq 2\max\{v, 1 - v\} \left[\frac{\Phi(x) + \Phi(y)}{2} - \Phi\left(\frac{x + y}{2}\right) \right], \tag{4.64}$$

for any $x, y \in \mathbb{R}$ and $v \in [0, 1]$. If we take the function $\Phi(x) = \exp(x)$, then we get from (4.64)

$$2\min\{v, 1 - v\} \left[\frac{\exp(x) + \exp(y)}{2} - \exp\left(\frac{x + y}{2}\right) \right]$$

$$\leq v \exp(x) + (1 - v) \exp(y) - \exp\left[vx + (1 - v)y\right]$$

$$\leq 2\max\{v, 1 - v\} \left[\frac{\exp(x) + \exp(y)}{2} - \exp\left(\frac{x + y}{2}\right) \right], \tag{4.65}$$

for any $x, y \in \mathbb{R}$ and $v \in [0, 1]$. Further, we denote $\exp(x) = a$, $\exp(y) = b$ with a, b > 0, then from the (4.65) we obtain the desired result (4.62).

From the refinement and its reverse of Young's inequality (4.62), we have the following corollary (see [216]).

Corollary 74 For any $x, y \ge 0$ and p > 1 with 1/p + 1/q = 1, we have

$$2\min\left\{\frac{1}{q}, \frac{1}{p}\right\} \left(\frac{x^{q} + y^{p}}{2} - x^{q/2}y^{p/2}\right)$$

$$\leq \frac{x^{q}}{q} + \frac{y^{p}}{p} - xy$$

$$\leq 2\max\left\{\frac{1}{q}, \frac{1}{p}\right\} \left(\frac{x^{q} + y^{p}}{2} - x^{q/2}y^{p/2}\right). \tag{4.66}$$

Proof. The proof follows by choosing $a=x^q$, $b=y^p$, v=1/q, 1-v=q/p in Lemma 73. \blacksquare

Remark 75 The first inequality in (4.62) provides Kittaneh and Manasrah's result (2.85) in [250] as well as a reverse of that result. One can also see that it is equivalent with Aldaz's result (2.86) [10].

First, utilizing the inequality (2.81) for functions defined by the real power series with positive coefficients, we obtain the following result (see [216]).

Theorem 76 Let $f(x) = \sum_{n=0}^{\infty} p_n x^n$ be a power series with positive coefficients and convergent on (0,R). Then for $v \in [0,1]$, $x, y \geq 0$ such that $y, xy, x^v y, x^{1-v}y \in (0,R)$, we have

$$f(x^{v}y) f(x^{1-v}y) \le f(xy)f(y).$$
 (4.67)

Proof. The proof follows by choosing in (2.81) $a = x^j$ and $b = x^k$ for $j, k \in \{0, 1, ..., n\}$. Thus, we have

$$x^{vj}x^{(1-v)k} \le vx^j + (1-v)x^k, \tag{4.68}$$

for any $x \ge 0$ and $v \in [0,1]$. If now we multiply the inequality (4.68) with the positive quantity, i.e., $p_j y^j p_k y^k \ge 0$, $j, k \in \{0,1,\ldots,n\}$, $y \in (0,R)$, and sum over j and k from 0 to n, then we get

$$\sum_{j=0}^{n} p_{j} (x^{v}y)^{j} \sum_{k=0}^{n} p_{k} (x^{1-v}y)^{k}$$

$$\leq v \sum_{j=0}^{n} p_{j} (xy)^{j} \sum_{k=0}^{n} p_{k}y^{k} + (1-v) \sum_{j=0}^{n} p_{j}y^{j} \sum_{k=0}^{n} p_{k} (xy)^{k}.$$
(4.69)

Since all the series whose partial sums are involved in the inequality (4.69), are convergent on the interval (0, R), by taking the limit as $n \to \infty$ in (4.69), we deduce the desired result (4.67).

Remark 77 (i) If xy = z in (4.67), then we have

$$f(y^{v}z^{1-v}) f(y^{1-v}z^{v}) \le f(y)f(z),$$
 (4.70)

for y, z, $y^v z^{1-v}$, $y^{1-v} z^v \in (0, R)$ and $v \in [0, 1]$.

(ii) If y = x in (4.67), then we also have

$$f(x^{1+v}) f(x^{2-v}) \le f(x^2) f(x),$$
 (4.71)

for x, x^2 , x^{1+v} , $x^{2-v} \in (0, R)$ and $v \in [0, 1]$.

Some applications of the inequality (4.70) for particular functions of interest are as follows:

(1) If we apply the inequality (4.70) for the function f(x) = 1/(1-x), $x \in (0,1)$, then we get

$$(1-y)(1-z) \le (1-y^v z^{1-v}) (1-y^{1-v} z^v), \qquad (4.72)$$

for $y, z, y^v z^{1-v}, y^{1-v} z^v \in (0, 1)$ and $v \in [0, 1]$.

(2) If we consider the logarithm function $f(x) = \ln [1/(1-x)], x \in (0,1)$, and apply the inequality (4.70), then we get

$$\ln\left(1 - y^{v}z^{1-v}\right)\ln\left(1 - y^{1-v}z^{v}\right) \le \ln\left(1 - y\right)\ln\left(1 - z\right),\tag{4.73}$$

for $y, z, y^v z^{1-v}, y^{1-v} z^v \in (0, 1)$ and $v \in [0, 1]$.

Next, we prove the following inequality based on the refinement and the reverse of Young's inequality (4.62) (see [216]).

Theorem 78 (Ibrahim, Dragomir and Darus [216]) Let f(x) be as in Theorem 76. Then, one has the inequality

$$2\min\{v, 1 - v\} \left[f(xy) f(y) - f^{2} \left(x^{1/2} y \right) \right]$$

$$\leq f(xy) f(y) - f(x^{v}y) f\left(x^{1-v}y \right)$$

$$\leq 2\max\{v, 1 - v\} \left[f(xy) f(y) - f^{2} \left(x^{1/2} y \right) \right]$$
(4.74)

for $x, y \ge 0$ such that $xy, y, x^{1/2}y, x^vy, x^{1-v}y \in (0, R)$ and $v \in [0, 1]$.

Proof. We use the inequality (4.62) for $a=x^j$ and $b=x^k$, $j, k \in \{0,1,\ldots,n\}$ to get

$$2\min\{v, 1 - v\} \left(\frac{x^{j} + x^{k}}{2} - x^{j/2}x^{k/2}\right)$$

$$\leq vx^{j} + (1 - v)x^{k} - x^{vj}x^{(1-v)k}$$

$$\leq 2\max\{v, 1 - v\} \left(\frac{x^{j} + x^{k}}{2} - x^{j/2}x^{k/2}\right)$$
(4.75)

for any $x, y \ge 0$ and $v \in [0,1]$. Then, multiplying the inequality (4.75) with $p_j y^j p_k y^k \ge 0$, $j, k \in \{0,1,\ldots,n\}$ and summing over j and k from 0 to n, we have

$$2t \left[\frac{1}{2} \left(\sum_{j=0}^{n} p_{j} x^{j} y^{j} \sum_{k=0}^{n} p_{k} y^{k} + \sum_{j=0}^{n} p_{j} y^{j} \sum_{k=0}^{n} p_{k} x^{k} y^{k} \right) \right.$$

$$\left. - \sum_{j=0}^{n} p_{j} x^{j/2} y^{j} \sum_{k=0}^{n} p_{k} x^{k/2} y^{k} \right]$$

$$\leq v \sum_{j=0}^{n} p_{j} x^{j} y^{j} \sum_{k=0}^{n} p_{k} y^{k} + (1-v) \sum_{j=0}^{n} p_{j} y^{j} \sum_{k=0}^{n} p_{k} x^{k} y^{k}$$

$$\left. - \sum_{j=0}^{n} p_{j} x^{vj} y^{j} \sum_{k=0}^{n} p_{k} x^{(1-v)k} y^{k} \right.$$

$$\leq 2T \left[\frac{1}{2} \left(\sum_{j=0}^{n} p_{j} x^{j} y^{j} \sum_{k=0}^{n} p_{k} y^{k} + \sum_{j=0}^{n} p_{j} y^{j} \sum_{k=0}^{n} p_{k} x^{k} y^{k} \right) \right.$$

$$\left. - \sum_{j=0}^{n} p_{j} x^{j/2} y^{j} \sum_{k=0}^{n} p_{k} x^{k/2} y^{k} \right], \qquad (4.76)$$

where $t = \min\{v, 1 - v\}$ and $T = \max\{v, 1 - v\}$.

Since all the series whose partial sums are involved in the inequality (4.76) are convergent on the interval (0, R), by taking the limit as $n \to \infty$ in (4.76), we deduce the desired result (4.74).

Remark 79 (a) If xy = z in (4.74), then we have

$$2\min\{v, 1 - v\} \left[f(y)f(z) - f^{2}(\sqrt{yz}) \right]$$

$$\leq f(y)f(z) - f(y^{v}z^{1-v}) f(y^{1-v}z^{v})$$

$$\leq 2\max\{v, 1 - v\} \left[f(y)f(z) - f^{2}(\sqrt{yz}) \right]$$
(4.77)

for $y, z, z^v y^{1-v}, z^{1-v} y^v \in (0, R)$ and $v \in [0, 1]$. This result provides somehow a symmetric form for the inequality (4.74) and has some nice applications as well, see (4.79).

(b) If y = x in (4.74), then we also have

$$2\min\{v, 1 - v\} \left[f(x)f(x^{2}) - f^{2}(x^{3/2}) \right]$$

$$\leq f(x)f(x^{2}) - f(x^{1+v}) f(x^{2-v})$$

$$\leq 2\max\{v, 1 - v\} \left[f(x)f(x^{2}) - f^{2}(x^{3/2}) \right]$$
(4.78)

for x, x^2 , $x^{3/2}$, x^{1+v} , $x^{2-v} \in (0, R)$ and $v \in [0, 1]$.

Now, if we consider the exponential function $f(x) = \exp(x)$, $x \in \mathbb{R}$ and apply the inequality (4.77), then we get

$$2\min\{v, 1 - v\} \left[\exp(y + z) - \exp(2\sqrt{yz}) \right]$$

$$\leq \exp(y + z) - \exp(y^{v}z^{1-v} + y^{1-v}z^{v})$$

$$\leq 2\max\{v, 1 - v\} \left[\exp(y + z) - \exp(2\sqrt{yz}) \right]$$
(4.79)

for any $y, z \ge 0$ and $v \in [0, 1]$.

The second improvement of Hölder's inequality (2.94) via the refinement and the reverse of Young's inequality (4.66) is incorporated in the following theorem (see [216]).

Theorem 80 (Ibrahim, Dragomir and Darus [216]) Let f(x) be as in Theorem 76. If p > 1, 1/p + 1/q = 1 and $x, y \ge 0$ such that xy, x^q , y^p , $x^{q/2}y^{p/2} \in (0, R)$, then

$$2t\left[\frac{1}{2}\left[f\left(x^{q}\right)+f\left(y^{p}\right)\right]-f\left(x^{q/2}y^{p/2}\right)\right]$$

$$\leq \frac{1}{q} f(x^{q}) + \frac{1}{p} f(y^{p}) - f(xy)
\leq 2T \left[\frac{1}{2} \left[f(x^{q}) + f(y^{p}) \right] - f(x^{q/2} y^{p/2}) \right]$$
(4.80)

where $t = \min \{1/q, 1/p\}$ and $T = \max \{1/q, 1/p\}$.

Proof. If we choose $x = x^j$ and $y = y^j$, $j \in \{0, 1, 2, ..., n\}$, then we have from (4.66) that

$$2t\left(\frac{x^{qj}+y^{pj}}{2}-x^{(q/2)j}y^{(p/2)j}\right) \leq \frac{x^{qj}}{q}+\frac{y^{pj}}{p}-(xy)^{j}$$

$$\leq 2T\left(\frac{x^{qj}+y^{pj}}{2}-x^{(q/2)j}y^{(p/2)j}\right) \quad (4.81)$$

for any $x, y \ge 0$ and p > 1 with 1/p + 1/q = 1. If we multiply this inequality (4.81) with the positive quantities $p_j > 0, j \in \{0, 1, 2, ..., n\}$, and sum over j from 0 to n, then we get

$$2t \left(\frac{1}{2} \left[\sum_{j=0}^{n} p_{j} x^{qj} + \sum_{j=0}^{n} p_{j} y^{pj} \right] - \sum_{j=0}^{n} p_{j} \left(x^{q/2} y^{p/2} \right)^{j} \right)$$

$$\leq \frac{1}{q} \sum_{j=0}^{n} p_{j} x^{qj} + \frac{1}{p} \sum_{j=0}^{n} p_{j} y^{pj} - \sum_{j=0}^{n} p_{j} \left(xy \right)^{j}$$

$$\leq 2T \left(\frac{1}{2} \left[\sum_{j=0}^{n} p_{j} x^{qj} + \sum_{j=0}^{n} p_{j} y^{pj} \right] - \sum_{j=0}^{n} p_{j} \left(x^{q/2} y^{p/2} \right)^{j} \right). \tag{4.82}$$

Since all the series whose partial sums are involved in the inequality (4.82) are convergent on the interval (0, R), by taking the limit as $n \to \infty$ in (4.82), we deduce the desired inequality (4.80).

Corollary 81 If y = x in (4.80), then for any x^2 , x^q , x^p , $x^{pq/2} \in (0, R)$ and p > 1 with 1/p + 1/q = 1 we have

$$2t \left[\frac{1}{2} \left[f\left(x^{q}\right) + f\left(x^{p}\right) \right] - f\left(x^{pq/2}\right) \right]$$

$$\leq \frac{1}{q} f\left(x^{q}\right) + \frac{1}{p} f\left(x^{p}\right) - f\left(x^{2}\right)$$

$$\leq 2T \left[\frac{1}{2} \left[f(x^q) + f(x^p) \right] - f(x^{pq/2}) \right].$$
 (4.83)

In the following, we give some applications of the inequality (4.83) for particular functions of interest:

(1) If we apply the inequality (4.83) for the function f(x) = 1/(1-x), $x \in (0,1)$, then we get

$$t \left[\frac{1}{1 - x^q} + \frac{1}{1 - x^p} - \frac{2}{1 - x^{pq/2}} \right]$$

$$\leq \frac{1}{q(1 - x^q)} + \frac{1}{p(1 - x^p)} - \frac{1}{1 - x^2}$$

$$\leq T \left[\frac{1}{1 - x^q} + \frac{1}{1 - x^p} - \frac{2}{1 - x^{pq/2}} \right]$$
(4.84)

for x^2 , x^q , x^p , $x^{pq/2} \in (0,1)$, p > 1 with 1/p + 1/q = 1.

(2) If we apply the inequality (4.83) for the function $f(x) = \ln [1/(1-x)]$, $x \in (0,1)$, then we get

$$\left(\frac{\left(1-x^{pq/2}\right)^{2}}{\left(1-x^{q}\right)\left(1-x^{p}\right)}\right)^{t} \leq \frac{1-x^{2}}{\left(1-x^{q}\right)^{1/q}\left(1-x^{p}\right)^{1/p}} \\
\leq \left(\frac{\left(1-x^{pq/2}\right)^{2}}{\left(1-x^{q}\right)\left(1-x^{p}\right)}\right)^{T}, \tag{4.85}$$

for x^2 , x^q , x^p , $x^{pq/2} \in (0,1)$, p > 1 with 1/p + 1/q = 1.

(3) If we consider the hyperbolic function $f(x) = \sinh(x)$, $x \in \mathbb{R}$ and apply the inequality (4.83), then we get

$$t \left[\sinh\left(x^{q}\right) + \sinh\left(x^{p}\right) - 2\sinh\left(x^{pq/2}\right) \right]$$

$$\leq \frac{1}{q} \sinh\left(x^{q}\right) + \frac{1}{p} \sinh\left(x^{p}\right) - \sinh\left(x^{2}\right)$$

$$\leq T \left[\sinh\left(x^{q}\right) + \sinh\left(x^{p}\right) - 2\sinh\left(x^{pq/2}\right) \right] \tag{4.86}$$

for any $x \ge 0, p > 1$ with 1/p + 1/q = 1.

(4) A similar result can be obtained for the function $f(x) = \cosh(x)$, namely

$$t \left[\cosh\left(x^{q}\right) + \cosh\left(x^{p}\right) - 2\cosh\left(x^{pq/2}\right)\right]$$

$$\leq \frac{1}{q}\cosh\left(x^{q}\right) + \frac{1}{p}\cosh\left(x^{p}\right) - \cosh\left(x^{2}\right)$$

$$\leq T \left[\cosh f\left(x^{q}\right) + \cosh\left(x^{p}\right) - 2\cosh\left(x^{pq/2}\right)\right] \tag{4.87}$$

for any $x \ge 0$, p > 1 with 1/p + 1/q = 1.

Further, we utilize the inequality (4.66) to improve the results from Section 4.1.2, giving the refinements and the reverses of Hölder's inequality for two functions defined by the real power series with positive coefficients. First, we obtain the following result (see [216]).

Theorem 82 (Ibrahim, Dragomir and Darus [216]) Let $f(x) = \sum_{n=0}^{\infty} p_n x^n$ and $g(x) = \sum_{n=0}^{\infty} q_n x^n$ be two power series with positive coefficients and convergent on (0,R). If p > 1, 1/p + 1/q = 1 and $x, y \ge 0$ such that $xy, x^p, x^q, y^p, y^q, x^{q/2}y^{p/2}, x^{p/2}y^{q/2}, (xy)^{q/2}, (xy)^{p/2}, xy^{q-1}, xy^{p-1} \in (0,R)$, then

$$t \left[f(x^{q})g(y^{q}) + g(x^{p})f(y^{p}) - 2f\left(x^{q/2}y^{p/2}\right)g\left(x^{p/2}y^{q/2}\right) \right]$$

$$\leq \frac{1}{q}f(x^{q})g(y^{q}) + \frac{1}{p}g(x^{p})f(y^{p}) - f\left(xy\right)g\left(xy\right)$$

$$\leq T \left[f(x^{q})g(y^{q}) + g(x^{p})f(y^{p}) - 2f\left(x^{q/2}y^{p/2}\right)g\left(x^{p/2}y^{q/2}\right) \right]$$
(4.88)

and

$$t \left[f(x^{q})g(y^{p}) + g(x^{p})f(y^{q}) - 2f(x^{q/2}y^{q/2})g(x^{p/2}y^{p/2}) \right]$$

$$\leq \frac{1}{q}f(x^{q})g(y^{p}) + \frac{1}{p}g(x^{p})f(y^{q}) - f(xy^{q-1})g(xy^{p-1})$$

$$\leq T \left[f(x^{q})g(y^{p}) + g(x^{p})f(y^{q}) - 2f(x^{q/2}y^{q/2})g(x^{p/2}y^{p/2}) \right]. \tag{4.89}$$

Proof. If we choose in (4.66) $x = x^j y^k$ and $y = x^k y^j$, $j, k \in \{0, 1, 2, ..., n\}$, then we have

$$2t\left(\frac{x^{qj}y^{qk} + x^{pk}y^{pj}}{2} - x^{(q/2)j}y^{(p/2)j}x^{(p/2)k}y^{(q/2)k}\right)$$

$$\leq \frac{1}{q} \left(x^{qj} y^{qk} \right) + \frac{1}{p} \left(x^{pk} y^{pj} \right) - (xy)^j (xy)^k
\leq 2T \left(\frac{x^{qj} y^{qk} + x^{pk} y^{pj}}{2} - x^{(q/2)j} y^{(p/2)j} x^{(p/2)k} y^{(q/2)k} \right)$$
(4.90)

for any $x, y \ge 0$ and p > 1 with 1/p + 1/q = 1. Multiplying this inequality (4.90) with $p_j q_k \ge 0$, $j, k \in \{0, 1, 2, ..., n\}$ and summing over j and k from 0 to n, we get

$$2t\left(\frac{1}{2}\left[\sum_{j=0}^{n}p_{j}x^{qj}\sum_{k=0}^{n}q_{k}y^{qk} + \sum_{k=0}^{n}q_{k}x^{pk}\sum_{j=0}^{n}p_{j}y^{pj}\right]\right]$$

$$-\sum_{j=0}^{n}p_{j}x^{(q/2)j}y^{(p/2)j}\sum_{k=0}^{n}q_{k}x^{(p/2)k}y^{(q/2)k}\right)$$

$$\leq \frac{1}{q}\left(\sum_{j=0}^{n}p_{j}x^{qj}\sum_{k=0}^{n}q_{k}y^{qk}\right) + \frac{1}{p}\left(\sum_{k=0}^{n}q_{k}x^{pk}\sum_{j=0}^{n}p_{j}y^{pj}\right)$$

$$-\sum_{j=0}^{n}p_{j}\left(xy\right)^{j}\sum_{k=0}^{n}q_{k}\left(xy\right)^{k}$$

$$\leq 2T\left(\frac{1}{2}\left[\sum_{j=0}^{n}p_{j}x^{qj}\sum_{k=0}^{n}q_{k}y^{qk} + \sum_{k=0}^{n}q_{k}x^{pk}\sum_{j=0}^{n}p_{j}y^{pj}\right]$$

$$-\sum_{j=0}^{n}p_{j}x^{(q/2)j}y^{(p/2)j}\sum_{k=0}^{n}q_{k}x^{(p/2)k}y^{(q/2)k}\right),$$

$$(4.91)$$

where p > 1, 1/p + 1/q = 1.

Further, if we choose in (4.66) $x = x^j/y^j$ and $y = x^k/y^k$, $y \neq 0$, $j, k \in \{0, 1, 2, ..., n\}$ and repeat the same method as above, then we get

$$2t \left(\frac{1}{2} \left[\sum_{j=0}^{n} p_j x^{qj} \sum_{k=0}^{n} q_k y^{pk} + \sum_{k=0}^{n} q_k x^{pk} \sum_{j=0}^{n} p_j y^{qj} \right] - \sum_{j=0}^{n} p_j x^{(q/2)j} y^{(q/2)j} \sum_{k=0}^{n} q_k x^{(p/2)k} y^{(p/2)k} \right)$$

$$\leq \frac{1}{q} \sum_{j=0}^{n} p_j x^{qj} \sum_{k=0}^{n} q_k y^{pk} + \frac{1}{p} \sum_{k=0}^{n} q_k x^{pk} \sum_{j=0}^{n} p_j y^{qj}$$

$$-\sum_{j=0}^{n} p_{j} x^{j} y^{(q-1)j} \sum_{k=0}^{n} q_{k} x^{k} y^{(p-1)k}$$

$$\leq 2T \left(\frac{1}{2} \left[\sum_{j=0}^{n} p_{j} x^{qj} \sum_{k=0}^{n} q_{k} y^{pk} + \sum_{k=0}^{n} q_{k} x^{pk} \sum_{j=0}^{n} p_{j} y^{qj} \right]$$

$$-\sum_{j=0}^{n} p_{j} x^{(q/2)j} y^{(q/2)j} \sum_{k=0}^{n} q_{k} x^{(p/2)k} y^{(p/2)k} \right)$$

$$(4.92)$$

where p > 1, 1/p + 1/q = 1.

Since all the series whose partial sums are involved in the inequalities (4.91) and (4.92) are convergent on the interval (0, R), by taking the limit as $n \to \infty$ in the both inequalities (4.91) and (4.92), we deduce the desired results (4.88) and (4.89).

The natural consequence of Theorem 82 is as follows:

Corollary 83 If q(x) = f(x) in (4.88) and (4.89), then we have

$$t \left[f(x^{q}) f(y^{q}) + f(x^{p}) f(y^{p}) - 2f \left(x^{q/2} y^{p/2} \right) f \left(x^{p/2} y^{q/2} \right) \right]$$

$$\leq \frac{1}{q} f(x^{q}) f(y^{q}) + \frac{1}{p} f(x^{p}) f(y^{p}) - f^{2} \left(xy \right)$$

$$\leq T \left[f(x^{q}) f(y^{q}) + f(x^{p}) f(y^{p}) - 2f \left(x^{q/2} y^{p/2} \right) f \left(x^{p/2} y^{q/2} \right) \right]$$
(4.93)

and

$$t \left[f(x^{q})f(y^{p}) + f(x^{p})f(y^{q}) - 2f(x^{q/2}y^{q/2})f(x^{p/2}y^{p/2}) \right]$$

$$\leq \frac{1}{q}f(x^{q})f(y^{p}) + \frac{1}{p}f(x^{p})f(y^{q}) - f(xy^{q-1})f(xy^{p-1})$$

$$\leq T \left[f(x^{q})f(y^{p}) + f(x^{p})f(y^{q}) - 2f(x^{q/2}y^{q/2})f(x^{p/2}y^{p/2}) \right]$$
(4.94)

respectively, for any $x, y \ge 0$ such that $xy, x^p, x^q, y^p, y^q \in (0, R)$ and p > 1 with 1/p + 1/q = 1.

The above results (4.93) and (4.94) have some natural applications for some particular functions of interest. For example, if we apply the inequalities (4.93) and (4.94) for the exponential function f(x) given by (1.25), $x \in \mathbb{R}$, then we get the following inequalities:

$$t \left[\exp(x^{q} + y^{q}) + \exp(x^{p} + y^{p}) - 2 \exp\left(x^{q/2}y^{p/2} + x^{p/2}y^{q/2}\right) \right]$$

$$\leq \frac{1}{q} \exp(x^{q} + y^{q}) + \frac{1}{p} \exp(x^{p} + y^{p}) - \exp\left(2xy\right)$$

$$\leq T \left[\exp(x^{q} + y^{q}) + \exp(x^{p} + y^{p}) - 2 \exp\left(x^{q/2}y^{p/2} + x^{p/2}y^{q/2}\right) \right]$$
(4.95)

and

$$t \left[\exp(x^{q} + y^{p}) + \exp(x^{p} + y^{q}) - 2 \exp(x^{q/2}y^{q/2} + x^{p/2}y^{p/2}) \right]$$

$$\leq \frac{1}{q} \exp(x^{q} + y^{p}) + \frac{1}{p} \exp(x^{p} + y^{q}) - \exp(xy^{q-1} + xy^{p-1})$$

$$\leq T \left[\exp(x^{q} + y^{p}) + \exp(x^{p} + y^{q}) - 2 \exp(x^{q/2}y^{q/2} + x^{p/2}y^{p/2}) \right], \tag{4.96}$$

respectively, for any $x \ge 0$, y > 0 and p > 1 with 1/p + 1/q = 1.

We also obtain the following result (see [216]).

Theorem 84 (Ibrahim, Dragomir and Darus [216]) Let f(x) and g(x) be as in Theorem 82. Then, one has the inequalities

$$t \left[f(x^{p})g(y^{q}) + g(x^{p})f(y^{q}) - 2g\left(x^{p/2}y^{q/2}\right) f\left(x^{p/2}y^{q/2}\right) \right]$$

$$\leq \frac{1}{q} f(x^{p})g(y^{q}) + \frac{1}{p} g(x^{p})f(y^{q}) - f\left(x^{p-1}y^{q-1}\right) g\left(xy\right)$$

$$\leq T \left[f(x^{p})g(y^{q}) + g(x^{p})f(y^{q}) - 2g\left(x^{p/2}y^{q/2}\right) f\left(x^{p/2}y^{q/2}\right) \right]$$
(4.97)

and

$$t \left[g\left(x^{2} \right) f\left(y^{q} \right) + f\left(x^{p} \right) g\left(y^{2} \right) - 2f\left(x^{p/2}y^{q/2} \right) g\left(xy \right) \right]$$

$$\leq \frac{1}{q} g\left(x^{2} \right) f\left(y^{q} \right) + \frac{1}{p} f\left(x^{p} \right) g\left(y^{2} \right) - f\left(xy \right) g\left(x^{2/q}y^{2/p} \right)$$

$$\leq T \left[g\left(x^{2} \right) f\left(y^{q} \right) + f\left(x^{p} \right) g\left(y^{2} \right) - 2f\left(x^{p/2}y^{q/2} \right) g\left(xy \right) \right]. \tag{4.98}$$

Proof. If we choose in (4.66) $x = y^k/y^j$, $y = x^k/x^j$, $x, y \neq 0$ and $x = x^{(2/q)k}y^j$, $y = x^jy^{(2/p)k}$, $j,k \in \{0,1,2,\ldots,n\}$, then we have the following inequalities:

$$t\left(x^{pj}y^{qk} + x^{pk}y^{qj} - 2x^{(p/2)k}y^{(q/2)k}x^{(p/2)j}y^{(q/2)j}\right)$$

$$\leq \frac{1}{q} x^{pj} y^{qk} + \frac{1}{p} x^{pk} y^{qj} - x^{(p-1)j} y^{(q-1)j} x^k y^k
\leq T \left(x^{pj} y^{qk} + x^{pk} y^{qj} - 2x^{(p/2)k} y^{(q/2)k} x^{(p/2)j} y^{(q/2)j} \right)$$
(4.99)

and

$$t \left(x^{2k} y^{qj} + x^{pj} y^{2k} - 2x^{(p/2)j} y^{(q/2)j} x^k y^k \right)$$

$$\leq \frac{1}{q} x^{2k} y^{qj} + \frac{1}{p} x^{pj} y^{2k} - x^j y^j x^{(2/q)k} y^{(2/p)k}$$

$$\leq T \left(x^{2k} y^{qj} + x^{pj} y^{2k} - 2x^{(p/2)j} y^{(q/2)j} x^k y^k \right), \tag{4.100}$$

respectively, for any $x, y \ge 0$ and p > 1 with 1/p + 1/q = 1. Now, repeating the same method as in Theorem 82, we obtain the desired inequalities (4.97) and (4.98).

Corollary 85 If g(x) = f(x) in (4.97) and (4.98), then we have

$$2t \left[f(x^p) f(y^q) - f^2 \left(x^{p/2} y^{q/2} \right) \right] \le f(x^p) f(y^q) - f \left(x^{p-1} y^{q-1} \right) f(xy)$$

$$\le 2T \left[f(x^p) f(y^q) - f^2 \left(x^{p/2} y^{q/2} \right) \right]$$
(4.101)

and

$$t \left[f\left(x^{2}\right) f\left(y^{q}\right) + f\left(x^{p}\right) f\left(y^{2}\right) - 2f\left(xy\right) f\left(x^{p/2}y^{q/2}\right) \right]$$

$$\leq \frac{1}{q} f\left(x^{2}\right) f\left(y^{q}\right) + \frac{1}{p} f\left(x^{p}\right) f\left(y^{2}\right) - f\left(xy\right) f\left(x^{2/q}y^{2/p}\right)$$

$$\leq T \left[f\left(x^{2}\right) f\left(y^{q}\right) + f\left(x^{p}\right) g\left(y^{2}\right) - 2f\left(xy\right) f\left(x^{p/2}y^{q/2}\right) \right]. \tag{4.102}$$

The above inequalities also provide some natural applications for particular functions of interest. We give some examples here as follows:

(1) If we apply the inequality (4.97) for the hyperbolic functions $f(x) = \sinh(x)$ and $g(x) = \cosh(x)$, $x \in \mathbb{R}$, then we get

$$t \left[\sinh(x^{p} + y^{q}) - \sinh\left(2x^{p/2}y^{q/2}\right) \right]$$

$$\leq \frac{1}{q} \sinh(x^{p}) \cosh(y^{q}) + \frac{1}{p} \cosh(x^{p}) \sinh(y^{q})$$

$$- \sinh\left(x^{p-1}y^{q-1}\right) \cosh\left(xy\right)$$

$$\leq T \left[\sinh(x^{p} + y^{q}) - \sinh\left(2x^{p/2}y^{q/2}\right) \right], \tag{4.103}$$

for any $x, y \ge 0$ and p > 1 with 1/p + 1/q = 1.

(2) Further, if we consider the exponential function $f(x) = \exp(x)$, $x \in \mathbb{R}$ and apply the inequality (4.101), then we get

$$2t \left[\exp(x^{p} + y^{q}) - \exp\left(2x^{p/2}y^{q/2}\right) \right]$$

$$\leq \exp(x^{p} + y^{q}) - \exp\left(x^{p-1}y^{q-1} + xy\right)$$

$$\leq 2T \left[\exp(x^{p} + y^{q}) - \exp\left(2x^{p/2}y^{q/2}\right) \right], \tag{4.104}$$

for any $x, y \ge 0$ and p > 1 with 1/p + 1/q = 1.

We end this section with the following result (see [216]).

Theorem 86 (Ibrahim, Dragomir and Darus [216]) Let f(x) and g(x) be as in Theorem 82. Then one has the inequality

$$t \left[f(x^{p}) g(y^{2}) + g(x^{2}) f(y^{q}) - 2f(x^{p/2}y^{q/2}) g(xy) \right]$$

$$\leq \frac{1}{q} f(x^{p}) g(y^{2}) + \frac{1}{p} g(x^{2}) f(y^{q}) - f(x^{p-1}y^{q-1}) g(x^{2/p}y^{2/q})$$

$$\leq T \left[f(x^{p}) g(y^{2}) + g(x^{2}) f(y^{q}) - 2f(x^{p/2}y^{q/2}) g(xy) \right]$$
(4.105)

and

$$t \left[f\left(x^{2}\right) g\left(y^{q}\right) + g\left(x^{2}\right) f\left(y^{p}\right) - 2f\left(xy^{p/2}\right) g\left(xy^{q/2}\right) \right]$$

$$\leq \frac{1}{q} f\left(x^{2}\right) g\left(y^{q}\right) + \frac{1}{p} g\left(x^{2}\right) f\left(y^{p}\right) - f\left(x^{2/q}y\right) g\left(x^{2/p}y\right)$$

$$\leq T \left[f\left(x^{2}\right) g\left(y^{q}\right) + g\left(x^{2}\right) f\left(y^{p}\right) - 2f\left(xy^{p/2}\right) g\left(xy^{q/2}\right) \right]. \tag{4.106}$$

Proof. Again, the proof follows by using the same method as in Theorem 82, on choosing in (4.66) as $x = y^{(2/q)k}/y^j$, $y = x^{(2/p)k}/x^j$, $x, y \neq 0$ and $x = x^{(2/q)j}y^k$, $y = x^{(2/p)k}y^j$, $j, k \in \{0, 1, 2, ..., n\}$ respectively. The details are omitted.

Corollary 87 If g(x) = f(x) in (4.105) and (4.106), then we have

$$t \left[f(x^p) f(y^2) + f(x^2) f(y^q) - 2f(xy) f(x^{p/2}y^{q/2}) \right]$$

$$\leq \frac{1}{q} f(x^{p}) f(y^{2}) + \frac{1}{p} f(x^{2}) f(y^{q}) - f(x^{p-1}y^{q-1}) f(x^{2/p}y^{2/q})
\leq T \left[f(x^{p}) f(y^{2}) + f(x^{2}) f(y^{q}) - 2f(xy) f(x^{p/2}y^{q/2}) \right]$$
(4.107)

and

$$t \left[f\left(x^{2}\right) \left[f\left(y^{q}\right) + f\left(y^{p}\right) \right] - 2f\left(xy^{p/2}\right) f\left(xy^{q/2}\right) \right]$$

$$\leq f\left(x^{2}\right) \left[\frac{1}{q} f\left(y^{q}\right) + \frac{1}{p} f\left(y^{p}\right) \right] - f\left(x^{2/q} y\right) f\left(x^{2/p} y\right)$$

$$\leq T \left[f\left(x^{2}\right) \left[f\left(y^{q}\right) + f\left(y^{p}\right) \right] - 2f\left(xy^{p/2}\right) f\left(xy^{q/2}\right) \right]. \tag{4.108}$$

4.2 Some Results Via Convexity and Jensen's Type Inequalities

4.2.1 Introduction and Preliminary Results

In 1989, Pečarić and Dragomir [351] established a refinement of Jensen's inequality (2.95) as follows:

$$f\left(\frac{1}{P_{n}}\sum_{j=1}^{n}p_{j}x_{j}\right) \leq \frac{1}{P_{n}^{m+1}}\sum_{j_{1},j_{2},\dots,j_{m+1}}p_{j_{1}}\dots p_{j_{m}}f\left(\frac{x_{j_{1}}+\dots+x_{j_{m+1}}}{m+1}\right)$$

$$\leq \frac{1}{P_{n}^{m}}\sum_{j_{1},j_{2},\dots,j_{m}=1}p_{j_{1}}\dots p_{j_{m}}f\left(\frac{x_{j_{1}}+\dots+x_{j_{m}}}{m}\right)$$

$$\leq \dots \leq \frac{1}{p_{n}}\sum_{j=1}^{n}p_{j}f\left(x_{j}\right), \tag{4.109}$$

for a convex function $f: I \subseteq \mathbb{R} \to \mathbb{R}$, $x_j \in I$, $j \in \{1, 2, ..., n\}$ and $p_j \geq 0$ with $P_n = \sum_{j=1}^n p_j > 0$, where $m \geq 1$, $m \in \mathbb{N}$. The following refinement of the weighted Jensen's inequality was obtained by Dragomir and Scarmozzino [137] based on the properties of modulus of differentiable convex functions. It states that (see also [117, p. 204])

$$\frac{1}{p_n} \sum_{j=1}^n p_j f(x_j) - f\left(\frac{1}{P_n} \sum_{j=1}^n p_j x_j\right) \\
\ge \left| \frac{1}{p_n} \sum_{j=1}^n p_j \left| f(x_j) - f\left(\frac{1}{p_n} \sum_{j=1}^n p_j x_j\right) \right| \\
- \left| f'\left(\frac{1}{P_n} \sum_{j=1}^n p_j x_j\right) \right| \cdot \frac{1}{P_n} \sum_{j=1}^n p_j \left| x_j - \frac{1}{P_n} \sum_{j=1}^n p_j x_j \right| \ge 0, \tag{4.110}$$

for any $x_j \in I$, $j \in \{1, 2, ..., n\}$ and $p_j \ge 0$ with $P_n = \sum_{j=1}^n p_j > 0$. In particular, we have from (4.110) the following result for unweighted means:

$$\frac{f(x_1) + \dots + f(x_n)}{n} - f\left(\frac{x_1 + \dots + x_n}{n}\right)$$

$$\geq \left|\frac{1}{n}\sum_{j=1}^n \left| f(x_j) - f\left(\frac{x_1 + \dots + x_n}{n}\right) \right|$$

$$- \left| f'\left(\frac{x_1 + \dots + x_n}{n}\right) \right| \cdot \frac{1}{n}\sum_{j=1}^n \left| x_j - \frac{1}{n}\sum_{k=1}^n x_k \right| \quad (\geq 0), \tag{4.111}$$

for any $x_j \in I$, $j \in \{1, 2, ..., n\}$. If we apply the inequality (4.110) for the convex function $f(x) = x^2$, $x \in \mathbb{R}$, and choose $x_j = a_j/b_j$, $p_j = b_j^2$, $j \in \{1, 2, ..., n\}$, then we get a refinement of the classical (CBS)-inequality for real numbers.

Meanwhile, in 1994, Dragomir and Ionescu [139] proved the following reverse of the Jensen's type inequality for a differentiable convex function,

$$0 \le \sum_{j=1}^{n} q_{j} f(x_{j}) - f\left(\sum_{j=1}^{n} q_{j} x_{j}\right)$$

$$\le \sum_{j=1}^{n} q_{j} x_{j} f'(x_{j}) - \sum_{j=1}^{n} q_{j} x_{j} \sum_{j=1}^{n} q_{j} f'(x_{j}),$$

where $q_j \ge 0$, $j \in \{1, 2, ..., n\}$ such that $\sum_{j=1}^n q_j = 0$, or in more general case, they have

$$0 \le \frac{\sum_{j=1}^{n} p_j f(x_j)}{\sum_{j=1}^{n} p_j} - f\left(\frac{\sum_{j=1}^{n} p_j x_j}{\sum_{j=1}^{n} p_j}\right)$$

$$\leq \frac{\sum_{j=1}^{n} p_{j} x_{j} f'(x_{j})}{\sum_{j=1}^{n} p_{j}} - \frac{\sum_{j=1}^{n} p_{j} x_{j}}{\sum_{j=1}^{n} p_{j}} \cdot \frac{\sum_{j=1}^{n} p_{j} f'(x_{j})}{\sum_{j=1}^{n} w p_{j}}, \tag{4.112}$$

provided that $f: I \to \mathbb{R}$ is a differentiable convex function on the interval $I \subset \mathbb{R}$, $x_j \in I$, $p_j \geq 0$ for $j \in \{1, 2, ..., n\}$ such that $\sum_{j=1}^n p_j > 0$. If f is strictly convex on I (i.e., the interior of I), then the equality holds in (4.112) if and only if $x_1 = x_2 = \cdots = x_n$.

For other refinements, generalizations and applications of the Jensen's type inequalities, see for instance ([117, p. 190-213], [126], [129], [130], [132], [141]) and the references which are cited therein.

Utilising the celebrated Jensen's discrete inequality (2.95) and its reverse (4.112) for particular real convex functions, we establish in this section, some new and interesting inequalities for functions defined by the real power series (2.65) with real coefficients and convergent on the interval (-R, R), R > 0. Applications for some fundamental functions of interest are also presented.

4.2.2 Power Series Inequalities Via Convexity

First, we state the following result that has been obtained on utilising the convexity properties of certain underlying functions (see [213]).

Theorem 88 (Ibrahim, Dragomir, Cerone and Darus [213]) Assume that $f(x) = \sum_{n=0}^{\infty} p_n x^n$ is a function with nonnegative coefficients defined on (-R, R), R > 0. If a, b, c > 0 are such that $ac, bc \in (0, R)$, then

$$(\ln a - \ln b) acf'(ac) \ge f(ac) - f(bc)$$

$$\ge (\ln a - \ln b) bcf'(bc). \tag{4.113}$$

Proof. It is well known that if $f: I \subset \mathbb{R} \to \mathbb{R}$ is a differentiable convex function on \mathring{I} , then for any $x, y \in \mathring{I}$ we have

$$f'(y)(y-x) \ge f(y) - f(x)$$
. (4.114)

Now, if we apply the property (4.114) for the function $f(t) = -\ln t$, t > 0, then we get

$$\frac{x}{y} - 1 \ge \ln x - \ln y,\tag{4.115}$$

for any x, y > 0. If in (4.115) we choose $x = a^n, y = b^n, n \ge 0$, then we obtain

$$a^n - b^n \ge nb^n \ln a - nb^n \ln b, \tag{4.116}$$

for any a, b > 0, $n \ge 0$. If we multiply (4.116) by $p_n c^n \ge 0$, $n \ge 0$ and sum over n from 0 to k, then we derive

$$\sum_{n=0}^{k} p_n c^n a^n - \sum_{n=0}^{k} p_n b^n c^n \ge \ln a \sum_{n=0}^{k} n p_n b^n c^n - \ln b \sum_{n=0}^{k} n p_n b^n c^n.$$
 (4.117)

Since

$$\sum_{n=0}^{\infty} p_n c^n a^n = f(ac), \qquad \sum_{n=0}^{\infty} p_n c^n b^n = f(cb)$$
 (4.118)

and

$$\sum_{n=0}^{\infty} n p_n b^n c^n = bc \sum_{n=1}^{\infty} n p_n b^{n-1} c^{n-1} = bc f'(bc), \qquad (4.119)$$

then by letting $k \to \infty$ in (4.117), we deduce

$$f(ac) - f(bc) \ge (\ln a - \ln b) bc f'(bc)$$
. (4.120)

Now, replacing a with b in (4.120), we have

$$f(bc) - f(ac) > (\ln b - \ln a) acf'(ac),$$
 (4.121)

which is equivalent with

$$f(ac) - f(bc) \le (\ln a - \ln b) acf'(ac). \tag{4.122}$$

Thus, from the inequalities (4.120) and (4.122), we derive the desired result (4.113).

Corollary 89 With the assumptions of Theorem 88 and if $a, b \in (0, R)$, then

$$(\ln a - \ln b) \, a f'(a) > f(a) - f(b) > (\ln a - \ln b) \, b f'(b) \,. \tag{4.123}$$

Corollary 90 With the assumptions of Theorem 88 and if a, c > 0 and $a, ac \in (0, R)$, then

$$ac \ln a \ f'(ac) \ge f(ac) - f(c) \ge c \ln a \ f'(c)$$
. (4.124)

The above results (4.123) and (4.124) have some natural applications for particular real functions of interest as follows:

(1) If we apply the inequality (4.123) for the function f(x) = 1/(1-x), $x \in (-1,1)$, then for any $a,b \in (0,1)$ we get

$$(\ln a - \ln b) \cdot \frac{a}{(1-a)^2} \ge \frac{1}{1-a} - \frac{1}{1-b}$$

$$\ge (\ln a - \ln b) \cdot \frac{b}{(1-b)^2}.$$
(4.125)

Hence

$$\frac{(1-b)(a-b)}{b(1-a)} \ge \ln a - \ln b \ge \frac{(1-a)(a-b)}{a(1-b)},\tag{4.126}$$

for any $a, b \in (0, 1)$.

(2) If we apply the inequality (4.123) for the function $f(x) = x/(1-x)^2$, $x \in (-1,1)$, then for any $a,b \in (0,1)$ we obtain

$$(\ln a - \ln b) \cdot \frac{a(1+a)}{(1-a)^3} \ge \frac{a}{(1-a)^2} - \frac{b}{(1-b)^2}$$

$$\ge (\ln a - \ln b) \cdot \frac{b(1+b)}{(1-b)^3}, \qquad (4.127)$$

which implies that

$$\frac{(1-b)(1-ab)(a-b)}{b(1+b)(1-a)^2} \ge \ln a - \ln b$$

$$\ge \frac{(1-a)(1-ab)(a-b)}{a(1+a)(1-b)^2}, (4.128)$$

for any $a, b \in (0, 1)$.

(3) If in (4.123), we choose the logarithm function $f(x) = -\ln(1-x)$, $x \in (-1,1)$, then for any $a,b \in (0,1)$ we get the inequality

$$\ln\left(\frac{a}{b}\right)^{a/(1-a)} \ge \ln\left(\frac{1-b}{1-a}\right) \ge \ln\left(\frac{a}{b}\right)^{b/(1-b)}.\tag{4.129}$$

Hence

$$\left(\frac{a}{b}\right)^{a/(1-a)} \ge \frac{1-b}{1-a} \ge \left(\frac{a}{b}\right)^{b/(1-b)},$$
 (4.130)

for any $a, b \in (0, 1)$. In particular, if in (4.130) we choose b = 1 - a, then we obtain the following result:

$$\left(\frac{a}{1-a}\right)^{a/(1-a)} \ge \frac{a}{1-a} \ge \left(\frac{a}{1-a}\right)^{(1-a)/a},$$
 (4.131)

for any $a \in (0,1)$.

(4) If in (4.123) we choose the function $f(x) = \frac{1}{2} \ln \left[(1+x) / (1-x) \right], x \in (-1,1)$, then for any $a,b \in (0,1)$ we get the following inequality:

$$(\ln a - \ln b) \cdot \frac{a}{1 - a^2} \ge \frac{1}{2} \left[\ln \left(\frac{1 + a}{1 - a} \right) - \ln \left(\frac{1 + b}{1 - b} \right) \right]$$

$$\ge (\ln a - \ln b) \frac{b}{1 - b^2},$$

$$(4.132)$$

which is equivalent to

$$\ln\left(\frac{a}{b}\right)^{a/(1-a^2)} \ge \ln\left(\frac{(1+a)(1-b)}{(1-a)(1+b)}\right)^{1/2} \ge \ln\left(\frac{a}{b}\right)^{b/(1-b)}.$$
 (4.133)

Hence

$$\left(\frac{a}{b}\right)^{2a/(1-a^2)} \ge \frac{(1+a)(1-b)}{(1-a)(1+b)} \ge \left(\frac{a}{b}\right)^{2b/(1-b^2)},$$
(4.134)

for any $a, b \in (0, 1)$.

(5) If we apply the same inequality (4.123) for the hyperbolic function $f(x) = \cosh(x)$, $x \in \mathbb{R}$, then we obtain that

$$\ln\left(\frac{a}{b}\right)^{a\sinh(a)} \ge \cosh\left(a\right) - \cosh\left(b\right) \ge \ln\left(\frac{a}{b}\right)^{b\sinh(b)},\tag{4.135}$$

for any $a, b \in (0, R), R > 0$.

(6) Further, if we apply the inequality (4.124) for the function $f(x) = \cosh(x)$, $x \in \mathbb{R}$, then for any $a, b \in \mathbb{R}$ we have

$$\ln a^{ab\sinh(ab)} \ge \cosh(ab) - \cosh(b) \ge \ln a^{b\sinh(b)}. \tag{4.136}$$

In particular, if in (4.136) we choose b = 1, then we obtain that

$$\ln a^{2ea\sinh(a)} > 2e\cosh(a) - e^2 - 1 > \ln a^{e^2 - 1},\tag{4.137}$$

for any $a \in \mathbb{R}$.

Now, we prove the following reverse of the Jensen type inequality that has been obtained by utilising the inequality (4.123) for a differentiable convex function (see [213]).

Theorem 91 (Ibrahim, Dragomir, Cerone and Darus [213]) Assume that f is as in Theorem 88 and R=1 or $R=+\infty$. If $a_i \in (0,R)$ and $p_k \geq 0$, $k \in \{1,\ldots,n\}$ with $\sum_{k=1}^{n} p_k = 1$, then we have the inequalities

$$\ln\left[\frac{\prod_{k=1}^{n} a_k^{p_k a_k f'(a_k)}}{\left(\prod_{j=1}^{n} a_j^{p_j}\right)^{\sum_{k=1}^{n} p_k a_k f'(a_k)}}\right] \ge \sum_{k=1}^{n} p_k f\left(a_k\right) - f\left(\prod_{j=1}^{n} a_j^{p_j}\right) \ge 0. \tag{4.138}$$

Proof. We use the inequality (4.123) for the choices $a = a_i$ and $b = \prod_{j=1}^n a_j^{p_j}$ to get

$$\left[\ln a_k - \ln \left(\prod_{j=1}^n a_j^{p_j}\right)\right] a_k f'(a_k)$$

$$\geq f(a_k) - f\left(\prod_{j=1}^n a_j^{p_j}\right)$$

$$\geq \left[\ln a_k - \ln \left(\prod_{j=1}^n a_j^{p_j}\right)\right] \left(\prod_{j=1}^n a_j^{p_j}\right) f'\left(\prod_{j=1}^n a_j^{p_j}\right), \tag{4.139}$$

for any $k \in \{1, ..., n\}$. Now, if we multiply (4.139) by $p_k \ge 0$ and sum over k from 1 to n, then we get

$$\ln\left(\prod_{k=1}^{n} a_k^{p_k a_k f'(a_k)}\right) - \sum_{k=1}^{n} p_k a_k f'\left(a_k\right) \cdot \ln\left(\prod_{j=1}^{n} a_j^{p_j}\right)$$

$$\geq \sum_{k=1}^{n} p_k f\left(a_k\right) - f\left(\prod_{j=1}^{n} a_j^{p_j}\right) \geq 0,$$

$$(4.140)$$

which is clearly equivalent with the desired result (4.138).

Remark 92 The second inequality in (4.138) shows that

$$\lambda f(x) + (1 - \lambda) f(y) \ge f(x^{\lambda} y^{1 - \lambda}) \tag{4.141}$$

for any $\lambda \in [0, 1]$ and $x, y \in (0, R)$ $(R = 1 \text{ or } R = \infty)$, i.e., f is an GA-convex function in the sense of terminology introduced by Anderson et al. in [22]. For other similar results, see [23], [47], [79, p.], [304], [333].

The following result is also established, which provides a refinement as well as a reverse of the Jensen's type for the power series.

Theorem 93 (Ibrahim, Dragomir, Cerone and Darus [213]) Assume that f, a_k and p_k are as in Theorem 91. Then, we have the inequalities:

$$\ln\left[\left(\prod_{k=1}^{n} a_{k}^{p_{k} a_{k} f'(a_{k})}\right) \cdot \left(\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}\right)^{\sum_{k=1}^{n} p_{k} a_{k} f'(a_{k})}\right]$$

$$\geq \sum_{k=1}^{n} p_{k} f\left(a_{k}\right) - f\left[\left(\sum_{k=1}^{n} \frac{p_{k}}{a_{k}}\right)^{-1}\right]$$

$$\geq \ln\left[\left(\prod_{k=1}^{n} a_{k}^{p_{k}}\right) \cdot \left(\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}\right)\right]^{f'\left[\left(\sum_{k=1}^{n} \frac{p_{k}}{a_{k}}\right)^{-1}\right]\left(\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}\right)^{-1}}$$

$$\geq 0. \tag{4.142}$$

Proof. From the inequality (4.123), we have for the choices $a = a_k$, $b = \left(\sum_{j=1}^n \frac{p_j}{a_j}\right)^{-1}$ that

$$\left[\ln a_k - \ln \left(\frac{1}{\sum_{j=1}^n \frac{p_j}{a_j}}\right)\right] a_k f'(a_k)$$

$$\geq f(a_k) - f\left(\frac{1}{\sum_{j=1}^n \frac{p_j}{a_j}}\right)$$

$$\geq \left[\ln a_k - \ln \left(\frac{1}{\sum_{j=1}^n \frac{p_j}{a_j}}\right)\right] \frac{1}{\sum_{j=1}^n \frac{p_j}{a_j}} f'\left(\frac{1}{\sum_{j=1}^n \frac{p_j}{a_j}}\right), \tag{4.143}$$

for any $k \in \{1, ..., n\}$. Now, if we multiply (4.143) by $p_k \ge 0$ and sum over k from 1 to n, then we deduce

$$\ln \left[\frac{\prod_{k=1}^{n} a_{k}^{p_{k} a_{k} f'(a_{k})}}{\left(\prod_{j=1}^{n} a_{j}^{p_{j}}\right)^{-\sum_{k=1}^{n} p_{k} a_{k} f'(a_{k})}} \right] \\
\geq \sum_{k=1}^{n} p_{k} f(a_{k}) - f\left(\frac{1}{\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}}\right) \\
\geq \ln \left[\frac{\prod_{k=1}^{n} a_{k}^{p_{k}}}{\left(\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}\right)^{-1}} \right]^{f'\left(1/\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}\right)\left(\sum_{j=1}^{n} \frac{p_{j}}{a_{j}}\right)^{-1}} \geq 0 \tag{4.144}$$

giving the desired result (4.142).

Finally, we establish the following result.

Theorem 94 (Ibrahim, Dragomir, Cerone and Darus [213]) With the assumptions in Theorem 88, we have the inequality:

$$f\left(\left(\prod_{k=1}^{n} a_{k}^{p_{k}a_{k}f'(a_{k})}\right)^{\frac{1}{\sum_{k=1}^{n} p_{k}a_{k}f'(a_{k})}}\right) \geq \sum_{k=1}^{n} p_{k}f\left(a_{k}\right). \tag{4.145}$$

Proof. From the inequality (4.123) we have

$$f(a) - f(a_k) \ge (\ln a - \ln a_k) a_k f'(a_k),$$
 (4.146)

for any $k \in \{1, ..., n\}$. Now, if we multiply (4.146) by $p_k \ge 0$ and sum over k from 1 to n, then we get

$$f(a) - \sum_{k=1}^{n} p_k f(a_k) \ge \ln a \cdot \sum_{k=1}^{n} p_k a_k f'(a_k) - \sum_{k=1}^{n} p_k a_k f'(a_k) \ln a_k.$$
 (4.147)

Now, if we choose a so that

$$\ln a = \frac{\sum_{k=1}^{n} p_k a_k f'(a_k) \ln a_k}{\sum_{k=1}^{n} p_k a_k f'(a_k)} = \ln \left(\prod_{k=1}^{n} a_k^{p_k a_k f'(a_k)} \right)^{\frac{1}{\sum_{k=1}^{n} p_k a_k f'(a_k)}}, \quad (4.148)$$

then we get the desired result (4.145).

4.2.3 Power Series Inequalities Via Jensen Type

In this section, we employ the inequality (2.95) and also the reverse of the Jensen inequality (4.112) due to Dragomir and Ionescu [139], to derive some new and interesting inequalities for functions defined by the real power series with positive coefficients.

First, we get the following result by utilising Jensen's inequality (2.95) for a particular convex function.

Theorem 95 (Ibrahim, Dragomir, Cerone and Darus [213]) Assume that the function $f(x) = \sum_{n=0}^{\infty} p_n x^n$ is defined on the interval (-R, R), R > 0 and $p_0 > 0$, $p_n \ge 0$ for $n \ge 1$. If a, b > 0 are such that $a, ab \in (0, R)$, then

$$b^{af'(a)/f(a)} \le \frac{f(ab)}{f(a)} \le b^{abf'(ab)/f(ab)}.$$
 (4.149)

Proof. If we apply Jensen's inequality (2.95) for the convex function $g(t) = -\ln t$, t > 0, then we have

$$\ln\left(\frac{\sum_{n=0}^{k} p_n a^n b^n}{\sum_{n=0}^{k} p_n a^n}\right)$$

$$\geq \frac{\sum_{n=0}^{k} p_n a^n \ln b^n}{\sum_{n=0}^{k} p_n a^n} = \frac{\ln b \sum_{n=0}^{k} n p_n a^n}{\sum_{n=0}^{k} p_n a^n} = \frac{a \ln b \sum_{n=1}^{k} n p_n a^{n-1}}{\sum_{n=0}^{k} p_n a^n}, \tag{4.150}$$

for any a, b > 0 with $a, ab \in (0, R)$. Taking the limit over $k \to \infty$ and since

$$\sum_{n=0}^{\infty} p_n a^n b^n = f(ab), \quad \sum_{n=1}^{\infty} n p_n a^{n-1} = f'(a), \quad \sum_{n=0}^{\infty} p_n a^n = f(a), \quad (4.151)$$

then from (4.150) we have

$$\ln\left(\frac{f(ab)}{f(a)}\right) \ge \frac{a\ln bf'(a)}{f(a)},\tag{4.152}$$

for a, b > 0 with $a, ab \in (0, R)$, which is equivalent with the first inequality in (4.149).

Now, if we apply Jensen's inequality (2.95) for the convex function $g(t) = t \ln t$, t > 0, then we can state that

$$\left(\frac{\sum_{n=0}^{k} p_n a^n b^n}{\sum_{n=0}^{k} p_n a^n}\right) \cdot \ln \left(\frac{\sum_{n=0}^{k} p_n a^n b^n}{\sum_{n=0}^{k} p_n a^n}\right)
\leq \frac{\sum_{n=0}^{k} p_n a^n b^n \ln b^n}{\sum_{n=0}^{k} p_n a^n} = \frac{\ln b \sum_{n=0}^{k} n p_n a^n b^n}{\sum_{n=0}^{k} p_n a^n}
= \frac{ab \ln b \sum_{n=1}^{k} n p_n a^{n-1} b^{n-1}}{\sum_{n=0}^{k} p_n a^n},$$
(4.153)

for a, b > 0 with $a, ab \in (0, R)$. Taking the limit over $k \to \infty$ and since

$$\sum_{n=1}^{\infty} n p_n a^{n-1} b^{n-1} = f'(ab), \qquad (4.154)$$

then from (4.153) we have

$$\frac{f(ab)}{f(a)}\ln\left(\frac{f(ab)}{f(a)}\right) \le \frac{ab\ln bf'(ab)}{f(a)},\tag{4.155}$$

which is clearly equivalent with the second inequality in (4.149).

Corollary 96 With the assumptions in Theorem 95, and if $a, c \in (0, R)$, then

$$\left(\frac{c}{a}\right)^{af'(a)/f(a)} \le \frac{f(c)}{f(a)} \le \left(\frac{c}{a}\right)^{cf'(c)/f(c)}.$$
(4.156)

Proof. It follows from Theorem 95 on choosing c = ab.

Some applications of the inequality (4.156) for particular functions of interest are given in the following:

(1) If we apply the inequality (4.156) for the function $f(x) = -\ln(1-x)$, $x \in (-1,1)$, then we get

$$\left(\frac{c}{a}\right)^{-a/(1-a)\ln(1-a)} \le \frac{\ln(1-c)}{\ln(1-a)} \le \left(\frac{c}{a}\right)^{-c/(1-c)\ln(1-c)},\tag{4.157}$$

for any $a, c \in (0, 1)$.

(2) If in (4.156) we choose the function $f(x) = \exp(x)$, $x \in \mathbb{R}$, then we can state that

$$\left(\frac{c}{a}\right)^a \le \exp\left(c - a\right) \le \left(\frac{c}{a}\right)^c,$$
 (4.158)

for any $a, c \in (0, R)$.

(3) If we apply the same inequality (4.156) for the function $f(x) = \sinh(x)$, $x \in \mathbb{R}$, then we obtain

$$\left(\frac{c}{a}\right)^{a \coth(a)} \le \frac{\sinh(c)}{\sinh(a)} \le \left(\frac{c}{a}\right)^{c \coth(c)} \tag{4.159}$$

for any $a, c \in (0, R)$.

Next, we utilize the reverse of the Jensen type inequality (4.112) for a particular convex function to obtain the following result.

Theorem 97 (Ibrahim, Dragomir, Cerone and Darus [213]) Let f be as in Theorem 95. If a, b > 0 are such that $a, ab, ab \in (0, R)$, then

$$\frac{f\left(ab\right)}{f\left(a\right)} \le b^{af'(a)/f(a)} \cdot \exp\left[\frac{f\left(ab\right)f\left(\frac{a}{b}\right)}{f^{2}\left(a\right)} - 1\right]. \tag{4.160}$$

Proof. Utilizing the inequality (4.112), for the convex function $g(t) = -\ln t$, t > 0, we can write that

$$\ln\left(\frac{\sum_{n=0}^{k} p_n a^n b^n}{\sum_{n=0}^{k} p_n a^n}\right) - \frac{\sum_{n=0}^{k} p_n a^n \ln b^n}{\sum_{n=0}^{k} p_n a^n}$$

$$\leq \frac{\sum_{n=0}^{k} p_n a^n b^n}{\sum_{n=0}^{k} p_n a^n} \cdot \frac{\sum_{n=0}^{k} \frac{p_n a^n}{b^n}}{\sum_{n=0}^{k} p_n a^n} - 1,$$
(4.161)

for a, b > 0 with $a, ab, ...ab \in (0, R)$. Taking the limit over $k \to \infty$ and since

$$\sum_{n=0}^{\infty} p_n \left(\frac{a}{b}\right)^n = f\left(\frac{a}{b}\right),\tag{4.162}$$

then by (4.161) we deduce

$$\ln\left(\frac{f\left(ab\right)}{f\left(a\right)}\right) - \frac{a\ln bf'\left(a\right)}{f\left(a\right)} \le \frac{f\left(ab\right)f\left(\frac{a}{b}\right)}{f^{2}\left(a\right)} - 1,\tag{4.163}$$

which is clearly equivalent with the desired result (4.160).

The particular case of Theorem 97 can be stated as well.

Corollary 98 With the assumptions in Theorem 97 and if $a, c \in (0, R)$ with $a^2/c \in (0, R)$, then we have the inequality

$$\frac{f(c)}{f(a)} \le \left(\frac{c}{a}\right)^{af'(a)/f(a)} \exp\left[\frac{f(c)f\left(\frac{a^2}{c}\right)}{f^2(a)} - 1\right]. \tag{4.164}$$

If $c^2/a \in (0,R)$, then we also have

$$\left(\frac{c}{a}\right)^{cf'(c)/f(c)} \exp\left[1 - \frac{f(a)f\left(\frac{c^2}{a}\right)}{f^2(c)}\right] \le \frac{f(c)}{f(a)}.$$
(4.165)

Some examples for particular inequalities that are generated by the real power series with positive coefficients are as follows.

(1) If we apply the inequality (4.164) for the function f(x) = 1/(1-x), $x \in (-1,1)$, then for any $a, c \in (0,1)$ we get

$$\frac{1-a}{1-c} \le \left(\frac{c}{a}\right)^{a/(1-a)} \exp\left[\frac{c(1-a)^2}{(1-c)(c-a^2)} - 1\right]. \tag{4.166}$$

(2) If in (4.164) we choose the function $f(x) = \exp(x)$, $x \in \mathbb{R}$, then for any $a, c \in (0, R)$ we can state that

$$\exp\left(c - a\right) \le \left(\frac{c}{a}\right)^a \exp\left[\exp\left(\frac{\left(c - a\right)^2}{c}\right) - 1\right]. \tag{4.167}$$

The following result is also established.

Theorem 99 (Ibrahim, Dragomir, Cerone and Darus [213]) With the assumptions of Theorem 4.149 for the function f and if $x_k \in (0, R)$ and $p_k \geq 0$ with $\sum_{k=1}^{n} p_k = 1$, then we have the inequality

$$\sum_{k=1}^{n} p_{k} \left(\frac{x_{k}}{\sum_{j=1}^{n} p_{j} x_{j}} \right)^{\frac{\sum_{j=1}^{n} p_{j} x_{j} f'\left(\sum_{j=1}^{n} p_{j} x_{j}\right)}{f\left(\sum_{j=1}^{n} p_{j} x_{j}\right)}}$$

$$\leq \frac{\sum_{k=1}^{n} p_{k} f\left(x_{k}\right)}{f\left(\sum_{j=1}^{n} p_{j} x_{j}\right)} \leq \sum_{k=1}^{n} p_{k} \left(\frac{x_{k}}{\sum_{j=1}^{n} p_{j} x_{j}}\right)^{\frac{x_{k} f'\left(x_{k}\right)}{f\left(x_{k}\right)}}.$$
(4.168)

Proof. It is obvious by the inequality (4.156) on noticing that for $c = x_k$ and $a = \sum_{j=1}^{n} p_j x_j$. Thus, we have

$$\left(\frac{x_k}{\sum_{j=1}^n p_j x_j}\right)^{\frac{\sum_{j=1}^n p_j x_j f'\left(\sum_{j=1}^n p_j x_j\right)}{f\left(\sum_{j=1}^n p_j x_j\right)}}$$

$$\leq \frac{f\left(x_k\right)}{f\left(\sum_{j=1}^n p_j x_j\right)} \leq \left(\frac{x_k}{\sum_{j=1}^n p_j x_j}\right)^{\frac{x_k f'\left(x_k\right)}{f\left(x_k\right)}}, \tag{4.169}$$

for all $k \in \{1, ..., n\}$. On multiplying the inequality (4.169) by $p_k \geq 0$ and summing over k from 1 to n, we get the desired result.

A more natural result of the Jensen type inequality is incorporated in the following theorem.

Theorem 100 (Ibrahim, Dragomir, Cerone and Darus [213]) With the assumptions of Theorem 4.168, we have

$$1 \le \frac{G_p(f(x))}{f(G_p(x))} \le \frac{G_p\left(x^{\frac{xf'(x)}{f(x)}}\right)}{\left[G_p(x)\right]A_p\left(\frac{xf'(x)}{f(x)}\right)},\tag{4.170}$$

where

$$A_{p}\left(h\left(x\right)\right) := \sum p_{j}h\left(x_{j}\right) \tag{4.171}$$

and

$$G_p(h(x)) := \prod_{j=1}^n h^{p_j}(x_j), h: (0, \infty) \to (0, \infty).$$
 (4.172)

Proof. If we choose in (4.156) $c = x_j$, $a = G_p(x) = \prod_{j=1}^n x_j^{p_j}$, then we have

$$\left(\frac{x_j}{G_p(x)}\right)^{\frac{G_p(x)f'\left(G_p(x)\right)}{f\left(G_p(x)\right)}} \le \frac{f\left(x_j\right)}{f\left(G_p\left(x\right)\right)} \le \left(\frac{x_j}{G_p(x)}\right)^{\frac{x_jf'\left(x_j\right)}{f\left(x_j\right)}},$$
(4.173)

which by taking the power p_j gives us that

$$\left(\frac{x_j}{G_p(x)}\right)^{p_j A} \le \left[\frac{f(x_j)}{f(G_p(x))}\right]^{p_j} \le \left(\frac{x_j}{G_p(x)}\right)^{\frac{p_j x_j f'(x_j)}{f(x_j)}},$$
(4.174)

with

$$A := \frac{G_p(x) f'(G_p(x))}{f(G_p(x))}.$$
(4.175)

Now, multiplying the inequality (4.174) over j from 1 to n, we get

$$1 \leq \frac{\prod_{j=1}^{n} f^{p_{j}}(x_{j})}{f(G_{p}(x))} \leq \frac{\prod_{j=1}^{n} x_{j}^{\frac{p_{j}x_{j}f'(x_{j})}{f(x_{j})}}}{\left[G_{p}(x)\right]^{\sum_{j=1}^{n} \frac{p_{j}x_{j}f'(x_{j})}{f(x_{j})}}},$$
(4.176)

and the proof is completed. \blacksquare

For some other results related to the Jensen's type inequalities and their applications, see for instance [27], [57], [129], [130], [131], [132], [133], [134], [139], [273], [303], [351], [421] and the references which are cited therein.

Chapter 5

Applications to Special Functions

Special functions are another important type of functions that are defined by the power series. They arise in almost all areas of Modern Mathematics, with physics, engineering, chemistry, statistics and computer science as the most well-known application areas of these functions. In this chapter, we provide some inequalities involving the special functions, such as polylogarithm, hypergeometric, Bessel and modified Bessel functions of the first kind, that have been obtained on utilising the Buzano, Schwarz, Young, Hölder and Jensen type inequalities.

Section 5.1 briefly introduces some general facts and basic concepts of the special functions. We address some of their important relations to the fundamental functions. Some inequalities involving the polylogarithm, hypergeometric, Bessel and modified Bessel functions of the first kind are presented subsequently in Section 5.2, Section 5.3 and Section 5.4.

The author's research papers in collaboration with Dragomir, Darus and Cerone (see [208], [209], [213], [214], [215], [216]) are the main contributions to this chapter. The survey research paper published by the author and Dragomir [217], is also incorporated with some of the results.

5.1 Definitions and Basic Concepts

Before we state our results for the polylogarithm, hypergeometric, Bessel and modified Bessel functions, we recall in this section, the definitions and some basic concepts of these functions. The details about these types of special functions can be found in the work of (see for instance [4], [25], [102], [296], [342, Chapt. 60]), and numerous classical and modern texts devoted to the special functions.

We start by introducing the classical gamma, zeta and their related functions. The gamma function is useful in defining some other types of special functions.

5.1.1 Gamma, Zeta and Related Functions

The gamma function, denoted by $\Gamma(z)$, is a function defined by the Euler's integral [102, p. 221], i.e.,

$$\Gamma(z) = \int_0^\infty t^{z-1} e^{-t} dt, \qquad (5.1)$$

for any $z \in \mathbb{C}$ such that Re(z) > 0. It is clear that from (5.1) we have

$$\Gamma\left(1\right) = 1,\tag{5.2}$$

and it also satisfies the following recurrent formulas:

(i)
$$\Gamma(1+z) = z\Gamma(z)$$
, (5.3)

(ii)
$$\Gamma(1-z) = -z\Gamma(-z),$$
 (5.4)

for all $z \in \mathbb{C}$ with Re (z) > 1. The gamma function (5.1) reduces to the factorial function when its argument is a positive real integer, i.e.,

$$\Gamma(n) = (n-1)! \tag{5.5}$$

for $n \in \mathbb{N}$. For positive half-integer arguments, the gamma function has the special form as follows:

$$\Gamma\left(n+\frac{1}{2}\right) = \frac{(2n-1)!!\sqrt{\pi}}{2^n},\tag{5.6}$$

where n!! denotes the double factorial function¹ of integer $n, n \ge -1$. It can be seen that the relations in (5.4), (5.5) and (5.6), define the gamma functions in whole real numbers.

¹For integers $k \ge -1$, the double factorial k!! are defined by (-1)!! = 0!! = 1, $(2k)!! = \prod_{j=1}^{k} 2j$, $(2k+1)!! = \prod_{j=0}^{k} (2j+1)$ and it is undefined for $k \le -2$.

Some common values of the gamma function are given in the following [4, Chapt. 6]:

$$\Gamma\left(-\frac{3}{2}\right) = \frac{4\sqrt{\pi}}{3}, \qquad \Gamma\left(-\frac{1}{2}\right) = -2\sqrt{\pi},$$

$$\Gamma\left(\frac{1}{2}\right) = \sqrt{\pi}, \qquad \Gamma\left(\frac{3}{2}\right) = \frac{\sqrt{\pi}}{2},$$

$$\Gamma(2) = 1, \qquad \Gamma\left(\frac{5}{2}\right) = \frac{3\sqrt{\pi}}{4},$$

$$\Gamma(3) = 2, \qquad \Gamma(4) = 6.$$

$$(5.7)$$

The derivative of the gamma function is

$$\Gamma'(z) = \int_0^\infty e^{-t} t^{z-1} \log(t) dt, \qquad (5.8)$$

and the n-th derivative is given by

$$\Gamma^{(n)}(z) = \int_0^\infty e^{-t} t^{z-1} \log^n(t) dt.$$
 (5.9)

These can be derived by differentiating the integral form of the gamma function (5.1) with respect to z, and using the technique of differentiation under the integral sign. Closely related to the derivative of the gamma function is the digamma or psi function $\Psi(z)$, which is defined by

$$\Psi(z) = \frac{d}{dz} \left[\ln \Gamma(z) \right] = \frac{\Gamma'(z)}{\Gamma(z)}, \ z \neq 0, 1, 2, \dots$$
 (5.10)

Further differentiations of (5.8) lead to the family of the polygamma functions $\Psi_n(z)$,

$$\Psi_n(z) = \Gamma^{(n+1)} \left[\log \Gamma(z) \right], \tag{5.11}$$

where $\Psi_0(z) = \Psi(z)$. The incomplete gamma function $\gamma(\alpha, z)$ and the complementary incomplete gamma function $\Gamma(\alpha, z)$, which are defined by

$$\gamma(\alpha, z) = \int_0^z t^{\alpha - 1} e^{-t} dt, \quad \text{Re}(\alpha) > 0$$
 (5.12)

and

$$\Gamma(\alpha, z) = \int_{z}^{\infty} t^{\alpha - 1} e^{-t} dt, \quad \text{Re}(\alpha) > 0$$
 (5.13)

respectively, have a relation to the gamma function as follows:

$$\Gamma(\alpha) = \gamma(\alpha, z) + \Gamma(\alpha, z). \tag{5.14}$$

Although, the gamma function does not have a power series representation, the incomplete gamma functions can be represented by the following power series:

$$\gamma(\alpha, z) = z^{\alpha} \sum_{n=0}^{\infty} \frac{(-1)^n}{(n+\alpha)} \frac{z^n}{n!}.$$
 (5.15)

Hence,

$$\Gamma(\alpha, z) = \Gamma(\alpha) - z^{\alpha} \sum_{n=0}^{\infty} \frac{(-1)^n}{(n+\alpha)} \frac{z^n}{n!},$$
(5.16)

for all $\alpha, z \in \mathbb{C}$ with $\operatorname{Re}(\alpha) > 0$.

Another important function related to the gamma function is the *beta function* B(x, y), where

$$B(x,y) = \frac{\Gamma(x)\Gamma(y)}{\Gamma(x+y)},$$
(5.17)

for all $x, y \in \mathbb{R}$ such that x, y > 0. The beta function is a real function of two variables, and it is defined by the formula

$$B(x,y) = \int_0^1 e^{x-1} (1-t)^{y-1} dt,$$
 (5.18)

for x, y > 0.

The gamma function (5.5) with the natural extension of the factorial of an integer was first introduced by Euler in 1930 (see [111]). It is of fundamental importance to many areas of science including Probability Theory, Mathematical Physics, Number Theory and Special Functions Theory. For more on its properties and some historical remarks, we may refer to [25, Chapt. 1], [34], [102, Chapt. 12], [167], [296, Chapt. 2], [395] and the references which are cited therein.

The Riemann zeta function or Euler-Riemann zeta function $\zeta(z)$, is a function defined by [187, p. 60]

$$\zeta(z) = \frac{1}{\Gamma(z)} \int_0^\infty \frac{t^{z-1}}{e^t - 1} dt, \tag{5.19}$$

for $z \in \mathbb{C}$ with Re(z) > 1, where $\Gamma(z)$ is the gamma function defined by (5.1). Another useful definition of the Riemann zeta function is given by the infinite series [4, p. 807], namely

$$\zeta(z) = \sum_{k=1}^{\infty} \frac{1}{k^z},\tag{5.20}$$

for any $z \in \mathbb{C}$ such that Re(z) > 1. For even and negative integer arguments, the Riemann zeta function can be expressed in terms of the Bernoulli numbers, that is,

$$\zeta(2k) = (-1)^k \frac{|B_{2k}| (2\pi)^{2k}}{2(2k)!}$$
 and $\zeta(-k) = -\frac{B_{k+1}}{k+1}$ (5.21)

for $k \in \mathbb{N}$.

Some common values of the Riemann zeta function are as follows:

$$\zeta(-3) = \frac{1}{120}, \qquad \zeta(-2) = 0, \qquad \zeta(-1) = -\frac{1}{12},
\zeta(0) = -\frac{1}{2}, \qquad \zeta(1) = \infty \qquad \zeta(2) = \frac{\pi^2}{6}
\zeta(3) = Z, \qquad \zeta(4) = \frac{\pi^4}{90}, \qquad \zeta(\infty) = 1,$$
(5.22)

where Z is known as the $Ap\acute{e}ry$'s $constant^2$ (cf. Apéry [30]).

The Hurwitz zeta function $\zeta(s,q)$ is the generalization of the Riemann zeta function, that is,

$$\zeta(s,q) = \sum_{k=0}^{\infty} \frac{1}{(k+q)^s},$$
 (5.23)

which coincides with the Riemann zeta function (5.20) when q=1 (see [4, p. 807], [149, p. 24-27]). Some useful functions related to the Riemann zeta function are the *Dirichlet beta* $\beta(s)$, *Dirichlet eta* $\eta(s)$ and lambda $\lambda(s)$ functions, where

²The Apéry's constant $Z:=\sum_{k=0}^{\infty}\frac{1}{k^3}=1.20205690\ldots$. It appears in a number of physical problems, for instance, in the electron's gyromagnetic ratio, computed using quantum electrodynamics, Debye model, Stefen-Boltzman law, etc.

$$\frac{\zeta\left(s\right)}{2^{s}} = \frac{\eta\left(s\right)}{2^{s} - 1} = \frac{\lambda\left(s\right)}{2^{s} - 2}, \qquad 2\lambda\left(s\right) = \zeta\left(s\right) + \eta\left(s\right), \tag{5.24}$$

and

$$\beta(s) = 2^{-2s} \left[\zeta\left(s, \frac{1}{4}\right) - \zeta\left(s, \frac{3}{4}\right) \right], \tag{5.25}$$

in which $\beta(s)$, $\eta(s)$ and $\lambda(s)$ are defined by the following series:

$$\beta(s) = \sum_{k=0}^{\infty} \frac{(-1)^k}{(2k+1)^s}, \ \eta(s) = \sum_{k=1}^{\infty} \frac{(-1)^{k-1}}{k^s}, \ \lambda(s) = \sum_{k=0}^{\infty} \frac{1}{(2k+1)^s},$$
 (5.26)

respectively, for Re(s) > 0. In the following, we give some special values of the $\beta(s)$, $\eta(s)$ and $\lambda(s)$ functions.

$$\beta(0) = \frac{1}{2}, \qquad \beta(1) = \frac{\pi}{4}, \qquad \beta(2) = G,$$

$$\beta(3) = \frac{\pi^3}{32}, \qquad \eta(-1) = \frac{1}{4}, \qquad \eta(0) = \frac{1}{2},$$

$$\eta(1) = \ln 2, \qquad \eta(2) = \frac{\pi^2}{12}, \qquad \eta(3) = \frac{3}{4}Z,$$

$$\eta(4) = \frac{7\pi^4}{720}, \qquad \lambda(2) = \frac{\pi^2}{8}, \qquad \lambda(4) = \frac{\pi^4}{96},$$
(5.27)

where Z and G, respectively represent the Apèry's constant and Catalan's constant³ (see [59, p. 153]).

Similarly, the Riemann zeta function is also one of the most important classical functions in analysis. It is widely used in various areas such as asymptotic series, definite integration, Number Theory, hypergeometric series, etc., and with applications even in physics, Probability Theory and Applied Statistic. Further reading on this subject can be found in the various textbooks devoted to the Theory of Special Functions, see for instance [4, p. 807-808], [146], [218], [224, Chapt. 9], [356] and the references which are cited therein.

 $^{{}^{3}}G := \sum_{n=0}^{\infty} \frac{(-1)^{n}}{(2n+1)^{2}} = 0.915965594...$

5.1.2 Polylogarithm Functions

The classical s-th polylogarithm function, or for short, the polylogarithms $Li_s(z)$ is a function defined by the power series (see [281, p. 239])

$$Li_s(z) = \sum_{k=1}^{\infty} \frac{z^k}{k^s}.$$
 (5.28)

This series (5.28) converges absolutely for all complex values of the order s and the argument z, where Re(s) > 0 and $z \in D(0,1)$. The polylogarithm function, which is also known in the literature as Jonqui'ere's function (cf. Jonquiere [225]), appears in many different branches of mathematics such as combinatorics, Algebraic Number Theory and Knot Theory (see [1], [71], [272]). It also plays an important role in physics and engineering including Quantum Field Theory, dynamical system, Feynman integrals, thermodynamic, network analysis, etc., (see [48], [154], [311], [443, p. 497-512]).

The alternative definition of the polylogarithm function is given by the following integral representation [364, p. 762]:

$$Li_s(z) = \frac{z}{\Gamma(s)} \int_0^\infty \frac{t^{s-1}}{e^t - z} dt, \qquad (5.29)$$

for all $s, z \in \mathbb{C}$ such that $\operatorname{Re}(s) > 0$ and |z| < 1, where $\Gamma(s)$ is the gamma function defined by (5.1).

The polylogarithm functions (5.28) reduce to the elementary functions for the orders s = 0 and 1, that is,

$$Li_0(z) = \frac{z}{1-z} \tag{5.30}$$

and

$$Li_1(z) = \ln\left(\frac{1}{1-z}\right),\tag{5.31}$$

respectively, for $z \in D(0,1)$, while for n=2, we have

$$Li_2(z) = \sum_{k=1}^{\infty} \frac{z^k}{k^2} = -\int_0^z \frac{\ln(1-t)}{t} dt, \quad |z| < 1,$$
 (5.32)

which is called the dilogarithm or Spence's function (cf. Spence [405, p. 24-36]). The function (5.32) was first introduced by Leibniz (cf. Leibniz [280, p. 336-339]) in 1696, and it has bee extensively studied in its own right, see for instance [71, p. 103-104], [142], [196], [202], [267], [281, Chapt. 1], [306] and [442]. For the cases of the integer orders $s = 3, 4, \ldots$, they are known, respectively as the trilogarithm, quadrilogarithm, and so on. However, for the higher of integer orders $s \geq 2$, the functions $Li_s(z)$ cannot be expressed in terms of the simple elementary functions.

The polylogarithm functions can also be defined through successive steps of their integration by

$$Li_{s}(z) = \int_{0}^{z} \frac{Li_{s-1}(t)}{t} dt,$$
 (5.33)

for s = 1, 2, 3..., with $Li_0(z)$ is the well-known function as defined in (5.30). Then, the differential relation follows from (5.33) that

$$\frac{d\left[Li_s\left(z\right)\right]}{ds} = \frac{Li_{s-1}\left(z\right)}{z} \tag{5.34}$$

or, in general,

$$\frac{z^{m}d^{(m)}\left[Li_{s}\left(z\right)\right]}{ds^{m}} = Li_{s-m}\left(z\right) \tag{5.35}$$

for m = 1, 2, ... and $d^{(m)}$ is denoted as the m-th of the derivative. Thus, by applying the recurrence formulas given by (5.34) and the equation (5.30) for s = ..., -3, -2, -1, 0, we have the following relations [277]:

$$Li_{-1}(z) = \frac{z}{(1-z)^2}, \qquad Li_{-2}(z) = \frac{z(z+1)}{(1-z)^3},$$

$$Li_{-3}(z) = \frac{z(1+4z+z^2)}{(1-z)^4}, \qquad Li_{-4}(z) = \frac{z(1+z)(1+10z+z^2)}{(1-z)^5}$$
(5.36)

and so on, for all $z \in \mathbb{C}$ such that |z| < 1. It is clearly seen that for all nonnegative integer order n, $Li_{-n}(z)$ is a rational function in z, whose the denominator is $(1-z)^{n+1}$. Hence, for any desired function of the lower orders $n=1,2,3,\ldots$ and |z|<1, we have the following expression [99, p. 245]:

$$Li_{-n}(z) = \frac{1}{(1-z)^{n+1}} \sum_{k=0}^{n} E_{n,k} z^{n-k},$$
 (5.37)

where the coefficients $E_{n,k}$, $n \in \mathbb{N}$, $k \in \mathbb{N}_0$ are Eulerian number⁴, namely, we recall that these numbers can be obtained by the following series (see [85], [99, p. 243]):

$$E_{n,k} = \sum_{j=0}^{k+1} (-1)^j \binom{n+1}{j} (k+1-j)^n$$
 (5.38)

for integers $n \geq 1$. Polylogarithms also arise in the sum of generalized harmonic numbers $H_{n,r}$ as

$$\sum_{n=1}^{\infty} H_{n,r} z^n = \frac{Li_r(z)}{1-z}, \quad |z| < 1$$
 (5.39)

for $z \in D(0,1)$, where

$$H_{n,r} := \sum_{k=1}^{\infty} \frac{1}{k^r}$$
 and $H_n := H_{n,1} = \sum_{k=1}^{\infty} \frac{1}{k}$. (5.40)

There are some special cases of polylogarithms, with first remarkable values (see [146, p. 11], [443, p. 497-512])

$$Li_s(1) = \zeta(s), \tag{5.41}$$

for Re(s) > 1 and

$$Li_s(-1) = (2^{1-s} - 1)\zeta(s),$$
 (5.42)

for $s \neq 1$, where $\zeta(s)$ is the Riemann zeta function defined by (5.20). Clearly, from the equation (5.41), the polylogarithm function (5.28) can be viewed as a generalization of the Riemann zeta function as well. Instead of (5.41), the polylogarithm also has relationships to other functions for some special cases of argument z. For instance:

$$Li_n(-1) = -\eta(n), (5.43)$$

$$Li_n(\pm i) = \frac{1}{2^s} \eta(n) \pm i\beta(n),$$
 (5.44)

where $\beta(n)$ and $\eta(n)$ are the Dirichlet beta and Dirichlet eta functions, which

⁴Some particular cases of Eulerian numbers are $E_{1,0}=E_{2,0}=E_{3,0}=E_{4,0}=1$, $E_{2,1}=1, E_{3,1}=4, E_{3,2}=1, E_{4,1}=11$, etc. Eulerian numbers can also be obtained by the recurrence equation: $E_{m,k}=(m+1-k)\,E_{m-1,k-1}+kE_{m-1,k}$.

were introduced in the previous section in (5.26).

There are some numerical values of $Li_2(z)$ and $Li_3(z)$ for the special arguments z, for instance [282, Chapt. 1, 13],

$$Li_{1}\left(\frac{1}{2}\right) = \ln 2, \qquad Li_{2}\left(-1\right) = -\frac{\pi^{2}}{12}, \qquad Li_{2}\left(0\right) = 0,$$

$$Li_{2}\left(\frac{1}{2}\right) = \frac{\pi^{2}}{12} - \frac{1}{2}\ln^{2}\left(2\right), \qquad Li_{2}\left(1\right) = \frac{\pi^{2}}{6}, \qquad Li_{2}(2) = \frac{\pi^{2}}{4},$$

$$Li_{2}\left(i\right) = -\frac{\pi^{2}}{48} + iG, \qquad Li_{2}\left(\frac{\sqrt{5}-1}{2}\right) = \frac{\pi^{2}}{10} - \frac{1}{2}\ln^{2}\left(\frac{\sqrt{5}+1}{2}\right),$$

$$Li_{3}\left(i\right) = -\frac{3}{48}Z + i\pi^{3}, \qquad Li_{3}\left(\frac{1}{2}\right) = \frac{1}{24}\left[4\ln^{3}\left(2\right) - 2\pi^{2}\ln\left(2\right) + 21Z\right],$$

where Z and G are the Apéry and Catalan constant, respectively. For similar equations containing the sum of dilogarithms of other arguments, see [249, p. 69-74], and see also [143], [179], [281, Chapt. 7], [363, p. 762-763] for some other formulae involving $Li_s(z)$.

Next, we shall introduce another important class of the special functions called hypergeometric functions.

5.1.3 Hypergeometric Functions

There are varieties of the hypergeometric-type functions ${}_{p}F_{q}(a_{1},\ldots,a_{p};b_{1},\ldots,b_{q};z)$, where p and q are nonnegative integers. However, the most important are the functions of the form ${}_{2}F_{1}\left(a,b;c;z\right)$ and ${}_{1}F_{1}\left(a;c;z\right)$, which are called the Gauss hypergeometric function and the confluent hypergeometric function of the first kind. In this section, we shall discuss some basic properties of these types of hypergeometric functions. For the details on this topic, we may refer to the work of Erdélyi et al. [149], and some other well-known textbooks such as Abramowitz and Stegun [4, p. 556-565], Andrews [26, p. 85-101] and Oldham et al. [342]. See also [25], [46], [186], [368] for some historical overview of hypergeometric functions.

First, we give the definition of the hypergeometric series with any positive integer of parameters p and q, which is called the *generalized hypergeometric* series ${}_{p}F_{q}(a_{1},\ldots,a_{p};b_{1},\ldots,b_{q};z)$. This series is defined by (see [368, p. 73])

$$_{p}F_{q}(a_{1},\ldots,a_{p};b_{1},\ldots,b_{q};z) = \sum_{n=0}^{\infty} \frac{(a_{1})_{n}\cdots(a_{p})_{n}}{(b_{1})_{n}\cdots(b_{q})_{n}} \frac{z^{n}}{n!}$$
 (5.45)

for arbitrary $z \in \mathbb{C}$, a_j , $b_j \in \mathbb{R} \setminus \mathbb{Z}_0^-$, $1 \leq i \leq p$, $1 \leq j \leq q$, and $(t)_n$, $n \in \mathbb{N}_0$ is the Pochhammer symbol, which is defined by (2.4). The subscripts p and q indicate the number of parameters in numerator and denominator of the coefficients in its power series expansion. The series (5.45) converges absolutely for all $z \in \mathbb{C}$ when p < q + 1, and it converges only for z = 0 when p > q + 1. If p = q + 1, then the series (5.45) converges absolutely for |z| < 1, and it diverges for |z| > 1, while, for z = 1, it converges absolutely when the Re $(\sum_{k=1}^q b_k - \sum_{k=1}^p a_k) > 0$. If the hypergeometric series converges, then it corresponds to the hypergeometric function.

The Gauss hypergeometric function or simply hypergeometric function, is the special case of the series (5.45) in which p=2 and q=1, that is,

$$_{2}F_{1}(a,b;c;z) = \sum_{n=0}^{\infty} \frac{(a)_{n}(b)_{n}}{(c)_{n}} \frac{z^{n}}{n!}$$
 (5.46)

for any $a,b,c\in\mathbb{R},\ z\in\mathbb{C}$ such that |z|<1 and $c\notin\{0,-1,-2,\ldots\},\ n\in\mathbb{N}$ (see [53], [162]). This series arises in many areas of mathematics, physics and engineering with several notations, for instance $F\left(a,b;c;z\right),\ F\left(\begin{smallmatrix}a,&b\\c\end{smallmatrix}\right|z\right),$ ${}_{2}F_{1}\left(\begin{smallmatrix}a,&b\\c\end{smallmatrix}\right|z\right)$ and $\frac{\Gamma(c)}{\Gamma(a)\Gamma(b)}\left[z\frac{a-1,b-1}{0,c-1}\right]$.

In general, the hypergeometric function defined by (5.46), is the solution of the Euler's hypergeometric differential equation, which is

$$z(1-z)w''(z) + [c - (a+b+1)z]w'(z) - abw = 0$$
 (5.47)

with the initial conditions w(0) = 1 and w'(0) = ab/c.

It is obvious that ${}_{2}F_{1}(a,b;c;z) = {}_{2}F_{1}(b,a;c;z)$. That is, the hypergeometric function (5.46) is symmetrical with respect to the interchange of its two parameter of the numerators a and b. The function (5.46) is also an analytic and

univalent in the unit disk D. For the fixed b, c and z, it is an entire function of a, while, if a, c and z are fixed, then it is an entire function of b. If a or b are zero or negative integers, then the series in (5.46) terminates after a finite number of its terms, thus the power series reduces to a polynomial in z. For example, if a = -m, $m \in \{0, 1, 2, \ldots\}$ and $c \notin \{\ldots, -2, -1, 0\}$, then the functions

$${}_{2}F_{1}\left(-m,b;c;z\right) = \sum_{n=0}^{m} \frac{(-m)_{n}(b)_{n}}{(c)_{n}} \frac{z^{n}}{n!} = \sum_{n=0}^{\infty} (-1) \binom{m}{n} \frac{(b)_{n}}{(c)_{n}} z^{n}$$

$$(5.48)$$

are the polynomials in z.

Some values of (5.46) for the special arguments z are given as follows:

$$_{2}F_{1}(a, b; c; 0) = 1,$$
 $_{2}F_{1}(a, b; c; -\infty) = 1,$ for a or $b = 0;$
 $_{2}F_{1}(a, b; c; -\infty) = 0,$ for $a < 0$ and $b < 0;$
 $_{2}F_{1}(a, b; c; -\infty) = \infty,$ for a or $b > 0.$

$$(5.49)$$

Many fundamental functions in a real or complex variable can be expressed in terms of the hypergeometric function as well, and some of their typical examples are given in the following [4, p. 556]:

$$(1+z)^{p} = {}_{2}F_{1}(-p,1;1;-z), p \notin \mathbb{Z}_{0}^{-};$$

$$(1-z)^{-1} = {}_{2}F_{1}(1,p;p;z);$$

$$(1-z)^{-2} = {}_{2}F_{1}(1,2;1;z);$$

$$\frac{1}{(1-z)^{a}} = {}_{2}F_{1}(a,b;b;z), a \in \mathbb{Z}^{+};$$

$$z \ln\left(\frac{1}{1-z}\right) = {}_{2}F_{1}(1,1;2;z),$$

$$\frac{1}{z}\ln(1+z) = {}_{2}F_{1}(1,1;2;-z),$$

$$\frac{1}{z}\ln\left(\frac{1+z}{1-z}\right) = {}_{2}F_{1}\left(\frac{1}{2},1;\frac{3}{2};z^{2}\right),$$

$$e^{z} = \lim_{b\to\infty} {}_{2}F_{1}\left(1,b;1;\frac{z}{b}\right),$$

$$\cos(z) = {}_{2}F_{1}\left(\frac{1}{2},-\frac{1}{2};\frac{1}{2};\sin^{2}(z)\right),$$

$$\frac{1}{z}\arcsin(z) = {}_{2}F_{1}\left(\frac{1}{2}, \frac{1}{2}; \frac{3}{2}; z^{2}\right),$$

$$\arccos(z) = {}_{2}F_{1}\left(\frac{1}{2}, 1; \frac{3}{2}; -z^{2}\right),$$

$$\frac{1}{z}\arctan(z) = {}_{2}F_{1}\left(\frac{1}{2}, 1; \frac{3}{2}; -z^{2}\right),$$
(5.51)

The derivative of the hypergeometric function (5.46) is given by

$$\frac{d}{dz}\left[{}_{2}F_{1}\left(a,b;c;z\right)\right] = \frac{ab}{c} {}_{2}F_{1}\left(a+1,b+1;c+1;z\right)$$
(5.52)

or

$$\frac{d^{k}}{dz^{k}} \left[{}_{2}F_{1}\left(a,b;c;z\right) \right] = \frac{(a)_{k}(b)_{k}}{(c)_{k}} {}_{2}F_{1}\left(a+k,b+k;c+k;z\right), \tag{5.53}$$

for all $k \in \{1, 2, ..., n\}$, $a, b, c, z \in \mathbb{C}$ such as |z| < 1, $c \notin \mathbb{Z}_0^-$, while the integration is defined by

$$\int_0^z \left[{}_2F_1\left({a,b;c;t} \right) \right]dt = \frac{{c - 1}}{{\left({a - 1} \right)\left({b - 1} \right)}} \left[{\,}_2F_1\left({a - 1,b - 1;c - 1;z} \right) - 1 \right],\quad (5.54)$$

for all $a,b,c,z\in\mathbb{C}$ with $|z|<1,\,c\notin\mathbb{Z}_{0}^{-}$ and $\operatorname{Re}\left(a\right),\,\operatorname{Re}\left(b\right),\,\operatorname{Re}\left(c\right)\neq1.$

Like the polylogarithms, the hypergeometric functions (5.46) can also be represented in terms of integrals by

$${}_{2}F_{1}(a,b;c;z) = \frac{\Gamma(c)}{\Gamma(b)\Gamma(c-b)} \int_{0}^{1} t^{b-1} (1-t)^{c-b-1} (1-tz)^{-a} dt, \qquad (5.55)$$

where $z \in \mathbb{C} \setminus [1, \infty)$, $a \in \mathbb{C}$, $b, c - b \in \mathbb{C} \setminus \mathbb{Z}_0^-$ and 0 < Re(b) < Re(c), as proved by Euler in 1748 (see [46, p. 4-5]). Other integral representations and similar formulas defining analytic functions can be found in (see for instance [25, p. 65], [149, p.], [268]). If the argument z = 1 and the Re(c - a - b) > 0, then we have from (5.55) that

$${}_{2}F_{1}\left(a,b;c;1\right) = \frac{\Gamma\left(c\right)\Gamma\left(c-a-b\right)}{\Gamma\left(c-a\right)\Gamma\left(c-a\right)}.$$
(5.56)

The functions of the form ${}_{1}F_{1}\left(a;b;z\right)$ are called the *confluent hypergeometric* function of the first kind, which is corresponding to p=q=1, i.e.,

$$_{1}F_{1}(a;b;z) = \sum_{n=0}^{\infty} \frac{(a)_{n}}{(b)_{n}} \frac{z^{n}}{n!},$$
 (5.57)

where $a, z \in \mathbb{C}$, $b \in \mathbb{C} \setminus \mathbb{Z}_0^-$. It can be derived from the Gauss hypergeometric function (5.46) by taking the limit as follows:

$$\lim_{a \to \infty} {}_{2}F_{1}\left(a, b; c; \frac{z}{a}\right) = {}_{1}F_{1}\left(b; c; z\right). \tag{5.58}$$

This function ${}_{1}F_{1}\left(a;b;z\right)$ or $F\left(\begin{smallmatrix} a\\b \end{smallmatrix} \mid z\right)$ is sometimes written as $M\left(a,b,z\right)$ or $\Phi\left(a;b;z\right)$ with its corresponding second-order differential equation called the confluent hypergeometric differential equation or Kummer's differential equation [4, p. 504], namely,

$$z\frac{d^2w}{dz^2} + (b-z)\frac{dw}{dz} - aw = 0, (5.59)$$

for any $a \in \mathbb{C}$, $b \in \mathbb{C} \setminus \mathbb{Z}_0^-$ with w(0) = 0 and w'(0) = a/b. The derivative of the function (5.57) is given by [4, p. 507]

$$\frac{d}{dz} {}_{1}F_{1}(a;b;z) = \frac{a}{b} {}_{1}F_{1}(a+1;b+1;z).$$
 (5.60)

There are many elementary functions connects to (5.57), for example,

$$_{1}F_{1}(a;a;z) = e^{z}, \quad _{1}F_{1}(a;2;2z) = \frac{e^{z}}{z}\sinh(z).$$
 (5.61)

The Kummer's function U(a, b, z) or $\Psi(a; b; z)$ is also related to the hypergeometric functions of type ${}_{2}F_{0}(a, b; ; z)$, where

$$_{2}F_{0}\left(a,1+a-b;;-\frac{1}{z}\right)=z^{a}U\left(a,b,z\right).$$
 (5.62)

Many relations of more complicated hypergeometric functions can be found in (see for example [4, Chapt. 15], [149, p. 686-706], [298, Chapt. II]).

5.1.4 Bessel and Modified Bessel Functions

The Bessel functions of the first kind $J_{\alpha}(z)$, are defined as the solutions to the Bessel's differential equation, i.e.,

$$z^{2} \frac{d^{2}y}{dz^{2}} + z \frac{dy}{dz} + (z^{2} - \alpha^{2}) y = 0, \operatorname{Re}(\alpha) \ge 0,$$
 (5.63)

for an arbitrary real or complex order α . The solution of (5.63) is an analytic function of z in \mathbb{C} , except for a point z=0 when α is not an integer. The solutions, denoted by $J_{\alpha}(z)$ are defined by Taylor series expansion around the origin (see [4, p. 360], [26, p. 57]), i.e.,

$$J_{\alpha}(z) = \sum_{k=0}^{\infty} \frac{\left(-1\right)^{k}}{\Gamma\left(k+\alpha+1\right)k!} \left(\frac{z}{2}\right)^{2k+\alpha}, \tag{5.64}$$

for any $\alpha, z \in \mathbb{C}$ with $\operatorname{Re}(\alpha) \geq 0$ and $\Gamma(z)$ is the gamma function.

The most common of the Bessel functions that occur in practice are those of the integer orders $\alpha = n \in \mathbb{N}$. For such functions, we have

$$J_n(z) = \sum_{k=0}^{\infty} \frac{(-1)^k}{(k+n)!k!} \left(\frac{z}{2}\right)^{2k+n}.$$
 (5.65)

For a non-integer order α , $J_{\alpha}(z)$ and $J_{-\alpha}(z)$ are linearly independent, and therefore the two solutions of the differential equation (5.63). On the other hand, for an integer order α , the $J_{\alpha}(z)$ and $J_{-\alpha}(z)$ are linearly dependent solutions of (5.63), hence the following relationship is valid [4, p. 358]:

$$J_{-\alpha}(z) = (-1)^{\alpha} J_{\alpha}(z).$$
 (5.66)

The Bessel functions satisfy a large number of identities and integral relations, some of which are provided below [26, p. 58]:

$$J_0(0) = 1, (5.67)$$

$$J_{\alpha}(0) = 0, \quad \alpha > 0, \tag{5.68}$$

$$\frac{2\alpha}{x}J_{\alpha}(x) = J_{\alpha-1}(x) + J_{\alpha+1}(x), \qquad (5.69)$$

$$2J'_{\alpha}(x) = J_{\alpha-1}(x) - J_{\alpha+1}(x), \qquad (5.70)$$

$$J_0(x) = \frac{1}{2\pi} \int_0^{2\pi} e^{\pm ix \cos \theta} d\theta, \qquad (5.71)$$

$$J_n(x) = \frac{1}{\pi} \int_0^{2\pi} \cos(n\theta - x\sin\theta) d\theta, \ n \in \mathbb{N}_0, \tag{5.72}$$

$$J_{\frac{1}{2}}(x) = \sqrt{\frac{2}{\pi x}}\sin(x), \quad x > 0,$$
 (5.73)

$$J_{-\frac{1}{2}}(x) = \sqrt{\frac{2}{\pi x}}\cos(x), \quad x > 0.$$
 (5.74)

If the argument z in (5.63) is replaced by $\pm iz$, then the solutions of the second order differential equations are called the *modified Bessel functions of the first kind*. We denoted these functions by $I_{\alpha}(z)$. It is easy to verify from (5.64) that the modified Bessel functions $I_{\alpha}(z)$ are defined by the following power series [4, p. 375]:

$$I_{\alpha}(z) = \sum_{k=0}^{\infty} \frac{1}{\Gamma(k+\alpha+1)k!} \left(\frac{z}{2}\right)^{2k+\alpha}, \tag{5.75}$$

for all $\alpha, z \in \mathbb{C}$ such as $\text{Re}(\alpha) \geq 0$. We observe that the function $I_{\alpha}(z)$ has all the nonnegative coefficients. If $\alpha = n \in \mathbb{N}_0$, then we have

$$I_n(z) = \sum_{k=0}^{\infty} \frac{1}{(k+n)!k!} \left(\frac{z}{2}\right)^{2k+\alpha}.$$

Similar to the Bessel functions, the modified Bessel functions of the first kind (5.75) also satisfy the following relations,

$$I_{\alpha}(-z) = (-1)^{\alpha} I_{\alpha}(z)$$
 and $I_{-\alpha}(z) = I_{\alpha}(z)$ (5.76)

holding for all $\alpha \in \mathbb{Z}$, $z \in \mathbb{C}$ such as |z| < 1. The modified Bessel functions of the first kind of order α , $I_{\alpha}(z)$, can be expressed by Bessel functions of the first kind as follows:

$$J_{\alpha}(iz) = i^{\alpha} I_{\alpha}(z). \tag{5.77}$$

Most of the properties of the modified Bessel functions are analogous to the ordinary Bessel functions, some of which are provided below [26, p. 62]:

$$I_0(0) = 1, I_{\alpha}(0) = 0, \alpha > 0,$$
 (5.78)

$$\frac{2\alpha}{r}I_{\alpha}(x) = I_{\alpha-1}(x) - I_{\alpha+1}(x), \qquad (5.79)$$

$$2I'_{\alpha}(x) = I_{\alpha-1}(x) + I_{\alpha+1}(x),$$
 (5.80)

$$I_0(x) = \frac{1}{2\pi} \int_0^{2\pi} e^{\pm x \cos \theta} d\theta, \qquad (5.81)$$

$$I_{-\frac{1}{2}}(x) = \sqrt{\frac{2}{\pi x}} \cosh(x), \quad x > 0,$$
 (5.82)

$$I_{\frac{1}{2}}(x) = \sqrt{\frac{2}{\pi x}} \sinh(x), \quad x > 0.$$
 (5.83)

In the following section, we provide some results involving the special functions. We start by presenting some inequalities for the polylogarithm functions.

5.2 Inequalities for Polylogarithm Functions

It is clearly seen that the polylogarithm function (5.28) is the power series with nonnegative coefficients and convergent on the open disk D(0,1), so that all the results established in Section 3.2 - Section 3.4 of Chapter 3, and Section 4.1.2, Section 4.1.3, Section 4.2.2, Section 4.2.3 of Chapter 4 hold true. Therefore, for the examples, and for the main purpose of this section, we derive some inequalities involving the polylogarithm functions by considering the inequalities (3.9), (3.56), (3.95), (3.118), (4.29), (4.70) and (4.156).

First, we give the following result proved by Cerone and Dragomir [89].

Corollary 101 If $Li_n(z)$ is the polylogarithm function, then we have

$$|Li_n(az)|^2 \le \frac{1}{2}Li_n(a^2)\left[Li_n(|z|^2) + |Li_n(z^2)|\right],$$
 (5.84)

for $a \in (-1,1)$, $z \in \mathbb{C}$ with z, az, a^2 , z^2 , $|z|^2 \in D(0,1)$ and n is a negative or positive integer.

Utilising the inequalities (3.9) and (3.56) for the polylogarithm functions (5.28), we obtain the following results (see [208], [209], [217]).

Corollary 102 If $Li_n(z)$ is the polylogarithm function, then we have

$$\left| Li_{n}\left(\alpha \overline{x}\right) Li_{n}\left(\overline{\beta}x\right) \right|$$

$$\leq \frac{1}{2} \left[\left[Li_{n}\left(|\alpha|^{2}\right) Li_{n}\left(|\beta|^{2}\right) \right]^{1/2} + \left| Li_{n}\left(\alpha \overline{\beta}\right) \right| \right] Li_{n}\left(|x|^{2}\right), \qquad (5.85)$$

for any $\alpha, \beta, x \in \mathbb{C}$ with $|\alpha|^2$, $|\beta|^2$, $|x|^2$, $\alpha \overline{\beta}$, $\alpha \overline{x}$, $\overline{\beta}x \in D(0,1)$ and n is a negative or positive integer.

Corollary 103 If $Li_n(z)$ is the polylogarithm function, then we have

$$\left[Li_{n}\left(\left|x\right|^{2}\right)Li_{n}\left(\left|y\right|^{2}\right)\right]^{1/2}Li_{n}\left(\left|z\right|^{2}\right)-\left|Li_{n}\left(x\overline{z}\right)Li_{n}\left(z\overline{y}\right)\right|$$

$$\geq\left|Li_{n}\left(x\overline{y}\right)Li_{n}\left(\left|z\right|^{2}\right)-Li_{n}\left(x\overline{z}\right)Li_{n}\left(z\overline{y}\right)\right|,$$
(5.86)

for any x, y, $z \in \mathbb{C}$ such that $|x|^2$, $|y|^2$, $|z|^2$, $x\overline{z}$, $z\overline{y}$, $x\overline{y} \in D(0,1)$ and n is a negative or a positive integer.

We also obtain the following results from (3.95) and (3.118), see [214], [217].

Corollary 104 If $Li_n(z)$ is the polylogarithm function, then we have

$$Li_{n}(|x||z|^{2})Li_{n}(|x|) - |Li_{n}(|x|z)|^{2}$$

$$\geq |Li_{n}(x)Li_{n}(x|z|z) - Li_{n}(xz)Li_{n}(x|z|)|, \qquad (5.87)$$

for any $x, z \in \mathbb{C}$ such that $x, xz, |x| |z|^2 \in D(0,1)$ and n is a negative or positive integer.

Corollary 105 If $Li_n(z)$ is the polylogarithm function, then we have

$$Li_{n}(|x|^{2}) Li_{n}(|y|^{2}) - |Li_{n}(xy)|^{2}$$

$$\geq |Li_{n}(|x|x)Li_{n}(|y|\overline{y}) - Li_{n}(|y|x)Li_{n}(|x|\overline{y})|, \qquad (5.88)$$

for any $x, y \in \mathbb{C}$ with $|x|^2$, $|y|^2 < R$ and n is a negative or positive integer.

As a consequence of the above results, we get the following inequalities, which incorporate the Riemann zeta function $\zeta(z)$, see ([214],

Corollary 106 (i) If we choose x = 1 in (5.87), then we get the following inequality, which incorporates the zeta function, i.e.,

$$\zeta(n) Li_n(|z|^2) - |Li_n(z)|^2$$

 $\geq |\zeta(n) Li_n(|z|z) - Li_n(z) Li_n(|z|)|,$ (5.89)

for any $z \in D(0,1)$, where n is a positive or negative integer. In particular, if n = 2, then by the inequality (5.89) we get

$$\frac{\pi^{2}}{6}Li_{2}(|z|^{2}) - |Li_{2}(z)|^{2} \ge \left| \frac{\pi^{2}}{6}Li_{2}(|z|z) - Li_{2}(z)Li_{2}(|z|) \right|, \tag{5.90}$$

for any $z \in D(0,1)$ and the $Li_2(z)$ is the dilogarithm or Spence's function which is defined in (5.32).

(ii) If we choose x = i in (5.87), then we have

$$\zeta(n)Li_{n}(|z|^{2}) - |Li_{n}(z)|^{2}$$

$$\geq \left| \left(\frac{1}{2^{n}} \eta(n) + i\beta(n) \right) Li_{n}(i|z|z) - Li_{n}(iz) Li_{n}(i|z|) \right|, \tag{5.91}$$

for any $z \in D(0,1)$ and n is a positive or negative integer. In particular, for n=2, we have from (5.91) that

$$\frac{\pi^{2}}{6}Li_{2}(|z|^{2}) - |Li_{2}(z)|^{2}
\geq \left| \left(\frac{\pi^{2}}{48} + iG \right) Li_{2}(i|z|z) - Li_{2}(iz) Li_{2}(i|z|) \right|,$$
(5.92)

for any $z \in D(0,1)$, where G is the Catalan's constant.

Corollary 107 (i) If we choose x = 1 in (5.88), then we have

$$\zeta(n) \operatorname{Li}_{n}(|y|^{2}) - |\operatorname{Li}_{n}(y)|^{2} \ge |\zeta(n) \operatorname{Li}_{n}(|y| \overline{y}) - \operatorname{Li}_{n}(|y|) \operatorname{Li}_{n}(\overline{y})|$$
(5.93)

for any $z \in D(0,1)$, where n is a positive or negative integer. In particular, for n=2, we get from the inequality (5.93)

$$\frac{\pi^{2}}{6}Li_{2}(|y|^{2}) - |Li_{2}(y)|^{2} \ge \left|\frac{\pi^{2}}{6}Li_{2}(|y|\overline{y}) - Li_{2}(|y|)Li_{2}(\overline{y})\right|$$
(5.94)

for any $z \in D(0,1)$.

(ii) If we choose x = i in (5.88), then we have

$$\zeta(n) Li_n(|y|^2) - |Li_n(iy)|^2$$

$$\geq \left| \left(\frac{1}{2^n} \eta(n) + i\beta(n) \right) Li_n(|y|\overline{y}) - Li_n(i|y|) Li_n(\overline{y}) \right|, \tag{5.95}$$

for any $y \in D(0,1)$, where n is a positive or negative integer. In particular, we get the following inequality by choosing n = 2 in (5.95),

$$\frac{\pi^{2}}{6}Li_{2}(|y|^{2}) - |Li_{2}(iy)|^{2}
\geq \left| \left(\frac{\pi^{2}}{48} + iG \right) Li_{2}(|y|\overline{y}) - Li_{2}(i|y|)Li_{2}(\overline{y}) \right|,$$
(5.96)

for any $z \in D(0,1)$.

Similarly, if we apply the inequalities (4.29), (4.70) and (4.156) for the polylogarithm functions (5.28) for, then we get the following results (see [213], [215], [216], [217]).

Corollary 108 Let $Li_n(z)$ be the polylogarithm function. Then, we have

$$Li_n(|x|^p)Li_n(|y|^q) \ge |Li_n(xy)Li_n(|x|^{p-1}|y|^{q-1})|,$$
 (5.97)

for any $x, y \in \mathbb{C}$, $x, y \neq 0$ with xy, $|x|^p$, $|y|^q \in D(0,1)$ and p, q > 1, 1/p + 1/q = 1. In particular, if n = 0 in (5.97), then we get the following inequality:

$$|1 - xy| |1 - |x|^{p-1} |y|^{q-1}| \ge (1 - |x|^p) (1 - |y|^q),$$
 (5.98)

for any $x, y \neq 0$, xy, $|x|^p$, $|y|^q \in D(0,1)$ and p, q > 1 with 1/p + 1/q = 1. If we take n = 1 in (5.97), then we get the inequality (4.38) for all $x, y \neq 0$ with xy, $|x|^p$, $|y|^q \in D(0,1)$ and p, q > 1, 1/p + 1/q = 1. Also, if we choose in (5.97) n = 2, then we obtain

$$Li_2(|x|^p)Li_2(|y|^q) \ge |Li_2(xy)Li_2(|x|^{p-1}|y|^{q-1})|,$$
 (5.99)

for any $x, y \neq 0, xy, |x|^p, |y|^q \in D(0,1)$ and p > 1 with 1/p + 1/q = 1.

Corollary 109 If $Li_n(x)$ is the polylogarithm function, then we have

$$Li_n(y^{\nu}z^{1-\nu})Li_n(y^{1-\nu}z^{\nu}) \le Li_n(y)Li_n(z),$$
 (5.100)

for any y, z, $y^{\nu}z^{1-\nu}$, $y^{1-\nu}z^{\nu} \in (0,1)$, $\nu \in [0,1]$ and $n \in \mathbb{Z}$. In particular, if n = 0 in (5.100), then we get the inequality (4.72). Also, if n = 1, then we get from (5.100) the inequality (4.73). Further, we obtain the following inequality by choosing n = 2 in (5.100):

$$Li_2(x^{\nu}y^{1-\nu})Li_2(x^{1-\nu}y^{\nu}) \le Li_2(x)Li_2(y),$$
 (5.101)

for y, z, $y^{\nu}z^{1-\nu}$, $y^{1-\nu}z^{\nu} \in (0,1)$ and $\nu \in [0,1]$.

Corollary 110 If $Li_n(x)$ is the polylogarithm function, then we have

$$\left(\frac{c}{a}\right)^{\frac{Li_{n-1}(a)}{Li_{n}(a)}} \le \frac{Li_{n}\left(c\right)}{Li_{n}\left(a\right)} \le \left(\frac{c}{a}\right)^{\frac{Li_{n-1}(c)}{Li_{n}(c)}},$$
(5.102)

for any $a, c \in (0,1)$ and n is positive or negative integer. In particular, if n = 0 in (5.102), then we have the simpler inequality

$$\left(\frac{c}{a}\right)^{\frac{a}{1-a}} \le \frac{1-a}{1-c} \le \left(\frac{c}{a}\right)^{\frac{c}{1-c}},\tag{5.103}$$

for any $a, c \in (0,1)$. Also, if we take n = 1 in (5.102), then we get

$$\left(\frac{c}{a}\right)^{-\frac{a}{(1-a)\ln(1-a)}} \le \frac{\ln\left(1-c\right)}{\ln\left(1-a\right)} \le \left(\frac{c}{a}\right)^{-\frac{c}{(1-c)\ln(1-c)}},$$
(5.104)

for any $a, c \in (0,1)$. Further, we obtain the following inequality by choosing n = 2 in (5.102):

$$\left(\frac{c}{a}\right)^{-\frac{\ln(1-a)}{Li_2(a)}} \le \frac{Li_2\left(c\right)}{Li_2\left(a\right)} \le \left(\frac{c}{a}\right)^{-\frac{\ln(1-c)}{Li_2(c)}},$$
(5.105)

for any $a, c \in (0, 1)$.

Further, we establish some inequalities involving the polylogarithms for different order of integers p, q and r. In [89], Cerone and Dragomir proved the following result via the de Bruijn inequality.

Theorem 111 (Cerone and Dragomir [89]) $Li_n(z)$ is the polylogarithm function, $a \in (-1,1)$, $z \in D(0,1)$ and p, q, r integers such that the following series exist. Then

$$|Li_{r+p+q}(az)|^2 \le \frac{1}{2}Li_{r+2p}(a^2)\left[Li_{r+2q}(|z|^2) + \left|Li_{r+2q}(z^2)\right|\right].$$
 (5.106)

We obtain the similar inequality to (5.106), by utilizing the Buzano's result (3.6) in inner product spaces (see [208], [217]). The result is given as follows:

Theorem 112 (Ibrahim and Dragomir [208]) Let α , β , $x \in \mathbb{C}$ with $\alpha \overline{x}$, $\overline{\beta}x$, $|\alpha|^2$, $|\beta|^2$, $\alpha \overline{\beta}$, $|x|^2 \in D(0,1)$ and p, q, r integers such that the following series exist. Then

$$\left| Li_{r+p+q} \left(\alpha \overline{x} \right) Li_{r+p+q} \left(\overline{\beta} x \right) \right|$$

$$\leq \frac{1}{2} \left(\left[Li_{r+2q} \left(|\alpha|^2 \right) Li_{r+2q} \left(|\beta|^2 \right) \right]^{1/2} + \left| Li_{r+2q} \left(\alpha \overline{\beta} \right) \right| \right) Li_{r+2p} \left(|x|^2 \right).$$
 (5.107)

Proof. Utilising the Buzano inequality (3.6) for $p_k = 1/k^r$, $c_k = \alpha^k/k^q$, $b_k = \beta^k/k^q$ and $x_k = x^k/k^p$, we have

$$\begin{aligned} &|Li_{r+p+q}\left(\alpha\overline{x}\right)Li_{r+p+q}\left(\overline{\beta}x\right)| \\ &= \left|\sum_{k=1}^{\infty} \frac{1}{k^{r}} \frac{\alpha^{k}}{k^{q}} \frac{\left(\overline{x}\right)^{k}}{k^{p}} \sum_{k=1}^{\infty} \frac{1}{k^{r}} \frac{\left(\overline{\beta}\right)^{k}}{k^{q}} \frac{\left(x\right)^{k}}{k^{p}} \right| \\ &\leq \frac{1}{2} \left(\left[\sum_{k=1}^{\infty} \frac{1}{k^{r}} \frac{\left(|\alpha|^{2}\right)^{k}}{k^{2q}} \sum_{k=1}^{\infty} \frac{1}{k^{r}} \frac{\left(|\beta|^{2}\right)^{k}}{k^{2q}} \right]^{1/2} \\ &+ \left|\sum_{k=1}^{\infty} \frac{1}{k^{r}} \frac{\left(\alpha\overline{\beta}\right)^{k}}{k^{2q}} \right| \sum_{k=1}^{\infty} \frac{1}{k^{r}} \frac{\left(|x|^{2}\right)^{k}}{k^{2p}} \\ &= \frac{1}{2} \left(\left[Li_{r+2q}\left(|\alpha|^{2}\right)Li_{r+2q}\left(|\beta|^{2}\right) \right]^{1/2} \\ &+ \left|Li_{r+2q}\left(\alpha\overline{\beta}\right)\right| \right) Li_{r+2p}\left(|x|^{2}\right). \end{aligned} (5.108)$$

Thus, the desired inequality (5.107) is proved.

Also, on utilising the refinement of the Schwarz inequality (3.54) in inner product spaces, we obtain the following result (see [209], [217]).

Theorem 113 (Ibrahim and Dragomir [209]) Let $x, y, z \in \mathbb{C}$ with $|x|^2$, $|y|^2$, $|z|^2$, $x\overline{z}$, $z\overline{y}$, $x\overline{y} \in D(0,1)$ and p, q, r integers such that the following series exist. Then

$$\left[Li_{r+2q}\left(|x|^{2}\right)Li_{r+2q}\left(|y|^{2}\right)\right]^{1/2}Li_{r+2p}\left(|z|^{2}\right)
-\left|Li_{r+p+q}\left(x\overline{z}\right)Li_{r+p+q}\left(z\overline{y}\right)\right|
\geq\left|Li_{r+2q}\left(x\overline{y}\right)Li_{r+2p}\left(|z|^{2}\right)-Li_{r+p+q}\left(x\overline{z}\right)Li_{r+p+q}\left(z\overline{y}\right)\right|.$$
(5.109)

Proof. Utilising the discrete inequality (3.54) for $p_k = 1/k^r$, $x_k = x^k/k^q$, $y_k = y^k/k^q$, $z_k = z^k/k^p$, $k \in \{1, 2, ..., m\}$, we have

$$\left(\sum_{k=1}^{m} \frac{1}{k^{r}} \left| \frac{x^{k}}{k^{q}} \right|^{2}\right)^{1/2} \left(\sum_{k=1}^{m} \frac{1}{k^{r}} \left| \frac{y^{k}}{k^{q}} \right|^{2}\right)^{1/2} \sum_{k=1}^{m} \frac{1}{k^{r}} \left| \frac{z^{k}}{k^{p}} \right|^{2} \\
- \left| \sum_{k=1}^{n} \frac{1}{k^{r}} \frac{x^{k}}{k^{q}} \frac{(\overline{z})^{k}}{k^{p}} \sum_{k=1}^{n} \frac{1}{k^{r}} \frac{z^{k}}{k^{p}} \frac{(\overline{y})^{k}}{k^{q}} \right| \\
\ge \left| \sum_{k=1}^{m} \frac{1}{k^{r}} \frac{x^{k}}{k^{q}} \frac{(\overline{y})^{k}}{k^{q}} \sum_{k=1}^{n} \frac{1}{k^{r}} \left| \frac{z^{k}}{k^{p}} \right|^{2} - \sum_{k=1}^{m} \frac{1}{k^{r}} \frac{x^{k}}{k^{q}} \frac{(\overline{z})^{k}}{k^{p}} \sum_{k=1}^{m} \frac{1}{k^{r}} \frac{z^{k}}{k^{q}} \frac{(\overline{y})^{k}}{k^{q}} \right|.$$
(5.110)

Hence

$$\left(\sum_{k=1}^{m} \frac{1}{k^{r+2q}} \left(|x|^{2}\right)^{k}\right)^{1/2} \left(\sum_{k=1}^{m} \frac{1}{k^{r+2q}} \left(|y|^{2}\right)^{k}\right)^{1/2} \sum_{k=1}^{m} \frac{1}{k^{r+2p}} \left(|z|^{2}\right)^{k} \\
-\left|\sum_{k=1}^{m} \frac{1}{k^{r+p+q}} \left(x\overline{z}\right)^{k} \sum_{k=1}^{m} \frac{1}{k^{r+p+q}} \left(z\overline{y}\right)^{k}\right| \\
\geq \sum_{k=1}^{m} \frac{(x\overline{y})^{k}}{k^{r+2q}} \sum_{k=1}^{m} \frac{(|z|^{2})^{k}}{k^{r+2p}} - \sum_{k=1}^{m} \frac{(x\overline{z})^{k}}{k^{r+p+q}} \sum_{k=1}^{m} \frac{(z\overline{y})^{k}}{k^{r+p+q}}, \tag{5.111}$$

for $m \geq 0$. Taking the limit as $m \to \infty$ in the inequality (5.111), then we deduce the desired result (5.109). \blacksquare

On making use of the results (5.107) and (5.109), we obtain some more simpler inequalities as follows:

(a) If $\alpha = z$, $\beta = \overline{z}$, then from (5.107) we can state that

$$|Li_{r+p+q}(z\overline{x}) Li_{r+p+q}(zx)| \le \frac{1}{2} \left[Li_{r+2q}(|z|^2) + \left| Li_{r+2q}(z^2) \right| \right] Li_{r+2p}(|x|^2).$$
 (5.112)

Further, if $x = a \in \mathbb{R}$, then from (5.112) we deduce the inequality (5.106).

(b) If $y = \overline{x}$, then from (5.109) we can state that

$$Li_{r+2q}(|x|^{2}) Li_{r+2p}(|z|^{2}) - |Li_{r+p+q}(xz) Li_{r+p+q}(x\overline{z})|$$

$$\geq |Li_{r+2q}(x^{2}) L_{r+2p}(|z|^{2}) - Li_{r+p+q}(xz) Li_{r+p+q}(x\overline{z})|, \qquad (5.113)$$

for $x, z \in \mathbb{C}$. Moreover, if $z = a \in \mathbb{R}$, then from (5.113) we deduce the inequality

$$Li_{r+2q}(|x|^2)Li_{r+2p}(a^2) - |Li_{r+p+q}(ax)|^2$$

 $\geq |Li_{r+2q}(x^2)Li_{r+2p}(a^2) - Li_{r+p+q}^2(ax)|,$ (5.114)

for any $x \in \mathbb{C}$ and $a \in \mathbb{R}$.

In particular, we can state the following inequalities, which incorporate the Riemann zeta function $\zeta\left(z\right)$:

Corollary 114 Let $z \in D(0,1)$ and p, q, r integer such that r = 2p > 1. Then

$$|Li_{p+q+r}(z)|^2 \le \frac{1}{2}\zeta(r+2p)\left[Li_{r+2q}(|z|^2) + \left|Li_{r+2q}(z^2)\right|\right].$$
 (5.115)

Proof. The proof follows by Theorem 111 for a = 1.

Corollary 115 Let α , $\beta \in D(0,1)$ and p, q, r integers such that r+2p > 1. Then

$$\left| Li_{r+p+q} \left(-\alpha i \right) Li_{r+p+q} \left(\overline{\beta} i \right) \right|$$

$$\leq \frac{1}{2} \zeta \left(r + 2p \right) \left(\left[Li_{r+2q} \left(|\alpha|^2 \right) Li_{r+2q} \left(|\beta|^2 \right) \right]^{1/2} + \left| Li_{r+2q} \left(\alpha \overline{\beta} \right) \right| \right).$$
(5.116)

Proof. The proof follows by Theorem 112 for x = i.

Corollary 116 Let α , $\beta \in D(0,1)$ and p, q, r integers such that r + 2p > 1. Then

$$\zeta(r+2q) Li_{r+2q}(|x|^2) - |Li_{r+p+q}(x)|^2$$

$$\ge |\zeta(r+2q) Li_{r+2q}(x^2) - Li_{r+p+q}^2(x)|, (5.117)$$

for any $x \in \mathbb{C}$.

Proof. The proof follows by utilising the inequality (5.114) for a = 1.

Next, on utilising the inequalities given by (5.115), (5.116) and (5.117), and taking into account that some particular values of ζ (n) are known such as ζ (2) and ζ (4) as mentioned in (5.22), we derive some particular inequalities involving the polylogarithm functions. The results are summarized as follows:

(a)
$$|Li_{q+1}(z)|^2 \le \frac{\pi^2}{12} \left[Li_{2q}(|z|^2) + |Li_{2q}(z^2)| \right],$$

(b)
$$|Li_{q+2}(z)|^2 \le \frac{\pi^4}{180} \left[Li_{2q}(|z|^2) + |Li_{2q}(z^2)| \right],$$

(c)
$$\left| Li_{q+1} \left(-\alpha i \right) Li_{q+1} \left(\overline{\beta} i \right) \right|$$

$$\leq \frac{\pi^2}{12} \left(\left[Li_{2q} \left(|\alpha|^2 \right) Li_{2q} \left(|\beta|^2 \right) \right]^{1/2} + \left| Li_{2q} \left(\alpha \overline{\beta} \right) \right| \right),$$

(d)
$$\left| Li_{q+2} \left(-\alpha i \right) Li_{q+2} \left(\overline{\beta} i \right) \right|$$

$$\leq \frac{\pi^2}{180} \left(\left[Li_{2q} \left(|\alpha|^2 \right) Li_{2q} \left(|\beta|^2 \right) \right]^{1/2} + \left| Li_{2q} \left(\alpha \overline{\beta} \right) \right| \right), \tag{5.118}$$

(e)
$$|Li_{q+3}(-\alpha i)Li_{q+3}(\overline{\beta}i)|$$

 $\leq \frac{\pi^4}{180} \left(\left[Li_{2(q+1)}(|\alpha|^2) Li_{2(q+1)}(|\beta|^2) \right]^{1/2} + \left| Li_{2(q+1)}(\alpha \overline{\beta}) \right| \right),$

(f)
$$\left| \frac{\pi^2}{6} Li_{2q}(x^2) - Li_{q+1}^2(x) \right| \le \frac{\pi^2}{6} Li_{2q}(|x|^2) - |Li_{q+1}(x)|^2$$
,

(g)
$$\left| \frac{\pi^4}{90} Li_{2q}(x^2) - Li_{q+2}^2(x) \right| \le \frac{\pi^4}{90} Li_{2q}(|x|^2) - |Li_{q+2}(x)|,$$

(h)
$$\left| \frac{\pi^4}{90} Li_{2(q+1)}(x^2) - Li_{q+3}^2(x) \right| \le \frac{\pi^4}{90} Li_{2(q+1)}(|x|^2) - |Li_{q+3}(x)|^2$$
,

for any α , β , x, $z \in D(0,1)$ and q is an integer.

Other studies related to the polylogarithm functions for further reading can be found in the literature, see for example ([1], [89], [105], [143], [295], [312], [391]) and the references cited therein.

5.3 Inequalities for Hypergeometric Functions

Since, the hypergeometric function (5.46) is a power series with nonnegative coefficients and convergent on the open disk D(0,1), then all the results, which are established in Section 3.2 - Section 3.4 of Chapter 3, and Section 4.1.2, Section 4.1.3, Section 4.2.2 and Section 4.2.3 of Chapter 4 hold true. Therefore, for instance and thus, for the purpose of this section, we present some inequalities involving the hypergeometric functions by applying some of those results such as (3.118), (4.29), (4.70) and (4.156) for this function.

First, we obtain the following result by applying the inequality (3.118) for the hypergeometric functions (5.46), see ([214], [217]).

Corollary 117 Let ${}_2F_1(a,b;c;z)$ be the hypergeometric function. Then

$${}_{2}F_{1}(a,b;c;|x|^{2}) {}_{2}F_{1}(a,b;c;|y|^{2}) - |{}_{2}F_{1}(a,b;c;xy)|^{2}$$

$$\geq |{}_{2}F_{1}(a,b;c;|x|x) {}_{2}F_{1}(a,b;c;|y|\overline{y})$$

$$-{}_{2}F_{1}(a,b;c;|y|x) {}_{2}F_{1}(a,b;c;|x|\overline{y})|, \qquad (5.119)$$

for any $a, b, c \in \mathbb{R}$, with $c \notin \{\ldots, -2, -1, 0\}$ and $x, y \in \mathbb{C}$ such that |x|, |y| < 1.

As a natural consequence, the following result holds:

Corollary 118 (i) If in (5.119), we choose c = b, then we have

$$\frac{1}{\left[\left(1-|x|^{2}\right)\left(1-|y|^{2}\right)\right]^{a}} - \frac{1}{|1-xy|^{2a}}$$

$$\geq \left|\frac{1}{\left[\left(1-|x|x\right)\left(1-|y|\overline{y}\right)\right]^{a}} - \frac{1}{\left[\left(1-|y|x\right)\left(1-|x|\overline{y}\right)\right]^{a}}\right|, \tag{5.120}$$

for any $a \in \mathbb{R}$, $x, y \in \mathbb{C}$ such that |x|, |y| < 1.

(ii) Also, if in (5.119), we choose a = b = 1, c = 2, then we get

$$\ln\left(\frac{1}{1-|x|^2}\right)\ln\left(\frac{1}{1-|y|^2}\right) - \left|\ln\left(\frac{1}{1-xy}\right)\right|^2$$

$$\geq \left|\ln\left(\frac{1}{1-|x|x}\right)\ln\left(\frac{1}{1-|y|\overline{y}}\right)\right|$$

$$-\ln\left(\frac{1}{1-|y|x}\right)\ln\left(\frac{1}{1-|x|\overline{y}}\right), \qquad (5.121)$$

for $x, y \in \mathbb{C}$ with |x|, |y| < 1.

Remark 119 For a = 1, the inequality (5.120) reduces to (3.125).

On applying the inequality (4.29) for the hypergeometric function (5.46), we also have (see [215]):

Corollary 120 If ${}_2F_1(a,b;c;z)$ is a hypergeometric function, then for any $a, b, c \in \mathbb{R}$ we have

$${}_{2}F_{1}(a,b;c;|x|^{p}) {}_{2}F_{1}(a,b;c;|y|^{q})$$

$$\geq |{}_{2}F_{1}(a,b;c;xy) {}_{2}F_{1}(a,b;c;|x|^{p-1}|y|^{q-1})|, \qquad (5.122)$$

where $x, y \neq 0$ with $xy, |x|^p, |y|^q \in D(0,1)$ and p > 1, 1/p + 1/q = 1. In particular, if we choose a = 1, c = b in (5.122), then we get the inequality (4.34). Also, if we choose a = b = 1 and c = 2, then the inequality (5.122) reduces to (4.38).

Next, from the inequality (4.70) we obtain (see [216]):

Corollary 121 If ${}_{2}F_{1}\left(a,b;c;x\right)$ is the hypergeometric function, then we have

$${}_{2}F_{1}\left(a,b;c;y^{\nu}z^{1-\nu}\right) {}_{2}F_{1}\left(a,b;c;y^{1-\nu}z^{\nu}\right)$$

$$\leq {}_{2}F_{1}\left(a,b;c;y\right) {}_{2}F_{1}\left(a,b;c;z\right),$$
(5.123)

for $y, z, y^{\nu}z^{1-\nu}, y^{1-\nu}z^{\nu} \in (0,1)$ and $\nu \in [0,1]$. In particular, if we choose a = b = c = 1, then the inequality (5.123) reduces to (4.72). In fact, the inequality (5.123) reduces to (4.72) for any $a, b, c \in \mathbb{R}$ such that $c = b \notin \{0, -1, -2, \ldots\}$.

Finally, we get from (4.156) the following result (see [213]):

Corollary 122 If ${}_2F_1(a,b;c;x)$ is the hypergeometric function, then for any a, b, $c \in \mathbb{R}$ and x, $y \in (0,1)$ we have

$$\left(\frac{x}{y}\right)^{\frac{aby\ _{2}F_{1}(a+1,b+1;c+1;y)}{c\ _{2}F_{1}(a,b;c;y)}} \le \frac{_{2}F_{1}\left(a,b;c;x\right)}{_{2}F_{1}\left(a,b;c;y\right)} \le \left(\frac{x}{y}\right)^{\frac{abx\ _{2}F_{1}(a+1,b+1;c+1;x)}{c\ _{2}F_{1}(a,b;c;x)}}.$$
(5.124)

In particular, if we choose b = c in (5.124), then we obtain that

$$\left(\frac{x}{y}\right)^{y/(1-y)} \le \frac{1-y}{1-x} \le \left(\frac{x}{y}\right)^{x/(1-x)},\tag{5.125}$$

for any $a \in \mathbb{R}$, $x, y \in (0, 1)$.

For other results devoted to the hypergeometric functions, see for example ([51], [86], [191], [260], [263], [366], [396]) and the references which are cited therein.

5.4 Inequalities for Bessel Functions

Similar to the polylogarithm and hypergeometric functions, the modified Bessel functions of the first kind (5.75) are also defined by the power series with nonnegative coefficients and convergent on the open disk D(0,1), so that all the results in Section 3.2 - Section 3.4 of Chapter 3, and Section 4.1.2, Section 4.1.3, Section 4.2.2 and Section 4.2.3 of Chapter 4 hold true. Therefore, for instance from (3.118), (4.29), (4.70) and (4.156), we have the following inequalities involving the Bessel and modified Bessel functions of the first kind (see [213], [214], [215], [216]).

Corollary 123 If $J_{\alpha}(x)$ and $I_{\alpha}(x)$ are the Bessel and modified Bessel functions of the first kind respectively, then we have

$$I_{\alpha}(|x|^{2}) I_{\alpha}(|y|^{2}) - |I_{\alpha}(xy)|^{2}$$

$$\geq |J_{\alpha}(|x|x)J_{\alpha}(|y|\overline{y}) - J_{\alpha}(|y|x)J_{\alpha}(|x|\overline{y})| \qquad (5.126)$$

for $\alpha \in \mathbb{R}$, $x, y \in \mathbb{C}$ with |x|, |y| < 1. In particular, if $y = \alpha = 0$, then we obtain from (5.126) that

$$I_0(|x|^2) - 1 \ge |J_0(|x|x) - 1|,$$
 (5.127)

where

$$J_0(z) = \sum_{k=0}^{\infty} \frac{(-1)^n}{(k!)^2} \left(\frac{z}{2}\right)^{2k} \quad and \quad I_0(z) = \sum_{k=0}^{\infty} \frac{1}{(k!)^2} \left(\frac{z}{2}\right)^{2k}$$
 (5.128)

for |z| < 1.

Corollary 124 If $J_{\alpha}(z)$ and $I_{\alpha}(z)$ are the Bessel and modified Bessel functions of the first kind respectively, then for any $\alpha, x, y \in \mathbb{C}$, we have

$$I_{\alpha}(|x|^{p})I_{\alpha}(|y|^{q}) \ge |J_{\alpha}(xy)J_{\alpha}(|x|^{p-1}|y|^{q-1})|,$$
 (5.129)

where $x, y \neq 0$, xy, $|x|^p$, $|y|^q \in D(0,1)$ and p > 1 with 1/p + 1/q = 1. In particular, if $\alpha = 0$ in (5.129), then for p, q > 1 with 1/p + 1/q = 1, we get

$$J_0(i|x|^p) J_0(i|y|^q) \ge \left| J_0(xy) J_0(|x|^{p-1}|y|^{q-1}) \right|. \tag{5.130}$$

Corollary 125 If $I_{\alpha}(x)$ is the modified Bessel function of the first kind, then we have

$$I_{\alpha}\left(y^{\nu}z^{1-\nu}\right)I_{\alpha}\left(y^{1-\nu}z^{\nu}\right) \leq I_{\alpha}(y)I_{\alpha}(z),\tag{5.131}$$

for $y, z, y^{\nu}z^{1-\nu}, y^{1-\nu}z^{\nu} \in (0,1)$ and $\nu \in [0,1]$. In particular, if $\alpha = 0$, then from (5.131) we get

$$I_0(y^{\nu}z^{1-\nu})I_0(y^{1-\nu}z^{\nu}) \le I_0(y)I_0(z),$$
 (5.132)

for $y, z, y^{\nu}z^{1-\nu}, y^{1-\nu}z^{\nu} \in (0,1)$ and $\nu \in [0,1]$, where

$$I_0(x) = \sum_{k=0}^{\infty} \frac{x^{2k}}{4^k (k!)^2}.$$
 (5.133)

If in (5.131) we choose $\alpha = \frac{1}{2}$, then we we obtain

$$\sinh\left(y^{\nu}z^{1-\nu}\right)\sinh\left(y^{1-\nu}z^{\nu}\right) \le \sinh\left(y\right)\sinh\left(z\right),\tag{5.134}$$

for $y, z, y^{\nu}z^{1-\nu}, y^{1-\nu}z^{\nu} \in (0,1)$ and $\nu \in [0,1]$. If we take $\alpha = 3/2$, then we get from (5.131) that

$$[y^{\nu}z^{1-\nu}\cosh(y^{\nu}z^{1-\nu}) - \sinh(y^{\nu}z^{1-\nu})]$$

$$\times [y^{1-\nu}z^{\nu}\cosh(y^{1-\nu}z^{\nu}) - \sinh(y^{1-\nu}z^{\nu})]$$

$$\leq [y\cosh(y) - \sinh(y)][z\cosh(z) - \sinh(z)], \qquad (5.135)$$

for $y, z, y^{\nu}z^{1-\nu}, y^{1-\nu}z^{\nu} \in (0,1)$ and $\nu \in [0,1]$.

Remark 126 For $\alpha = -1/2$ and -3/2 in (5.131), we get the dual results, namely

$$\cosh\left(y^{\nu}z^{1-\nu}\right)\cosh\left(y^{1-\nu}z^{\nu}\right) \le \cosh\left(y\right)\cosh\left(z\right) \tag{5.136}$$

and

$$[y^{\nu}z^{1-\nu}\sinh(y^{\nu}z^{1-\nu}) - \cosh(y^{\nu}z^{1-\nu})]$$

$$\times [y^{1-\nu}z^{\nu}\sinh(y^{1-\nu}z^{\nu}) - \cosh(y^{1-\nu}z^{\nu})]$$

$$\leq [y\sinh(y) - \cosh(y)][z\sinh(z) - \cosh(z)]$$
(5.137)

respectively, for $y, z, y^{\nu}z^{1-\nu}, y^{1-\nu}z^{\nu} \in (0,1) \text{ and } \nu \in [0,1].$

Corollary 127 If $I_{\alpha}(x)$ is the modified Bessel function of the first kind, then for any $a, c, \alpha \in \mathbb{R}$ we have

$$\left(\frac{c}{a}\right)^{\frac{a\left[I_{\alpha-1}(a)+I_{\alpha+1}(a)\right]}{2I_{\alpha}(a)}} \leq \frac{I_{\alpha}\left(c\right)}{I_{\alpha}\left(a\right)} \leq \left(\frac{c}{a}\right)^{\frac{c\left[I_{\alpha-1}(c)+I_{\alpha+1}(c)\right]}{2I_{\alpha}(c)}}.$$
(5.138)

In particular, if $\alpha = 0$ in (5.138), then for any $a, c \in \mathbb{R}$, we get

$$\left(\frac{c}{a}\right)^{\frac{aI_1(a)}{I_0(a)}} \le \frac{I_0(c)}{I_0(a)} \le \left(\frac{c}{a}\right)^{\frac{cI_1(c)}{I_0(c)}},$$
(5.139)

where

$$I_1(z) = \sum_{k=0}^{\infty} \frac{1}{k! (k+1)!} \left(\frac{z}{2}\right)^{2k+1}.$$
 (5.140)

If in (5.138), we choose $\alpha = 1/2$, then we obtain

$$\left(\frac{c}{a}\right)^{a \coth(a) - \frac{1}{2}} \le \frac{\sinh(c)}{\sinh(a)} \le \left(\frac{c}{a}\right)^{c \coth(c) - \frac{1}{2}},\tag{5.141}$$

for any $a, c \in \mathbb{R}$.

Remark 128 For $\alpha = -1/2$ in (5.138), we get the dual result, namely

$$\left(\frac{c}{a}\right)^{a\tanh(a)-\frac{1}{2}} \le \frac{\cosh\left(c\right)}{\cosh\left(a\right)} \le \left(\frac{c}{a}\right)^{c\tanh(c)-\frac{1}{2}},\tag{5.142}$$

for any $a, c \in \mathbb{R}$.

Other inequalities related to the Bessel and Modified Bessel functions of the first kind for further reading can be found in the literature, see for example ([50], [227], [331], [447]) and the references which are cited therein.

Part II

SOME INEQUALITIES INVOLVING ANALYTIC AND UNIVALENT FUNCTIONS

Chapter 6

Elementary Theory of Univalent Functions

This chapter is of an introductory nature and gives the necessary background to Univalent Function Theory. Univalent functions are studied by many, not only in the area of Geometric Function Theory, but also in the classical studies of Complex Analysis as a whole.

Some fundamental concepts in the field of Univalent Function Theory, such as a simply connected domain, analyticity, conformality property, normalization conditions of analytic functions, and the Riemann Mapping Theorem, are given in Section 6.1. In Section 6.2, we introduce one of the most important subclasses of analytic functions, which is called the class \mathcal{P} of all analytic functions with a positive real part. Some important and useful results related to this class are also given. The class S of analytic, univalent and normalized functions in a unit disk, and some of its well-known subclasses such as starlike, convex, close-to-convex, etc., are briefly discussed in Section 6.3 and Section 6.4. Some important properties of functions in these classes are given for completeness.

In Section 6.5, we state some of the classical results related to the class S. One of the famous differential operators, which is known as the Sălăgean differential operator, is mentioned at the end of this chapter. Several subclasses of S, which are characterized by this operator, are mentioned as well.

Some of the well-known results are mostly adopted from the standard texts

of Ahlfors [6], Duren [144], Pommerenke [357], Graham and Kohr [170], Kühnau ([265], [266]) and other references, which are cited therein. Further reading and some additional results on this subject can also be found in the wide variety of classical and modern textbooks devoted to the Univalent Functions Theory and related topics, including ([5], [74], [169], [221], [262], [330], [386], [406]).

6.1 Basic Concepts

In a complex plane, an open set is said to be *connected* if every two points in the set can be joined by a continuous piecewise smooth curve (or polygonal path) that lies entirely in the set. An open connected set is a *domain*, while, a domain together with some, none or all of its boundary points, is often called a region. A domain E is said to be simply connected if it has the property that any simple closed curve¹ γ (or Jordan curve), which is parametrized by z(t) = x(t) + iy(t) for $t \in [a, b], z \in E$, lies completely in E. This means that the domain interior to γ lies wholly in E [5, p. 77-80]. In a simple language, a simply connected domain is one which is free of holes or cuts in its interior.

The open unit disk D, which is defined by (2.3), is one of the most important examples of a simply connected domain in \mathbb{C} . It plays an important role in the study of Univalent Function Theory, which is primarily concerned with analytic and univalent functions on this domain.

A set Λ_0 in a complex plane is called *starlike* (star domain or star-shaped) with respect to the fixed point $x_0 \in \Lambda_0$, for every $x \in \Lambda_0$, the line segment joining the point x_0 to x lies entirely in the set Λ_0 . Geometrically, for any $x_0, x \in \Lambda_0$ with x_0 is a fixed point, then $tx + (1 - t) x_0 \in \Lambda_0$, for every $t \in [0, 1]$. In the case if $x_0 = 0$, then we say that the set Λ_0 is a starlike with respect to the origin. The convex set can be described in a similar way. A set $\Lambda_1 \in \mathbb{C}$ is said to be convex if it is starlike with respect to each of its points; that is, the line segment joining any two points $x, y \in \Lambda_1$.

A complex-valued function $f: \Delta \to \mathbb{C}$ of a complex variable is differentiable

¹A curve is said to be simple if it does not cross itself. It is called a simple closed curve if it closed and it does not cross itself except at the end points.

at a point $z_0 \in \Delta \subset \mathbb{C}$ if it has a derivative at z_0 , i.e.,

$$f'(z_0) = \lim_{z \to z_0} \frac{f(z) - f(z_0)}{z - z_0}.$$
 (6.1)

A function f(z) is said to be analytic at the point $z_0 \in \Delta$ if it is differentiable at every point in some neighborhood of $z_0 \in \Delta$. The function f(z) is to be said analytic on the domain Δ provided that it is analytic at every point of $z \in \Delta$. We say that f(z) is an entire function if is analytic on the whole complex plane.

One of the most important facts of complex analysis is that, if f(z) is an analytic function on a certain domain in a complex plane, then it must have derivatives of all orders at z_0 in its domain, and f(z) has a Taylor series expansion, which converges in an open disk centered at z_0 (see [144, p. 2], [166, p. 33]).

Theorem 129 ([166, p. 33]) Suppose that the power series $\sum_{k=0}^{\infty} a_k z^k$ has radius of convergence R, R > 0. For |z| < R, let $f(z) = \sum_{k=0}^{\infty} a_k z^k$. Then, the power series $\sum_{k=0}^{\infty} k a_k z^k$ has radius of convergence R and $f'(z) = \sum_{k=0}^{\infty} k a_k z^k$ for all |z| < R. Thus, a power series defines a function, which is analytic in its disk of convergence.

This means that every analytic function is locally represented by its convergent power series in an open disk. Explicitly, we have [144, p. 2]

$$f(z) = \sum_{n=0}^{\infty} a_n (z - z_0)^n,$$
 (6.2)

which converges for all z sufficiently close to z_0 . More precisely, if f(z) is analytic in the disk $D(z_0, R)$, then the Taylor series expansion (6.2) converges to f(z) for all $z \in D(z_0, R)$. Conversely, if the power series defined in (6.2) converges for every $z \in D(z_0, R)$, then the series (6.2) represents a function that is analytic on $D(z_0, R)$.

Besides the analyticity, the *conformality* of functions is another remarkable geometric property that must be possessed by the analytic complex-valued functions. A function f, which is analytic and univalent on D, also maps conformally

the unit disk D onto the another domain in \mathbb{C} . In the other words, a complex-valued function is conformal if and only if it is analytic, and it has everywhere nonzero derivative on its domain. In particular, we have the following theorem [5, p. 259] (see also [44, p. 170]).

Theorem 130 ([5, p. 259]) If the function f(z) is analytic at $z_0 \in D$ and $f'(z_0) \neq 0$, then f(z) is conformal at z_0 . If f(z) is analytic in D, then f(z) is conformal at every point in D. The converse is also valid.

The conformal mapping, which preserves the magnitude and the orientation of the angles, is also called the conformal transformation, angle-preserving transformation or biholomorphic maps, with its applications found in many areas of physics and engineering, electronics, medicines and other branches of applied mathematics (see for instance [153], [238], [245], [402]).

Riemann Mapping Theorem is one of the remarkable results in the Geometric Function Theory, which asserts that any simply connected domain in \mathbb{C} can be mapped conformally onto the unit disk D (see [144, p. 11], [170, p. 5], [357, p. 10]). More precisely, we have (see also [6, p. 222])

Theorem 131 ([144, p. 11]) Let U be a simply connected domain, which is a proper subset of a complex plane. Let ζ_0 be a given point in U. Then, there exists a unique analytic and one-to-one function $f: U \to D$, which maps U conformally onto the unit disk D, and has the properties $f(\zeta_0) = 0$, $f'(\zeta_0) > 0$.

This theorem guarantees the existence of a conformal mapping, which maps the simply connected domain D onto any simply connected domain in \mathbb{C} . We note that since the inverse image of a conformal mapping is also conformal, then the Riemann Mapping Theorem implies that any two simply connected domains are conformally equivalent. That is, if $D \subsetneq \mathbb{C}$ and $U \not\subseteq \mathbb{C}$ are simply connected with $z \in D$ and $w \in U$, then there exists a unique conformal transformation $f: D \to U$ with f(z) = w and f'(z) > 0.

An analytic function w = f(z) that maps conformally a simply connected domain onto another domain in \mathbb{C} is not unique. In view of the Riemman Mapping Theorem, the function f(z) needs to be unique. Therefore, such functions need to satisfy the conditions for uniqueness, which are called *normalization*.

The standard normalization conditions of analytic functions in a complex plane are that, for $f: D_1 \to D_2$, where D_1 and D_2 are simply connected domains in \mathbb{C} , then the function f(z) is called normalized such that [5, p. 261]

$$f(0) = 0$$
 and $f'(0) = 1$. (6.3)

Other normalizations are possible, but the conditions in (6.3) are the most usual and the ones that we will be used throughout this study.

The next section discusses one of the important classes of analytic functions, which is called the class \mathcal{P} consisting of all analytic functions with a positive real part.

6.2 Functions With a Positive Real Part

The analytic functions, which map the open unit disk D onto the right halfplane, are of particular interest in the study of Univalent Function Theory. In this section, we introduce the class \mathcal{P} of analytic functions in the open unit disk D having a positive real part. Let us denote $\mathcal{H}(D)$, the class of all analytic functions in the unit disk D.

Definition 132 Let \mathcal{P} be the family of all analytic functions. Then,

$$\mathcal{P} = \{ p \in \mathcal{H}(D) : \text{Re}[p(z)] > 0, \ p(0) = 1, \ z \in D \}.$$
 (6.4)

The function in the class \mathcal{P} is often called the *Carathédory function* [84]. It can be noted that all functions in the class \mathcal{P} can be represented by the following power series:

$$p(z) = 1 + \sum_{k=1}^{\infty} p_k z^k, \tag{6.5}$$

for all $z \in D$, which is obviously that p(0) = 1. Due to Herglots [194], the family \mathcal{P} is characterized by the set of functions p(z), where they are represented by the Herglotz formula as follows:

$$p(z) = \int_{0}^{2\pi} \left(\frac{1 + e^{-it}z}{1 - e^{-it}z} \right) d\mu(t), \tag{6.6}$$

for $z \in D$, $\mu(t)$ is a real-valued nondecreasing function on $[0, 2\pi]$ and $\mu(2\pi) - \mu(0) = 1$. The most important example of a function in the class \mathcal{P} is called the *Möbius transform*, which is of the form

$$m_0(z) = \frac{1+z}{1-z} = 1 + \sum_{n=1}^{\infty} 2z^n.$$
 (6.7)

The function (6.7), which maps the unit disk D one-to-one onto the right halfplane, also provides one of the extremal functions of this class.

The following properties might be useful in the subsequent chapters (see [84], [144, p. 41], [240], [284], [357, p. 41]), with the first lemma due to Carathéodory [84]:

Lemma 133 ([84]) If $p(z) \in \mathcal{P}$ and p(z) is defined by the series (6.5), then $|p_n| \leq 2$ for $n \in \{1, 2, 3, ...\}$. Equality is attained if and only if p(z) is the Möbius function (6.7).

Lemma 134 Let $p(z) \in \mathcal{P}$ and be of the form (6.5). Then,

$$\left| p_2 - \frac{p_1^2}{2} \right| \le 2 - \frac{\left| p_1 \right|^2}{2}.$$
 (6.8)

This result is sharp, with equality holding for the function

$$p(z) = \frac{1 + \frac{1}{2} (p_1 + \xi \overline{p_1}) z + \xi z^2}{1 - \frac{1}{2} (p_1 - \xi \overline{p_1}) z - \xi z^2}, \quad |\xi| = 1.$$
 (6.9)

Lemma 135 (see [180]) Let $p(z) \in \mathcal{P}$ and be of the form (6.5). Then,

(i)
$$|p_n| \leq 2$$
, for $n \geq 1$,

(ii)
$$|p_2 - \mu p_1^2| \le 2 \max\{1, |1 - 2\mu|\}$$
, for any $\mu \in \mathbb{C}$. (6.10)

6.3 Analytic and Univalent Functions

A single-valued function $f: D \to \mathbb{C}$ is said to be $univalent^2$ in a domain $D \subset \mathbb{C}$ if it never takes the same value twice: that is, $f(z_1) \neq f(z_2)$ for all pairs of distinct points z_1 and z_2 in D. In other words, f(z) is one-to-one (or injective) mapping of D onto another domain in a complex plane [144, p. 26]. A function f(z) is called *locally univalent* at a point $z_0 \in D$, if it is univalent in some neighborhood of $z_0 \in D$. For an analytic function f(z), the condition $f'(z_0) \neq 0$ is equivalent to the local univalence at z_0 [6, p. 131]. The locally univalent function is also known as conformal mapping because of its angle and sense preserving property (see [6, p. 10]).

As mentioned in the previous section, an analytic function f(z) defined on the unit disk D has a Taylor series representation given by (6.2). Thus, we may assume without loss of generality that the series (6.2) is centered at the origin, and has been normalized by the conditions (6.3). Hence, each an analytic function has the power series of the form

$$f(z) = z + \sum_{k=2}^{\infty} a_k z^k,$$
 (6.11)

for $z \in D$ and $a_k \in \mathbb{C}$, $k \in \{2, 3, \ldots\}$.

Let A denote the subclass of $\mathcal{H}(D)$ consisting of all analytic functions of the form (6.11), which are analytic and normalized by the conditions (6.3). We denote by S, the subclass of A consisting of all analytic, univalent and normalized functions in the unit disk D, and f(z) has the form (6.11), i.e.,

$$S = \left\{ f \in A : f(z) = z + \sum_{n=2}^{\infty} a_n z^n, \ z \in D \right\}.$$

Remark 136 Throughout this study, we shall be primarily concerned with the function f(z) in the class A and S, and of the form (6.11), which are analytic and univalent in the unit disk D, and normalized by the conditions (6.3).

Example 137 The leading example of a function in the class S is called the Koebe function, which is given by

²The terminologies such as 'simple' or 'Schlicht' are also used for univalent.

$$k(z) = \frac{z}{(1-z)^2} = \frac{1}{4} \left(\frac{1+z}{1-z}\right)^2 - \frac{1}{4} = \sum_{n=1}^{\infty} nz^n, \ z \in D.$$
 (6.12)

This function (6.12) maps the unit disk D, one-to-one and conformally onto the entire complex plane except for the negative real axis from -1/4 to ∞ . The functions

$$e^{-i\beta}k_0\left(e^{i\beta}z\right) = \frac{z}{\left(1 - e^{i\beta}z\right)^2} = \sum_{n=1}^{\infty} ne^{i(n-1)\beta}z^n,$$
 (6.13)

also belong to S and they are referred to as the rotations of Koebe function.

Example 138 Other examples of functions belonging to the class S include

- (a) The identity map, f(z) = z.
- (b) f(z) = z/(1-z), which maps the unit disk D conformally onto the halfplane, Re(z) > -1/2.
- (c) $f(z) = \frac{1}{2} \log [(1+z)/(1-z)]$, which maps the unit disk D onto the horizontal strip, $-\pi/4 < \text{Im}(z) < \pi/4$.

6.4 Subclasses of Analytic and Univalent Functions

In this section, we briefly discuss some of the well-known subclasses S, namely the starlike, convex and close-to-convex functions. These functions are defined by geometrical considerations, but they are very useful for analytic characterizations as well. The first subclass of S that is worthy in the study, is the class of starlike functions.

The class of starlike and convex functions: A function $f(z) \in S$ is said to be *starlike* in D with respect to origin if the image of D under f, i.e., f(D) with $w = 0 \in f(D)$, is a starlike domain: that is, the line segment connecting the point w = 0 to any point of f(D) lies entirely in f(D). Similarly, a function

 $f(z) \in S$ is said to be *convex* in D if the image f(D) is a convex domain (see [357, p. 44]). We shall denote these classes of functions by S^* and C, which are starlike with respect to the origin and convex in the unit disk D, respectively. The class S^* was first studied by Alexander [15], later by Gronwall [174], Nevanlinna [332] and others (see [376], [319]).

The necessary and sufficient conditions for functions $f \in S$ to be starlike and convex have been given by Robertson [376] as follows (see also [144, p. 41], [319], [357, p. 42]):

Theorem 139 ([376]) A function $f \in S$ of the form (6.11) is starlike in the unit disk D if and only if it satisfies

$$z\frac{f'(z)}{f(z)} \in \mathcal{P}$$
, i.e., $\operatorname{Re}\left\{z\frac{f'(z)}{f(z)}\right\} > 0, z \in D.$ (6.14)

Theorem 140 ([376]) A function $f \in S$ of the form (6.11) is convex in the unit disk D if and only if it satisfies

$$1 + z \frac{f''(z)}{f'(z)} \in \mathcal{P}$$
, i.e., $\operatorname{Re} \left\{ 1 + z \frac{f''(z)}{f'(z)} \right\} > 0, z \in D.$ (6.15)

The representation formula for functions in the class of starlike, $f(z) \in S^*$ of the form (6.11) is given by

$$f(z) = z \exp\left[\int_0^{2\pi} \log\left(\frac{1}{1 - e^{-i\tau}z}\right) du(\tau)\right], \ z \in D, \tag{6.16}$$

for some increasing function $u(\tau)$ with $u(2\pi) - u(0) = 1$ [357, p. 43].

Alexander [15] observed that there exists a close connection between the convex and starlike functions: that is, a function $f \in S$ maps the unit disk D onto a convex domain if and only if zf'(z) maps the unit disk D onto a starlike domain: i.e., $f(z) \in C$ if and only if $zf'(z) \in S^*$, $z \in D$. Thus, we have the inclusion $C \subset S^* \subset S$. Note that the Koebe function defined by (6.12) is starlike with respect to origin but not convex. The function f(z) given by (1.21) is one of the examples of the convex functions, which maps the unit disk D onto a half plane, and it plays a central role in the class C.

In [376], Robertson also introduced the classes $S^*(\alpha)$ and $C(\alpha)$ of starlike functions of order α and convex functions of order α , for some α ($0 \le \alpha < 0$), which are defined by

$$S^*\left(\alpha\right) = \left\{ f \in S : \operatorname{Re}\left\{z\frac{f'\left(z\right)}{f\left(z\right)}\right\} > \alpha, \ 0 \le \alpha < 1, \ z \in D \right\}, \tag{6.17}$$

$$C(\alpha) = \left\{ f \in S : \operatorname{Re} \left\{ 1 + z \frac{f''(z)}{f(z)} \right\} > \alpha, \ 0 \le \alpha < 1, \ z \in D \right\}.$$
 (6.18)

In particular, we have $S^*(\alpha) \subset S^*(0) = S^* \subset S$ and $C(\alpha) \subset C(0) = C \subset S$. Strochhäcker [411] proved that every convex function is starlike of order 1/2.

Further, an interesting unification of functions in the classes S^* and C was provided by Miller, Mocannu and Reade in [309]. They introduced the class \mathcal{C}_{λ} of λ -convex functions (or λ -starlike functions) via the linear combinations of the representations of starlike and convex functions studied by Moccanu [319]. In [309], Miller, Mocannu and Reade showed that if f(z) is an analytic function in the unit disk D with $f(z)f'(z)/z \neq 0$, and λ is a real number, then f(z) is said to be in the class \mathcal{C}_{λ} if and only if

$$\operatorname{Re}\left\{ (1-\lambda)\frac{zf'(z)}{f(z)} + \lambda\left(1 + \frac{zf''(z)}{f'(z)}\right) \right\} > 0, \tag{6.19}$$

for $z \in D$. This more general class C_{λ} reduces to the standard starlike functions S^* and convex functions C when $\lambda = 0$ and 1, respectively.

Meanwhile, in [328], Nasr and Aouf introduced the class of starlike and convex functions of complex order b ($b \neq 0$). Such classes of functions are denoted by S_b^* and C_b , and they are defined as follows:

$$S_b^* = \left\{ f \in S : \operatorname{Re} \left\{ 1 + \frac{1}{b} \left(\frac{zf'(z)}{f(z)} - 1 \right) \right\} > 0 \ (z \in U) \right\}, \quad (6.20)$$

$$C_b = \left\{ f \in S : \text{Re} \left\{ 1 + \frac{1}{b} \left(\frac{zf''(z)}{f'(z)} \right) \right\} > 0 \ (z \in U) \right\},$$
 (6.21)

where $f(z)/z \neq 0$.

The class of close-to-convex functions: Another important subclass of S is the class of close-to-convex functions, which was introduced by Kaplan [235] in 1952. The necessary and sufficient condition of functions in this class is stated in the following. A function $f(z) \in S$ and of the form (6.11) is said to be close-to-convex in D if and only if there exists a convex function g(z) in D such that

$$\operatorname{Re}\left\{\frac{f'(z)}{g'(z)}\right\} > 0, \ z \in D. \tag{6.22}$$

Since $h(z) = zg'(z) \in S^*$ with $g \in S$, then (6.22) is equivalent with

$$\operatorname{Re}\left\{\frac{zf'(z)}{h(z)}\right\} > 0, \ z \in D. \tag{6.23}$$

We shall denote by \mathcal{K} , the class of all functions, which are close-to-convex in D. It is clear that every convex function is close-to-convex, and every starlike function is close-to-convex as well. In [357, p. 51], Pommerenke showed that every close-to-convex function is univalent in D. Therefore, it can be seen that the proper inclusions $\mathcal{C} \subset S^* \subset \mathcal{K} \subset S$ hold.

The more general class $\mathcal{K}(\alpha, \beta)$ of close-to-convex functions of order α and type β are characterized as follows. A function $f(z) \in S$ is in the class $\mathcal{K}(\alpha, \beta)$ if and only if

$$\operatorname{Re}\left\{\frac{zf'(z)}{h(z)}\right\} > \beta, \ z \in D, \tag{6.24}$$

for $h(z) \in S^*(\alpha)$, $0 \le \alpha < 1$, $0 \le \beta < 1$.

The class K and its generalizations have been studied by numerous researchers (see for instance [108], [109], [110], [164], [165], [206], [219], [292]) and recently by ([233], [261], [390], [430], [434], [446]).

The class of Bazilević functions: A more general class of analytic functions was introduced by Bazilevič [54] in 1955, which is called the class of *Bazilevič* functions of type α , and is denoted by $\mathcal{B}(\alpha)$, $\alpha > 0$. The functions in this class are characterized by

$$\operatorname{Re}\left(\frac{zf'(z)f(z)^{\alpha-1}}{g(z)^{\alpha}}\right) > 0 \ (z \in D), \tag{6.25}$$

for $f(z) \in A$ and of the form (6.11), where $g(z) \in S^*$ in D, $\alpha > 0$. The class of Bazilevič functions has been studied by many authors with Thomas (see [418], [419]), Halim [182], Noor [336], Singh [400], Keogh and Miller [241] among others. We note that with specific choices of the associated parameter, the class $\mathcal{B}(\alpha)$ reduces to the well-known classes of starlike, convex and close-to-convex functions. Furthermore, if in (6.25) we choose the starlike function g(z) = z, then we have the family $\mathcal{B}_1(\alpha)$ of function satisfying [400]

$$\operatorname{Re}\left(\frac{zf'(z)f(z)^{\alpha-1}}{z^{\alpha}}\right) > 0, \ z \in D.$$
(6.26)

A more general class $\mathcal{B}(\alpha, \beta)$ of Bazilevič functions of order β and type α is defined by

$$\mathcal{B}(\alpha,\beta) = \left\{ f \in A : \operatorname{Re}\left(\frac{zf'(z)f(z)^{\alpha-1}}{g(z)^{\alpha}}\right) > \beta \quad (z \in D) \right\},\tag{6.27}$$

for some α ($\alpha > 0$) and β ($0 \le \beta < 1$), where $g(z) \in S^*$. In particular, if $\beta = 0$ and 1, then we have $\mathcal{B}(\alpha, 0) = \mathcal{B}(\alpha)$ and $\mathcal{B}(1, 0) = \mathcal{B}(1) = \mathcal{K}$. In particular, we denote by $\mathcal{B}_1(\alpha, \beta)$ the subclass of $\mathcal{B}(\alpha, \beta)$ for which $g(z) = z \in S^*$, namely

$$\mathcal{B}_1(\alpha,\beta) = \left\{ f \in A : \operatorname{Re}\left(\frac{zf'(z)f(z)^{\alpha-1}}{z^{\alpha}}\right) > \beta, \ z \in D \right\}, \tag{6.28}$$

for some α ($\alpha > 0$) and β ($0 \le \beta < 1$). Also, we note that $\mathcal{B}(0,0) = \mathcal{B}_1(0,0) = S^*$, $\mathcal{B}(0,\beta) = \mathcal{B}_1(0,\beta) = S^*(\beta)$ and that $\mathcal{B}_1(1,\beta)$ is a subclass of A consisting of all functions for which $\text{Re}\{f'(z)\} > \beta$, for $z \in D$.

Other Subclasses: Some other well-known subclasses of analytic and univalent functions that have been studied repeatedly by many authors, are given in the following:

$$S_0 = \left\{ f(z) \in A : \operatorname{Re}\left\{ \frac{f(z)}{z} \right\} > 0, \ z \in D \right\}, \tag{6.29}$$

$$B(\beta) = \left\{ f(z) \in A : \operatorname{Re}\left\{ \frac{f(z)}{z} \right\} > \beta, \ 0 \le \beta < 1, \ z \in D \right\}, \quad (6.30)$$

$$\delta(\beta) = \{ f(z) \in A : \text{Re} \{ f'(z) \} > \beta, \ 0 \le \beta < 1, \ z \in D \}.$$
 (6.31)

6.5 Some Classical Results

One of the most classical results in the Theory of Univalent Functions is Bieber-bach Conjecture, proposed by Bieberbach [67] in 1961. This conjecture states the upper bounds for the coefficients of functions in the class S. Bieberbach was the first who established the bound for the second coefficient of functions in the class S: that is, $|a_2| \leq 2$ for $f(z) \in S$ and of the form (6.11). He conjectured that if $f(z) \in S$, then the coefficients a_n of f(z) satisfied $|a_n| \leq n$, for all $n \geq 2$, with the equality holding if and only if f(z) is the Koebe function (6.12) or one of its rotations (6.13). For many years, this conjecture has stood as a challenge to many mathematicians and has inspired the development of important new methods in complex analysis. However, the Bieberbach Conjecture was completely proved by de Brange [75] in 1985.

Since then, the estimation of the coefficients $|a_n|$, for $n \in \{2, 3, 4, ...\}$ have been investigated for various subclasses of S, in order to provide some basic properties of univalent functions (see [18], [58], [83], [115], [236], [294], [357, p. 53], [371], [379]). For instance, the following results showed that the Bieberbach conjecture also holds for functions in the subclass of S.

Theorem 141 ([357, p. 46]) If $f \in S^*$ of the form (6.11), then $|a_n| \leq n$. Equality occurs if and only if f(z) is the rotation of Koebe function.

Theorem 142 ([371]) If $f \in K$ of the form (6.11), then $a_n \leq n$.

The necessary and sufficient condition for functions in the starlike and convex classes that has been proved by Alexander [15], is given as follows (see also [372]):

Theorem 143 ([15]) Let f(z) be of the form (6.11). If $\sum_{n=2}^{\infty} n |a_n| \le 1$, then f(z) is univalent in |z| < 1 and maps that region onto a region that is starlike

with respect to the origin. If $\sum_{n=2}^{\infty} n^2 |a_n| \le 1$, then f(z) is univalent in |z| < 1 and maps that region onto a convex region.

Silverman [398] provided the following properties of the starlikeness and convexity of functions in the class $S^*(\alpha)$ and $C(\alpha)$.

Theorem 144 ([398]) Let f(z) be of the form (6.11). If $\sum_{n=2}^{\infty} (n-\alpha) |a_n| \le 1 - \alpha$, $0 \le \alpha < 1$, then $f \in S^*(\alpha)$. If $\sum_{n=2}^{\infty} n(n-\alpha) |a_n| \le 1 - \alpha$, $0 \le \alpha < 1$, then $f \in C(\alpha)$.

Another closely related problem in the Theory of Univalent Functions is the determination of the sharp upper bounds for the nonlinear functional $|a_3 - ua_2^2|$, for both real and complex values of the parameter u. It is popularly known as the Fekete-Szegő problem, and also arise in the investigation of univalency of the analytic functions. The Fekete-Szegő problem has its origin in a conjecture of Littlewood and Parley [289] in 1932 that the bound on the coefficients $|a_3 - a_2^2|$ of an odd univalent function is 1. However, this result was disproved by Fekete and Szegő [152] (see also [144, p. 104]). Since them, the Fekete-Szegő problem has continued to receive attention of large number of researchers in Geometric Function Theory.

In 1933, Fekete and Szegő [152] first obtained the sharp upper bounds of the functional $|a_3 - a_2^2|$ for $f \in S$, for each fixed u in the interval $0 \le \lambda < 1$. The result is stated as follows:

Theorem 145 ([152]) If f(z) in S and of the form (6.11), then

$$|a_3 - ua_2^2| \le \begin{cases} 3 - 4u, & \text{for } u \le 0; \\ 1 + 2\exp\left(\frac{-2u}{1 - u}\right), & \text{for } 0 \le u < 1; \\ 4u - 3, & \text{for } u \ge 1. \end{cases}$$
 (6.32)

This bound is sharp when u is real.

Pfluger (see [353], [354]) has considered the Fekete-Szegő problems for $f \in S$ when u is complex numbers. In the cases of functions in the classes S^* , C and \mathcal{K} of starlike, convex and close-to-convex functions, the above inequality (6.32) has been improved by several researchers. For instance, Keogh and Merker [240] proved the Fekete-Szegő problem for the close-to-convex functions as follows:

Theorem 146 ([240]) For $f(z) \in \mathcal{K}$ and of the form (6.11), we have

$$|a_{3} - ua_{2}^{2}| \leq \begin{cases} 3 - 4u, & \text{if } u \leq \frac{1}{3}, \\ \frac{1}{3} + \frac{4}{9u}, & \text{if } \frac{1}{3} < u \leq \frac{2}{3}, \\ 1, & \text{if } \frac{2}{3} < u \leq 1, \\ 4u - 3, & \text{if } u \geq 1, \end{cases}$$

$$(6.33)$$

where for each u real, there is a function in K such that the equality holds.

They also proved the Fekete-Szegö problem for functions in the class of star-like of order α as follows:

Theorem 147 ([240]) Let $f \in S^*(\alpha)$, $0 \le \alpha < 1$ and of the form (6.11), then for u real

$$\left| a_3 - ua_2^2 \right| \le (1 - \alpha) \max \left\{ 1, \left| 3 - 2\alpha - 4u \left(1 - \alpha \right) \right| \right\}.$$
 (6.34)

The study of the Fekete-Szegő functional has continued to receive attention from many researchers for more general classes of functions with varied success. For instance, Koepf (see [254], [255]) established the Fekete-Szegő inequalities for the class $\mathcal{K}(\alpha)$, while London [292] extended the results of Koepf to the strongly class of close-to-convex functions of order α . Meanwhile, Darus and Thomas (see [108], [109], [110]) solved the Fekete-Szegő problems for the strongly class of close-to-convex functions and the class M^{α} of α -logarithmic convex functions. A similar result for the close-to-convex functions of order α and type β has been proved by Ibrahim and Darus [206]. For other results related to the Fekete-Szegő problem, see ([63], [155], [234], [314], [347], [370], [392]) and the references which are cited therein.

The remainder of this chapter discusses the Sălăgean differential operator and its related subclasses of functions defined involving this operator.

6.6 Sălăgean Differential Operator and Related Subclasses

For functions f(z) belonging to the class A of analytic functions in the unit disk D, Sălăgean [383] was introduced the differential operator $D^n f$, $n \in \mathbb{N}_0$, which is popularly known as Sălăgean differential operator. In this section, we discuss this well-known differential operator and its generalization, and provide some subclasses of analytic functions that are characterized involving this operator.

Definition 148 ([383]) For a function $f(z) \in A$, $n \in \mathbb{N}_0$, the Sălăgean differential operator $D^n: A \to A$, is defined by

$$D^{n} f(z) = D\left[D^{n-1} f(z)\right] = z\left[D^{n-1} f(z)\right]', \tag{6.35}$$

where

$$D^{0}f(z) = f(z)$$
 and $D^{1}f(z) = zf'(z)$. (6.36)

The operator D^n has been employed by various authors to defined several subclasses of analytic and univalent functions (see for instance [41], [69], [180], [232], [339], [383], [343]).

The following operator was introduced by Al-Oboudi [17], which generalized the Sălăgean differential operator.

Definition 149 ([17]) For a function $f(z) \in A$, $n \in \mathbb{N}_0$ and $\lambda \geq 1$, the generalized Sălăgean operator $D_{\lambda}^n: A \to A$, is defined by

$$D_{\lambda}^{n} f(z) = D_{\lambda} \left[D_{\lambda}^{n-1} f(z) \right]$$
$$= (1 - \lambda) D_{\lambda}^{n-1} f(z) + \lambda z \left[D_{\lambda}^{n-1} f(z) \right]', \tag{6.37}$$

where

$$D_{\lambda}^{0}f\left(z\right) = f\left(z\right), \tag{6.38}$$

$$D_{\lambda}^{1} f(z) = D_{\lambda} f(z) = (1 - \lambda) f(z) + \lambda z f'(z).$$
 (6.39)

We note that $D_1^n f(z) = D^n f(z)$. Some properties of these operators have been investigated by several authors (see for instance [207], [228], [243]).

We observe that for $f(z) \in A$ and of the form (6.11), and applying the operator (6.35) for (6.11), we have

$$D^{n} f(z) = z + \sum_{k=2}^{\infty} k^{n} a_{k} z^{k}, \qquad (6.40)$$

for $z \in D$, $n \in \mathbb{N}_0$. Also, suppose that $\alpha > 0$, then form (6.11) we can write

$$f(z)^{\alpha} = \left(z + \sum_{k=2}^{\infty} a_k z^k\right)^{\alpha}.$$
 (6.41)

Using binomial expansion for the function (6.41), we get

$$f(z)^{\alpha} = z^{\alpha} + \sum_{k=2}^{\infty} A_k(\alpha) z^{\alpha+k-1}, \qquad (6.42)$$

where the coefficients $A_k(\alpha)$, $k \in \{2, 3, ...\}$ with $A_k(1) = a_k$, depend on the coefficients a_k of f(z) and the parameter $\alpha > 0$. Now, if we apply the operator (6.35) for the function (6.42), then we obtain

$$D^{n} f(z)^{\alpha} = \alpha^{n} z^{\alpha} + \sum_{k=2}^{\infty} (\alpha + k - 1)^{n} A_{k}(\alpha) z^{\alpha + k - 1},$$
 (6.43)

for $z \in D$, $\alpha > 0$, $n \in \mathbb{N}_0$, where $D^0 f(z)^{\alpha} = f(z)^{\alpha}$ and $D^1 f(z)^{\alpha} = z [f(z)^{\alpha}]'$.

Similarly, for $f(z) \in A$ and of the form (6.11), we observe that from (6.37) - (6.39)

$$D_{\lambda}^{n} f(z) = z + \sum_{k=2}^{\infty} \left[1 + (k-1)\lambda \right]^{n} a_{k} z^{k}, \tag{6.44}$$

for $z \in D$, $\lambda \ge 1$, $n \in \mathbb{N}_0$. Also, on utilizing the differential operator (6.37) for the function (6.42), we get

$$D_{\lambda}^{n} f(z)^{\alpha} = [1 + (\alpha - 1) \lambda]^{n} z^{\alpha} + \sum_{k=2}^{\infty} [1 + (\alpha + k - 2) \lambda]^{n} A_{k}(\alpha) z^{\alpha + k - 1},$$
 (6.45)

for $\alpha > 0$, $\lambda \ge 1$, $n \in \mathbb{N}_0$, $z \in D$.

Many subclasses of analytic functions, which are characterized involving the Sălăgean differential operator (6.35), have been explored by numerous researchers. Sălăgean [383] used the operator introduced the class S_n , $n \in \mathbb{N}_0$, which generalized the convexity and starlikeness of analytic and univalent functions in the unit disk. He defined that a function $f(z) \in A$ belongs to the class S_n , $n \in \mathbb{N}_0$ if and only if it satisfies the condition

$$\operatorname{Re}\left(\frac{D^{n+1}f(z)}{D^nf(z)}\right) > 0,\tag{6.46}$$

for $z \in D$. We note that $S_0 = S^*$ and $S_1 = C$, that is the standard class of starlike and convex functions, respectively. The class S_n of order α , i.e., $S_n(\alpha)$, $0 \le \alpha < 1$, has been studied by Owa et al. ([349], [350]), Lin and Owa [285] and others.

In 2005, Oros [348] defined the class $M_n(\alpha)$ consisting of all analytic functions in D and satisfying the inequality

$$\operatorname{Re}\left[D^{n}f(z)\right]' > \alpha,\tag{6.47}$$

for some α ($0 \le \alpha < 1$), $z \in D$, $n \in \mathbb{N}_0$. This class has been further studied by Tăut, Oros and Şendruţiu [415]. Halim in [180] introduced and studied the a generalization of analytic functions satisfying (6.26) as follows:

$$\operatorname{Re}\left(\frac{D^{n}f\left(z\right)^{\alpha}}{z^{\alpha}}\right) > 0,$$
 (6.48)

for $\alpha > 0$, $z \in D$. He denoted this class of functions by $\mathcal{B}_n(\alpha)$. Obviously, this class reduces to the class of the Bazilević functions (6.26) with logarithmic growth studied in ([338], [400], [418], [419]), when n = 1. The class $\mathcal{B}_0(\alpha)$ was initiated by Yamaguchi [435].

Opoola [343] further generalized functions defined by geometric condition (6.48) by choosing a real number β for $0 \le \beta < 1$ that will be discussed in details in the following chapter.

Chapter 7

Properties for Certain Subclasses of Analytic Functions

Chapter 7 is devoted to some results concerning analytic, univalent and normalized functions on a unit disk. The class $T_n^{\alpha}(\beta)$, $\alpha > 0$, $0 \le \beta < 1$, $n \in \mathbb{N}_0$, of analytic and univalent functions, which was introduced by Opoola [343] in 1994, is one of the interested subclasses of functions throughout this chapter. The properties of functions in the class $T_n^{\alpha}(\beta)$ have been explored by numerous researchers in the last few decades. In this chapter, we present further properties of functions in this class.

Section 7.1 discusses some known results concerning the class $T_n^{\alpha}(\beta)$ that have been proved by Opoola [343], Opoola et al. [344], Babalola and Opoola [42] and others. In Section 7.2, we provide some other properties of functions in the this class. The properties of functions in the classes $\tilde{T}_n^{\alpha}(\beta)$ and $T_n^{\alpha}(\beta,\lambda)$, $\lambda \geq 1$ are also investigated. The coefficient bounds for functions in the strongly class of Opoola's functions, $\tilde{T}_n^{\alpha}(\beta)$, $\alpha > 0$, $0 < \beta \leq 1$, $n \in \mathbb{N}_0$, have been presented in Section 7.3. Finally, we end this chapter by providing some results on Fekete-Szegő problem for a certain subclass of analytic and univalent functions.

All the results presented in this chapter, are mainly taken from the author's research papers in collaboration with Darus, Dragomir and Joseph (see [210], [211], [212]).

7.1 Introduction

In [343], Opoola introduced the subclass $T_n^{\alpha}(\beta)$ of analytic functions, which are defined involving the Sălăgean differential operator (6.35) as follows (see also [39], [40], [42]):

Definition 150 ([343]) A function $f(z) \in T_n^{\alpha}(\beta)$ if and only if it satisfies the condition

$$\operatorname{Re}\left\{\frac{D^{n}f\left(z\right)^{\alpha}}{\alpha^{n}z^{\alpha}}\right\} > \beta,\tag{7.1}$$

for $\alpha > 0$ is real, $0 \le \beta < 1$, $z \in D$, where D^n , $n \in \mathbb{N}_0$ is the Sălăgean differential operator and the powers in (7.1) meaning principal determinations only.

The geometric condition (7.1) slightly modifies the one given originally in [343] (see also [41], [345]) with the number α^n is considered. The class $T_n^{\alpha}(\beta)$ was studied by Opoola [343] as well as by Babalola [39], Babalola and Opoola (see [40], [41]), Halim [181], Oladipo et al. [341] and others (see [344], [345], [346]). Some interesting properties of functions in this class were established in the literature. For instance, Opoola [343] proved that the class T_n^{α} is a subclass of univalent functions, and the members of the family $T_n^{\alpha}(\beta)$ satisfied the inclusion relations, $T_{n+1}^{\alpha}(\beta) \subset T_n^{\alpha}(\beta)$, $\alpha > 0$, $n \in \mathbb{N}_0$. In [344], Opoola et al. showed that if f(z) belongs to the class $T_n^{\alpha}(\beta)$, then the coefficient bounds are given by

$$|a_2| \leq \frac{2(1-\beta)\alpha^{n-1}}{(\alpha+1)^n} \text{ for } \alpha > 0, \tag{7.2}$$

$$|a_{3}| \leq \begin{cases} 2(1-\beta)\alpha^{n-1} \left[\frac{1}{(\alpha+2)^{n}} - \frac{(\alpha-1)^{n-1}(1-\beta)}{(\alpha+1)^{2n}}\right], & \text{for } 0 < \alpha < 1, \\ \frac{2(1-\beta)\alpha^{n-1}}{(\alpha+2)^{n}}, & \text{for } \alpha \geq 1. \end{cases}$$
(7.3)

The coefficient bounds for $|a_4|$ and $|a_5|$ are investigated as well. Earlier results, Halim [180] provided the coefficient bounds for $|a_n|$, n = 2, 3, 4 for the class

 $T_n^{\alpha}(0) = B_n(\alpha)$. Recently, Babalola and Opoola [42] obtained certain coefficient inequalities and solved the Fekete-Szegö problem for functions in the class $T_n^{\alpha}(\beta)$ as well.

One of the aims of this chapter is to establish some other properties concerning functions in the class $T_n^{\alpha}(\beta)$. Instead of this class, in this chapter, we also studied the subclasses of analytic functions, which are defined as follows.

Definition 151 Let $\widetilde{T}_n^{\alpha}(\beta)$ denote the subclass of A consisting of analytic functions, which satisfy the condition

$$\left| \arg \left\{ \frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} \right\} \right| \le \frac{\beta \pi}{2}, \tag{7.4}$$

for some $\alpha > 0$, $0 < \beta \le 1$, $z \in D$, where D^n , $n \in \mathbb{N}_0$ is the Sălăgean differential operator, and all the index in (7.4) meant principal determination only.

Using the Al-Oboudi differential operator (6.37), we give the definition of a more general class of Opoola's functions.

Definition 152 Let $T_n^{\alpha}(\beta, \lambda)$ denote the subclass of A consisting of analytic functions, which satisfy the inequality

$$\operatorname{Re}\left\{\frac{D_{\lambda}^{n} f\left(z\right)^{\alpha}}{\alpha^{n} z^{\alpha}}\right\} > \beta, \tag{7.5}$$

for some $\alpha > 0$, $0 \le \beta < 1$, $z \in D$ and D_{λ}^{n} , $\lambda \ge 1$, $n \in \mathbb{N}_{0}$ is the Al-Oboudi differential operator.

The class $T_n^{\alpha}(\beta, \lambda)$ serves as a new generalization of many known subclasses of analytic functions in the direction by means of various choices of the parameters involved. For instance, several subclasses of functions are given in the following remark.

Remark 153 (a) We note that $\widetilde{T}_{n}^{\alpha}(1) := T_{n}^{\alpha}(0)$.

- (ii) For $\lambda = 1$ in (7.5), we have $T_n^{\alpha}(\beta, 1) := T_n^{\alpha}(\beta)$, which is the class of functions studied by Opoola [343] and others (see [41], [341], [345]).
- (iii) For $\lambda = 1$, $\beta = 0$ in (7.5), we have $T_n^{\alpha}(0,1) := T_n^{\alpha}(0) = B_n(\alpha)$, which is the class of functions studied by Halim in [180].

- (iv) For $\lambda = 1$, $\beta = 0$, n = 1 in (7.5), we have $T_1^{\alpha}(0, 1) := T_1^{\alpha}(0) = B_1(\alpha)$, which is the class of functions studied by Bazilević [54] and Singh [400].
- (v) For $\alpha = 1$, $\lambda = 1$, $\beta = 0$, n = 0 in (7.5), we have $T_0^{\alpha}(0,1) := T_0^{\alpha}(0) = S_0$, which is the class of functions studied by Yamaguchi in [435].

In order to prove the main results, we shall need the following well-known lemma due to Hayami et al. [188].

Lemma 154 A function $p(z) \in \mathcal{P}$ satisfies the condition

$$\operatorname{Re}\left[p(z)\right] > 0, \ z \in D \tag{7.6}$$

if and only if

$$p(z) \neq \frac{\psi - 1}{\psi + 1}, \ z \in D, \ \psi \in \mathbb{C}, \ |\psi| = 1.$$
 (7.7)

The following result is well-known as $Schwarz\ Lemma$ (see [6, p. 135], [144, p. 3]).

Lemma 155 Let f(z) be an analytic function in the unit disk D, with f(0) = 0 and |f(z)| < 1. Then, $|f'(0)| \le 1$ and $|f(z)| \le |z|$ in D. Strict inequality holds in both estimates unless f(z) is a rotation of the disk: $f(z) = e^{i\theta}z$. If |f(z)| = |z| for some $z \ne 0$, then f(z) = cz, with a constant c of absolute value 1.

7.2 The Properties

In this section, we establish some other properties of functions in the class $T_n^{\alpha}(\beta)$. The properties of functions in the class $\widetilde{T}_n^{\alpha}(\beta)$ and $T_n^{\alpha}(\beta,\lambda)$ are also presented. First, we prove a sufficient condition for functions in the class $T_n^{\alpha}(\beta)$.

Theorem 156 (Ibrahim, Darus, Dragomir and Joseph [212]) A function $f(z) \in A$ is in the class $T_n^{\alpha}(\beta)$ if and only if

$$1 + \sum_{k=2}^{\infty} Q_k z^{k-1} \neq 0, \tag{7.8}$$

where

$$Q_{k} = \frac{(\psi + 1)}{2(1 - \beta)} \left(\frac{\alpha + k - 1}{\alpha}\right)^{n} A_{k}(\alpha), \qquad (7.9)$$

for some $\alpha > 0, \ 0 \le \beta < 1, \ \psi \in \mathbb{C}$ with $|\psi| = 1$.

Proof. From Definition 150, it suggests that there exists a function $p(z) \in \mathcal{P}$ such that

$$\frac{D^n f(z)^{\alpha}}{z^{\alpha}} = \alpha^n \left[\beta + (1 - \beta) p(z) \right], \tag{7.10}$$

for $z \in D$, $\alpha > 0$, $0 < \beta \le 1$, $n \in \mathbb{N}_0$. Upon setting

$$p(z) = \frac{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - \beta}{1 - \beta},$$
(7.11)

then from Lemma 154, we have

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - \beta = \frac{\psi - 1}{\psi + 1}, \tag{7.12}$$

for $z \in D$, $\psi \in \mathbb{C}$ with $|\psi| = 1$, which is equivalent with

$$(\psi + 1) D^{n} f(z)^{\alpha} - [(\psi + 1) \beta + (1 - \beta) (\psi - 1)] \alpha^{n} z^{\alpha} \neq 0.$$
 (7.13)

Substituting (6.43) into (7.13), we get

$$(\psi+1)\left[\alpha^{n}z^{\alpha} + \sum_{k=2}^{\infty} (\alpha+k-1)^{n} A_{k}(\alpha) z^{\alpha+k-1}\right] - \left[(\psi+1)\beta + (1-\beta)(\psi-1)\right] \alpha^{n}z^{\alpha} \neq 0,$$

$$(7.14)$$

which gives us that

$$2(1-\beta)\alpha^{n}z^{\alpha} + \sum_{k=2}^{\infty} (\psi+1)(\alpha+k-1)^{n} A_{k}(\alpha) z^{\alpha+k-1} \neq 0.$$
 (7.15)

Now, dividing both side of (7.15) by $2(1-\beta)\alpha^n z^{\alpha} \neq 0$, we obtain

$$1 + \sum_{k=2}^{\infty} \frac{(\psi+1)}{2(1-\beta)} \left(\frac{\alpha+k-1}{\alpha}\right)^n A_k(\alpha) z^{k-1} \neq 0, \tag{7.16}$$

for any $\psi \in \mathbb{C}$ such that $|\psi| = 1$, $\alpha > 0$, $0 \le \beta < 1$, $z \in D$, $n \in \mathbb{N}_0$.

Corollary 157 A function $f(z) \in A$ is in the class $\mathcal{B}_n(\alpha) := T_n^{\alpha}(0)$ if and only if

$$1 + \sum_{k=2}^{\infty} Q_k z^{k-1} \neq 0,$$

where

$$Q_{k} = \frac{(\psi + 1)}{2} \left(\frac{\alpha + k - 1}{\alpha}\right)^{n} A_{k}(\alpha), \qquad (7.17)$$

for any $\psi \in \mathbb{C}$ with $|\psi| = 1$, $\alpha > 0$, $z \in D$, $n \in \mathbb{N}_0$.

Remark 158 The Corollary 157 has been proved by Singh et al. [401].

The following property of functions in the class $T_n^{\alpha}(\beta)$ is also established.

Theorem 159 (Ibrahim, Darus, Dragomir and Joseph [212]) Let f(z) belongs to the class $T_n^{\alpha}(\beta)$. Then, there exists an analytic function $\phi(z)$ with $|\phi(z)| \leq 1$, $z \in D$ such that

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} = 2\beta - 1 + \frac{2(1-\beta)}{1 - z\phi(z)},\tag{7.18}$$

for $z \in D$, $\alpha > 0$, $0 \le \beta < 1$, $n \in \mathbb{N}_0$.

Proof. Let us define the functions A(z) and B(z) as follows:

$$A(z) = \frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - \beta \tag{7.19}$$

and

$$B(z) = z \left(\frac{A(z) - (1 - \beta)}{A(z) + (1 - \beta)} \right), \tag{7.20}$$

for $\alpha > 0$, $0 \le \beta < 1$, $z \in D$, $n \in \mathbb{N}_0$. Substituting the equation (7.19) into (7.20), we get

$$B(z) = z \left(\frac{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - 1}{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - (2\beta - 1)} \right), \tag{7.21}$$

for any $z \in D$. B(z) is an analytic function for $z \in D$. Also, since f(0) - 0 and f'(0) = 1, we have that B(0) = 0 and |B(z)| < 1, for $z \in D$. Hence, by Schwarz's Lemma (Lemma 155), |B(z)| < |z| for $z \in D$, which gives that

$$\left| \frac{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - 1}{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - (2\beta - 1)} \right| < |z|, \ z \in D, \tag{7.22}$$

or equivalently,

$$\left(\frac{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - 1}{\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - (2\beta - 1)}\right) = z\phi(z),$$
(7.23)

where $\phi(z)$ is analytic, and $|\phi(z)| \leq 1$ for $z \in D$. Therefore

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - 1 = z\phi(z) \frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} - (2\beta - 1) z\phi(z). \tag{7.24}$$

Solving for $D^n f(z)^{\alpha}/\alpha^n z^{\alpha}$, we get

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} = \frac{1 - (2\beta - 1) z\phi(z)}{1 - z\phi(z)} = (2\beta - 1) + \frac{2(1 - \beta)}{1 - z\phi(z)}.$$
 (7.25)

Thus, we obtain the desired result (7.18).

The following result is called the integral representation theorem for functions in the class $T_n^{\alpha}(\beta)$, which provides further property of functions in this class.

Theorem 160 (Ibrahim, Darus, Dragomir and Joseph [212]) Let f(z) belongs to the class $T_n^{\alpha}(\beta)$. Then, there exists an analytic function $\phi(z)$, with $|\phi(z)| \leq 1$, $z \in D$ such that

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} = \frac{1}{\alpha^n z^{\alpha}} \exp \int_0^z \left(\frac{\alpha}{t} - C(t)\right) dt, \tag{7.26}$$

for $\alpha > 0$, $0 \le \beta < 1$, $n \in \mathbb{N}_0$, where

$$C(t) = \frac{2[\beta - 1][z\phi'(t) + \phi(t)]}{1 - 2\beta t\phi(t) + (2\beta - 1)t^2\phi^2(t)}.$$
 (7.27)

Proof. Let $f(z) \in T_n^{\alpha}(\beta)$. Then, from Theorem 159, we have

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} = (2\beta - 1) + \frac{2(1 - \beta)}{1 - z\phi(z)} = \frac{1 - (2\beta - 1)z\phi(z)}{1 - z\phi(z)},\tag{7.28}$$

Taking the logarithmic differentiation, we get from (7.28)

$$\frac{[D^n f(z)^{\alpha}]'}{D^n f(z)^{\alpha}} = \frac{\alpha}{z} - \frac{2[\beta - 1][z\phi'(z) + \phi(z)]}{1 - 2\beta z\phi(z) + (2\beta - 1)z^2\phi^2(z)}.$$
 (7.29)

Now, integrating both sides of (7.29) along the line segment from 0 to z, we obtain

$$\ln\left[D^n f(z)^{\alpha}\right] = \int_0^z \left(\frac{\alpha}{t} - \frac{2\left[\beta - 1\right]\left[t\phi'(t) + \phi(t)\right]}{1 - 2\beta t\phi(t) + (2\beta - 1)t^2\phi^2(t)}\right) dt. \tag{7.30}$$

This gives us that

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} = \frac{1}{\alpha^n z^{\alpha}} \exp \int_0^z \left(\frac{\alpha}{t} - C(t)\right) dt, \tag{7.31}$$

for $z \in D$, $\alpha > 0$, $0 \le \beta < 1$, $n \in \mathbb{N}_0$, where

$$C(t) = \frac{2[\beta - 1][t\phi'(t) + \phi(t)]}{1 - 2\beta t\phi(t) + (2\beta - 1)t^2\phi^2(t)},$$
(7.32)

for an analytic function $\phi(z)$ with $|\phi(z)| \leq 1$, $z \in D$. Thus, the proof of Theorem 160 is completed. \blacksquare

Theorem 161 (Ibrahim, Darus and Dragomir [211]) A function $f(z) \in A$ is in the class $\widetilde{T}_n^{\alpha}(\beta)$ if and only if

$$1 + \sum_{k=2}^{\infty} Q_k z^{k-1} \neq 0, \tag{7.33}$$

where

$$Q_{k} = \frac{(\psi+1)^{\beta}}{\left[(\psi+1)^{\beta} - (\psi-1)^{\beta}\right]} \left(\frac{\alpha+k-1}{\alpha}\right)^{n} A_{k}(\alpha), \qquad (7.34)$$

for some α ($\alpha > 0$), $0 < \beta \le 1$, $\psi \in C$, $|\psi| = 1$.

Proof. From (7.4), it suggests that there exists a function $p(z) \in \mathcal{P}$ such that

$$\frac{D^n f(z)^{\alpha}}{z^{\alpha}} = \alpha^n \left[p(z) \right]^{\beta}, \tag{7.35}$$

for $z \in D$, $\alpha > 1$, $0 < \beta \le 1$. Upon setting

$$p(z) = \left(\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}}\right)^{1/\beta},\tag{7.36}$$

then from Lemma 154, we have

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} \neq \frac{(\psi - 1)^{\beta}}{(\psi + 1)^{\beta}},\tag{7.37}$$

for $z \in D$, $\psi \in \mathbb{C}$ with $|\psi| = 1$. It is equivalent with

$$(\psi + 1)^{\beta} D^{n} f(z)^{\alpha} - (\psi - 1)^{\beta} \alpha^{n} z^{\alpha} \neq 0.$$
 (7.38)

Substituting (6.43) into (7.38) yields

$$(\psi + 1)^{\beta} \left[\alpha^{n} z^{\alpha} + \sum_{k=2}^{\infty} (\alpha + k - 1)^{n} A_{k}(\alpha) z^{\alpha + k - 1} \right] - (\psi - 1)^{\beta} \alpha^{n} z^{\alpha} \neq 0, \quad (7.39)$$

which give us that

$$\left[(\psi + 1)^{\beta} - (\psi - 1)^{\beta} \right] \alpha^{n} z^{\alpha}$$

$$+ (\psi + 1)^{\beta} \sum_{k=2}^{\infty} (\alpha + k - 1)^{n} A_{k}(\alpha) z^{\alpha + k - 1} \neq 0.$$
(7.40)

Now, dividing both sides of (7.40) by $\left[(\psi + 1)^{\beta} - (\psi - 1)^{\beta} \right] \alpha^n z^{\alpha} \neq 0$, we obtain

$$1 + \sum_{k=2}^{\infty} \frac{(\psi+1)^{\beta}}{\left[(\psi+1)^{\beta} - (\psi-1)^{\beta}\right]} \left(\frac{\alpha+k-1}{\alpha}\right)^{n} A_{k}(\alpha) z^{k-1} \neq 0, \tag{7.41}$$

for all $z \in D$, $\psi \in \mathbb{C}$ such that $|\psi| = 1$. Thus, the proof of Theorem 161 is completed. \blacksquare

The similar results to (7.9), (7.18) and (7.26) for more general class $T_n^{\alpha}(\beta, \lambda)$ also have been obtained.

Theorem 162 (Ibrahim, Darus, Dragomir and Joseph [212]) A function $f(z) \in A$ is in the class $T_n^{\alpha}(\beta, \lambda)$ if and only if

$$1 + \sum_{k=2}^{\infty} Q_k z^{k-1} \neq 0, \tag{7.42}$$

where

$$Q_{k} = \frac{(\psi+1) \left[1 + (\alpha + k - 2) \lambda\right]^{n}}{(\psi+1) \left[1 + (\alpha - 1) \lambda\right]^{n} - (2\beta + \psi - 1) \alpha^{n}} A_{k}(\alpha), \qquad (7.43)$$

for some $\alpha > 0$, $\lambda \ge 1$, $0 < \beta \le 1$, $\psi \in \mathbb{C}$ with $|\psi| = 1$.

Proof. From Definition 152, it suggests that there exists a function $p(z) \in \mathcal{P}$ such that

$$\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\alpha^{n} z^{\alpha}} = \left[\beta + (1 - \beta) p(z)\right], \tag{7.44}$$

for $z \in D$, $\alpha > 0$, $\lambda \ge 1$, $0 \le \beta < 1$, $n \in \mathbb{N}_0$. Upon setting

$$p(z) = \frac{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\alpha^{n} z^{\alpha}} - \beta}{1 - \beta},$$
(7.45)

Hence from Lemma 154, we have

$$\frac{D_{\lambda}^{n} f(z)^{\alpha}}{1 - \beta} - \beta \neq \frac{\psi - 1}{\psi + 1},\tag{7.46}$$

for $z \in D$, $\psi \in \mathbb{C}$ with $|\psi| = 1$, which is equivalent with

$$(\psi + 1) D_{\lambda}^{n} f(z)^{\alpha} - [(\psi + 1) \beta + (\psi - 1) (1 - \beta)] \alpha^{n} z^{\alpha} \neq 0.$$
 (7.47)

Substituting (6.45) into (7.47), we get

$$(\psi + 1) \left[[1 + (\alpha - 1)\lambda]^n z^{\alpha} + \sum_{k=2}^{\infty} [1 + (\alpha + k - 2)\lambda]^n A_k(\alpha) z^{\alpha + k - 1} \right] - (2\beta + \psi - 1)\alpha^n z^{\alpha} \neq 0.$$
(7.48)

Hence, it gives us that

$$[(\psi + 1) [1 + (\alpha - 1) \lambda]^{n} - (2\beta + \psi - 1) \alpha^{n}] z^{\alpha}$$

$$+ \sum_{k=2}^{\infty} (\psi + 1) [1 + (\alpha + k - 2) \lambda]^{n} A_{k}(\alpha) z^{\alpha + k - 1} \neq 0.$$
(7.49)

Now, dividing both side of (7.49) by

$$[(\psi + 1) [1 + (\alpha - 1) \lambda]^n - (2\beta + \psi - 1) \alpha^n] z^{\alpha} \neq 0,$$

we obtain

$$1 + \sum_{k=2}^{\infty} \frac{(\psi+1) \left[1 + (\alpha+k-2)\lambda\right]^n}{(\psi+1) \left[1 + (\alpha-1)\lambda\right]^n - (2\beta+\psi-1)\alpha^n} A_k(\alpha) z^{k-1} \neq 0,$$
 (7.50)

for all $\psi \in \mathbb{C}$ such that $|\psi| = 1$, $\alpha > 0$, $\lambda \ge 1$, $0 \le \beta < 1$, $z \in D$, $n \in \mathbb{N}_0$.

Remark 163 In particular, if $\lambda = 1$ in Theorem 162, then it reduce to the result as shown in Theorem 156. Also, we have Corollary 157 for $\lambda = 1$ and $\beta = 0$.

The following corollaries also hold for some particular values of the parameters involved in Theorem 162.

Corollary 164 A function $f(z) \in A$ is in the class $B_1(\alpha)$ if and only if

$$1 + \sum_{k=2}^{\infty} Q_k z^{k-1} \neq 0,$$

where

$$Q_{k} = \frac{(\psi + 1)\left[\alpha + k - 1\right]}{2\alpha} A_{k}(\alpha), \qquad (7.51)$$

for any $\psi \in \mathbb{C}$ with $|\psi| = 1$, $\alpha > 0$, $z \in D$.

Corollary 165 A function $f(z) \in A$ is in the class S_0 if and only if

$$1 + \sum_{k=2}^{\infty} Q_k z^{k-1} \neq 0,$$

where

$$Q_k = \frac{(\psi + 1)}{2\alpha} A_k(\alpha) , \qquad (7.52)$$

for any $\psi \in \mathbb{C}$ with $|\psi| = 1$, $\alpha > 0$, $z \in D$, $n \in \mathbb{N}_0$.

Theorem 166 (Ibrahim, Darus, Dragomir and Joseph [212]) If f(z) belong to the class $T_n^{\alpha}(\beta, \lambda)$, then there exists an analytic function $\phi(z)$ with $|\phi(z)| \leq 1$, $z \in D$ such that

$$\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} = (2\beta - 1) + \frac{2(1-\beta)}{1 - z\phi(z)},\tag{7.53}$$

for $z \in D$, $\alpha > 0$, $\lambda \ge 1$, $0 \le \beta < 1$, $n \in \mathbb{N}_0$.

Proof. Define the functions F(z) and G(z) as follows:

$$F(z) = \frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - \beta \tag{7.54}$$

and

$$G(z) = z \left(\frac{F(z) - (1 - \beta)}{F(z) + (1 - \beta)} \right). \tag{7.55}$$

Substituting the equation (7.54) into (7.55), we get

$$G(z) = z \left(\frac{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - 1}{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - (2\beta - 1)} \right), \ z \in D.$$
 (7.56)

G(z) is an analytic function for $z \in D$. Also, since f(0) - 0 and f'(0) = 1, we have that G(0) = 0 and |G(z)| < 1 for $z \in D$. Hence, by Schwarz's Lemma (Lemma 155), |G(z)| < |z|, $z \in D$, gives us that

$$\left| \frac{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - 1}{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - (2\beta - 1)} \right| < |z|, \quad z \in D,$$

$$(7.57)$$

or equivalently,

$$\left(\frac{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - 1}{\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} - (2\beta - 1)}\right) = z\phi(z), \tag{7.58}$$

where $\phi(z)$ is analytic, and $|\phi(z)| \leq 1$ for $z \in D$. Therefore

$$\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} = \frac{\left[1 - (2\beta - 1) z \phi(z)\right]}{1 - z \phi(z)}$$

$$= 2\beta - 1 + \frac{2(1 - \beta)}{1 - z \phi(z)}, \tag{7.59}$$

which completes the proof of Theorem 166.

Theorem 167 (Ibrahim, Darus, Dragomir and Joseph [212]) If f(z) belongs to the class $T_n^{\alpha}(\beta, \lambda)$, then there exists an analytic function $\phi(z)$ with $|\phi(z)| \leq 1$, $z \in D$ such that

$$\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\alpha^{n} z^{\alpha}} = \frac{1}{\alpha^{n} z^{\alpha}} \exp \int_{0}^{z} \left(\frac{\alpha \lambda}{t} - E(t)\right) dt, \tag{7.60}$$

for $\alpha > 0$, $\lambda \geq 1$, $0 \leq \beta < 1$, $n \in \mathbb{N}_0$, where

$$E(t) = \frac{2[t\phi'(t) + \phi(t)][\beta - 1]}{1 - 2\beta t\phi(t) + (2\beta - 1)t^2\phi^2(t)}.$$
 (7.61)

Proof. Let $f(z) \in T_n^{\alpha}(\beta, \lambda)$. Then, from Theorem 166, we have

$$\frac{D_{\lambda}^{n} f(z)^{\alpha}}{\lambda \alpha^{n} z^{\alpha}} = \frac{1 - (2\beta - 1) z\phi(z)}{1 - z\phi(z)}.$$
(7.62)

Taking the logarithmic differentiation, we get from (7.62)

$$\frac{[D^n f(z)^{\alpha}]'}{D^n f(z)^{\alpha}} = \frac{\alpha \lambda}{z} - \frac{[z\phi'(z) + \phi(z)][2\beta - 2]}{1 - 2\beta z\phi(z) + (2\beta - 1)z^2\phi^2(z)}.$$
 (7.63)

Integrating the both sides of (7.29) along the line segment from 0 to z, we obtain

$$\frac{D^n f(z)^{\alpha}}{\alpha^n z^{\alpha}} = \frac{1}{\alpha^n z^{\alpha}} \exp \int_0^z \left(\frac{\alpha}{t} - E(t)\right) dt, \tag{7.64}$$

where

$$E(t) = \frac{2 \left[t \phi'(t) + \phi(t) \right] [\beta - 1]}{1 - 2\beta t \phi(t) + (2\beta - 1) t^2 \phi^2(t)}.$$
 (7.65)

Thus, the proof of Theorem 167 is completed. ■

7.3 Coefficient Bounds

In this section, we establish certain coefficient bounds for functions in the Class $\widetilde{T}_n^{\alpha}(\beta)$.

Theorem 168 (Ibrahim, Darus and Dragomir [211]) Let f(z) given by (6.11) belongs to the class $\widetilde{T}_n^{\alpha}(\beta)$, $\alpha > 0$, $0 < \beta \le 1$, $n \in \{0, 1, 2, ...\}$. Then, the following inequalities hold

$$|a_2| \leq \frac{2\beta\alpha^{n-1}}{(\alpha+1)^n}; \quad for \, \alpha > 0, \tag{7.66}$$

$$|a_3| \leq \begin{cases} \frac{2\beta\alpha^{n-1}}{(\alpha+2)^n} \\ \times \left(1 + \frac{(\alpha+2)^n}{(\alpha+1)^{2n}} (1-\alpha)\alpha^{n-1}\beta\right); & \text{for } 0 < \alpha < 1, \\ \frac{2\beta\alpha^{n-1}}{(\alpha+2)^n}; & \text{for } \alpha \geq 1, \end{cases}$$

$$|a_4| \le \begin{cases} F_1 + F_2 + F_3; & for \ 0 < \alpha < 1, \\ F_1 + F_2 + F_4 + F_5; & for \ 1 \le \alpha < 2, \\ F_1 + F_2 + F_4; & for \ \alpha \ge 2, \end{cases}$$
 (7.68)

where

$$F_1 = \frac{2\beta\alpha^{n-1}}{3(\alpha+3)^n} [3 + 2(\beta-1)(\beta-5)], \qquad (7.69)$$

$$F_2 = \frac{4\beta^2 (1 - \alpha)^2 \alpha^{3n - 3}}{(1 + \alpha)^{3n}},\tag{7.70}$$

$$F_3 = \frac{4\beta^2 (1 - \alpha) \alpha^{2n-2}}{(\alpha + 1)^n (\alpha + 2)^n},$$
(7.71)

$$F_4 = \frac{4\beta^2 (\beta - 1) (1 - \alpha) \alpha^{2n - 2}}{(\alpha + 1)^n (\alpha + 2)^n},$$
(7.72)

$$F_5 = \frac{4\beta^3 (1 - \alpha) (\alpha - 2) \alpha^{3n - 3}}{3 (\alpha + 1)^{3n}}.$$
 (7.73)

Proof. From Definition 151, for $f \in \widetilde{T}_n^{\alpha}(\beta)$, it suggests that there exists an analytic function $p(z) \in \mathcal{P}$ such that

$$\frac{D^n f(z)^{\alpha}}{z^{\alpha}} = \alpha^n \left[p(z) \right]^{\beta}, \tag{7.74}$$

for $\alpha > 0$, $0 < \beta \le 1$, $z \in D$, $n \in \mathbb{N}_0$. Making use the Sălăgean differential operator (6.35), i.e., $D^n f(z)^{\alpha}$ as $z \left[D^{n-1} f(z)^{\alpha} \right]'$ and p(z) given by the series (6.5), we get

$$[D^{n-1}f(z)^{\alpha}]' = \alpha^n z^{\alpha-1} \left[1 + \sum_{i=1}^{\infty} c_i z^i \right]^{\beta}$$
$$= \alpha^n \left[z^{\alpha-1} + \mu_1 z^{\alpha} + \mu_2 z^{\alpha+1} + \mu_3 z^{\alpha+2} + \mu_4 z^{\alpha+3} + \dots \right], \quad (7.75)$$

where

$$\mu_1 = \beta_1 p_1, \tag{7.76}$$

$$\mu_2 = \beta_1 p_2 + \beta_2 p_1^2, \tag{7.77}$$

$$\mu_3 = \beta_1 p_3 + 2\beta_2 p_1 p_2 + \beta_3 p_1^3, \tag{7.78}$$

$$\mu_4 = \beta_1 p_4 + 2\beta_2 p_1 p_3 + \beta_2 p_2^2 + 3\beta_3 p_1^2 p_2 + \beta_4 p_1^4, \tag{7.79}$$

and

$$\beta_j = {\beta \choose j} = \frac{\beta!}{j! (\beta - j)!}, \ j \in \{1, 2, 3, ...\}.$$
 (7.80)

Integrating both sides of (7.75) along the line segment from 0 to z, we obtain

$$\frac{D^{n-1}f(z)^{\alpha}}{z^{\alpha}} = \alpha^{n-1} \left[1 + \frac{\mu_1 \alpha}{(\alpha+1)} z + \frac{\mu_2 \alpha}{(\alpha+2)} z^2 + \frac{\mu_3 \alpha}{(\alpha+3)} z^3 + \frac{\mu_4 \alpha}{(\alpha+4)} z^4 + \dots \right],$$
(7.81)

where f(0) = 0.

Noting that $D^{n-1}f(z)^{\alpha} = z [D^{n-2}f(z)^{\alpha}]'$ and so on, and repeating the process as above, we are able to produce the following relation, which holds in general for any $k \in \{0, 1, 2, ...\}$, that is,

$$\frac{D^{n-k}f(z)^{\alpha}}{z^{\alpha}} = \alpha^{n-k} \left[1 + \frac{\mu_1 \alpha^k}{(\alpha+1)^k} z + \frac{\mu_2 \alpha^k}{(\alpha+2)^k} z^2 + \frac{\mu_3 \alpha^k}{(\alpha+3)^k} z^3 + \frac{\mu_4 \alpha^k}{(\alpha+4)^k} z^4 + \dots \right].$$

In particular, for n = k, we have the following equation:

$$\left(\frac{f(z)}{z}\right)^{\alpha} = 1 + \frac{\mu_1 \alpha^n}{(\alpha+1)^n} z + \frac{\mu_2 \alpha^n}{(\alpha+2)^n} z^2 + \frac{\mu_3 \alpha^n}{(\alpha+3)^n} z^3 + \frac{\mu_4 \alpha^n}{(\alpha+4)^n} z^4 + \dots$$
(7.82)

Comparing the coefficients of z, z^2 and z^3 in (7.82), we get

$$\alpha_1 a_2 = \frac{\mu_1 \alpha^n}{(\alpha + 1)^n},\tag{7.83}$$

$$\alpha_1 a_3 = \frac{\mu_2 \alpha^n}{(\alpha + 2)^n} - \alpha_2 a_2^2, \tag{7.84}$$

$$\alpha_1 a_4 = \frac{\mu_3 \alpha^n}{(\alpha + 3)^n} - 2\alpha_2 a_2 a_3 - \alpha_3 a_2^3, \tag{7.85}$$

where

$$\alpha_j = \binom{\alpha}{j} = \frac{\alpha!}{j! (\alpha - j)!}, \ j \in \{1, 2, 3, ...\}.$$
 (7.86)

Since $|p_1| \le 2$ from Lemma 133, thus it follows from (7.83) that

$$|a_2| \le \frac{2\beta\alpha^{n-1}}{(1+\alpha)^n},\tag{7.87}$$

for $\alpha > 0$, $0 < \beta \le 1$.

Next, we eliminate a_2 in (7.84) to get

$$a_3 = \frac{\beta \alpha^{n-1}}{(\alpha+2)^n} p_2 + \frac{\beta (\beta-1) \alpha^{n-1}}{2 (\alpha+2)^n} p_1^2 + \frac{(1-\alpha) \beta^2 \alpha^{2n-2}}{(\alpha+1)^{2n}} p_1^2.$$
 (7.88)

Again, from Lemma 133 with $|p_2| \le 2$ and $0 < \beta \le 1$, we have

$$|a_3| \le \frac{2\beta\alpha^{n-1}}{(\alpha+2)^n} \left(1 + (\beta-1) + \frac{2|1-\alpha|(\alpha+2)^n\beta\alpha^{n-1}}{(\alpha+1)^{2n}} \right). \tag{7.89}$$

Considering the cases of α in the intervals $0 < \alpha < 1$ and $\alpha \ge 1$, then the both desired inequalities in (7.67) are easily obtained.

Now, we prove the inequalities in (7.68). Substituting the equalities (7.83) and (7.84) into (7.85), gives us that

$$a_{4} = \frac{\beta \alpha^{n-1}}{(\alpha+3)^{n}} p_{3} + \frac{\beta (\beta-1) \alpha^{n-1}}{(\alpha+3)^{n}} p_{1} p_{2}$$

$$+ \frac{\beta (\beta-1) (\beta-2) \alpha^{n-1}}{6 (\alpha+3)^{n}} p_{1}^{3}$$

$$+ \frac{\beta^{2} (1-\alpha) \alpha^{2n-2}}{(\alpha+1)^{n} (\alpha+2)^{n}} p_{1} p_{2} + \frac{\beta^{2} (\beta-1) (1-\alpha) \alpha^{2n-2}}{2 (\alpha+1)^{n} (\alpha+2)^{n}} p_{1}^{3}$$

$$+ \frac{\beta^{3} (1-\alpha)^{2} \alpha^{3n-3}}{2 (\alpha+1)^{3n}} p_{1}^{3} + \frac{\beta^{3} (1-\alpha) (\alpha-2) \alpha^{3n-3}}{6 (\alpha+1)^{3n}} p_{1}^{3}.$$
 (7.90)

Hence, for $\alpha > 0$, $0 < \beta \le 1$, we obtain

$$|a_{4}| \leq \frac{2\beta\alpha^{n-1}}{3(\alpha+3)^{n}} \left[3 + 2(\beta-1)(\beta-5)\right] + \frac{4\beta^{3}(1-\alpha)^{2}\alpha^{3n-3}}{(1+\alpha)^{3n}} + \frac{4\beta^{2}|(1-\alpha)|\alpha^{2n-2}}{(\alpha+1)^{n}(\alpha+2)^{n}} - \frac{4\beta^{2}|(1-\alpha)|(\beta-1)\alpha^{2n-2}}{(\alpha+1)^{n}(\alpha+2)^{n}} + \frac{4\beta^{3}|(1-\alpha)(\alpha-2)|\alpha^{3n-3}}{3(\alpha+1)^{3n}},$$

$$(7.91)$$

by applying the well-known inequality $|p_k| \leq 2$, $k \in \{1, 2, ...\}$ from Lemma 133. Thus, the coefficient bounds $|a_4|$ depends on the parameter α in the intervals (0,1), [1,2) and $[2,\infty)$. Let us denote

$$F_{1} = \frac{2\beta\alpha^{n-1}}{3(\alpha+3)^{n}} [3+2(\beta-1)(\beta-5)],$$

$$F_{2} = \frac{4\beta^{2}(1-\alpha)^{2}\alpha^{3n-3}}{(1+\alpha)^{3n}},$$

$$F_{3} = \frac{4\beta^{2}(1-\alpha)\alpha^{2n-2}}{(\alpha+1)^{n}(\alpha+2)^{n}},$$

$$F_{4} = \frac{4\beta^{2}(\beta-1)(1-\alpha)\alpha^{2n-2}}{(\alpha+1)^{n}(\alpha+2)^{n}},$$

$$F_5 = \frac{4\beta^3 (1 - \alpha) (\alpha - 2) \alpha^{3n - 3}}{3 (\alpha + 1)^{3n}}.$$

Then, from (7.91) we get the desired inequalities (7.68), and the proof of Theorem 168 is completed. \blacksquare

Remark 169 If we choose $\beta = 1$ in Theorem 168, we deduce the result proved by Halim in [181].

7.4 Fekete-Szegő Inequalities

7.4.1 Introduction and Preliminary results

In this section, we consider functions $f: D \to G \subset \overline{\mathbb{C}}$ that are univalent in D, analytic at zero and have the expansion given by (6.11). The class of those functions, which map D into the domain whose complement with respect to G is convex, is denoted by C_0 and we call it the class of concave univalent functions. Since ∂G is a curve, the function f(z) has continuous extension to \overline{D} onto G, which is infinite at exactly one point on $T = \partial D$. We can choose the map such that $f(1) = \infty$. The class $C_0(1)$ consists of all concave univalent functions of the form (6.11) and normalized such that $f(1) = \infty$.

Another case of interest is when $\overline{\mathbb{C}}\backslash G$ is a bounded convex set. Then, the conformal map f(z) of D onto G has a pole in D. The class $C_0(p)$ consists of all such functions of the form (6.11) and $f(p) = \infty$, where 0 . For the details discussion about this class of functions, we refer to the works of ([35], [36], [103], [291], [431]) and the references which are cited therein. In a recent work by Avkhadiev and Wirths [37] (see also [36], [64]), they consider a concave domain <math>G with given angle $\pi\alpha$ at ∞ , where $\alpha \in [1, 2]$, and if $\alpha = 1$, then G is a halfplane.

To be precise, we say that a function $f: D \to \overline{\mathbb{C}}$ belongs to the family $C_0(\alpha)$, $\alpha > 1$ if f(z) satisfies the following conditions: (a) f(z) is analytic in D with the standard normalization conditions (6.3). In addition, it satisfies $f(1) = \infty$; (b) f(z) maps conformally onto a set whose complement with respect to $\overline{\mathbb{C}}$ is convex;

(c) The opening angle of f(D) at ∞ is less than or equal to $\pi\alpha$, $\alpha \in (1,2]$. We observe that $C_0(2)$ contains the classes $C_0(\alpha)$, $\alpha \in (1,2]$.

In [36], Avkhadiev and Wirths showed that an analytic function in the class $C_0(\alpha)$, $\alpha \in (1,2]$ if and only if it satisfies the following condition

$$\operatorname{Re}\left(\frac{2}{\alpha - 1} \left[\frac{(\alpha + 1)}{2} \frac{1 + z}{1 - z} - 1 - z \frac{f''(z)}{f'(z)} \right] \right) > 0. \tag{7.92}$$

The class $C_0(\alpha)$ was recently studied by Bhowmik, Ponnusamy and Wirths (see [64], [65]). In [65], they have used the above characterization (7.92) and proved the following theorem.

Theorem 170 ([65]) Let $\alpha \in (1,2]$. A function $f(z) \in C_0(\alpha)$ if and only if there exists a starlike function $\phi \in S^*$ such that $f(z) = \Lambda_{\phi}(z)$ where

$$\Lambda_{\phi}(z) = \int_{0}^{z} \frac{1}{(1-t)^{\alpha+1}} \left(\frac{t}{\phi(t)}\right)^{(\alpha-1)/2} dt, \tag{7.93}$$

and S^* denote the family of starlike functions $g \in S^*$ satisfying (6.14).

Theorem 171 ([65]) Let $\alpha \in (1,2]$. A function $f(z) \in C_0(\alpha)$ if and only if there exists a function $s \in \Pi_{\alpha-1}$ such that

$$f(z) = \int_0^z \frac{s(t)}{(1-t)^{\alpha+1}} dt.$$
 (7.94)

In this section, we introduce the class $C_0(\alpha, n)$, $\alpha \in (1, 2]$, $n \in \mathbb{N}_0$ consisting of all the concave univalent functions, which are defined as follows. A function $f(z) \in C_0(\alpha, n)$ if and only if there exists a function $\psi \in S_n$ such that $f(z) = \Lambda_{\psi}(z)$ where

$$\Lambda_{\psi}(z) = \int_{0}^{z} \frac{1}{(1-t)^{\alpha+1}} \left(\frac{t}{\psi(t)}\right)^{(\alpha-1)/2} dt, \tag{7.95}$$

and S_n denote the family of functions $g \in S_n$, which satisfy the condition (6.46). We note that $C_0(\alpha, 0) = C_0(\alpha)$.

The purpose of this section is to determine the sharp upper bounds of the Fekete-Szegő functional $|a_3 - \mu a_2^2|$, for functions in the class $C_0(\alpha, n)$. In particular, we extend the result of the Fekete-Szegő problem established by Bhowmik

et al. in [64], for a real and complex parameter μ . In order to prove our main results, we need to recall the following results.

Corollary 172 (See [207]) Let
$$g(z) = z + \sum_{k=2}^{\infty} b_k z^k \in S_n$$
. Then $|b_k| \le k^{1-n}$, for $k = 2, 3, ...$ (7.96)

and

$$|b_3 - \mu b_2^2| \le \max\{1, |3 - 4\mu|\}.$$
 (7.97)

7.4.2 Main Results and Their Proofs

The Fekete-Szegő inequalities $|a_3 - \lambda a_2^2|$ for a real and complex parameter λ are incorporated in the following theorems. First for a real λ , we have

Theorem 173 (Ibrahim, Darus and Dragomir [210]) Let $f \in C_0(\alpha, n)$, $\alpha \in (1,2]$, $n \in \{0,1,2,...\}$, have the expansion given by (6.11). If λ is a real number, then we have

$$12 |a_3 - \lambda a_2^2|$$

$$\begin{cases}
(3+2^{2n})(2-3\lambda)\alpha^{2} \\
+3(1-2^{2n})(1-2\alpha)\lambda & if \lambda \leq \lambda_{0}; \\
+6(1-3^{n})\alpha+2(3^{n+1}-2^{2n}), \\
4[(2-3\lambda)\alpha^{2}+1], & if \lambda_{0} \leq \lambda \leq \frac{2}{3\alpha}(\alpha-1); \\
\frac{4[(10-9\lambda)\alpha+(2-3\lambda)]}{3(2-\lambda)-(2-3\lambda)\alpha}, & if \frac{2}{3\alpha}(\alpha-1) \leq \lambda \leq \frac{2}{3}; \\
12(1-\lambda)\alpha\sqrt{\frac{12(1-\lambda)}{(4-3\lambda)^{2}-(3\lambda-2)^{2}\alpha^{2}}}, & if \frac{2}{3} \leq \lambda \leq \lambda_{2}; \\
4[(3\lambda-2)\alpha^{2}-1], & if \lambda_{2} \leq \lambda \leq \frac{2(\alpha+2)}{3(\alpha+1)}; \\
(3+2^{2n})(3\lambda-2)\alpha^{2} \\
+3(2^{2n}-1)(1-2\alpha)\lambda & if \lambda \geq \frac{2(\alpha+2)}{3(\alpha+1)}; \\
+6(3^{n}-1)\alpha+2(2^{2n}-3^{n+1}), & if \lambda \geq \frac{2(\alpha+2)}{3(\alpha+1)};
\end{cases}$$

where

$$\lambda_0 = \frac{2^{2n-2} (\alpha + 1) - 3^n}{3 (2^{2n-3}) (\alpha - 1)} \text{ and } \lambda_2 = \frac{2}{3} + \frac{1}{6\alpha^2} \left(\sqrt{8\alpha^2 + 1} - 1 \right). \tag{7.99}$$

The inequalities are sharp.

We have the following result for the complex parameter λ .

Theorem 174 (Ibrahim, Darus and Dragomir [210]) Let $f \in C_0(\alpha, n)$ have the expansion given by (6.11), $\alpha \in (1, 2]$, $n \in \{0, 1, 2, ...\}$. If λ is a complex number, then we have

$$\left|a_3 - \lambda a_2^2\right| \le \max\left\{1, \ \frac{1}{12}\left(\alpha + 1\right)\nu\left(\alpha, \lambda\right)\right\},$$
 (7.100)

where

$$\nu(\alpha, \lambda) = |(2 - 3\lambda)(\alpha + 1) + 2| + 2(\alpha - 1)|3\lambda - 2| + \left(\frac{\alpha - 1}{\alpha + 1}\right)|6 + [2 - 3(\alpha - 1)\lambda]|.$$
 (7.101)

Proof. We recall that for $f \in C_0(\alpha, n)$ if and only if there exist a function $\psi(z) = z + \sum_{n=2}^{\infty} \psi_n z^n \in S_n, n \in \{0, 1, 2, ...\}$ such that

$$f'(z) = \frac{1}{(1-z)^{\alpha+1}} \left(\frac{z}{\psi(z)}\right)^{(\alpha-1)/2},$$
 (7.102)

where f(z) has the form given by (6.11). Comparing the coefficients of z and z^2 on the both sides of the series expansion (7.102), we obtain

$$a_2 = \frac{(\alpha+1)}{2} - 2^{n-2} (\alpha - 1) \psi_2 \tag{7.103}$$

and

$$a_{3} = \frac{1}{6} (\alpha + 1) (\alpha + 2) - \frac{2^{n-1}}{3} (\alpha^{2} - 1) \psi_{2}$$
$$- \frac{3^{n-1}}{2} (\alpha - 1) \psi_{3} + \frac{2^{2n-3}}{3} (\alpha - 1) \psi_{2}^{2}, \tag{7.104}$$

respectively. A computation yields that

$$a_{3} - \lambda a_{2}^{2} = \frac{(\alpha+1)^{2}}{4} \left[\frac{2(\alpha+2)}{3(\alpha+1)} - \lambda \right]$$

$$+2^{n-2} \left(\alpha^{2} - 1\right) \left(\lambda - \frac{2}{3}\right) \psi_{2} - \frac{3^{n-1}}{2} (\alpha - 1)$$

$$\times \left[\psi_{3} - \left(\frac{2^{2n-2} (\alpha+1) - 3\lambda \left(2^{2n-3}\right) (\alpha - 1)}{3^{n}}\right) \psi_{2}^{2} \right]. \quad (7.105)$$

Hence, for $\alpha > 1$, we get

$$|a_{3} - \lambda a_{2}^{2}| \leq \frac{(\alpha+1)^{2}}{4} \left| \frac{2(\alpha+2)}{3(\alpha+1)} - \lambda \right|$$

$$+2^{n-2} (\alpha^{2}-1) \left| \lambda - \frac{2}{3} \right| |\psi_{2}| - \frac{3^{n-1}}{2} (\alpha-1)$$

$$\times \left| \psi_{3} - \left(\frac{2^{2n-2} (\alpha+1) - 3\lambda (2^{2n-3}) (\alpha-1)}{3^{n}} \right) \psi_{2}^{2} \right|. (7.106)$$

Now, we need to investigate the maximum values of the functional $|a_3 - \lambda a_2^2|$ by considering several cases of λ real.

Case 1: First, consider the case for all $\lambda \leq \frac{2^{2n-2}(\alpha+1)-3^n}{3(2^{2n-3})(\alpha-1)}$. We observe that the assumption on λ is seen to be equivalent to

$$2^{2n-2} (\alpha + 1) - 3\lambda (2^{2n-3}) (\alpha - 1) \ge 1, \tag{7.107}$$

and the first term in the right hand-side of (7.105) is nonnegative. Hence, from (7.97) we have that

$$\left| \psi_3 - \left(\frac{2^{2n-2} (\alpha + 1) - 3\lambda (2^{2n-3}) (\alpha - 1)}{3^n} \right) \psi_2^2 \right|$$

$$\leq \frac{2^{2n} (\alpha + 1) - 3\lambda (2^{2n-1}) (\alpha - 1)}{3^n} - 3,$$
(7.108)

and also, noticing that from (7.96), for $\psi(z) \in S_n$, $|\psi_k| \le k^{1-n}$ for $k \in \{2, 3, ...\}$. Hence we have from (7.106) that

$$|a_{3} - \lambda a_{2}^{2}| \leq \frac{(\alpha+1)^{2}}{4} \left[\frac{2(\alpha+2)}{3(\alpha+1)} - \lambda \right] + 2^{n-2} \left(\alpha^{2} - 1\right) \left(\frac{2}{3} - \lambda \right) |\psi_{2}|$$

$$+ \frac{3^{n-1}}{2} (\alpha - 1) \left| \psi_{3} - \left(\frac{2^{2n-2} (\alpha+1) - 3\lambda (2^{2n-3}) (\alpha - 1)}{3^{n}} \right) \psi_{2}^{2} \right|$$

$$= \frac{(\alpha+1) (\alpha+2)}{6} - \frac{\lambda}{4} (\alpha+1)^{2} + \frac{(\alpha^{2}-1)}{2} \left(\frac{2}{3} - \lambda \right)$$

$$+ \frac{3^{n-1}}{2} (\alpha-1) \left(\frac{2^{2n} (\alpha+1) - 3\lambda (2^{2n-1}) (\alpha - 1)}{3^{n}} - 3 \right).$$
 (7.109)

It can be simplified to

$$|a_3 - \lambda a_2^2| \le \frac{1}{12} \left[\left(3 + 2^{2n} \right) (2 - 3\lambda) \alpha^2 + 3 \left(1 - 2^{2n} \right) (1 - 2\alpha) \lambda + 6 \left(1 - 3^n \right) \alpha + 2 \left(3^{n+1} - 2^{2n} \right) \right], \tag{7.110}$$

for any
$$\lambda \in \left(\infty, \frac{2^{2n-2}(\alpha+1)-3^n}{3(2^{2n-3})(\alpha-1)}\right)$$
.

Case 2: Let $\lambda \geq \frac{2(\alpha+2)}{3(\alpha+1)}$. For this case, the first term of right hand-side of (7.105) is nonnegative. The condition on λ in particular, gives $\lambda \geq 2/3$, and therefore, our assumption on λ implies that

$$\frac{2^{2n-2}(\alpha+1) - 3\lambda(2^{2n-3})(\alpha-1)}{3^n} \le \frac{2^{2n}}{3^n} \left(\frac{1}{2}\right). \tag{7.111}$$

Again, it follows from (7.97) that

$$\left| \psi_3 - \left(\frac{2^{2n-2} (\alpha + 1) - 3\lambda (2^{2n-3}) (\alpha - 1)}{3^n} \right) \psi_2^2 \right|$$

$$\leq 3 - \frac{2^{2n} (\alpha + 1) - 3\lambda (2^{2n-1}) (\alpha - 1)}{3^n}.$$
(7.112)

In view of these observation and the use of the inequality that $|\psi_2| \leq 2^{1-k}$, the inequality (7.106) gives

$$\left| a_3 - \lambda a_2^2 \right| \le \frac{(\alpha + 1)^2}{4} \left[\lambda - \frac{2(\alpha + 2)}{3(\alpha + 1)} \right] + 2^{n-2} \left(\alpha^2 - 1 \right) \left(\lambda - \frac{2}{3} \right) \left(2^{1-n} \right)$$

$$+ \frac{3^{n-1}}{2} \left(\alpha - 1 \right) \left(3 - \frac{2^{2n} \left(\alpha + 1 \right) - 3\lambda \left(2^{2n-1} \right) \left(\alpha - 1 \right)}{3^n} \right). \tag{7.113}$$

Thus, simplifying the right hand-side expression in (7.113), we obtain

$$\left| a_3 - \lambda a_2^2 \right| \le \frac{1}{12} \left[\left(3 + 2^{2n} \right) \left(3\lambda - 2 \right) \alpha^2 + 3 \left(2^{2n} - 1 \right) \left(1 - 2\alpha \right) \lambda + 6 \left(3^n - 1 \right) \alpha - 2 \left(3^{n+1} - 2^{2n} \right) \right], \tag{7.114}$$

for any $\lambda \in [2(\alpha+2)/3(\alpha+1), \infty)$.

Case 3: Consider λ , where $\lambda \in \left(\frac{2^{2n-2}(\alpha+1)-3^n}{3(2^{2n-3})(\alpha-1)}, \frac{2(\alpha+2)}{3(\alpha+1)}\right)$. Now we deal with the case by using the formulas (7.102) and (7.105) together with the representation formula for $\psi(z) \in S_n$. Let us define the function w(z) by

$$\frac{D^{n+1}\psi(z)}{D^{n}\psi(z)} = \frac{1+zw(z)}{1-zw(z)}; \ (w(z) \neq 1),$$
 (7.115)

where $w: D \to \overline{D}$ is a function analytic in D with the Taylor series

$$w(z) = \sum_{n=0}^{\infty} c_n z^n, \tag{7.116}$$

and D^n , $n \in \mathbb{N}_0$ is the Sălăgean differential operator defined by (6.35). Comparing the coefficients of z and z^2 in the series expansion (7.115), we get

$$\psi_2 = 2^{1-n}c_0 \tag{7.117}$$

and

$$\psi_3 = \frac{1}{3^n} \left(c_1 + 3c_0^2 \right). \tag{7.118}$$

Inserting these resulting formulas (7.117) and (7.118), into (7.105) yields

$$a_{3} - \lambda a_{2}^{2} \leq \frac{(\alpha+1)^{2}}{4} \left[\frac{2(\alpha+2)}{3(\alpha+1)} - \lambda \right]$$

$$+ 2^{n-2} \left(\alpha^{2} - 1 \right) \left(\lambda - \frac{2}{3} \right) \left(2^{1-n} c_{0} \right)$$

$$+ \frac{3^{n-1}}{2} \left(\alpha - 1 \right) \left[\frac{1}{3^{n}} \left(c_{1} + 3c_{0}^{2} \right) \right]$$

$$- \left(\frac{2^{2n-2} \left(\alpha+1 \right) - 3\lambda \left(2^{2n-3} \right) \left(\alpha-1 \right)}{3^{n}} \right) \left(2^{2-2n} \right) c_{0}^{2}$$

$$= A + Bc_{0} + Cc_{0}^{2} + Dc_{1},$$

$$(7.119)$$

where

$$A = \frac{1}{6} (\alpha + 2) (\alpha + 1) - \frac{\lambda}{4} (\alpha + 1)^{2}, \qquad (7.120)$$

$$B = \frac{1}{6} (\alpha^2 - 1) (3\lambda - 2), \qquad (7.121)$$

$$C = -\frac{1}{12} (\alpha - 1) [4 - 2\alpha + 3\lambda (\alpha - 1)], \qquad (7.122)$$

$$D = -\frac{1}{6}(\alpha - 1). (7.123)$$

Hence, by using the well-known inequalities that $|c_0| \le 1$ and $|c_1| \le 1 - |c_0|^2$, we obtain from (7.119) that

$$\left|a_3 - \lambda a_2^2\right| \le \left|A + Bc_0 + Cc_0^2\right| + \frac{1}{6} (\alpha - 1) \left(1 - \left|c_0\right|^2\right).$$
 (7.124)

To find the maximum value of $|a_3 - \lambda a_2^2|$ in (7.124), we let $c_0 = re^{i\theta}$, and consider the following quadratic expression:

$$f(r,\theta) = |A + Bc_0 + Cc_0^2|^2$$

= $(A - Cr^2)^2 + B^2r^2 + 2Br(A + Cr^2)\cos\theta + 4ACr^2\cos^2\theta$, (7.125)

where $\cos \theta \in [-1, 1]$, $r \in (0, 1]$. For getting the upper bounds of $|a_3 - \lambda a_2^2|$, we have to find the maximum value of $f(r, \theta)$ for r in the interval (0, 1]. So, let $x = \cos \theta$, then from (7.125) we have

$$h(x) = (A - Cr^{2})^{2} + B^{2}r^{2} + 2Br(A + Cr^{2})x + 4ACr^{2}x^{2}.$$
 (7.126)

In order to determine the maximum value of h(x) for $x \in [-1,1]$, we need to consider several subintervals of λ , where

$$\lambda \in \left(\frac{2^{2n-2}(\alpha+1) - 3^n}{3(2^{2n-3})(\alpha-1)}, \frac{2(\alpha+2)}{3(\alpha+1)}\right). \tag{7.127}$$

Case 3A: First, consider $\lambda \in \left(\frac{2^{2n-2}(\alpha+1)-3^n}{3(2^{2n-3})(\alpha-1)}, \frac{2(\alpha-2)}{3(\alpha-1)}\right)$. We observe that for λ in this interval, we have A>0, B<0, C>0 and $A+Cr^2>0$ for $r\in(0,1]$, and (7.126) attains its maximum value at x=-1. Therefore, it gives that

$$\left| a_3 - \lambda a_2^2 \right| \le g(r) = A - Br + Cr^2 + \frac{1}{3} (\alpha - 1) (1 - r^2).$$
 (7.128)

By a simple calculation, we show that the maximum value of (7.128) attains at the boundary of r, i.e., r = 1. Thus, we have

$$g(r) \le g(1) = A - B + C = \frac{1}{3} [(2 - 3\lambda) \alpha^2 + 1].$$
 (7.129)

Case 3B: Let $\lambda = 2(\alpha - 2)/3(\alpha - 1)$. In this case, we have C = 0, therefore from (7.126), h(x) becomes a linear function as follows,

$$h(x) = A^2 + B^2 r^2 + 2BrAx. (7.130)$$

It is easy to show that the maximum value of (7.130) occurs at x = -1 and r = 1. Again we get the maximum value of $|a_3 - \lambda a_2|$ as given in the previous case.

Case 3C: Let $\lambda \in \left(\frac{2(\alpha-2)}{3(\alpha-1)}, \frac{2(\alpha-1)}{3\alpha}\right)$. In this interval, the quadratic function (7.126) has the maximum value at

$$x(r) = -\frac{B}{4} \left(\frac{1}{Cr} + \frac{r}{A} \right), \tag{7.131}$$

where x(r) is a monotonic increasing function in $r \in (0,1]$, and x(1) < -1. Hence we get the upper bound of $|a_3 - \lambda a_2^2|$ as given in Cases 3A and 3B. As conclusion, Cases 3A, 3B and 3C give

$$\left| a_3 - \lambda a_2^2 \right| \le \frac{1}{3} \left[(2 - 3\lambda) \alpha^2 + 1 \right],$$
 (7.132)

for all
$$\lambda \in \left(\frac{2^{2n-2}(\alpha+1)-3^n}{3(2^{2n-3})(\alpha-1)}, \frac{2(\alpha-1)}{3\alpha}\right)$$
.

Case 3D: Let $\lambda \in \left[\frac{2(\alpha-1)}{3\alpha}, \frac{2}{3}\right)$. From the Case 3C, the inequality x(1) < -1 gives us that

$$\frac{2(3\lambda + 4\alpha^2 - 12\alpha^2\lambda + 9\alpha^2\lambda^2 - 4)}{[(3\lambda - 4) + \alpha(3\lambda - 2)][\alpha(3\lambda - 2) - (3\lambda - 4)]} < 0,$$
(7.133)

hence it shows that

$$p(\lambda) = 9\alpha^2 \lambda^2 + (3 - 12\alpha^2)\lambda + 4(\alpha^2 - 1) < 0, \tag{7.134}$$

where $\lambda < 2/3$. Factorizing $p(\lambda)$ in (7.134), we have

$$\lambda_1 = \frac{2}{3} - \frac{1}{6\alpha^2} \left(1 + \sqrt{8\alpha^2 + 1} \right) \tag{7.135}$$

and

$$\lambda_2 = \frac{2}{3} - \frac{1}{6\alpha^2} \left(1 - \sqrt{8\alpha^2 + 1} \right). \tag{7.136}$$

It is clear that $\lambda_1 < \lambda_2$. Therefore, for $\lambda \in [2(\alpha - 1)/3\alpha, \lambda_1)$, the functions (7.126) and (7.128) have their maximum values at

$$x = -1 \text{ and } r_m = \frac{-3B}{-6C + \alpha - 1} \in (0, 1],$$
 (7.137)

respectively. Hence the upper bound of the Fekete-Szegő functional is given by

$$|a_3 - \lambda a_2^2| \le g(r_m) = A - Br_m + Cr_m^2 + \frac{1}{3}(\alpha - 1)(1 - r_m^2)$$

$$= \frac{4[(10 - 9\lambda)\alpha + (2 - 3\lambda)]}{3(2 - \lambda) - (2 - 3\lambda)\alpha}.$$
(7.138)

Next, we consider $\lambda \in [\lambda_1, 2/3)$. In this interval, the quadratic equation (7.126) attains its maximum value at

$$x(r) = \frac{-B(A + Cr^2)}{4ACr},$$
(7.139)

with

$$h(x(r)) = -\frac{1}{4AC} (B^2 - 4AC) (A - Cr)^2.$$
 (7.140)

Hence, the Fekete-Szegő functional satisfies the following inequality

$$|a_{3} - \lambda a_{2}^{2}| \leq \sqrt{h(x(r))} + \frac{(\alpha - 1)}{6} (1 - r^{2})$$

$$= (A - Cr) \sqrt{1 - \frac{B^{2}}{4AC}} + \frac{(\alpha - 1)}{6} (1 - r^{2}) = k(r).$$
(7.141)

The maximum value of g(r), where

$$g(r) = A - Br + Cr^{2} + \frac{(\alpha - 1)}{6} (1 - r^{2})$$
 (7.142)

and the function (7.141) occur at

$$r_m = \frac{-B}{-2C + \frac{(\alpha - 1)}{3}} \text{ and } r_0 = \frac{B}{2C + \sqrt{1 - \frac{B^2}{4AC}}},$$
 (7.143)

respectively. It is easy to show that (7.141) is a monotonic decreasing function for $r \geq r_0$. Hence, the maximum value of $|a_3 - \lambda a_2^2|$ is also given by (7.138).

For $\lambda = 2/3$, we get B = 0 and $C = (1 - \alpha)/6$. Thus, the maximum value

$$\left|a_3 - \lambda a_2^2\right| = \frac{\alpha}{3},\tag{7.144}$$

occurs at $x = \cos \theta = 0$ and $r \in (0, 1]$. Therefore, from (7.138) and (7.144) we conclude that

$$\left| a_3 - \lambda a_2^2 \right| \le \frac{4 \left[(10 - 9\lambda) \alpha + (2 - 3\lambda) \right]}{3 (2 - \lambda) - (2 - 3\lambda) \alpha},$$
 (7.145)

for all $\lambda \in [2(\alpha - 1)/3\alpha, 2/3]$.

Case 3E: Let $\lambda \in (2/3, \lambda_2]$, where λ_2 is given by (7.136). In this interval, we have B > 0. So that the function (7.126) attains its maximum value at x = 1. Then, we have

$$l(r) = h(1) = A + Br + Cr^{2} + \frac{(\alpha - 1)}{6} (1 - r^{2}).$$
 (7.146)

Again, by a simple calculation shows that the maximum value of l(r) occurs at

$$r_n = \frac{B}{-2C + \frac{(\alpha - 1)}{3}}. (7.147)$$

Hence the maximum of the function (7.141) is attained at

$$r_1 = \frac{B}{-2C\left(1 + \sqrt{1 - \frac{B^2}{4AC}}\right)} \in (0, 1]. \tag{7.148}$$

It is easily to prove that $r_1 < r_n \le 1$. Since k(r) is monotonic increasing function, then

$$k(r) \le k(1) = (A - C)\sqrt{1 - \frac{B^2}{4AC}},$$
 (7.149)

which gives that

$$\left| a_3 - \lambda a_2^2 \right| \le k(1) = (1 - \lambda) \alpha \sqrt{\frac{12(1 - \lambda)}{(4 - 3\lambda)^2 - (3\lambda - 2)^2 \alpha^2}},$$
 (7.150)

for $\lambda \in (2/3, \lambda_2]$.

Case 3F: Finally, we consider the case for $\lambda \in (\lambda_2, 2(\alpha + 2)/3(\alpha + 1))$. For λ in this interval, we see that A < 0, B > 0, C < 0, $A + Cr^2 < 0$ and the maximum value of function (7.124) is attained for x = -1, i.e.,

$$\eta(x) = -A + Br - Cr^2 + \frac{(\alpha - 1)}{6} (1 - r^2).$$
 (7.151)

We get $\eta(r) \leq \eta(1)$ for all λ in these interval, and hence

$$\left| a_3 - \lambda a_2^2 \right| \le -A + B - C = \frac{1}{3} \left[(3\lambda - 2) \alpha^2 - 1 \right].$$
 (7.152)

Thus, the proof of Theorem 173 is completed.

Further, substituting (7.117) and (7.118) into (7.105), we get

$$12 (a_3 - \lambda a_2^2) = (\alpha + 1) [(2 - 3\lambda) (\alpha + 1) + 2]$$

$$+ 2 (\alpha^2 - 1) (3\lambda - 2) c_0 + 2 (1 - \alpha) c_1$$

$$+ (\alpha - 1) (6 + [2 - 3 (\alpha - 1) \lambda]) c_0^2.$$
(7.153)

Hence for λ complex numbers, we have

$$12 |a_{3} - \lambda a_{2}^{2}| \leq (\alpha + 1) |(2 - 3\lambda) (\alpha + 1) + 2|$$

$$+ 2 (1 - \alpha) |c_{1}| + 2 (\alpha^{2} - 1) |3\lambda - 2| |c_{0}|$$

$$+ (\alpha - 1) |6 + [2 - 3 (\alpha - 1) \lambda]| |c_{0}|^{2}.$$
(7.154)

Using the well-known inequality that $|c_0| \leq 1$ and $|c_1| \leq 1 - |c_0|^2$, then from (7.154) we get

$$12|a_3 - \lambda a_2^2| \le \frac{1}{12}(\alpha + 1)\nu(\alpha, \lambda),$$
 (7.155)

for Re $\{\nu(\alpha, \lambda)\} > 0$, where

$$\nu(\alpha, \lambda) = |(2 - 3\lambda)(\alpha + 1) + 2| + 2(1 - \alpha)|3\lambda - 2| + \frac{(\alpha - 1)}{\alpha + 1}|6 + [2 - 3(\alpha - 1)\lambda]|.$$
(7.156)

Thus, the proof of Theorem 174 is completed. ■

Remark 175 Taking n = 0 and λ is a real numbers in Theorem 173, we deduce a result of Bhowmik et al. [64]..

Other problems related to the Fekete-Szegő functional for further reading can be found in (see for instance [107], [148], [151], [206], [234], [244], [321], [347], [393], [394]).

Chapter 8

Summary and Conclusion

This dissertation is composed of eight chapters in which the research has been carried out. This work and the main achievements throughout this study are summarized in this chapter.

8.1 Summary

The power series f(z) in a complex variable with real or complex coefficients is defined by

$$f(z) = \sum_{k=1}^{\infty} a_k z^k, \tag{8.1}$$

which is convergent on the open disk $D(0,R) = \{z \in \mathbb{C} : |z| < R\}, R > 0$, where R is called the radius of convergence of the series (8.1). If an analytic function, which is defined by the convergent power series (8.1), is univalent on the open unit disk D(0,1) and normalized by the conditions f(0) = 0 and f'(0) = 1, then we have the power series of the form

$$f(z) = z + \sum_{k=2}^{\infty} a_k z^k, \tag{8.2}$$

for any $z \in D(0,1)$.

The classical Cauchy-Bunyakovsky-Schwarz (CBS) inequality for two sequences of complex numbers states that

$$\left| \sum_{k=1}^{n} a_k b_k \right|^2 \le \sum_{k=1}^{n} |a_k|^2 \sum_{k=1}^{n} |b_k|^2, \tag{8.3}$$

for $a_k, b_k \in \mathbb{C}$, $k \in \{1, 2, ..., n\}$, with equality holds in (8.3) if and only if there is a complex number $c \in \mathbb{C}$ such that $a_k = c\overline{b}_k$ for any $k \in \{1, 2, ..., n\}$. A direct generalization of the (CBS)-inequality (8.3) provided by Hölder [201] in 1889, is

$$\left| \sum_{k=1}^{n} a_k b_k \right| \le \left(\sum_{k=1}^{n} |a_k|^p \right)^{1/p} \left(\sum_{k=1}^{n} |b_k|^q \right)^{1/p}, \tag{8.4}$$

where p and q satisfy the equation 1/p + 1/q = 1 with p > 1. The equality occurs in (8.4) if and only if the sequences $\{|a_k|^p\}$ and $\{|b_k|^q\}$, $k \in \{1, 2, ..., n\}$ are proportional. This inequality is well-known as the Hölder inequality.

In the real case, the (CBS) and Hölder inequalities can be derived by utilising the Jensen inequality for certain underlying convex functions (see [422], [301, p. 457], [381, p. 63-64]). The Jensen inequality, which connects with the notion of convexity of functions, asserts that

$$f\left(\frac{1}{P_n}\sum_{j=1}^n p_j x_j\right) \le \frac{1}{P_n}\sum_{j=1}^n p_j f\left(x_j\right) \tag{8.5}$$

provided that $f: I \subseteq \mathbb{R} \to \mathbb{R}$ is a convex function on $I, x_j \in I$ such that $x_j \geq 0$, $j \in \{1, 2, ..., n\}$, $p_j \geq 0$ with $P_n := \sum_{j=1}^n p_j > 0$, $n \geq 2$. The equality holds in (8.5) if and only if $x_j = x_k$ for all $j, k \in \{1, 2, ..., n\}$.

The (CBS)-inequality has been generalized to integrals and inner product spaces. It can be generalized for functions defined by the power series as well. These analogous inequalities are highlighted in the following:

$$|f(zw)|^2 \le f(|z|^2) f(|w|^2),$$
 (8.6)

for $zw, |z|^2, |w|^2 \in D(0, R)$, where f is a function defined by the power series

(8.1) with nonnegative coefficients, and

$$|f(zw)|^2 \le f_A(|z|^2) f_A(|w|^2),$$
 (8.7)

for $zw, |z|^2, |w|^2 \in D(0, R)$, where f is a function defined by the power series (8.1) with real coefficients. Similarly, the analogous versions of the Hölder inequality for functions defined by the power series (8.1) can be stated as follows:

$$f(zw) = f^{1/p}(|z|^p) f^{1/q}(|w|^q)$$
(8.8)

and

$$f(zw) = f_A^{1/p}(|z|^p) f_A^{1/q}(|w|^q),$$
 (8.9)

for $zw, |z|^2, |w|^2 \in D(0, R)$.

In the literature, many results concerning generalizations, extensions, refinements, etc., of the classical (CBS), Hölder and Jensen type inequalities, have been established by a number of remarkable researchers over the years. However, most of the established results are discrete and involve finite sums. In this study, we derive some inequalities related to the (CBS), Hölder and Jensen type for the power series. In particular, some refinements, improvements, etc., of the inequalities (8.6) - (8.9) have been developed by utilising some tools that have been available in the literature.

8.2 Main Achievements

The main contributions of this dissertation are in Inequalities Theory and Univalent Function Theory: that is, to develop some inequalities involving the power series functions (8.1) and to investigate some properties of functions (8.2) in certain subclasses of analytic and univalent functions.

In the Theory of Inequalities, this dissertation contributes, firstly, in the improvements, as well as the refinements of the (CBS)-type inequality for functions defined by the power series with real or nonnegative coefficients. Applications for some fundamental functions such as exponential, logarithm, trigonometric and hyperbolic functions are highlighted. Some of the refinements, which are

contained in Chapter 3, generalize the results by Cerone and Dragomir [89]. The results in the first part of Chapter 4 also contribute to the improvements of the (CBS)-type inequality for the power series. Particularly, the results that are related to classical Young's inequality provide an improvement of the Hölder's type inequality for the power series with real coefficient. Whereas the results in the second part of Chapter 4 contribute to the Jensen type inequalities through their improvements and refinements, as well as their reverses of the Jensen type inequalities for the real power series with positive coefficients. Lastly, the results in Chapter 5 give some particular inequalities involving special functions such as polylogarithm, hypergeometric, Bessel and Modified Bessel functions of the first kind, that have been derived from some of the results in Chapter 3 and Chapter 4.

This dissertation has also contributed in the study of Univalent Function Theory. The results in Chapter 7 give some new properties of functions which are analytic and univalent in the unit disk. The coefficients inequalities and Fekete-Szegő theorem, which provide the univalence properties of certain subclasses of analytic and univalent functions, have been investigated.

In addition, the results established in this dissertation will contribute to the development of the new problems of these areas and the related topics, and will explore further applications in various fields of pure and applied mathematics.

- M. Abouzahra and L. Lewin, The polylogarithm in algebraic number fields,
 J. Numb. Theory 21 (1985), 214-244.
- [2] S. A. Abramov, Power series and linear differential equations, Proceed. of the 21st Internat. Symp. on Symbolic and Algebraic Computation (2008), 1-2.
- [3] S. Abramovich, B. Mond and J. E. Pečarić, Sharpening Hölder's inequality,
 J. Math. Anal. Appl. 196 (1995), 1131-1134.
- [4] M. Abramowitz and I. Stegun, Handbook of Mathematical Functions with Formulas, Graphs and Mathematical Tables, Edition 1972, Dover Publications, New York, 1972.
- [5] R. P. Agarwal, K. Perera and S. Pinelas, An Introduction to Complex Analysis, Springer, New York, 2011.
- [6] L. V. Ahlfors, Complex Analysis, 2nd Ed., McGraw Hill Book Comp., New York, 1966.
- [7] L. V. Ahlfors, Conformal Invariants. Topics in Geometric Function Theory, Mc Graw-Hill, New-York, 1973.
- [8] O. P. Ahuja and H. Silverman, Convolutions of prestarlike functions, Internat. J. Math. and Math. Sci. 6 (1) (1983), 59-68.
- [9] J. M. Aldaz, Strengthened Cauchy-Schwarz and Hölder inequalities, J. Ineq. Pure and Appl. Maths. **10** (4) (2009), 1-13.
- [10] J. M. Aldaz, A stability version of Hölder's inequality, J. Math. Anal. Appl. **342** (2) (2008), 842-852.

[11] J. M. Aldaz, Sharp bounds for the difference between the arithmetic and geometric means, Arch. Math. 99 (2012), 393-399.

- [12] J. M. Aldaz, Concentration of the ratio between the geometric and arithmetic means, J. Theoretical Probability 23 (2) (2010), 498-508.
- [13] J. M. Aldaz, Comparison of differences between arithmetic and geometric means, Tamkang J. Maths. **42** (4) (2011), 453-462.
- [14] J. M. Aldaz, Self-improvement of the inequality between arithmetic and geometric means, J. Math. Inequal. 3 (2) (2009), 213-216.
- [15] J. W. Alexander, Functions which map the interior of the unit circle upon simple regions, Ann. of. Math. 17 (1915-1916), 12-22.
- [16] J. W. Alexander, Functions which map the interior of the unit circle upon single region', Ph.D Thesis, Princeton Univ., New Jersey, USA, 1915.
- [17] F. M. Al-Oboudi, On univalent functions defined by a generalized Sălăgean operator, Internat. J. Math. Math. Sci. 27 (2004), 1429-1436.
- [18] O. Altintaş, H. Irmak, S. Owa and H. M. Srivastava, Coefficient bounds for some families of starlike and convex functions of complex order, Appl. Math. Letters 20 (12) (2007), 1218-1222.
- [19] H. Alzer, A refinement of the Cauchy-Schwarz inequality, J. Math. Anal. Appl. 168 (1992), 596-604.
- [20] H. Alzer, On the Cauchy-Schwarz inequality, J. Math. Anal. Appl. 234 (1999), 6-14.
- [21] T. Ando, Matrix Young inequality, Oper. Theory Adv. Appl. 75 (1995), 33-38.
- [22] G. D. Anderson, M. K. Vamanamurthy and M. Vuorinen, *Generalized convexity and inequalities*, J. Math. Anal. Appl. **335** (2) (2007), 1294-1308.
- [23] G. D. Anderson, R. W. Barnard, K. C. Richards, M. K. Vamanamurthy and M. Vuorinen, *Inequalities for zero-balanced hypergeometric functions*, Trans. Amer. Math. Soc. **347** (1995), 1713-1723.

[24] D. R. Anderson, S. Noren and B. Perreault, Young's integral inequality with upper and lower bounds, Elect. J. Different. Equat. 2011 (74) (2011), 1-10.

- [25] G. E. Andrews, R. Askey and R. Roy, Special Function, Encyclopedia of Mathematics and Its Application, Vol. 71, Cambridge Univ. Press, Cambridge, 1999.
- [26] L. C. Andrews, Field Guide to Special Functions for Engineers, Spie Press, Bellingham, 2011.
- [27] D. Andrica and I. Raşa, The Jensen inequality: refinements and applications, Anal. Numér. Théor. Approx. 14 (2) (1985), 105-108.
- [28] B. Anglès, On some p-adic power series attached to the arithmetic of $Q(\zeta_p)$, J. Numb. Theory **122** (2007), 221-246.
- [29] F. Antonelli and S. Scarlatti, *Pricing options under stochastic volatility: a power series approach*, Finance Stoch **13** (2009), 269-303.
- [30] R. Apéry, Irrationalité de $\zeta(2)$ et $\zeta(3)$, Astérisque **61** (1979), 11-13.
- [31] Archimedes, The Work of Archimedes: Quadrature of parabola (translated by T. L. Heath), Cambridge University Press, Cambridge, 1897.
- [32] J. Arnad and C. Haenel. Pi Unleashed, Springer-Verlag, Berlin, 2006.
- [33] F. Aroca, J. Cano and F. Jung, *Power series solutions for non-linear PDE's*, Proceed. of the 2003 Inter. Symp. on Symbolic and Algebraic Computation (2003), 15-22.
- [34] E. Artin, The Gamma Function, Rinehard and Winston, New York, 1964.
- [35] F. G. Avkhadiev, S. Pommerenke and K. -J. Wirths, *Sharp inequalities for coefficient of concave schlicht functions*, Commentarii Mathematici Helvetici **81** (2006), 801-807.
- [36] F. G. Avkhadiev and K. -J. Wirths, Concave schlicht functions with bounded opening angle at infinity, Lobachevskii J. 17 (2005), 3-10.
- [37] F. G. Avkhadiev and K-J. Wirths, The conformal radius as a function and its gradient image, Israel J. Math. 145 (2005), 439-374.

[38] L. K. Babadzanjanz and D. R. Sarkissian, Taylor series method for dynamic systems with control: convergence and error estimates, J. Math. Sci. 139 (2006), no. 6, 1-26.

- [39] K. O. Babalola, Bounds on the coefficients of certain analytic and univalent functions, Mathematica, Tome **50** (73) (2008), 39-148.
- [40] K. O. Babalola and T. O. Opoola, Radius problems for a certain class of analytic functions involving the Sălăgean derivative, in Advances in Inequalities for Series, edited by S. S. Dragmir and A. Sofo (2008), 15-21.
- [41] K. O. Babalola and T. O. Opoola, Integrated integral transform of Carathéodory functions and their applications to analytic and univalent functions, Tamkang J. Math. 37 (4) (2006), 355-366.
- [42] K. O. Babalola and T. O. Opoola, On the coefficients of a certain class of analytic functions, in Advances in Inequalities for Series, edited by S. S. Dragmir and A. Sofo (2008), 1-13.
- [43] A. K. Bag, Madhava's sine and cosine series, Indian J. History Sci. 11 (1) (1976), 54-57.
- [44] J. Bak and D. J. Newman, *Complex Analysis*, 3rd Ed., Springer, New York, 2011.
- [45] G. B. Arfken and H. J. Weber, *Mathematical Methods for Physicists*, Acad. Press, Burlington, 2005.
- [46] W. N. Bailey, *Generalized Hypergeometric Series*, Cambridge University Press, Cambridge, 1935.
- [47] R. Balasunramanian, S. Ponnusamy and M. Vourinen, Functional inequalities for the quotients of hyperbolic functions, J. Math. Anal. Appl. 218 (1998), 256-268.
- [48] R. Barbieri, J. A. Mignaco and E. Remiddi, Electron from factors up to fourth order, I and II, Nuovo Cimento A 11 (1972), 824-916.
- [49] A. Baricz, Functional inequalities involving special functions II, J. Math. Anal. Appl. 327 (2007), 1202-1213.

[50] A. Baricz, Functional inequalities involving Bessel and modified Bessel functions of the first kind, Expo. Math. 26 (2008), no. 3, 279-293.

- [51] R. W. Barnard and K. C. Kendall, On inequalities for hypergeometric analogues of the arithmetic-geometric mean, J. Inequal. Pure Appl. Math. 8 (2007), no. 3, 1-12.
- [52] R. W. Barnard, On applications of hypergeometric functions, J. Comp. Appl. Math. 105 (1999), 1-8.
- [53] E. W. Barnes, A new development in the theory of hypergeometric functions, Proc. London Math. Soc. 6 (1908), 141-177.
- [54] I. E. Bazilevič, On a case of integrability in quadratures of the Loewner-Kufarev equation (Russian), Mat. Sb **37** (79) (1955), 471-476.
- [55] E. F. Beckenbach and R. Bellman, *Inequalities*, Springer-Verlag, Berlin, 1961.
- [56] E. F. Beckenbach, An inequality of Jensen, Amer. Math. Monthly 53 (1946), 501-505.
- [57] P. R. Beesack, On inequalities complementary to Jensen's, Canad. J. Math. 35 (2) (1983), 324-338.
- [58] R. D. Berman and H. Silverman, Coefficient inequalities for a subclass of starlike functions, J. Math. Anal. Appl. 107 (1) (1985), 197-205.
- [59] B. C. Berndt, Ramanujan's Notebook, Part II, Springer-Verlag, New York, 1989.
- [60] S. D. Bernardi, A survey of the development of the theory of schlicht functions, Duke. Math. J. 19 (1952), 263-287.
- [61] S. D. Bernardi, A Bibliography of Schlicht Functions, New York Univ., Courant Inst. of Math. Sci., Part I in 1966, Part II in 1975.
- [62] S. D. Bernardi, Convex and starlike univalent functions, Trans. Amer. Math. Soc. 135 (1969), 429-446.

[63] R. Bhatia and C. Davis, A Cauchy-Schwarz inequality for operators with applications, Linear Algebra and Its Applications 223/224 (1995), 119-129.

- [64] B. Bhowmik, S. Ponnusamy and K. -J. Wriths, On the Fekete-Szegő problem for concave univalent functions, J. Math. Anal. Appl. 373 (2011), 432-438.
- [65] B. Bhowmik, S. Ponnusamy and K. -J. Wriths, Charaterization the pre-Schwarzian norm estimate for concave univalent functions, Monatsh Math. 161 (2010), 59-75.
- [66] A. M. Bica, V. A. Caus, I. Fechete and S. Muresan, Application of the Cauchy-Bunyakovsky-Schwarz's inequality to an optimal property for cubic splines, J. Comput. Anal. Appl. 9 (1) (2007), 43-53.
- [67] L. Bieberbach, Über die koeffizienten derjenigen potenzreihen, welche eine schlichte abbildung des einheitskreises vermitteln, Sitz. Ber. Preuss. Akad. Wiss (1916), 940-955.
- [68] M. Biernacki, Les Functions Multivalentes, Hermann, Paris, 1938.
- [69] D. Blesu, On the n-close-to-convex functions with respect to a convex set
 (1), Math. Revue d'analyse numerique et de theorie de l'approximation 28
 (51) (1) (1986), 9-19.
- [70] E. D. Bloch, *The Real Number and Real Analysis*, Springer, New York, 2011.
- [71] S. Bloch, Applications of the dilogarithm function in algebraic K-theory and algebraic geometry, in: M. Nagata ed., Proc. Int. Symp. Algeb. Geometry, Kinokuniya Book-Store Co., Kyoto, 1978.
- [72] V. I. Bogachev, Measure Theory I, Springer-Verlag, New York, 2007.
- [73] C. B. Boyer and U. C. Merzbach, A History of Mathematics, John Wiley & Sons, New York, 1991.
- [74] D. A. Brannan and J. G. Clunie (editors), Aspects of Contemporary Complex Analysis, Acad. Press, London, 1980.

[75] L. de Brange, A proof of the Bieberbach conjecture, Acta. Math. **154** (1985), 137-152.

- [76] M. K. Broadhead, First-order error propagation formulae for the Gassmann relations, Geophysical Prospecting **56** (5) (2008), 729-735.
- [77] I. N. Bronshtein, K. A. Semendyayev, G. Musiol and H. Muehlig, *Handbook of Mathematics*, 5th Ed., Springer, New York, 2005.
- [78] N. G. de Bruijn, *Problem 12*, Wisk. Opgaven **21** (1960), 12-14.
- [79] P. S. Bullen, *Handbook of Means and Their Inequalities*, Kluwer Acad. Publisher, Dordrecht/Boston/London, 2003.
- [80] P. S. Bullen, A Dictionary of Inequalities, Addison-Wesley Longman, London, 1998.
- [81] V. Y. Buniakowski, Sur quelques inégalités concernant les intégrales aux différences finies, Mem. Acad. Sci. St. Petersburg (VII) 1 (7) (1859), no. 9, 1-18.
- [82] M. L. Buzano, Generalizzazione della diseguaglianza di Cauchy-Schwarz (Italian), Rend. Sem. Mat. Univ. e Politech. Torino 31 (1971/73), 405-409 (1974).
- [83] M. Çağlar, E. Deniz and H. Orhan, Coefficient bounds for a subclass of starlike functions of complex order, Appl. Math. Comput. 218 (3) (2011), 693-698.
- [84] C. Carathéodory, Über den variabilitätsbereich der Fourier'schen konstanten von positiven harmonischen funktionen, Rend. Circ. Mat Palermo 32 (1911), 193-217.
- [85] L. Carlitz, Eulerian numbers and polynomials, Math. Mag. 33 (1959), 247-260.
- [86] B. C. Carlson, Some inequalities for hypergeometric functions, Proc. Amer. Math. Soc. 16 (1966), 32-39.
- [87] J. A. Carroll, R. Cordner and C. J. A. Evelyn, A new extension of Hölder's inequality, Einseignement Math. 16 (1970), 69-71.

[88] A. L. Cauchy, Cours d'Analyse del'École Royale Polytechnique, I^{re} partie, Analyse Algébrique, Paris, 1821.

- [89] P. Cerone and S. S. Dragomir, Some applications of de Bruijn's inequality for power series, Integral Transforms and Special Functions 18 (6) (2007), 387-396.
- [90] P. Cerone, On Young's inequality and its reverse for bounding the Lorenz curve and Gini mean, J. Math. Inequalities 3 (3) (2009), 369-381.
- [91] P. Cerone and S. S. Dragomir, Mathematical Inequalities, CRC Press, London/New York, 2011.
- [92] A. Chaieb, Formal power series, J. Autom Reasoning 47 (2011), 291-318.
- [93] F. C. Chang and H. Mott, Operations on power series and polynomials, J. Franklin Inst. **299** (1975), 227-243.
- [94] S. Chester, F. Charlette and C. Menevèau, Dynamic model for LES without test filtering: quantifying the accuracy of Taylor series approximations, Theoretical and Comptut. Fluid Dynamics, 15 (2001), No. 3, 165-181.
- [95] M. C. R. Cockett and G. Doggett, Maths for Chemist Vol 1, Royal Soc. Chemistry, Cambridge, 2003.
- [96] M. C. R. Cockett and G. Doggett, Maths for Chemist Vol II, Royal Soc. Chemistry, Cambridge, 2003.
- [97] H. Cohen, Numerical Approximations Methods, Springer, New York, 2011.
- [98] H. Cohen, *Mathematics for Scientists and Engineers*, Practice-Hall Publ. Co., Englewood Cliffs, 1992.
- [99] L. Comtet, Advanced Combinatorics: The Art of Finite and Infinite Series Expansion, rev. enl. ed., Netherlands: Reidal, Dordrecht/Holland, 1974.
- [100] G. M. Cordeiro, M. G. Andrade and M. de Castro, Power series generalized nonlinear models, Comput. Stat. Data Anal. 53 (2009), 1155-1166.
- [101] I. Csiszár, A note on Jensen's inequality, Studia Sci. Math. Hungar. 1 (1966), 185-188.

[102] A. Cuyt, V. B. Petersen, B. Verdonk, H. Waadeland and W. B. Jones, Handbook of Continued Fractions for Special Functions, Springer, New York, 2008.

- [103] L. Cruz and C. Pommerenke, On concave univalent functions, Complex Var. Elliptic Equat. **52** (2007), 153-159.
- [104] Z. Cvetkovski, *Inequalities: Theorem, Techniques and Selected Problems*, Springer, New York, 2012.
- [105] D. Cvijović, New integral representations of the polylogarithm function, Proc. R. Soc. A. 463 (2007), 897-905.
- [106] S. G. Dani, Ancient Indian mathematics, a conspectus, Resonance 17 (3) (2012), 236-246.
- [107] M. Darus, Fekete-Szegő functional for functions in $B_n(\beta)$, Inst. Math. Comput. Sci. Math. Ser. 17 (2) (2004), 129-135.
- [108] M. Darus and D. K. Thomas, On the Fekete-Szegő theorem for close-to-convex functions, Math. Japonica 44 (3) (1996), 507-511.
- [109] M. Darus and D. K. Thomas, The Fekete-Szegő theorem for close-to-convex functions, Math. Japonica 47 (1) (1998), 125-132.
- [110] M. Darus and D. K. Thomas, On the Fekete-Szegő theorem for strongly close-to-convex functions, Scientiae Mathematicae 3 (2) (2000), 201-212.
- [111] P. J. Davis, Leonhard Euler's integral: a historical profile of the gamma function, Amer. Math. Monthly 66 (1959), 849-869.
- [112] D. E. Daykin and C. J. Eliezer, Generalization of Hölder's and Minkowski's inequalities, Proc. Camb. Phil. Soc. **64** (1968), 1023-1027.
- [113] J. Denef and L. Lipshitz, Power series solutions of algebraic differential equations, Math. Ann. **267** (1984), 213-238.
- [114] J. B. Diaz and F. T. Metcalf, An analytic proof of Young's inequality, Amer. Math. Monthly 77 (6) (1970), 603-609.
- [115] J. Dieudonne, Sur les fonctions univalentes, C. R. Acad. Sci. Paris 192 (1931), 1148-1150.

[116] S. S. Dragomir, Some refinements of Cauchy-Schwarz's inequality, G. M. Metod. 10 (1980), 93-95.

- [117] S. S. Dragomir, Discrete Inequalities of the Cauchy-Bunyakovsky-Schwarz Type, Nova Sci. Publ., Inc., New York, 2004.
- [118] S. S. Dragomir, Some refinements of Schwarz inequality, Simpozionul de Mathematisi Aplicații, Timișoara, Romania, 1-2 Noiembrie (1985), 13-16.
- [119] S. S. Dragomir, Bounds for the normalized Jensen functional, Bull. Austral. Math. Soc. **74** (2006), no. 3, 417-478.
- [120] S. S. Dragomir, Advances in Inequalities of the Schwarz, Grüss and Bessel Type in Inner Product Spaces, Nova Sci. Publ., Inc., New York, 2005.
- [121] S. S. Dragomir, Advances in Inequalities of the Schwarz, Triangle and Heisenberg Type in Inner Product Spaces, Nova Sci. Publ., Inc., New York, 2007.
- [122] S. S. Dragomir, Inequalities of Cauchy-Buniakowski-Schwartz's type for positive linear functionals (Romanian), Gaz. Mat. Metod. (Bucharest) 9 (1988), 162-164.
- [123] S. S. Dragomir, Some inequalities for power series of selfadjoint operators in Hilbert spaces via reverses of the Schwarz inequality, Integ. Transforms Spec. Funct. **20** (10) (2009), 757-767.
- [124] S. S. Dragomir, Refinements of Buzano's and Kurepa's inequalities in inner product spaces, Ser. Math. Inform. **20** (2005), 65-73.
- [125] S. S. Dragomir, Some Schwarz type inequalities for sequences of operators in Hilbert spaces, Bull. Aust. Math. Soc. 73 (1) (2006), 17-26.
- [126] S. S. Dragomir, A new improvement of Jensen's inequality, Indian J. Pure Appl. Math. 26 (10) (1995), 959-968.
- [127] S. S. Dragomir, A survey on Cauchy-Bunyakovsky-Schwarz type discrete inequalities, J. Inequal. Pure Appl. Math. 4 (3) (63) (2003), 1-140.
- [128] S. S. Dragomir, On the Cauchy-Bunyakovsky-Schwarz inequality for sequences in inner product spaces, Math. Inequal. Appl. 3 (2000), 385-398.

[129] S. S. Dragomir, On some refinements of Jensen's inequality and applications, Utilitas Math. 43 (1993), 235-243.

- [130] S. S. Dragomir, An improvement of Jensen's inequality, Bull. Math. Soc. Sci. Math. Roumanie (N. S) 34 (82) (4) (1990), 291-296.
- [131] S. S. Dragomir, On a reverse of Jensen's inequality for isotonic linear functionals, J. Ineq. Pure Appl. Math. 2 (3) (2001), Article 36 (Online: http://jipam.vu.edu.au/v2n3/047_01.html).
- [132] S. S. Dragomir, A further improvement of Jensen inequality, Tamkang J. Math. 25 (1) (1994), 29-36.
- [133] S. S. Dragomir, Two mappings associated with Jensen's inequality, Extracta Math. (Spain) 8 (1993), 102-105.
- [134] S. S. Dragomir and C. J. Goh, A counterpart of Jensen's discrete inequality for differentiable convex mapping and applications in information theory, Math. Comput. Modelling 24 (2) (1996), 1-11.
- [135] S. S. Dragomir and B. Mond, On Cauchy-Bunyakovsky-Schwarz's inequality for sequences or linear operators in Hilbert space, Houston J. Math. 22 (2) (1996), 329-339.
- [136] S. S. Dragomir and J. Sándor, Some generalizations of Cauchy-Bunyakovsky-Schwarz's inequality (Romanian), Gaz. Mat. Metod. (Bucharest) 11 (1990), 104-109.
- [137] S. S. Dragomir and F. P. Scarmozzino, A refinement of Jensen's discrete inequality for differentiable convex functions, East Asian Math. J. 20 (2) (2004), 161-168.
- [138] S. S. Dragomir and N. M. Ionescu, Some refinements of Cauchy-Buniakowski-Schwartz inequality for sequences, Symp. Math. Appl. Timişoara (1989), 79-82.
- [139] S. S. Dragomir and N. M. Ionescu, Some converse of Jensen's inequality and applications, Rev. Anal. Numér. Théor. Approx. 23 (1) (1994), 71-78.

[140] S. S. Dragomir and C. E. M. Pearce, Selected topics on Hermite-Hadamard inequalities and applications, RGMIA Monographs (Victoria University, 2000), available online: http://www.rgmia.org/monographs).

- [141] S. S. Dragomir, J. E. Pečarić and L. E. Persson, *Properties of some functionals related to Jensen's inequality*, Acta Math. Hung. **70** (1-2) (1996),129-143.
- [142] J. L. Dupont, The dilogarithm as a characteristic class for flat bundles, J. Pure Appl. Algebra 44 (1987), 137- 168.
- [143] J. L. Dupont, On polylogarithms, Nagoya Math. J. 114 (1989), 1-20.
- [144] P. L. Duren, Univalent Functions, Springer-Verlag, New York, 1983.
- [145] P. L. Duren, Coefficients of univalent functions, Amer. Math. Soc. 83 (1977), no. 5, 891-911.
- [146] H. M. Edwards, Riemann Zeta Function, Academic Press, Inc., London, 1974.
- [147] C. J. Eliezer and D. E. Daykin, Generalizations and applications of Cauchy-Schwarz inequality, Quart. J. Math. Oxford Ser. 18 (2) (1967), 357-360.
- [148] E. A. Eljamal and M. Darus, On Fekete-Szegő problems for certain subclass of analytic functions, Inter. Jour. Pure and Appl. Math. **71** (4) (2011), 571-580.
- [149] A. Erdélyi, W. Magnus, F. Oberhettinger and F. G. Tricomi, *Higher Transcendental Functions*, Vol. 1, McGraw-Hill Book Comp. Inc., New York, 1953.
- [150] G. Faber, Neuer beweis eines Koebe-Bieberbachschen satzes über conforme abbildung, Sitzgsber. Bayer Acad. Wiss. München (1916), 39-42.
- [151] I. Faisal and M. Darus, On Fekete-Szegő theorem for certain analytic functions, Journal of Ouality Measurement and Analysis 5 (1) (2009), 85-91.
- [152] M. Fekete and G. Szegő, Eine bemerkung über ungerade schlichte funcktionen, J. London Math. Soc. 8 (1933), 85-89.

[153] A. S. Focas and L-Y. Sung, Generalized Fourier transforms, their non linearization and the imaging of the brain, Not. Amer. Math. Soc. **52** (2005), 1178-1192.

- [154] J. A. Fox and A. C. Hearn, Analytic computation of some integrals in fourth order quantum electrodynamics, J. Comp. Phys. 14 (1974), 301-317.
- [155] B. A. Frasin and M. Darus, On the Fekete-Szegö using the Hadamard product, Int. Math. J. 3 (12) (2003), 1297-1304.
- [156] A. Frey and V. Schmidt, Taylor series expansion for multivariate characteristics of classical risk processes, Insurance: Maths. and Economics 18 (1996), 1-12.
- [157] S. Furuichi, Refined Young inequalities with Specht's ratio, J. Egyptian Math. Soc. **20** (2012), 46-49.
- [158] S. Furuichi, On refined Young inequalities and reverse inequalities, J. Math. Inequal. 5 (1) (2011), 21-31.
- [159] S. Furuichi, A refinement of the arithmetic-geometric mean inequality (2009), Preprint: arXiv:0912.5227v1.
- [160] S. Furuichi and N. Minculete, Alternative reverse inequalities for Young's inequality, J. Math. Inequal. 5 (4) (2011), 595-600.
- [161] D. J. H. Garling, *Inequalities: A Journey Into Linear Analysis*, Cambridge Univ. Press, New York, 2007.
- [162] C. F. Gauss, Disquisitiones generales circa seriem infinitam, $1 + \frac{\alpha\beta}{\gamma}x + \frac{\alpha(\alpha+1)\beta(\beta+1)}{\gamma(\gamma+1)1\cdot 2}x^2 + etc.$ (in Latin), Comm. Soc. Regia Sci. Göttingen Rec. **2** (1812), 123-162. (Gauss's original paper can be found in Carl Friedrich Gauss Werke, p. 125).
- [163] M. C. Gauss, G. A. Marinho and M. de Castro, *Power series generalized nonlinear models*, Comput. Stat. Data Anal. **53** (2009), 1155-1166.
- [164] H. R. El-Gawad and D. K. Thomas, The Fekete-Szegö problem for strongly close-to-convex functions, Math. Japonica 44 (3) (1996), 8-13.

[165] H. R. El-Gawad and D. K. Thomas, A subclass of close-to-convex functions, Publ. Inst. Math. 49 (63) (1991), 61-66.

- [166] J. P. Gilman, I. Kra and R. E. Rodríguez, *Complex Analysis: In the Spirit of Lipman Bers*, Springer, New York, 2007.
- [167] M. Godefroy, La fonction Gamma; Théorie, Histoire, Bibliographie, Gauthier-Villars, Paris, 1901.
- [168] D. Gold and D. Pingree, A hitherto unknown Sanskrit work concerning Madhava's derivation of power series for sine and cosine, Historia Sci. 42 (1991), 49-65.
- [169] A. W. Goodman, Univalent Functions, Vol. I and II, Mariner Publ. Co., Tampa, Florida, 1983.
- [170] I. Graham and G. Kohr, Geometric Function Theory in One and Higher Dimension, Marcel Dekkar Inc., New York, 2003.
- [171] J. Gregory, Vera circuli et hyperbola quadratura, Codorini, Padova, 1667.
- [172] J. Gregory, Exercitations Geometricae, Godbid and Pitt, London, 1668.
- [173] A. Z. Grinshpan, Inequalities for formal power series and entire functions,
 J. Math. Anal. Appl. 338 (2008), 1418-1430.
- [174] T. H. Gronwall, Some remarks on conformal representation, Ann. of Math.2 (1914-1915), no. 16, 72-72.
- [175] G. M. Goluzin, Geometric Theory of Functions of a Complex Variable, GITTL, Moscow, 1952.
- [176] J. Hadamard, Etude sur les propriétés des fonctions entières et en particulier d'une fonction considérée par Riemann, J. Math. Pures Appl. 9 (1893), 171-215.
- [177] A. M. Hägglund and P. D. Folkow, Dynamic cylindrical shell equations by power series expansions, Inter. J. Solids and Structures 45 (2008), 4509-4522.
- [178] M. Haiman, Non-commutative rational power series and algebraic generating functions, Europ. J. Combinatorics 14 (1993), 335-339.

- [179] R. M. Hain, Classical polylogarithms, Pure Math. 55 (1994), 3-42.
- [180] S. A. Halim, Coefficients estimates for functions in $B_n(\alpha)$, Inter. J. of Maths and Math. Sci. **59** (2003), 3761-3767.
- [181] S. A. Halim, On a class of analytic functions involving the Sălăgean differential operator, Tamkang J. Maths. 23 (1) (1992), 51-58.
- [182] S. A. Halim, Some Bazilevič functions or order beta, Internat. J. Math. & Math. Sci. 14 (2) (1991), 238-288.
- [183] D. J. Hallenbeck and T. H. MacGregor, Linear Problems and Convexity Techniques in Geometric Function Theory, Pitman Adv. Publ. Program, Boston, 1984.
- [184] M. Harrison and P. Waldron, Mathematics for Economics and Finance, Routledge, New York, 2011.
- [185] G. H. Hardy, J. E. Littlewood and G. Pólya, *Inequalities*, Cambridge Univ. Press, New York, 1952.
- [186] G. H. Hardy, "Hypergeometric Series", Ch. 7 in Ramanujan: Twelve Lectures on Subjects Suggested by His Life and Work, 3rd ed., Chelsea, New York, p. 101-112, 1999.
- [187] J. Havil, *Gamma: Exploring Euler's Constant*, Princeton Univ. Press, Princeton, 2003.
- [188] T. Hayami, S. Owa and H. M. Srivastava, Coefficient inequalities for certain classes of analytic and univalent functions, J. Ineq. in Pure and Appl. Math. 8 (4) (2007), 1-21.
- [189] W. K. Hayman, Research Problems in Functions Theory, Athlone Press, London, 1967.
- [190] W. K. Hayman, Coefficients problems for univalent functions and related function classes, J. London Math. Soc. 40 (1965), 385-406.
- [191] B. He and B. Yang, On a Hilbert-type inequality with a hypergeometric function, Commun. Math. Anal. 9 (2010), no. 1, 84-92.

[192] W. S. He, Generalization of a sharp Hölder's inequality and its application,
 J. Math. Anal. Appl. 332 (2007), 741-750.

- [193] M. Heins, Complex Function Theory, Acad. Press, Burlington, 1968.
- [194] G. Herglotz, Über potenzreihen mit positivem, reelem teil in einheitskreis,
 S.-B. Sächs Akad. Wiss Leipzig Math. -Natur. KI. 63 (1911), 501-511.
- [195] Ch. Hermite, Sur deux limites d'une intégrale définie, Mathesis 3 (1883), 82.
- [196] C. J. Hill, Über die integration logarithmisch-rationaler differentiale, J. Reine Angew. Math. 3 (1828), 101-159.
- [197] J. Hirshleifer, On the theory of optimal investment decision, J. Political Economy 66 (1958), 329-352.
- [198] O. Hirzallah and F. Kittaneh, Matrix Young inequalities for the Hilbert-Schmidt norm, Linear Algebra Appl. 308 (2000), 77-84.
- [199] K. Hoechsmann, The Artin-Hasse power series and p-adic group rings, J. Numb. Theory 39 (1991), 117-128.
- [200] J. Van de Hoeven, Generalized power series solutions to linear partial differential equations, J. Symbolic Computation 42 (2007), 771-791.
- [201] O. L. Hölder, Über einen Mittelwerthssatz, Nachr. Ges. Wiss. Göttingen (1889), 38-47.
- [202] G. 't Hooft and M. Veltman, Scalar one-loop integrals, Nuclear Phys. B 153 (3-4) (1979), 365-401.
- [203] A. Hoorfar and F. Qi, A new refinement of Young's inequality, Math. Inequal. Appl. 11 (4) (2008), 689-692.
- [204] X-L Hu, An extension of Young's inequality and its application, Appl. Math. Comput. **219** (2013), 6393-6399.
- [205] C. Hwang, J. C. Guo and T. Y. Guo, An FFT-based algorithm for 2D power series expansions, Compt. Maths. Appl. 37 (1999), 19-27.

[206] A. Ibrahim and M. Darus, On the Fekete-Szegő problems: to determine the sharp upper bounds for close-to-convex functions, J. Inst. Math. & Comp. Sci. 14 (3) (2001), 207-216.

- [207] A. Ibrahim, S. Owa, M. Darus and Y. Nakamura, Generalization of Sălăgean operator for certain analytic functions, Banach J. Math. Anal. 2 (2008), 16-22.
- [208] A. Ibrahim and S. S. Dragomir, *Power series inequalities via Buzano's result and applications*, Integral Transforms and Special Functions **22** (12) (2011), 867-878.
- [209] A. Ibrahim and S. S. Dragomir, Power series inequalities via a refinement of the Schwarz inequality, Integral Transforms and Special Functions 23 (10) (2012), 769-781.
- [210] A. Ibrahim, M. Darus and S. S. Dragomir, Fekete-Szegő problem for concave univalent functions defined by Sălăgean operator, to appear. Preprint available at: RGMIA Research Report Collection, Vol. 14 (2012), 11 pp., http://www.rgmia.org/issues.php.
- [211] A. Ibrahim, M. Darus and S. S. Dragomir, Coefficient inequalities for a certain subclass of analytic functions, J. Ineq. Special Functions (accepted). Preprint available at: RGMIA Research Report Collection, Vol. 14 (2012), 9 pp., http://www.rgmia.org/issues.php.
- [212] A. Ibrahim, M. Darus, S. S. Dragomir and O. A. F-. Joseph, *Some properties for a certain subclass of univalent functions*, to appear.
- [213] A. Ibrahim, S. S. Dragomir, P. Cerone and M. Darus, *Inequalities for power series with positive coefficients*, J. Inequal. Spec. Funct. **3** (2) (2012), 1-15.
- [214] A. Ibrahim, S. S. Dragomir and M. Darus, *Some inequalities for power series with applications*, Integral Transforms and Special Functions **24** (5) (2013), 364-376.
- [215] A. Ibrahim, S. S. Dragomir and M. Darus, Power Series Inequalities Via Young's Inequality with Applications, Jour. Inequal. Appl. 2013 (2013): 314-326.

[216] A. Ibrahim, S. S. Dragomir and M. Darus, *Power Series Inequalities Related to Young's Inequality and Applications*, Integral Transforms and Speci. Funct. **24** (9) (2013), 700-714.

- [217] A. Ibrahim and S. S. Dragomir, A Survey on Cauchy-Bunyakovsky-Schwarz Inequality for Power Series, accepted as a chapter in the book of H. M. Srivastava Honorary Volume by Springer, to appear.
- [218] A. Ivic, *The Riemann Zeta Function*, John Wiley and Sons, New York, 1985.
- [219] J. M. Jahangiri, A coefficient inequality for a class of close-to-convex function, Math. Japon. 41 (3) (1995), 557-559.
- [220] J. Jaksetić and J. Pečarić, A note on Young inequality, Math. Inequal. Appl. 13 (1) (2010), 43-48.
- [221] J. A. Jenkins, Univalent Functions and Conformal Mappings, Springer-Verlag, Berlin, 1958.
- [222] J. L. W. V. Jensen, Om konvexe Funktioner og Uligheder Mellem Middelvaerdier, Nyt. Tidsskr. Math. 16B (1905), 49-69.
- [223] J. L. W. V. Jensen, Sur les fonctions convexes et les inégalités entre les valeurs moyennes, Acta Math 30 (1906), 175-193.
- [224] G. A. Jones and J. M. Jones, Elementary Number Theory, Springer-Verlag, Berlin, 1998.
- [225] A. Jonquière, Note sur la série $\sum_{n=1}^{\infty} \frac{x^n}{n^s}$, Bull. de la Société Mathématique de France **17** (1889), 142- 152.
- [226] B. D. Jordan, S. D. Ross and R. Westerfield, Fundamentals of Corporate Finance, Irwin/McGraw-Hill, Boston, 2000.
- [227] C. M. Joshi and S. K. Bissu, Some inequalities of Bessel and modified Bessel functions, J. Aust. Math. Soc. A 50 (1991), 333-342.
- [228] S. B. Joshi and N. D. Sangle, New subclass of univalent functions defined by using generalized Sălăgean operator, J. Indones. Math. Soc. 15 (2) (2009), 79-89.

[229] R. Jurasińska and J. Sokół, Some problems for certain family of starlike functions, Math. Comput. Model. **55** (2012), 2134-2140.

- [230] G. Jumarie, From Lagrangian mechanics fractal in space to space fractal Schrödinger's equation via fractional Taylor's series, Chaos, Solitons and Fractals 41 (2009), 1590-1604.
- [231] K. Th. Kalveram, Power series and neural net computing, Neurocomputing 5 (1993), 165-174.
- [232] S. Kanas, New subclasses of univalent functions, Folia Scientarium Univ. Tech. Resoviensis **60** (1989), 45-59.
- [233] S. Kanas and J. O-Puzio, Sharp distortion theorems for some subclass of close-to-convex functions, Appl. Math. Comput. 218 (3) (2011), 843-850.
- [234] S. Kanas and H. E. Darwish, Fekete-Szegö problem for starlike and convex functions of complex order, Appl. Math. Lett. 23 (2010), 777-782.
- [235] W. Kaplan, Close-to-convex schlicht functions, Michigan Math. J. 1 (1952), 169-185.
- [236] G. P. Kapoor and A. K. Mishra, Coefficient estimates for inverses of starlike functions of positive order, J. Math. Anal. Appl. 329 (2) (2007), 922-934.
- [237] J. N. Kapur, Inequalities: Theory, Applications and Measurements, Math. Sci. Trust Soc., New Delhi, 1997.
- [238] V. Karunakaran and K. Padma, Functions of bounded radius rotation, Indian J. Pure Appl. Math. 12 (1981), 621-627.
- [239] N. D. Kazarinoff, Analytic Inequalities, Dover Publication, Inc., Mineola/New York, 2003.
- [240] F. R. Keogh and E. P. Merkes, A coefficient inequality for certain classes of analytic functions, Proc. Amer. Math. Soc. **20** (1969), 8-12.
- [241] F. R. Keogh and S. S. Miller, On the coefficients of Bazilevič functions, Amer. Math. Soc. **30** (3) (1971), 492-496.

[242] J. P. Kennealy and W. M. Moore, A numerical method for chemical kinetics modelling based on the Taylor series expansion, J. Physical Chem. 81 (1977), no. 25, 2413-2419.

- [243] S. M. Khairnar and M. More, On a certain subclass of analytic functions involving the Al-Oboudi differential operator, J. Ineq. Pure Appl. Math. 10 (2) (2009), 11 pp.
- [244] A.A. Khaled and M. Darus, On the Fekete-Szegö theorem for the generalized Owa-Srivastava operator, *Proceedings of the Romanian Academy-series A: Mathematics*, **12**(3)(2011), 179-188.
- [245] L. Kharevych, B. Springborn and P. Schröder, *Discrete conformal map*pings via circle patterns, ACM Transactions on Graphics (2012), 1-27.
- [246] K. V. Kheruntsyan, J. C. Jaskula, P. Deuar, M. Bonneau, P. G. Partidge, J. Ruaudel, R. Lopes, D. Boiron and C. I. Westbrook, *Violation of the Cauchy-Schwarz inequality with matter waves*, Phys. Rev. Letters 108 (26) (2012), 5-9.
- [247] A. Khrennikov, Quantum mechanics as an approximation of statistical mechanics for classical fields, Reports on Math. Physics 60 (2007), No. 3, 453-484.
- [248] Y. I. Kim and X. Yang, Generalizations and refinements of Hölder's inequality, Appl. Maths Letters 25 (2012), 1094-1097.
- [249] A. N. Kirillov, *Dilogarithm identities*, Prog. Theor. Phys. **118** (1995), 61-142.
- [250] F. Kittaneh and Y. Manasrah, Improved Young and Heinz inequality for matrices, J. Math. Anal. Appl. 361 (2010), 262-269.
- [251] O. Kiymaz and S. Mirasyedioğlu, A new symbolic computation for formal integration with exact power series, Appl. Maths. Comput. **162** (2005), 215-224.
- [252] M. Kline, Mathematical Thought from Ancient to Modern Times, Vol. 1, Oxford University Press, New York, 1990.

[253] P. Koebe, Über die uniformisierung beliebiger analytischer kurven, Nachr. Kgl. Ges. Wiss. Göttingen Math. Phys. (1907), 191-210.

- [254] W. Koepf, On the Fekete-Szegö problem for close-to-convex functions, Proc. Amer. Math. Soc. **101** (1987), 89-95.
- [255] W. Koepf, On the Fekete-Szegö problem for close-to-convex functions II, Proc. Amer. Math. Soc. 49 (1987), 490-533.
- [256] W. Koepf, Power series in computer algebra, J. Symbolic Comput. 11 (1992), 581-603.
- [257] W. Koepf, Examples for the algorithm calculation of formal Puisuex, Laurent and power series, SIGSAM Bull. 27 (1993), 20-23.
- [258] G. C. Kokkorakis, Power series expansions for spheroidal wave functions with small arguments, J. Comput. Appl. Maths. 139 (2002), 95-127.
- [259] T. Kosem, Matrix versions of Young's inequality, Math. Inequal. Appl. 12 (2) (2009), 239-254.
- [260] K. M. Koti, Some inequalities in hypergeometric functions using statistical techniques, Indian J. Pure Appl. Math. 22 (1991), no. 5, 389-396.
- [261] J. Kowalczyk and E. Leś-Bomba, On a subclass of close-to-convex functions, Appl. Math. Letters 23 (10) (2010), 1147-1151.
- [262] S. G. Krantz, Geometric Function Theory, Explorations in Complex Analysis, Birkhäuser, Boston, 2006.
- [263] M. Krnic, Z. Tomovski and J. E. Pečarić, Hilbert inequalities related to generalized hypergeometric functions _mF_n, Math. Balkanica (N. S.) 22 (2008), no. 3-4, 307-322.
- [264] J. Krzyz and Z. Lewandowski, On some resent results in the theory of analytic functions, Colloq. Math. 11 (1964), 147-150.
- [265] R. Kühnau, Handbook of Complex Analysis: Geometric Function Theory Vol. 1, Elsevier, New York, 2002.
- [266] R. Kühnau, Handbook of Complex Analysis: Geometric Function Theory Vol. 11, Elsevier, New York, 2005.

[267] E. E. Kummer, Über die transzdenten, welche aus wiederholten integrationen rationaler funktionen entstehen, J. Math (Crelle) **21** (1840), 74-90, 193-225, 328-371.

- [268] E. E. Kummer, De integralibus definitis et seriebus infinitis, J. Reine Angew. Math. 17 (1837), 210-227.
- [269] S. Kurepa, On the Bunyakovsky-Cauchy-Schwarz inequality, Glasnik Mat. Ser. III 1 (21) (1966), 147-158.
- [270] N. Kurokawa and M. Wakayama, Cauchy-Schwarz type inequalities for categories, Kyushu J. Math **57** (2) (2003), 325-332.
- [271] E. G. Kwon, Extension of Hölder's inequality (1), Bull. Austral. Math. Soc. **51** (1995), 369-375.
- [272] G. Labelle and L. Laforest, Combinatorial variations on multidimensional quadtress, J. Combin. Theory Ser. A 69 (1) (1995), 1-16.
- [273] P. Lah and M. Ribarič, Converse of Jensen's inequality for convex functions, Mat. Fizika Series. No. 412-420 (1973), 201-205.
- [274] E. Landau, Darstellung and Begründung einiger neurerer Ergrbnisse der-Funktionenthrorie, Zweiten Aufl. Springer, Berlin, 1929.
- [275] M. Lara, A. Elipe and M. Palacios, Automatics programming of recurrent power series, Maths. Computers in Simulations 49 (1999), 351-362.
- [276] J. A. Leavitt, Methods and applications of power series, Math. Comput. **20** (1966), 46-52.
- [277] M. H. Lee, Polylogarithms and Riemann's ζ functions, Phys. Rev. E **56** (1997), 1518-1520.
- [278] G. W. Leibniz, De vera proportione circulis ad quadratum circumscriptum in numeris rationalibus expressa, Acta. Eruditorum (1682), 41-46.
- [279] G. W. Leibniz, Quadratura arithmetria cummunis cinicarum sectionum conicarum quaecentrum habent indeque ducta trigonomeria cononica ed quantumcumque in numeris exactitudiner a tabularm necessitate liberate,

- cum use speciali a linearm rhomborum nautuca, aptatuque illi, planisphearium, Acta. Eruditorum (1691), 178-182.
- [280] G. W. Leibniz, *Mathematische Schriften III/I* (edited by C. Gerhardt), Georg Olms Verlag, Hildescheim/New York, 1971.
- [281] L. Lewin, Polylogarithms and Associated Functions, North Holland Publ. Co., New York, 1981.
- [282] L. Lewin, *Dilogarithms and Associated Functions*, MacDonald, London, 1981.
- [283] R. J. Libera, Some classes of regular univalent functions, Proc. Amer. Math. Soc. 16 (1965), 755-758.
- [284] R. J. Libera and E. J. Zlotkiewicz, Coefficient bounds for the inverse of a function with derivative in ρ , Proceed. Amer. Math. Soc. 87 (2) (1983), 251-257.
- [285] L. J. Lin and S. Owa, Properties of the Sălăgean operator, Georgian Math. J. 5 (4) (1998), 361-366.
- [286] Y. Lin, Generalizations of Jensen's inequalities, J. Fuzhou Univ. Nat. Sci. Ed. 22 (1994), 13-19.
- [287] E. Lindelöf, Mémoire sur certaines inégalitis dans la théorie des functions monogénses et sur quelques propriétés nouvelles de ces fonctions dans le voisinage d'un point singulier essentiel, Ann. Soc. Sci. Fenn. **37** (7) (1909), 1-35.
- [288] J. E. Littlewood, On inequalities in the theory of functions, Proc. London Math. Soc. **23** (1925), 481-519.
- [289] J. E. Littlewood and R. E.A.C. Paley, A proof that and odd schlicht function has bounded coefficients, J. London Math. Soc. 7 (1932), 167-169.
- [290] S. Liu, Z. Fu, S. Liu and Q. Zhao, Power series expansion method and its applications to nonlinear wave equation, Physics Letters A 309 (2002), 234-239.

[291] A. E. Livingston, *Convex meromorphic mappings*, Annales Polonici Mathematici **59** (1994), 275-291.

- [292] A. A. London, Fekete-Szegö inequalities for close-to-convex functions, Proc. Amer. Math. Soc. 117 (1993), 947-950.
- [293] K. Löwner, Üntersuchubgen über schlichte konforme Abbildungen des Einheitsskreises, I, Math. Ann. 89 (1923), 103-121.
- [294] K. Löwner, Untersuchungen über die Verzerrung bei konformen Abbildungen des Einheitskreises |z| < 1, die durch Funktionen mit nichtverchwinderder Ableitung geliefert werden, S. -B. Sächs. Acad. Wiss. **69** (1917), 69-106.
- [295] L. G. Lucht and A. Schmalmack, Polylogarithms and arithmetic function spaces, Acta Aritmetica XCV 4 (2000), 361-382.
- [296] Y. L. Luke, The Special Functions and Their Approximations, Vol. I, Academic Press, Inc., New York, 1969.
- [297] C. Maclaurin, A Treatise of Fluxions in Two Books, Ruddimans, Edinburgh, 1742.
- [298] W. Magnus and F. Oberhettinger, Formulas and Theorems for the Special Functions of Mathematical Physics, Chelsea, New York, 1946.
- [299] S. M. Manjegani, Hölder and Young inequalities for the trace of operator, Positivity 11 (2007), 239-250.
- [300] A. M. Marino, V. Boyer and P. D. Lett, Violation of the Cauchy-Schwarz inequality in the macroscopic regime, Phys. Rev. Letters 100 (23) (2008), 4-7.
- [301] A. W. Marshall and I. Olkin, *Inequalities: Theory of Majorization and Its Applications*, Academic Press, New York, 1979.
- [302] M. Masjed-Jemai, S. S. Dragomir and H. M. Srivastava, Some generalizations of Cauchy-Schwarz and the Cauchy-Bunyakovsky inequalities involving four free parameters and their applications, Math. Comp. Model. 49 (2009), 1960-1968.

[303] A. Matković, J. Pečarić and I. Perić, A variant of Jensen's inequality od Mercer's type for operators with applications, Linear Algebra Appl. 418 (2006), No. 2-3, 551-564.

- [304] J. Matkowski, Convex functions with respect ro a mean and a characterization of quasiarithmetic means, Real Anal. Exchange **29** (2003/2004), 229-246.
- [305] J. Mathews and R. L. Walker, Mathematical Methods of Physics, 2nd ed., Addison-Wesley, Reading, 1970.
- [306] L. C. Maximon, The dilogarithm function for complex argument, Proc. Roy. Soc. London Ser. A **459** (2003), 2807-2819.
- [307] N. Mercator, Logarithmotechnia, sive Methodus Construendi Logarithmos Nove Accurate et Facilis, G. Godbit, London, 1668.
- [308] S. S. Miller and P. T. Mocannu, Differential Subordinations. Theory and Applications, Marcel Dekker, Inc., New York/Basel, 2000.
- [309] S. S. Miller, P. T. Mocanu and M. O. Reade, All α-convex functions are univalent and starlike, Proc. Amer. Math. Soc. 37 (2) (1973), 553-554.
- [310] N. Minculete, A result about Young's inequality and several applications, Scientia Magna 7 (1) (2011), 61-68.
- [311] H. N. Minh, Summations of polylogarithms via evaluation transform, Math. Comput. Simulation 1336 (1996), 707-728.
- [312] H. N. Minh, M. Petitot and J. Van Der Hoeven, *Shuffle algebra and polylogarithms*, Discrete Mathematics **225** (2000), 217-230.
- [313] H. Minkowski, Geometrie der Zahlen, Teubner, Leipzig, 1896.
- [314] A. K. Mishra and P. Gochhayat, The Fekete-Szegö problem for k-uniformly convex functions and for a class defined by the Owa-Srivastava operator, J. Math. Anal. Appl. **347** (2008), 563-572.
- [315] B. Mitchell, Taylor series methods for the solution of the point reactor kinetics equations, Annals of Nuclear Energy 4 (4-5) (1977), 169-179.

[316] D. S. Mitrinović, Analytic Inequalities, Springer-Verlag, New York, 1970.

- [317] D. S. Mitrinović and J. E. Pečarić, On an extension of Hölder's inequality, Bolletino U.M.1.4-A 7 (1990), 405-408.
- [318] D. S. Mitrinović, J. E. Pečarić and A. M. Fink, *Classical and New Inequalities in Analysis*, Kluwer Acad. Publ., Dordrecht, 1993.
- [319] P. T. Mocannu, Une propriété de convexité generalisée dans la théorie de la représentation conforme, Mathematica (Cluj) 11 (34) (1969), 127-133.
- [320] P. T. Mocanu, T. Bulboacă and Gr. Şt. Sălăgean, Teoria Geometrică a Funcțiilor Univalente, Casa Cărții de Ştiință, Cluj, 1999.
- [321] A. Mohammed and M. Darus, On Fekete-Szego problem for certain subclass of analytic functions, *Int. J. Open Problems Compt. Math.*, **3**(4) (2010), 510-520.
- [322] B. Mond and J. E. Pečarić, Remarks on Jensen's inequality for operator convex functions, Ann. Univ. Mariae Curie-Sklodowska Sect. A 47 (1993), 96-103.
- [323] Y. de Montaudouin, W. Tiller and H. Vold, Applications of power series in computational geometry, Computer-Aided Design 18 (1986), no. 10, 514-524.
- [324] P. Montel, Leçons sur les Fonctions Univalentes ou Multivalentes, Gauthier-Villars, Paris, 1933.
- [325] A. A. Nahla, Taylor's series method for solving the nonlinear point kinetics equations, Nuclear Engineering and Design 241 (2011), 1592-1595.
- [326] A. A. Nahla and E. M. E. Zayed, Solution of the nonlinear point nuclear reactor kinetics equations, Progress in Nuclear Energy **52** (8) (2010), 743-746.
- [327] M. A. Nasr and H. R. El Gawad, On the Fekete-Szegö for close-to-convex functions of order ρ , Proc. Int. Conference on New Trends in Geometric Functions Theory and Applications, World Scientific (1991), 66-74.

[328] M. A. Nasr and M. K. Aouf, Characterisation for convex and starlike functions of complex order in U, Bull. Fac. Sci. Assiut. Univ. A 11 (2)(1982), 117-121.

- [329] J. Needham, Science and Civilization in China: Mathematic and Sciences of the Heavens and the Earth, Vol. 3, Caves Books, Ltd., Taipei, 1986.
- [330] Z. Nehari, Conformal Mappings, Mc. Grraw-Hill Book Comp., New York, 1952.
- [331] E. Neuman, Inequalities involving modified Bessel functions of the first kind, J. Math. Anal. Appl. 171 (1992), 532-536.
- [332] R. Nevanlinna, Über die konforme abbildung von stergebieten, Oversikt av Finska-Vetenskaps. Soc. Forhandlingar A 63 (6) (1921), 48-408.
- [333] C. P. Niculescu, Convexity according to means, Math. Inequal. Appl. 6 (2003), 571-579.
- [334] C. P. Niculescu and L. -E. Persson, Convex Functions and Their Applications, Springer, New York, 2004.
- [335] O. A.Nielsen, Sharpness in Young's inequality for convolution products, Can. J. Math. **46** (6) (1994), 1287-1298.
- [336] K. I. Noor, Bazelevič functions of type β , Internat. J. Math. & Math. Sci. 5 (2) (1982), 411-415.
- [337] K. I. Noor, On a generalization of close-to-convexity, Internat. J. Math. Math. Sci. 6 (1983), 327-334.
- [338] M. Obradović, Some results on Bazilevič functions, Mat. Vesnik **37** (1) (1985), 92-96.
- [339] M. Obradović, On certain differential operator and some classes of univalent functions, Serija Mathematica 6 (1992), 107-112.
- [340] J. J. O'Connor and E. F. Robertson, A History of Calculus, University Of St. Andrews, Scotland, 1996 (http://www-groups.dcs.st.ac.uk/~history/HistTopics/The_rise_of_calculus. html).

[341] A. T. Oladipo, K. O. Babalola and T. O. Opoola, Generalized convolution properties for certain classes of analytic functions, Research J, Appl. Sci. 2 (3) (2007), 289-294.

- [342] K. B. Oldham, J. Myland and J. Spanier, *The Gauss Hypergeometric Function F* (a, b, c, x), Chapt. 60, p.627- 636 in An Atlas of Functions, Springer-Verlag, New York, 2009.
- [343] T. O. Opoola, On a new subclass of univalent functions, Mathematica (Cluj), **59** (2) (1994), 195-200.
- [344] T. O. Opoola, K. O. Babalola, O. A. Fadipe-Joseph and K. Rauf, On coefficient bounds of a subclass of univalent functions, Jour. of the Nigerian Math. Soc. 23 (2004), 87-92.
- [345] T. O. Opoola and J. A. Gbadeyan, On certain properties for a subclass of univalent functions, Nig. J. Pure & Appl. Sci. 11 (1996), 476-480.
- [346] T. O. Opoola and A. T. Oladipo, On some properties of a subclass of univalent functions, Nig. J. Pure & Appl. Sci. 12 (1997), 569-572.
- [347] H. Orhan and D. Răducanu, Fekete-Szegö problem for strongly starlike functions associated with generalized hypergeometric functions, Math. Comput. Modelling **50** (2009), 430-438.
- [348] G. I. Oros, A class of holomorphic functions defined using a differential operator, General Maths. 13 (4) (2005), 13-18.
- [349] S. Owa, C. Y. Shen and M. Obradović, Certain subclass of analytic functions, Tamkang J. Math. 20 (1989), 105-115.
- [350] S. Owa, M. Obradović and S. K. Lee, *Notes on a certain subclass of analytic functions introduced by Sălăgean*, Bull. Korean Math. Soc. **23** (1986), 133-140.
- [351] J. E. Pečarić and S. S. Dragomir, A refinement of Jensen inequality and applications, Studia Univ. Babeş-Bolyai, Mathematica 24 (1) (1989), 15-19.

[352] J. E. Pečarić, F. Proschan and Y. L. Tong, Convex Functions, Partial Orderings and Statistical Applications, Mathematics in Science and Engineering, 187, Academic Press, Inc., Bostan, MA, 1992.

- [353] A. Pfluger, The Fekete-Szegö inequality for complex parameters, Complex Variables Theory Appl. 7 (1986), 149-160.
- [354] A. Pfluger, On the functional $|a_3 ua_2^2|$ in the class S, Complex Variables Theory Appl. 10 (1988), 83-95.
- [355] Y. X. Ping, L. Liu, C. M. Zhang and Z. Cheng, Violations of Bell inequality, Cauchy-Schwarz inequality and entanglement in a two-mode three level atonic system, Comm. Theoritical Phys. **56** (2011), 151-156.
- [356] J. Plemelj, Über den Verzerrungssatz vom P. Koebe, Ges. Dtsch. Naturforschren Artze, 85, Versammlung Wien Zeiter Teie, Erste Hälfte, 1913.
- [357] C. Pommerenke, *Univalent Functions*, Vandenhoeck and Ruprecht, Göttingen, 1975.
- [358] S. Ponnusamy, Foundations of Mathematical Analysis, Birkhäuser, Boston, 2012.
- [359] S. Ponnusamy and H. Silverman, Complex Analysis with Applications, Birkhäuser, Boston, 2006.
- [360] T. Popoviciu, Les Fonctions Convexes, Hermann et Cie, Paris, 1944.
- [361] T. Precupanu, On a generalization of Cauchy-Buniakowski-Schwarz inequality, Anal. St. Univ. "Al. I. Cuza" Iaşi 22 (2) (1976), 173-175.
- [362] I. I. Privalov, On functions giving a univalent conformal mapping, Mathematicheski Sbornik. Izdatel'stvo Nauka, Moskva, Russian 32 (1924), 350-365...
- [363] A. P. Prudnikov, Yu. A. Brychkov and O. I. Marichev, *Integrals and Series*, Vols. 1 and 3, Gordon and Breach Sci. Publ., New York, 1986 and 1990.
- [364] A. P. Prudnikov, Yu. A. Brychkov and O. I. Marichev, *Integrals and Series*, Vol. 3, Gordon and Breach Sci. Publ., New York, 1990.

[365] H. Qiang and Z. Hu, Generalizations of Hölder's and some related inequalities, Computers. Maths. Appl. **61** (2011), 392-396.

- [366] S.-L. Qiu and M. Vuorinen, Landen inequalities for hypergeometric functions, Nagoya Math. J. **154** (1999), 31-56.
- [367] S. T. Rachev, Y. S. Kim, M. L. Bianchi and F. J. Fabozzi, Financial Models with Levy Processes and Volatility Clustering, John Wiley & Son, Inc., Hoboken/New Jersey, 2011.
- [368] E. Rainville, Special Functions, Macmillan, New York, 1960.
- [369] Y. Rakotondratsimba, Weighted Young inequalities for convolutions, Southeast Asian Bull. Math. 26 (2002), 77-99.
- [370] V. Ravichandran, M. Darus, M. H. Khan and K. G. Subramanian, Fekete-Szegö inequality for certain class of analytic functions, Aust. J. Math. Anal. Appl. 1 (2) (2004), 1-7.
- [371] M. O. Reade, On close-to-convex univalent functions, Michigan Math. J. 3 (1955), 59-62.
- [372] R. Remak, Ueber eine spezielle klasse schlichter konformer abbildungen des einheitekreises, Mathematica B 11 (1943), 175-192.
- [373] U. Richard, Sur des inégalités du type Wirtinger et leurs application aux équationes différebtielles ordinaires, Colloquium of Analysis held in Rio de Janeiro, August 1972 (1972), 233-244.
- [374] B. Riemann, Grundlagen Für Eine Allgemeine Theorie der Functionen Einer Veränderlichen Complexen Grösse, Inaugural-dissertation, Göttingen-Gesammelte Mathematische Werke 3rd ed., Springer-Verlag, Berlin 1851, 3-48.
- [375] A. W. Roberts and D. E. Varberg, Convex Functions, Acad. Press, New York/London, 1973.
- [376] G. S. Robertson, On the theory of univalent functions, Ann. Math. 37 (1936), 374-408.

[377] R. T. Rockafellar, *Convex Analysis*, Princeton Univ. Press, Princeton, 1970.

- [378] L. J. Rogers, An extension of a certain theorem in inequalities, Messenger Maths. 17 (1888), 145-150.
- [379] W. W. Rogosinski, Über positive harmonische Entwicklungen und typisch reele Potenzreihen, Math. Z. **35** (1932), 93-121.
- [380] A. De Rossi and L. Rodino, Strengthened Cauchy-Schwarz inequality for biorthogonal wavelets in Sobolev spaces, J. Math. Anal. Appl. 299 (1) (2004), 49-60.
- [381] W. Rudin, Real and Complex Analysis 3rd Ed., McGraw-Hill, New York, 1987.
- [382] S. Ruscheweyh, Neighborhoods for univalent functions, Proc. Amer. Math. Soc. 81 (1981), 521-527.
- [383] G. S. Sălăgean, Subclasses of Univalent Functions, Lecture Note in Math. 1013, Springer-Verlag, Berlin, Heidelberg and New York, 1983, 362-372.
- [384] A. Schild, On a class of functions schlicht in the unit circle, Proc. Amer. Math. Soc. 5 (1954), 115-120.
- [385] A. Schild, On a problem in conformal mapping of schlicht functions, Proc. Amer. Math. Soc. 4 (1954), 43-51.
- [386] G. Schober, *Univalent Function Selected Topic*. Lecture Notes in Math. No. 478, Springer-Verlag, Berlin, 1975.
- [387] P. Schreider, *The Cauchy-Bunyakovsky-Schwarz inequality*, Hermann Gra*B*mann, Lieschow, 1994, 64-70.
- [388] H. A. Schwarz, Üeber ein flächen kleinsten flächeninhalts betreffendes problem der variationsrechnung, Acta Societatis Scientiarum Fennicae XV (1888), 315-362.
- [389] H. A. Schwarz, Über ein die flächen kleinste flächeninhalts betreffende problem der variationsrechnung, Acta Sci. Scient. Fenn. 15 (1885), 315-362.

[390] B. Şeker, On certain new subclass of close-to-convex functions, Appl. Math. Comput. **218** (3) (2011), 1041-1045.

- [391] D. Shanks, *Polylogarithms, Dirichlet series and certain constants*, Math. Comput. **18** (86) (1964), 3220324.
- [392] T. N. Shanmugan, C. Ramachandran and V. Ravichandran, Fekete-Szegö problem for subclasses of starlike functions with respect to symmetric points, Bull. Korean Math. Soc. 43 (3) (2006), 589-598.
- [393] K. Al-Shaqsi and M. Darus, On Fekete-Szegö problems for certain subclass of analytic functions, Appl. Math. Sci. 2(2008), 431-441.
- [394] K. Al-Shaqsi and M. Darus, Fekete-Szegö problem for univalent functions with respect to k-symmetric points, The Australian J. Math. Anal. Appl. 5 (2008), 1-12.
- [395] W. Sibagaki, Theory and Application of the Gamma Function, Iwanami Syoten, Tokyo, 1952.
- [396] H. Silverman, Starlike and convexity properties for hypergeometric functions, J. Math. Anal. Appl. 172 (1993), no. 2, 574-581.
- [397] H. Silverman, Coefficient bounds for quotients of starlike functions, J. of Maths. 13 (1983), no. 4, 627-630.
- [398] H. Silverman, Univalent functions with negative coefficients, Proc. Amer. Math Soc. 51 (1975), no. 1, 109-116.
- [399] A. N. Singh, On the use of series in Hindu mathematics, Osiris 1 (1936), 606-628.
- [400] R. Singh, On Bazilevič functions, Proc. Amer. Math. Soc. 38 (2) (1973), 261-271.
- [401] S. Singh, S. Gupta and S. Singh, A general coefficient inequality, General Maths. 17 (3) (2009), 99-104.
- [402] G. Sobczyk, Conformal mappings in geometric algebra, Notices of the AMS **59** (2) (2012), 264-273.

[403] J. Sokół, A. W-. Wajnryb, On certain problem in the class of k-starlike functions, Comput. Math. Appl. **62** (2011), 4733-4741.

- [404] J. Spanier and K. B. Oldham, The Pochhammer Polynomials $(x)_n$, Ch. 18 in An Atlas of Functions, Hemisphere, Washington DC, 1987, pp. 149-165.
- [405] W. Spence, An Essay on Logarithmic Transcendents; with an inquiry into their applications to the integral calculus and the summation of series, J. Murray, London, 1809.
- [406] H. M. Srivastava and S. Owa, Current Topics in Analytic Function Theory, World Scientific Publ., London, 1992.
- [407] J. M. Steele, The Cauchy-Schwarz Master Class: An Introduction to the Art of Mathematical Inequalities, Cambridge University Press, New York, 2004.
- [408] W. L. Steiger, On a generalization of the Cauchy-Schwarz inequality, Amer. Math. Monthly **76** (1969), 815-816.
- [409] J. Stillwell, Mathematics and Its History, 2nd. ed., Springer, Berlin, 2004.
- [410] O. Stolz, Grundzüge der Differential- und Integralrechnung, Vol. 1, G. B. Teubner, Leipzig, 1893.
- [411] E. Strohhäcker, Beitrage zur theorie der schlichten funktionen, Math. Zeit. 37 (1933), 356-380.
- [412] W. T. Sulaiman, Notes on Young's inequality, Intern. Math. Forum 4 (24) (2009), 1173-1180.
- [413] H-S. Sun, Derivative of power series attached to Γ-Transform of p-adic measures, J. Number Theory 130 (2010), 10-26.
- [414] R. C. H. Tanner, Mathematics begins with inequality, The Math. Gazette 45 (1961), no. 354, 292-294.
- [415] A. O. Tăut, G. I. Oros and R. Şendruţiu, On a class of univalent functions defined by Sălăgean differential operator, Banch J. Math. Anal. 3 (1) (2009), 61-67.

[416] B. Taylor, Methodus Incrementorun Directa et Inversa, Innys, London, 1715.

- [417] F. Temurtas, A. Gulbag and N. Yumusak, A study on neural networks using Taylor series expansion of sigmoid activation function, Proceedings of International Conf. Comput. Sci. Appl. (ICCSA) **3046** (2004), 389-397.
- [418] D. K. Thomas, On a subclass of Bazilevič functions, Int. J. Math. Sci. 8 (4) (1985), 779-783.
- [419] D. K. Thomas, On Bazilevič functions, Amer. Math. Soc. 132 (2) (1968), 353-361.
- [420] T. Thirivikraman, On the contribution of Kerala to mathematics in the late nedieval and early modern periods, Math. Student **53** (1-4) (1985), 204-208.
- [421] G. Toader and S. S. Dragomir, *Refinement of Jensen's inequality*, Demonstratio Mathematics **28** (1995), 329-334.
- [422] E. Tolsted, An elementary derivation of the Cauchy, Hölder and Minkowski inequalities from Young's inequality, Math. Mag. 37 (1) (1964), 2-12.
- [423] M. Tominaga, Specht's ratio in the Young inequality, Sci. Math. Japon. 55 (2002), 538-588.
- [424] G. Turell, Mathematics for Chemistry and Physics, Acad. Press, New York/Boston/London, 2002.
- [425] M. A. Usábel, A note on the Taylor series expansions for multivariate characteristics of classical risk processes, Insurance: Math. and Economics 25 (1999), 37-47.
- [426] N. Uyanik and S. Owa, New extensions for classes of analytic functions associated with close-to-convex and starlike or order α, Math. Comput. Modelling 54 (2011), 359-366.
- [427] P. M. Vasić and J. E. Pečarić, On Jensen inequality, Univ. Beograd. Publ. Elektrotehn Fak. Ser. Mat. Fis. No. **639-677** (1979), 50-54.

[428] F. Viète, Variorum de Rebus Mathematicis Responsorum, Liber VIII, Tours 1593, in Opera mathematica: in unum volume congesta as recognita opera atque studio Francisci a Schooteen, Lyede B. et A. Elzeviriorum, 1646.

- [429] J. Wallis, Arithmetica Infinitorum sive Mova Methodus Inquirendi in Curvilineorum Quadraturam, aliaque difficiliora Matheseos Problemata, Oxon, Lichfield, 1656.
- [430] Z-G. Wang, Y. Sun and N. Xu, Some properties of certain meromorphic close-to-convex functions, Appl. Math. Letters **25** (3) (2012), 454-460.
- [431] K. -J. Wirths, The Koebe domain for concave univalent functions, Serdica J. Maths. 29 (2003), 355-360.
- [432] A. Witkowski, On Young's inequality, J. Math. Pure Appl. Math. 7 (2006), no. 5, Art. 164, available online at http://jipam.vu.edu.au/article.php-sid=782.
- [433] F-H. Wong, C-C. Yeh, S-L. Yu and C-H. Hong, Young's inequality and related results on time scales, Appl. Math. Letters 18 (2005), 983-988.
- [434] Q-H. Xu, H. M. Srivastava and Z. Li, A certain subclass of analytic and close-to-convex functions, Appl. Math. Letters 24 (3) (2011), 396-401.
- [435] K. Yamaguchi, On functions satisfing Re $\{f(z)/z\} > 0$, Proc. Amer. Math. Soc. 17 (1966), 588-591.
- [436] X. Yang, *Hölder's inequality*, Appl. Maths Letters **16** (2003), 897-903.
- [437] X. Yang, A generalization of Hölder's inequality, J. Math. Anal. Appl. 247 (2000), 328-330.
- [438] X. Yang, Refinement of Hölder's inequality and application to Ostrowski inequality, Appl. Maths. Comput. 138 (2003), 455-461.
- [439] N. Young, An Intoduction to Hilbert Space, Cambridge Univ. Press, New York, 1988.
- [440] W. H. Young, On classes of summable functions and their Fouries series, Proc. Roy. Soc. Lond. Series A 87 (594) (1912), 225-229.

[441] W. H. Young, On the multiplication of successions of Fourier constants, Proc. Roy. Soc. Lond. Series A 87 (596) (1912), 331-339.

- [442] D. Zagier, The remarkable dilogarithm, J. Math. Phys. Sci. 22 (1988), 131-145.
- [443] D. Zagier, Values of zeta functions and their applications, in: First European Congress of Mathematics Vol. 2, Birkhäuser, Basel, 1994, pp. 497-512.
- [444] A. Zaks, Annuities under random rates of interest, Insurence: Mathematics and Finance 28 (2001), 1-11.
- [445] L. Zheng, Remark on refinement of Cauchy-Schwarz inequality, J. Math. Anal. Appl. 218 (1998), 13-21.
- [446] P. Zhigang, On a subclass of close-to-convex functions, Acta Mathematica Scientia **30** (5) (2010), 1449-1456.
- [447] L. Zhu, Jordan type inequalities involving the Bessel and modified Bessel functions, Comp. Math. Appl. **59** (2010), 724-736.
- [448] L. Zhu, On Young's inequality, Intern. J. Math. Educt. Sci. Tech. **35** (4) (2004), 601-608.
- [449] H. Zuo, G. Shi and M. Fujii, Refined Young inequality with Kantorovich constant, J. Math. Ineq. 5 (4) (2011), 551-556.